APPLICATION OF MOLECULAR BIOLOGICAL TECHNIQUES TO THE IDENTIFICATION OF CYANOBACTERIA

WEIQUN LU

A thesis submitted in partial fulfilment of the requirements of Liverpool John Moores University for the degree of Doctor of Philosophy

School of Biomolecular Sciences Liverpool John Moores University Byrom Street Liverpool L3 3AF

June, 1998

To my family

.

-

-

Declaration

I declare that while registed as a candidate for the University's research degree of Doctor of Philosophy, no material contained in this thesis has been used in any other submission for another academic award of Liverpool John Moores University or other academic or professional Institution.

Publications

Parts of this thesis have been published as follows:

(1) W. Lu, V.A. Saunders, E. H. Evans and S. M. McColl (1995) Molecular comparisons of cyanobacterial free living and laboratory strains (poster), Third European Workshop on The Molecular Biology of Cyanobacteria, Sevilla, Spain, p: 76 (Abstract).

(2) W. Lu, E. H. Evans, S. M. McColl and V.A. Saunders (1995) Several Methods for isolation and Characterisation of plasmid DNA in Cyanobacteria (poster), Xth International Photosynthesis Congress, Montpellier, France, p: 143 (Abstract).

(3) W. Lu, E. H. Evans, S. M. McColl and V.A. Saunders (1997) Identification of cyanobacteria by polymorphisms of PCR-amplified ribosomal DNA spacer region, FEMS Microbiology Letters 153 : 141-149.

(4) V.A. Saunders, E. H. Evans, W. Lu and S. M. McColl (1998) Use of 16S-23S ribosomal DNA spacer region for identification of cyanobacteria (poster). VIIIth International Symposium on Microbial Ecology, Halifax, Canada. P: 292 (Abstract).

Contents

Acknowledge	ements	i
Abbreviation	IS	ii
Abstract		iv
Chapter 1	General Introduction	1
1.1	Cyanobacteria	2
1.2	Taxonomic classification	4
1.3	Molecular techniques used in taxonomy	13
1.4	Aims	26
Chapter 2	Materials and Methods	30
2.1	Materials	
	2.1.1. Chemicals	31
	2.1.2. Cyanobacterial strains	32
	2.1.3. Media	32
2.2	. Methods	
	2.2.1. Growth and Maintenance of Cyanobacteria	32
	2.2.2. Description of the sampling sites	39
	2.2.3. Sampling	48
	2.2.4. Isolation of cyanobacteria	50
	2.2.5. Recognition of cyanobacterial isolates	50
	2.2.6. Measurement of O_2 -evolution	51
	2.2.7. Determination of pigment content	51
	2.2.8. Purification of cyanobacteria	52
	2.2.9. Genus identification of cyanobacteria	53
	2.2.10.Determination of Nitrogenase Activity	53
	2.2.11. Determination of ability to grow	53

photoheterotrophically	
2.2.12. Electron microscopic and light microscopic	54
examination	
2.2.13. Determination of growth rates of cyanobacterial	55
cultures	
2.2.14. Preparation and purification of cyanobacterial DNA	56
2.2.14.1. Mini-methods for preparation of	56
plasmid DNA	
2.2.14.2. Large scale plasmid DNA preparation	59
2.2.14.3. Preparation of total genomic DNA	60
2.2.14.4. Phenol-chloroform extraction and ethanol precipitation of DNA	61
2.2.15. Agarose gel electrophoresis	62
2.2.16.Recovery of DNA from agarose gels	63
2.2.17. Measurement of DNA concentration	64
2.2.18.DNA restriction	65
2.2.19.Random Amplified Polymorphic DNA - Polymerase	65
Chain Reaction (RAPD-PCR)	
2.2.20. PCR amplification of rDNA spacer	68
2.2.21. Restriction digestion of PCR-amplified DNA	69
2.2.22. Data handling by Binary Matrix and	70
Similarity coefficients	
2.2.23. Phylogenetic analysis of PCR-amplified products	70
2.2.24. Purification of PCR-amplified product and	71
TA-Cloning	
2.2.25. Isolation and purification of recombinant	72
plasmid DNA	

		2.2.26. DNA sequencing	72
		2.2.27. DNA sequence analysis	73
		2.2.28. Primer design	74
Chapter	3	Isolation, Purification and Identification	75
		of Cyanobacteria	
	3.1.	Introduction	76
	3.2.	Results	85
		3.2.1. Isolation condition	85
		3.2.2. Simple recognition cyanobacteria	85
		3.2.3. Purification of isolates	92
		3.2.4. Classification of isolates from different	92
		sampling sites	
	3.3.	Discussion	116
Chapter	r 4	Isolation of DNA from Cyanobacteria	122
	4.1.	Introduction	123
	4.2.	Results	124
		4.2.1. Comparison of methods for plasmid isolation	124
		4.2.2. Analysis of extracted plasmid DNA	128
		4.2.3. Effect of culture age on plasmid isolation	128
		4.2.4. Plasmid content of selected cyanobacteria	131
		4.2.5. Isolation of total DNA	132
		4.2.6. Restriction analysis of plasmid and	133
		genomic DNA	
	4.3	Discussion	136
Chapte	r 5	RAPD Analysis of Cyanobacteria	139
	5.1	. Introduction	140
	5.2	. Results	145
		5.2.1. Reaction conditions for RAPD-PCR	145
		5.2.2. Primer specificity	157

		5.2.3. RAPDs of different cyanobacteria	161
	5.3.	Discussion	173
Chapter	6	Identification of cyanobacteria by polymorphisms of	178
		PCR-amplified ribosomal DNA spacer region	
	6.1.	Introduction	179
	6.2.	Results	185
		6.2.1. RFLP-PCR	185
		6.2.2. 16S-23S rDNA spacer sequencing	205
	6.3.	Discussion	237
Chapter	7	Morphological and genetic methods in identification	245
		and taxonomy of cyanobacteria	
	7.1.	Introduction	246
	7.2.	Results	250
	7.3.	Discussion	287
Chapter	8	General Discussion	295
	8.1.	General Discussion	296
	8.2.	Recommendations for Further work	298
Summai	rv		302
Referen	ces		305
Append	ix A	Preparation of Media	329
Append	ix B	Preparation of Reagents and Buffers	337
Append	ix C	Classification of cyanobacteria (after Rippka, 1988)	339

Content of Figures

Figure 1.1	Schematic diagram of ribosomal DNA repeat showing	24
	internal transcribed spacer region (ITS).	
Figure 1.2	Flow diagram summarising the schedule of work	29
Figure 2.1	Map of the United Kingdom showing collecting sites	40
Figure 2.2	Map of Wirral showing sampling location	41
Figure 2.3	Map of Bala lake showing sampling location	42
Figure 2.4	Map of Windermere lake sampling location	43
Figure 2.5	Map of China showing sampling site	44
Figure 2.6	Map of Taihu lake showing sampling location	45
Plate 1	A view of the Raby pool which was one of the freshwater	46
	sampling sites in this study	
Plate 2	A view of the Bala lake which was selected for freshwater	46
	sampling in this study.	
Plate 3	A view of the Bowness, Windermere Lake which was one	47
	of the freshwater sampling sites in this study.	
Plate 4	A view of the Liyuan Garden, Taihu lake which was selected	47
	for freshwater sampling in this study.	
Figure 3.1.	Flow diagram of morphological identification and	82
	taxonomy system	
Figure 3.2	Phycobiliproteins absorption spectra of cyanobacteria	87
Figure 3.3	The absorption spectrum of whole cell cyanobacteria,	89
	green-algae and prochloron	

Figure 3.4	The absorption spectrum of cell-free extracts cyanobacteria,	90
	green-algae and prochloron	
Figure 3.5	The number of cyanobacterial isolates and the total photosynthetic	91
	isolates isolated from different type of water samples of Raby pool	
	by using direct isolation	
Figure 3.6	Micrograph of Synechococcus LJM 002	93
Figure 3.7	Micrograph of Synechococcus LJM 003	94
Figure 3.8	Micrograph of Synechococcus LJM 004	95
Figure 3.9	Micrograph of Synechococcus LJM 020	96
Figure 3.10	Micrograph of Chroococcidiopsis LJM 013	97
Figure 3.11	Micrograph of Pseudanabaena LJM 001	98
Figure 3.12	Micrograph of Pseudanabaena LJM 005	99
Figure 3.13	Micrograph of Pseudanabaena LJM 006	100
Figure 3.14	Micrograph of Pseudanabaena LJM 007	101
Figure 3.15	Micrograph of Pseudanabaena LJM 009	102
Figure 3.16	Micrograph of Pseudanabaena LJM 010	103
Figure 3.17	Micrograph of Pseudanabaena LJM 016	104
Figure 3.18	Micrograph of Pseudanabaena LJM 019	105
Figure 3.19	Micrograph of Leptolyngbya LJM 012	106
Figure 3.20	Micrograph of Leptolyngbya LJM 014	107
Figure 3.21	Micrograph of Phormidium LJM 017	108
Figure 3.22.	The taxonomic relationships of cyanobacterial isolates	115
	from different sampling sites	
Figure 4.1	Agarose gel electrophoresis of plasmid DNAprepared by	126
	five rapid methods from Synechococcus PPC7942	

Figure 4.2.	Agarose gel electrophoresis of plasmid DNA preparation	127
Figure 4.3	Agarose gel electrophoresis of charaterisation plasmid DNA	129
	by heat-treated	
Figure 4.4	Effect of culture age on plasmid isolation	130
Figure 4.5	Agarose gel electrophoresis of total genome DNA	134
	from Synechococcus PCC 7942	
Figure 4.6	Agarose gel of Bam HI restricted plasmid and	135
	genomic DNA of Synechococcus PCC 7942	
Figure 5.1	A diagrammatic representation of the PCR reaction	142
Figure 5.2	Origins of random amplification fragment length	144
	polymorphisms	
Figure 5.3	Agarose gel electrophoresis showing the effect of	147
	DNA concentration on the RAPD assay	
Figure 5.4	Agarose gel electrophoresis of RAPD products of	148
	Synechococcus PCC7942 DNA obtained with	
	Different magnesium concentration	
Figure 5.5	Agarose gel electrophoresis of RAPD products using	149
	Different annealing temperature for the primers	
Figure 5.6.	Agarose gel electrophoresis showing the effect of	151
	deoxynucleotide triphosphate concentration	
	on the RAPD assay	
Figure 5.7.	Agarose gel electrophoresis showing the effect of primer	152
	concentration on the RAPD assay	
Figure 5.8	Agarose gel electrophoresis of RAPDs generated using	153
	different DNA polymerases	

Figure 5.9	Agarose gel electrophoresis of RAPDs generated	155
	using different PCR program	
Figure 5.10	Agarose gel electrophoresis of RAPDs generated	156
	using different PCR machines and different program	
Figure 5.11	Agarose gel elctrophoresis of RAPD products of	158
	Nostoc mus DNA obtained with different magnesium	
	concentrations and different annealing temperatures	
Figure 5.12	Agarose gel elctrophoresis of RAPD products of	159
	Synechococcus PCC7942 DNA obtained with	
	different magnesium concentrations and	
	primer combinations	
Figure 5.13	Agarose gel electrophoresis of RAPDs of Synechococcus	160
	PCC 7942 generated using different primer	
Figure 5.14	Agarose gel electrophoresis RAPDs generated using	162
	different cyanobacteria and different primers	
Figure 5.15	Agarose gel electrophoresis of RAPDs of	163
	different cyanobacteria	
Figure 5.16	Dendrogram indicating genetic distances among	169
	cyanobacterial strains	
Figure 6.1	Schematic diagram of ribosomal DNA repeat showing	181
	internal transcribed spacer region (ITS).	
Figure. 6.2	Agarose electrophresis gels of amplified rDNA spacer	187
	products of selected cyanobacterial strains.	
Figure 6.3.	Agarose electrophresis gels of single-strand conformation	191
	polymorphisms (SSCP) patterns of selected cyanobacterial	

strains.

Figure 6.4.	Agarose electrophresis gels of RFLP patterns of selected	195
	cyanobacterial strains.	
Figure 6.5	Diagramatic representation of RFLP patterns of amplified	196
	rDNA products	
Figure 6.6.	Dendrogram indicating genetic distances among cyanobacterial	201
	strains	
Figure 6.7.	Agarose electrophresis gel of amplified rDNA spacer products	203
	and RFLP patterns of selected cyanobacterial strains.	
Figure 6.8.	Agarose electrophresis gels of amplified partial 16S rDNA	204
	products and RFLP patterns of selected cyanobacterial strains	
Figure 6.9	Alignment of 16S-23S rDNA spacer sequences from Anabaena	206
	PCC 7118, Anabaena PCC 7120 and Nostoc ellipsora	
Figure 6.10	Sequencing electrophresis gels of amplified products of partial	208
	16S-23S rDNA spacer of selected cyanobacterial strains	
Figure 6.11	Fluorogram trace of automated sequencing data	209
Figure 6.12	Comparison of partial 16S-23S spacer sequence of	211
	Synechocystis sp. PCC6803	
Figure 6.13	Alignment of 16S-23S rDNA spacer sequences from various	214
	cyanobacterial species	
Figure 6.14	Diagrammatic representation of the position of restriction	219
	endonuclease sites in the cyanobacterial partial 16S-23S	
	rDNA spacer	
Figure 6.15	Consensus trees derived from parsimony method based	221
	on partial 16S-23S spacer region sequence analysis	

Figure 6.16	Maximum likelihood tree based on partial 16S-23S	226
	spacer region sequence analysis	
Figure 6.17	Agarose gel elctrophoresis of amplified partial 16S-23S	230
	rDNA spacer of Anabaena cylindrica DNA at different	
	magnesium concentrations and different annealing	
	temperatures	
Figure 6.18	Agarose gel elctrophoresis of amplified partial 16S-23S `	231
	rDNA spacer at different total genomic DNA concentration	
	using primers SF and SR.	
Figure 6.19	Agarose gel electrophoresis of partial 16S-23S rDNA	233
	spacer amplified using primers SF and SR	
Figure 6.20	Consensus trees derived from parsimony method based	234
	on analysis of the partial 16S-23S spacer region	
	sequence amplified using primers SF & SR	
Figure 6.21	Maximum likelihood tree based on SF/SR amplified partial	236
	16S-23S spacer region sequence analysis	
Figure 7.1	Time line for the planet Earth and the age of cyanobacteria from	247
	fossil evidence	
Figure 7.2	Diagramatic representation of RFLP patterns of amplified rDNA	251
	products (using primers 17 and 18) from selected strains of	
	cyanobacteria	
Figure 7.3	Dendrogram indicating genetic distances among cyanobacterial	253
	laboratory strains and isolates.	

•

Figure 7.4	Consensus trees trees derived from parsimony method	254
	based on partial 16S 23S spacer region sequence analysis	
Figure 7.5	Maximum likelihood tree based on partial 16S-23S spacer	256
	region (17/18) sequence analysis	
Figure 7.6	Distance tree based on partial 16S-23S spacer region (17/18)	257
	sequence analysis	
Figure 7.7	Transmission electron micrograph of a thin section	259
	Synechococcus CCAP1479/7 and Synechococcus WH 7803	
Figure 7.8	Micrograph of Synechococcus PCC7942.	260
Figure 7.9	Micrograph of Nostoc palaeo	263
Figure 7.10	Micrographs of trichome or clony of some heterocystous	265
	strains	
Figure 7.11	Consensus trees trees derived from parsimony method	266
	based on partial 16S 23S spacer region (SF/SR)	
Figure 7.12	Maximum likelihood tree based on partial 16S 23S spacer	267
	region (SF/SR)	
Figure 7.13	Distance tree based on partial 16S-23S spacer region (SF/SR)	268
Figure 7.14	Alignment of 16S-23S rDNA spacer sequences from various	270
	cyanobacterial species	
Figure 8.1	Flow diagram summarising the schedule of further work	299
Figure 8.2	Flow diagram summarising the schedule of further work	301
	of detection of the presence of toxigenic cyanobacteria	
	rapidly in water samples	

Contens of Tables

.

Table 1.1	Major names of cyanobacteria according to the International	5
	Code of Botanical Nomenture	
Table 1.2	Major diagnostic features of cyanobacteria as	11
	described by Rippka (1988)	
Table 1.3	The major chemotaxonomic and macromolecular	14
	markers of cyanobacteria	
Table 2.1	Cyanobacterial strains	33
Table 2.2	Media of cyanobacteria	36
Table 2.3	Location and type of water samples	49
Table 2.4	Selected 10-base single random primer	67
	codes and sequences	
Table 2.5	Primers used for PCR and sequence determination of	69
	rDNA regions	
Table 3.1	Comparison of organism growth in different conditions	86
Table 3.2	Properties of the cyanobacterial strains isolated from the lakes	109
Table 4.1	Plasmid content of selected cyanobacterial strains	132
Table 5.1	The similarity indices of RAPD for cyanobacteria strains	171
	with primer ABA-7 and primer ABA-10.	
Table 6.1.	Cyanobacteria used and the sizes of rDNA fragments amplified	189
	by primer pairs 1/18, 14/18 and 17/18.	
Table 6.2.	RFLP patterns and pattern-types obtained for selected	198
	cyanobacterial strains	
Table 6.3	The comparison of manual sequencing and automatic sequencing	212
Table 6.4	The similarity of partial 16S-23S rDNA spacer sequences	225
	from 16 cyanobacterial species	

Table 6.5	Specificity of cyanobacterial primers	228
Table 6.6	Sequence and physical characteristics of primers SF and SR	229
Table 7.1.	RFLP patterns and pattern-types obtained for selected	252
	cyanobacterial strains	

Acknowledgements

I would like to thank my supervisors, Professor E. H. Evans and Professor V. A. Saunders, for all their support, guidance and encouragement, and for advice on the preparation of this thesis.

I would especially like to thank Dr. G. Sharples for his excellent advice and help with electron and light microscopy. Thanks must go to Dr. D. Gatherer for his help on using the Phylip package. Thanks also go to Dr. S. Lambert for his enthusiasm, help and introduction on operating some of the instruments used in this research. I am grateful to Dr. T. Gaskell for her help with photography for light microscopy and guidance with one of the sampling sites. I would like to thank Mr. J. Bridson for his help.

Special thanks go to Dr. Yang Wu and Dr. M. Willcox for their advice and help with the use of primers and polymerase AmpliTaq. There are a lot of people I would like to thank who have helped me in this research.

Last but not least I would like to thank my wife, parents and all my family who have provided me with unceasing encouragement throughout my academic career.

i

Abbreviations

A	adenine
Ap	ampicillin
bp	base pairs
С	cytosine
ССАР	Culture Collection Algae and Protozoa
DAF	DNA amplification fingerprinting
dATP	adenosine 5'-triphosphate
dCTP、	cytosine 5'-triphosphate
dGTP	guanine 5'-triphosphate
DNA	deoxyribonucleic acid
dTTP 、	thymine 5'-triphosphate
dNTP	deoxynucleoside triphosphate
EDTA ·	ethylenediaminetetra-acetic acid
G 、	guanine
g 、	gram .
G+C、	guanine + cytosine
h 、	hour
ITS	internal transcribed spacer
Kb	Kilobase pairs
KDa	Kilodaltons
λ	lambda phage
LPP	Lyngbya Plectonema Phormidium
М	Molar
mA	milliamp
MD	megadalton
mg ,	milligram
min ,	minute

ml ·	millilitre
mm .	millimeter
mM .	milli-Molar
mRNA .	messenger RNA
nm ,	nanometer
OD .	optical density
PAGE	polyacrylamide gel electrophoresis
PCC	Pasteur Culture Collection
PCR .	polymerase chain reaction
RAPDs	Random Amplified Polymorphic DNAs
RFLP	restriction fragment length polymorphisim
RNA	ribonucleic acid
rRNA	ribosomal RNA
r.p.m	revolutions per minute
S	second
S	Svedberg unit - sedimentation coefficient
SD.	standard deviation
SDS	sodium dodecylsulphate
STRR	short tandemly repeated repetitive
Τ,	thymine
Tris ,	tris (hydroxymethyl) aminoethane
tRNA	transfer RNA
tRNA ^{ala}	alanine tRNA
tRNA ^{ile}	isoleucine tRNA
μg ·	microgram
μl .	microlitre
μm ,	micrometer
v/v .	volume for volume
w/v	weight for volume

Abstract

In order to identify cyanobacteria responsible for blooms in aquatic environments, methods are required that allow the rapid and reliable discrimination of specific cyanobacterial strains. Molecular methods have been applied to the identification and phylogenetic analysis of cyanobacteria.

A number of mini-methods for extraction of total genomic and plasmid DNA have been compared by using laboratory strains. The plasmid profile was not useful for establishing classification schemes because of limitations use in determining relationships between strains that contained plasmids. Polymerase chain reaction (PCR) based techniques involving random amplified polymorphic DNAs (RAPDs) and amplification of the 16S-23S ribosomal DNA (rDNA) spacer were optimised for selected cyanobacteria. RAPD profiles could be used to identify cyanobacteria to the strain level. Size heterogeneity and restriction fragment length polymorphisms (RFLPs) of the amplified rDNA spacer were useful in discriminating between different genera and species of cyanobacteria. Choice of primers proved critical to the use of spacer variation for species identification.

Variation was found between the rDNA spacer sequences of different genera and species. From sequence alignment two short (about 20 bp) high homology sequences were present, in addition to the transfer RNA genes of high sequence homology in the spacer region. Two primers designed from these high homology sequences were found to be cyano-specific and showed potential for use in the amplification of cyanobacterial DNA directly from environmental samples.

Cyanobacteria were isolated from water samples and identified, both on the basis of molecular characteristics (using the methods developed here) and morphological characteristics. Phylogenetic trees constructed on the basis of RAPDs, and of RFLPs and DNA sequences of the rDNA spacer separated the cyanobacteria into a number of clusters. The implications of these results for the phylogeny of the cyanobacteria are discussed. Chapter 1

General Introduction

1.1 Cyanobacteria

The cyanobacteria are one of the largest sub-groups of Gram-negative prokaryotes. They are photosynthetic prokaryotes which possess the ability to synthesise chlorophyll a and at least one phycobilin pigment, with water acting as the electron donor during photosynthesis and leading to the release of oxygen. Cyanobacteria are by far the largest group of photosynthetic prokaryotes, as judged by their widespread occurrence, frequent abundance, and morphological diversity. They are not only currently represented in most types of illuminated environment, except for those at low pH values, but they have one of the longest geological records (Schopf and Walter, 1982). Their fossil record is fragmentary and biased towards the formations where preservation of the morphology was possible. The earliest unicellular and filamentous forms attributed to the cyanobacteria were found in sedimentary rocks formed from the early Precambrian period, over three billion years old. At that time, much of the Earth's original atmospheric oxygen was probably formed by organisms quite similar to modern cyanobacteria (Schopf and Walter 1982, 1983; Knoll, 1985) and they are still responsible for a considerable proportion of photosynthetic oxygen evolution in the oceans. Thus cyanobacteria were probably responsible for a major evolutionary transformation leading to the development of aerobic metabolism and to the subsequent rise of higher plant and animal forms.

Cyanobacteria occupy a wide range of illuminated niches in freshwater, marine, terrestrial and hypersaline environments, where they often occur in such abundance that they are readily visible by eye. The colour of these forms may vary from green, blue-green, or olive-green to various shades of red to purple, or even black depending on pigments and light quality. In size the cyanobacteria range from sub-spherical cells, less than 1 μ m in diameter, to trichomes well over 100 μ m in diameter. The cells of the latter are always much shorter than wide, but nevertheless cell volume ranges over more than five orders of magnitude. Some species are capable of differentiating several distinct types of cell, and forms which combine these various cell types with various patterns of branching are morphologically the most complex prokaryotes known (Fay, 1983; Whitton, 1992).

Cyanobacteria are of considerable importance in the natural environment as initial colonisers of arid land and as primary producers of organic matter (Fay, 1983). Perhaps of greater importance is their role in the fundamental process of biological nitrogen fixation. Many of the free-living cyanobacteria and those engaged in symbiotic associations with plants and animals contribute significantly to the nitrogen fertility of aquatic and terrestrial habits, including cultivated lands, particularly in the tropics. Their usually high content of protein makes them prospective candidates for new, unconventional sources of animal and human food. Indeed, in the central African state of Chad dried cakes of *Spirulina*, which grows abundantly in soda lakes, constitute an important part of the diet of local people (Fay,1983; Rippka, 1988). In the recent search for new sources of energy, some researchers consider cyanobacteria as the most promising agents for the development of a biological solar energy conversion system, based on the concurrent light-driven generation of elemental O_2 and H_2 through the action of cyanobacteria.

The increase in planktonic cyanobacterial populations has sometimes been dramatic and has brought with it considerable practical problems. Dense populations present great difficulties to the water-supply industry, interfering with treatment processes and imparting tastes and odours to the water. The production of neurotoxins and hepatotoxins or the removal of oxygen by the respiration of living cells or by decomposition of dead cells can have serious consequences, causing human illness, death amongst fish, birds and occasionally cattle (Gibson and Smith, 1982; Carmichael, 1989, 1992). Cyanobacterial toxins that cause death in animals, birds and fish include the hepatotoxic heptapeptide microcystins and the pentapeptide nodularin, and neurotoxins including alkaloids and organophosphorus compounds (Lawton and Codd, 1991). Bathing in waters densely populated with *Aphanizomenon* can cause an unpleasant condition of the skin know as 'swimmers itch' (Schwimmer, 1964). Even without these unpleasant consequences, when the hitherto clear waters of lakes become clouded with plankton their aesthetic appeal is felt to diminish and the occurrence of surface scum (water-blooms) is still more objectionable.

1.2 Taxonomic classification

The first scientific descriptions of particular kinds of cyanobacteria did not appear until the early nineteenth century. The earliest description of the genus *Rivularia* was by Roth between 1797 and 1806, and Vaucher in 1803 described *Oscillatoria* and *Nostoc*. Vaucher placed these cyanobacteria in the animal kingdom, but nevertheless their subsequent study was undertaken mainly by botanists (Fogg, 1973). For about 150 years cyanobacteria were considered a special group of algae, the blue-green algae. According to the provisions of the International Code of Botanical Nomenclature, the name of a taxon above the family level is not subject to restrictions of priority. Consequently many names have been proposed for this group of microorganisms, as listed in Table 1.1 (Fogg, 1973; Rippka,1988).

Name	Year	Authors
Myxophyceae	1833	Wallroth
	1860	Stitzenberger
		Rabenhorst
Phycochromophyceae	1863	Rabenhorst
Cyanophyceae	1874	Sachs
Schizophyceae	1879	Cohn
Cyanophyta	1914	Pascher
	1938	Smith

Table 1.1 Major names of cyanobacteria according to the International Code of Botanical Nomenclature.*

* Taken from Fogg 1973, Rippka 1988

The name Cyanophyceae has survived the longest, and Cyanophyceae and Cyanophyta are still widely used by phycologists and botanists today. Bacteriologists believe that the name "Cyanobacteria", first proposed by Stanier (1978), is more appropriate than "Cyanophyceae", in view of the prokaryotic cellular properties of these phototrophs. Many names have been in circulation including Myxophyceae, Myxophyta, Cyanophyceae, Blue-green algae, Cyanophyta and Cyanobacteria. The term cyanobacteria will be used throughout this report.

The cyanobacteria have traditionally been treated as algae. They could be defined by broad morphological, cytological and less ecophysiological characters and separated into distinct division, classes, orders, and lower categories on the basis of morphological distinctions (Geitler, 1932; Anagnostidis and Komarek, 1985; Hoffmann, 1988). The complexity of simple multicellular and branched filaments and even multiseriate and thalliform "macrophytes" among the cyanobacteria would also lead most casual observers to the conclusion that these were algae, not bacteria, that were studied by botanists and phycologists. In nature, the cyanobacteria usually behave like algae. They possess chlorophyll a and phycobilins and perform oxygenic photosynthesis that is extremely similar to that of the red algae (division Rhodophyta). The morphological characteristics of the cyanobacteria and their ecological niches were so similar to those of many microalgae that few investigators doubted their place as the base or a branch of a true algae phylogenus (Fritsch, 1945). As early as 1853, Cohn indicated that these organisms were more related to bacteria and placed them in the kingdom prokaryota, division Schizophyta (cited in Castenholz and Waterbury, 1989). From early nineteenth century to the middle of the present century, cytological data was limited to impressions from light microscopy and only relatively crude chemical data was available. There were reasons to connect the blue-green algae with bacteria. However the evidence was not overwhelming, until electron microscopy and biochemical analyses showed convincingly that blue-green algae were unique and lacked the traditional nucleus and chloroplasts of the green and other algae. Recent nucleotide base sequence data from 16S and 5S rRNA place cyanobacteria within the Bacteria (Eubacteria), a group or domain distinct and apart from the Archaea (Archaeobacteria) and Eucarya (Eucaryotes) (Woese, 1987, 1994). For this

reason they should now be given extended treatment in the Bergey's Manual of Systematic Bacteriology. There are, however, considerable obstacles in the way of describing this group in the same manner as most other prokaryotes.

Cyanobacterial taxonomy has been approached in different ways, all with one major goal: defining those phenotypic properties that are taxonomically useful. Because the phenotype is the product of genetic and environmental factors, the stability of certain characteristics is a function of the properties of the environment. The development of an effective taxonomic system for cyanobacteria imposes problems for which there is no exact parallel with any other group of organisms since they have been treated under both the International Code of Botanical Nomenclature (as Blue-green algae) and the International Code of Bacteriological Nomenclature (as Cyanobacteria).

The building of the current morphology-based taxonomy of blue-green algae began in earnest with Bornet, Flahault and Gomont late last century (cited in Whitton, 1992). Under the International Code of Botanical Nomenclature, the starting point for the valid publication of names of heterocystous blue-green algae (family *Nostocaceae*) is with Bornet and Flahault, and that for the nonheterocystous, filamentous forms (family *Oscillatoriaceae*) is with Gomont (cited in Castenholz and Waterbury, 1989). The gathering of genera and species was continued, culminating with Geitler (1932), who produced a comprehensive treatise which recognized approximately 1300 species, 145 genera, 20 families, and 3 orders. This treatise, although regional in intent, in practice is worldwide, and gained the most widespread popularity among phycologists and bacteriologists. However, freshwater forms are more completely treated than marine forms, because

7

few marine forms had been studied. Other major gatherings of "blue-green" species on a regional basis are those by Tilden (1910), Desikachary (1959) and Starmach (1966), who reorganized the freshwater blue-greens at the familial and generic level. All followed a classification similar to that of Geitler (1932); this has become known as the "Geitlerian" system. In recent years, most "blue-green" specialists and nonspecialists have followed this organization. However, beginning with Drouet and Daily (1956), a major revision and consolidation of "blue-green" names was initiated (Drouet, 1968, 1973, 1978, 1981). The "Drouet" system, although gradually modified during the course of his work, resulted in an enormous reduction in the profusion of recognized generic and species names. Over 2000 species in over 140 genera were eventually reduced to 62 species in 24 genera. This system was not accepted by most "blue-green" taxonomists. However, several "blue-green" biochemists and physiologists have used the Drouet system because of its simplicity and the relative ease with which a binomial could be allocated. A series of papers provides detailed accounts of the Chroococcales (Komárek and Anagnostidis, 1986), Oscillatoriales (Anagnostidis and Komárek, 1988), Nostocales (Komárek and Anagnostidis, 1989), and Stigonematales (Anagnostidis and Komárek, 1990). The generic limits adopted tend to be quite narrow, leading to smaller and more homogeneous genera than in most other botanical floras. These papers provide the most detailed guides available to the morphological and ecological literature, considerable new insight into morphology, and accounts not only of well-known forms, but ones which are little known, though often widespread.

These and other more recent morphology-based systems have shown many improvements with time, but suffer to varying degrees from several problems. A large number of species and genera had been described over the years on the basis of a single difference, such as trichome width, presence or nature of sheath, number of trichomes per sheath, colour, and other features which later have been shown to have a great flexibility within clonal populations of some species, particularly when environmental conditions vary (Pearson and Kingsbury, 1966; Evans et al., 1976; Doers and Parker, 1988). However, some cyanobacterial clones in culture are quite conservative morphologically (Stam, 1978; Stulp, 1982; Stulp and Stam, 1982). Drouet (1981) presumed that relatively few cyanobacterial genotypes existed but that there were many phenotypes expressed in different environments (ecophenes). His decisions, however, were based mainly on his appreciation of the group through the microscopic examination of an extremely large number of collected specimens. No matter how much material is examined, decisions of this type are, of necessity, more arbitrary and less likely to reflect true relationship than decisions based on manipulations of clonal strains in culture. No clonal cultures were established by Drouet (1981); no physiological, biochemical or genetic information was included. The incorrectness of several decisions by Drouet (1981) has now been established by various means, including DNA/DNA hybridisation and DNA base composition comparisons, by Stam and associates (Stam and Holleman, 1975, 1979; Stam and Venema, 1977; Stam, 1980).

A second taxonomic approach was introduced by Stanier and his associates in the early 1970s (Stanier, 1970; cited in Rippka, 1988; cited in Whitton, 1992). This involved the long-term strategy of dealing with cyanobacteria in a similar way to other major groups of prokaryotes. The system is based on axenic, clonal cultures and the use of characters from a range of disciplines, rather than morphology alone; characters employed included: pigment composition, fatty acid analysis, chemoheterotrophic and photoheterotrophic capability and substrate sepecificity, nitrogenase activity, DNA base composition, and genome length (Kenyon, 1972; Herdman, 1979a,b; Rippka, 1979). Because the cyanobacteria are photoautotrophic organisms, physiological studies did not furnish many useful taxonomic characters. The basis of the bacteriological taxonomy of the cyanobacteria was published by Rippka *et al.* (1979). This taxonomic system, which still relies largely on morphology, allows the identification of the strains of the Pasteur Culture Collection at the generic level. The five major sections recognised by Rippka (1979) coincide broadly with orders of other classifications, as illustrated in Table 1.2. A modified version of this system is given by Castenholz, Waterbury and Rippka (1989). This treatment is more global and includes taxonomic information on well-known taxa which are not in the Pasteur Culture Collection, as well as ecological features.

The morphological and cytological data used by Stanier's group to distinguish genera were based on characteristics observable with light microscopy, except in the case of the presence or absence of thylakoids for which electron microscopy is required. The main characteristics include: the type of cellular division and the plane of subsequent divisions; the formation of baeocytes ("endospores" of the order blue-green algae terminology); the formation and structure of hormogonia; the presence and type of sheath or glycocalyx; the shape and dimensions of cells and the presence of constrictions between adjacent cells of trichomes; the nature of trichomes (helical or straight, unbranched, false branched or true branched, tapered or not); the presence and location of heterocysts and resting spores (akinetes); and the presence and location of gas vacuoles.

Table 1.2 Major diagnostic features (system of Geitler (1925, 1932), the revi	of cyanobacteria in the Sections described b ision by Anagnostidis and Komarek (1986, 19	y Rippka (19 988, 1989, 199	38), with the broadly equivalent orders following the botanical taxonomic 00), and Bergey's Manual of Systematic Bacteriology (1989)
Unicellular; cells single or forming colonial aggregates held together by additional outer cell wall layers	Reproduction by binary fission or by budding		Section I of Rippka et al. (1988) Orders Chroococcales + Chamaesiphonales sensu Geitler (1932) Orders Chroococcales sensu Waterbury and Rippka (1989)
Order C <i>hroococcales</i> sensu Komarek and Anagnostidis (1986)	Reproduction by multiple fission giving rise to small daughter cells (baeocytes), or by both multiple and binary fission		Section II of Rippka <i>et al.</i> (1988) Order <i>Pleurocapsales</i> sensu Geitler (1925), Waterbury (1989)
Filamentous; cells forming a multicellular unit, the trichome	Trichome always composed only of vegetative cells; reproduction only by intercalary division, followed by random trichome breakage	Division in only one plane.	Section III of Rippka <i>et al.</i> (1988) Order Oscillatoriales sensu Geitler (1932), Anagnostidis and Komarek (1988), Castenholz (1989)
	In the absence of combined nitrogen trichome contains both vegetative cells and heterocysts; reproduction occurs by intercalary binary fission and random trichome breakage; some	Division in only one plane	Section IV of Rippka <i>et al.</i> (1988) Order <i>Nostocales</i> sensu Geitler (1925), Komarek and Anagnostidis (1989), Castenholz (1989)
	of hormogonia or by germination of akinetes	Division in more than one plane	Section V of Rippka <i>et al.</i> (1988) Order <i>Stigonematales</i> sensu Geitler (1925), Anagnostidis and Komarek (1990) Castenholz (1989)

In view of the fact that the cyanobacteria have been researched by supporters of the "bacteriological" and "botanical" systems of classification, the problems for taxonomists are now largely practical ones. These two systems continue to use largely morphological criteria in description and distinction between genera. The main problem is the great array of existing genera and species that have been described solely on the basis of morphological characteristics obtained from fieldcollected samples, and the difficulty of relating these to living populations growing as clones in axenic culture. Some previous researchers already suspected that morphology may change depending on environmental conditions and the diversity of strains within a culture may be suppressed because of selective culturing conditions and period (Pearson and Kingsbury, 1966; Evans et al, 1976; Doers and Parker, 1988). Various approaches have been adopted to deal with these problems. There is a need for an extensive body of standardised information on field populations and laboratory strains, including morphological, cytological, ultrastructural, physiological and biochemical characters, which can be used for assessing similarity. The methodology employed in the bacteriological approach to cyanobacterial studies, however, opened the possibility to study cyanobacterial genotypes more directly. The new molecular genetics methodology has been successfully applied to cyanobacterial phylogenetic and taxonomic studies. Molecular methods such as DNA base composition (Herdman et al. 1979a), genome length (Herdman et al., 1979b), DNA-DNA hybridisation (Stam and Venema, 1977; Stam, 1980; Stam and Stulp, 1988) and restriction fragment length polymorphisms (RFLP) (Douglas and Carr, 1988; Wood and Townsend, 1990) have been applied to problems at lower taxonomic levels. The technique of partial 16S rRNA sequencing (Lane et al, 1985; Wilmotte, 1994), RFLP-PCR portions of rRNA genes (Vaneechoutte et al., 1992; Ward and Akrofi, 1994), and randomly amplified

polymorphic DNA (RAPD) markers (Williams *et al.*, 1990; Eskew *et al.*, 1993; Neilan, 1995) have also been applied to problems. These new applications provide the opportunity for a combined approach, using morphological criteria as a guide for molecular applications, in order to calibrate the molecular probes and optimise the selection of materials for study.

1.3 Molecular techniques used in taxonomy

The technology of genetic engineering has now become well established within the biological sciences. Molecular biology has provided new tools to decipher genetic information and can be used in attempts to reconstruct the evolution of organisms and determine their taxonomy. In the cyanobacteria, the use of molecular methods to study the genotypic relationships is underway, and initial results are promising (Wilmotte and Stam, 1984; Golden, *et al* 1989; Eskew, *et al*. 1993 and Wilmotte, 1994).

In its broadest sense the study of cyanobacterial taxonomy by molecular methods should include chemotaxonomic markers. Rapid methods for chemotaxonomy are generally available. However, data are relatively scarce and several studies are still preliminary (Wilmotte, 1994). Problems of consistency and variations, due to factors such as growth conditions, have not always been systematically investigated (Holton, 1981). Moreover, since the molecules employed are typically synthesised through complex pathways, their presence or absence can have different causes. The chemotaxonomic and macromolecular markers presented in Table 1.3 (concluded from Wilmotte, 1994) have been shown to be useful. However, genetic information obtained by using nucleic acid

Reference	Marker	Comments
Kenyon et al. (1972)	Lipid composition	A uniform fatty acid composition of 66 cyanobacterial strains was observed for the groups <i>Anabaena</i> and <i>Calothrix</i>
Sallar <i>et al.</i> (1990)	Lipid composition	A highly polar unknown glycolipid present only in the three heterocystous strains studied and alcohol glycosides are not restricted to nitrogen-fixing strains.
Merritt et al. (1991)	Lipid composition	Three marine, picoplanktonic <i>Synechococcuss</i> strains had a similar fatty acid composition to freshwater strain <i>Synechocystis sp.</i> strain PCC 6308
Caudales <i>et al.</i> (1992)	Lipid composition	The fatty acid compositions from free-living strains of the genera <i>Nostoc</i> and <i>Anabaena</i> and from symbionts of the water fern <i>Azolla</i> sp.
Hegewald and Knefal (1983)	Polyamines	Spermidine and sym-homospermidine were present in different strains assigned to the genera Oscillatoria, Phormidium, Calothrix and Chroococcus (Botanical taxonomy)
Hamana <i>et al.</i> (1983)	Polyamines	Higher concentrations of sym-homospermidine were present in nitrogen-fixing strains
Hamana and Matsuzaki (1992)	Polyamines	Polyamines as a chemotaxonomic marker in bacterial systematics
Hall (1982)	Aromatic amino acids	Biochemical diversity for biosynthesis of aromatic amino acids among the cyanobacteria
Healey (1968)	Carotenoids	Similar carotenoid patterns were observed for <i>Phormidium ectocarpi</i> strain PCC 7375 and <i>Phormidium persicinum</i> strain CCAP 1462/5

.

Table 1.3 contd		
Reference	Marker	Comments
Hertzberg et al. (1971)	Carotenoids	Variations of carotenoid patterns were probably useful for species identification.
Aakermann <i>et al.</i> (1992)	Carotenoids	The carotenoid content and composition was different in red or green isolates of the same species.
Klein et al. (1973)	Isozyme patterns	Shared α -esterase isozyme bands allowed the recognition of four clusters of related strains among the thirteen <i>Oscillatoriaceae</i> strains tested.
Schenk <i>et al.</i> (1973)	Malate-dehydrogenase patterns	Malate-dehydrogenase electrophoretic patterns were used to characterize eight cyanobacterial strains and shared bands were observed within the genera <i>Anabaena</i> and <i>Nostoc</i> , but <i>Synechococcus</i> sp. strain PCC 6301 and <i>Synechococcus elongatus</i> strain CCAP 1497/1, had no bands in common.
Stulp and Stam (1984)	Enzyme patterns	The electrophoretic patterns for five enzymes in axenic <i>Anabaena</i> sp. strains showed intra- and interspecific heterogeneity of enzymes in the strains tested.
Zimmerman and Rosen (1992)	Enzyme patterns	Zymograms for six enzymes were identical for twelve cyanobacterial symbionts from cycad
Philip et al. (1985)	Restriction randonucleases	Three site-specific endonucleases, <i>Af</i> II, <i>Af</i> III and <i>Af</i> III have been partially purified from the cyanobacterium <i>Anabaena flos-aquae</i> CCAP 1403/13f. <i>Af</i> III and <i>Af</i> III are new specificities.
Schenk and Kuhfitting (1983)	Phycobiliprotein	Phycobiliprotein patterns in twenty-one cyanobacterial species were studied. A conspicuous hetero- geneity was observed, even between strains assigned to the same species.
Guglielmi and Cohen-Bazire (1984)	Phycobiliprotein	Electrophoretic patterns of the phycobiliprotein subunits in <i>Pseudanabaena</i> were determined.
Table 1.3 contd		
---------------------------------	---------------------------	--
Reference	Marker	Comments
Bryant (1982)	Phycobiliprotein	Examined the entire Pasteur Culture Collection for strains capable of phycoerythrocyanin synthesis.
Ladha and Watanabe (1982)	Immunological proteins	Observed a high degree of antigenic similarity among cyanobacterial symbionts from different species of $Azolla$, but no crossreaction between the symbiots and free-living cultures.
Zilinskas and Howell (1987)	Immunological proteins	The antigenic determinants of two rod linkers polypeptide were very conserved in nine strains belonging to sections I, III, IV and V, whereas for a third polypeptide, only strains of section IV cross-reacted.
Bullerjahn <i>et al.</i> (1990)	Immunological proteins	The chlorophyll a/b-binding protein from <i>Prochlorothrix hollandica</i> was very similar to its counterpart in <i>Prochloron</i> sp., but not to light-harvesting chlorophyll a/b proteins of maize.

-

-

hybridisation, Polymerase Chain Reaction (PCR) - based techniques and rRNA or rDNA sequencing may provide the better tools to infer phylogenetic relationships. The genetic makers presented below have been shown to be worthy of further study.

DNA base composition is one of the few molecular characters that has been determined for almost 200 cyanobacterial strains (Herdman *et al.*, 1979a). It is however, a character marker. Large differences in DNA base composition indicate that the strains are not closely related, whereas similar Guanine + Cytosine (G+C) percentages give no clue concerning genotypic relationships (Stam and Venema, 1977; Wilmotte, 1994).

The method of DNA/DNA hybridisation allows the determination of sequence similarity between genomes of different organisms without the determination of exact nucleotide sequences. Generally DNA/DNA hybridisation work well for genomic comparisons at the species level (Stam and Venema, 1977; Stam, 1980; Stam and Stulp, 1988). It is based on the feature that denatured, single-stranded DNA will renature under suitable conditions. The extent of renaturation is proportional to the complementarity of the DNA strands involved. Furthermore, the stability of the renatured hybrid molecule is proportional to the accuracy of matching between the two strands. The thermal stability of a radioactively-labelled DNA hybrid can be established by cumulative plotting of the radioactivity eluted at each temperature against the elution temperature. The temperature at which 50% of the final eluted radioactivity is eluted is the thermal elution midpoint (Tm) for a hybrid. The difference between the Tm values of a homoduplex and a heteroduplex yields the Δ Tm. Low Δ Tm values indicate little mismatch in the heteroduplex, high Δ Tm values indicate much mismatch. For higher taxa the renaturation level rapidly

drops down to background levels; for lower taxons the method is insufficiently sensitive. The method has become particularly popular for genetic studies of prokaryotes, owing to the rather uncomplicated nature of prokaryotic genomes and the subsequent straight forwardness of the method (Stam and Venema, 1977; Stam, 1980; Stam and Stulp, 1988; Wilmotte, 1994). Several cyanobacterial genera have been studied by the use of this technique. DNA/DNA hybridisation results showing high relative hybridisation rates and low Δ Tm values, indicate that the cyanobacterial strains involved are genotypically closely related.

Wayne and colleagues (1987) proposed the use of DNA/DNA hybridisation as a criterion for the definition of eubacterial species. The phylogenetic definition of a species generally would include strains with approximately 70% or greater DNA-DNA relatedness and with 5 °C or less Δ Tm. The use of this method for cyanobacteria was pioneered by Stam (see Stam and Venema, 1977; Stam, 1980), who used a filter hybridisation technique. The genotypic relationships among twenty-nine filamentous strains which belonged to the LPP (*Lyngbya, Phormidium* and *Plectonema*) group (Rippka *et al.*, 1979) were determined. The results showed the genotypic homogeneity of a group of eighteen strains centred around the freshwater strain *Plectonema boryanum*. Other freshwater and marine LPP strains showed conspicuous genotypic differences. Gugielmi and Cohen-Bazire (1984) used DNA/DNA hybridisations to elucidate the relationships of nine strains belonging to the genus *Pseudanabaena* and the LPP group (Rippka *et al.* 1979).

Lachance (1981) investigated by DNA/DNA hybridisation the genotypic relationships among 45 strains of heterocyst-forming cyanobacteria assigned to eight genera by Rippka et al. (1979). The genera Nodularia, Cylindrospermum,

Chlorogloeopsis and *Fischerella* appeared to form tight genotypic groupings. On the other hand, the genera *Anabaena, Nostoc* and *Calothrix* contained strains or clusters of strains which showed little genotypic similarity. Four strains previously assigned to *Anabaena* appeared to belong to *Nostoc*. Stulp and Stam (1984) performed DNA/DNA hybridisation studies with twenty-one *Anabaena* sp. strains. Strains of the same species showed very high hybridisation percentages (about 100%). On the other hand, strains from different species showed intermediate hybridisation values (below 60%).

Wilmotte and Stam (1984) demonstrated by DNA/DNA hybridisation that Synechococcus PCC 7502 and PCC 6307 probably belong to mutually different genera. Strain 6312 is a member of the same genus as strain 6301 and two Anacystis nidulans strains PCC 7942 and PCC 7943 show high RB (relative binding) values and low Δ Tm values. This suggested that these strains belong to the same species as Synechococcus sp. strain PCC 6301. The result was later confirmed by restriction fragment length polymorphism (RFLP) analysis of the genomes of strains PCC 6301 and PCC 7942. The genomes of the two strains seemed identical, except for a rearrangement (Golden *et al.*, 1989; Wood and Townsend, 1990).

DNA/DNA reassociation studies have also been used to investigate the genotypic diversity of *Prochloron* sp. isolates from different didemnid species and locations. The fourteen *Prochloron* sp. isolates appeared to belong to a single species (Stam *et al.*, 1985; Holton *et al.*, 1990).

The RFLP technique for studying genomes is generally useful for identification and classification of organisms at the population or species level (Douglas and Carr, 1988; Wilmotte, 1994). By judicious choice of the restriction endonucleases used to cleave the DNA in a reproducible, site-specific manner, and of the hybridisation probes used to identify specific genes in the fragmented DNA, it is possible to detect fine structure differences in the genomes of organisms, differences which may pass undetected in bulk DNA-DNA hybridisation studies. This method has been productively used to study genomes of marine *Synechococcus* sp. strains and cyanobacterial symbionts.

The taxonomy of small, planktonic, phycoerythrin-containing Synechococcus sp. strains is problematic. Wood and Townsend (1990) tested the genetic homogeneity of a group of eight strains by using RFLP that showed cross-reactivity to the antiserum directed against Synechococcus sp. strain WH 7803. In the tree topology obtained, the eight strains were distributed into four different branches. Moreover, the genetic distances between terminal taxa on different branches were as large within the serogroup as the distances found between members of the serogroup and two freshwater Synechococcus sp. strains that showed no cross-reactivity to the antiserum. This conclusion was confirmed by an analysis of the previously published results of Douglas and Carr (1988), no marine Synechococcus isolate showed any similarity to the freshwater Synechococcus strain PCC 6301.

A filamentous, nitrogen-fixing symbiont lives in the leaf cavities of the water fern *Azolla* sp. and is generally identified as *Anabaena azollae*, although it is not clear whether one or more taxa are involved. The symbiont and its host are associated during all the life cycle. In total, the symbioses involving seven *Azolla* sp., belonging to the Sections Euazolla and Rhizosperma, have been studied. The probes were derived from heterologous genes (e.g. *nif*K, *nif*D, *nif*H, *nif*DH, 1.8kb

pAn207.3) from Anabaena sp. strain PCC 7120 (Franche and Cohen-Bazire, 1987; Meeks *et al.*, 1988) or from genomic and plasmid DNA clones of freshly isolated symbionts (Plazinski *et al.*, 1990, 1991). The results have shown that (i) the leaf cavities of Azolla sp. probably harboured one major symbiont, accompanied by minor species; (ii) there is a good correlation between the genotypic relationships among the freshly isolated symbionts and the classification of their hosts. The division into the Sections Euazolla and Rhizosperma was confirmed, except for the symbiont from Azolla nilotica which showed genotypic differences with the symbionts from both Sections; (iii) Only the minor constituents of the symbiosis could be established as free-living cultures.

Plazinski and colleagues (1990) found two DNA probes (pAm3 and pAc51) of which were in fact derived from plasmid sequences (Plazinski *et al.*, 1991), are strain-specific and hybridised only to *A. azollae* strains isolated from *Azolla microphylla* or *Azolla caroliniana*. In addition, three probes (pAc34, pAc50, and pAc60) could discriminate among symbionts from the same *Azolla* sp. collected in different geographical areas. The free-living *Anabaena* sp. strain PCC 7120 shared little similarity with the studied symbionts, but *Nostoc cycas* strain PCC 7422 seemed closely related to them.

The taxonomic affiliation of the symbionts to the genera Anabaena and Nostoc was also investigated. Using as taxonomic marker the presence of two conserved restriction enzyme sites in the nitrogen-fixing (nif) genes of four Nostoc sp. strains and their absence in Anabaena sp. strain PCC 7120 and Anabaena variabilis strain ATCC 29413 (PCC 7937), Meeks and colleagues(1988) proposed that the symbionts belonged to the genus Nostoc, rather than Anabaena sensu

Rippka *et al.* (1979). However, the delimitation of the two genera was not clear and the two *Anabaena* strains used as reference by the authors were later assigned to *Nostoc* (Caudales and Wells, 1992). Thus, the data from Meeks *et al.* (1988) only served to illustrate the genotypic variability existing within the genus *Nostoc*, and did not allow classification of the symbionts.

Three short tandemly repeated repetitive (STRR) sequences from the genome of heterocystous cyanobacteria *Calothrix* sp. strain PCC 7601 have been used to distinguish between cyanobacterial strains. Mazel and colleagues (1990) suggested that the banding patterns could be used to distinguish strains at the species and genus level. The three different STRR sequences were found as repetitive genomic DNA components specific to the heterocystous strains tested, and the hybridisation patterns also indicated that the strain '*Anabaena*' PCC 7120 belonged to the genus *Nostoc* on the basis of hybridisation patterns with highly repetitive DNA sequences (Mazel *et al.*, 1990). It is noteworthy that repetitive elements have also been used to generate species- and strain-specific fingerprints in eubacteria (Versalovic *et al.*, 1991). The taxonomic value of two insertion elements (IS701 and IS702) from *Calothrix* sp. strain PCC 7601 was investigated. The elements appeared to be restricted to the identification of identical strains (Mazel *et al.*, 1991).

Genetic analysis using random amplified polymorphic DNA markers (RAPDs) described by Williams *et al.* (1990), has permitted genetic mapping and diagnostics, and taxonomic studies. The RAPD technique is based on random DNA amplification by PCR with single, short primers of arbitrary nucleotide sequence. The assay is non-radioactive, requires only nanogram quantities of DNA, and is applicable to a broad range of species. Using genomic DNA, a set of DNA fragments is generated that can be resolved by electrophoresis to form a DNA banding pattern (DNA fingerprint). In studies with cyanobacteria, Eskew and colleagues (1993) used this method to study the cyanobacterial symbionts of *Azolla ferna*. DNA fingerprints (DAF) were generated that were unique to the symbionts from three different *Azolla* sp. and that showed the maternal transmission of one symbiont. The method is very promising for strain identification. Neilan (1995) demonstrated by using multiplex (more than one primer per PCR-amplification) randomly amplified polymorphic DNA markers (MRAPDs) that there is a clear distinction between the genera *Anabaena* and *Microcystis*. It was possible to discriminate among all toxigenic cyanobacteria studied to three taxonomic levels of genus, species and strain using MRAPD markers.

PCR technology has also been used in several recent studies as a rapid means of amplifying the 16S-23S spacer regions of ribosomal DNA (rDNA) for the specific purpose of detecting heterogeneity between and within species (Barry *et al.*, 1991; Jensen *et al.*, 1993). The bacterial rRNA operon is transcribed into one prerRNA transcript that contains the following components in the order (5' to 3'): 16S, spacer, tRNA, spacer, 23S, spacer and 5S rRNA sequences (Watson *et al.*,1987; see Fig 1.1). Variability in both the number and the length of the spacers raises the possibility that this methodology has broad applicability as a rapid, possibly automatable, general method for bacterial identification and typing (Gurtler and Stanisich, 1996).

For a rapid taxonomic survey, PCR-amplified portions of rRNA genes may be digested by restriction enzymes. This kind of RFLP analysis has been used with success for eubacterial taxonomy (Vaneechoutte *et al.*, 1992). Ward and Akrofi



Scale: 200bp

Figure 1.1. Schematic diagram of ribosomal DNA repeat showing internal transcribed spacer region (ITS). Transcription of the operon is from left to right. Some cyanobacteria have two tRNA genes as shown; others just have tRNA^{ile}.

(1994) used this method to identify the fungi in the *Gaeumannomyces-Phialophora* complex. In studies with cyanobacteria, Neilan and colleagues (1995) used RFLP analysis of the PCR-amplified phycocyanin locus to study toxigenic cyanobacteria. The assay was appropriate for the rapid identification of strains in freshwater samples and clearly discriminated among the eight genera (40 strains) tested. Bolch and co-workers (1996) investigated the genetic characterisation of the β - and α -phycocyanin subunits (*cpc* BA) intergenic spacer and flanking regions among 19 strains of three morphospecies of cyanobacteria by using PCR-RFLP. The presence of genetic variation between strains within these three divergent morphospecies, which span two orders of cyanobacteria (Chroococcales and Nostocales), may provide a versatile tool for discriminating cyanobacteria at a variety of taxonomic levels.

Information on the relatedness of phylogenetically more distant taxa can also be obtained by nucleotide sequencing of RNA and DNA. rRNA genes are at present the most useful molecular chronometers. They show a high degree of functional constancy, which assures relatively good clock-like behaviour (Woese,1982). They occur in all organisms, and different positions in their sequences change at very different rates, allowing most phylogenetic relationships (including the most distant) to be measured, which makes their range all-encompassing. Their sizes are large and they consist of many domains. There are about 50 helical stalks in the 16S rRNA secondary structure and roughly twice that number in the 23S rRNA (Noller, 1984; Gutell *et al.*, 1985), which makes them accurate chronometers on two counts. rRNAs can also be sequenced directly (Lane *et al.*, 1985). A technique of partial 16S rRNA sequencing was devised (Lane *et al.*, 1985), and applied to more than forty cyanobacterial strains (Tomioka and Sugiura, 1983; Giovannoni *et al.*, 1988, 1990; Turner et al., 1989; Britschgi and Giovannon, 1991; Ligon et al., 1991; Weller et al., 1991, 1992; Nelissen et al., 1992; Urbach et al., 1992; Wilmotte et al., 1992,1993; Neilan, 1994).

The rRNA gene sequence analyses and the phylogenetic reconstructions based on these analyses have proved useful in assessing a wide range of phylogenetic relationships including those between higher taxonomic categories. The 16S rRNA has been shown to be an adequate macromolecule to study the evolution of the eubacteria. According to Woese (1987), the 16S rRNA is a good 'molecular chronometer' measuring the overall rate of evolutionary change in a line of descent. In studies with cyanobacteria, more than 50 strains have been studied and grouped into eight branches (Wilmotte, 1994). On the other hand, 16S rRNA gene sequences are probably too conserved to investigate intraspecies variability (Ward, *et al.*, 1992). The sequence of the internal transcribed spacer (ITS), situated between the 16S rRNA and the 23S rRNA gene (Fig. 1.1), may be an adequate marker. Species-specific probes inferred from the ITS sequence have been used for eubacterial identification (e.g Rossau *et al.*, 1992).

1.4 Aims

The cyanobacteria are a diverse and widespread group of micro-organisms that are responsible for substantial productivity in both open ocean and freshwater courses (Fogg *et al.*, 1973). Problems with toxic blooms in British freshwater lakes (Carmichael, 1989) have given increasing impetus to studying the organisms involved. Rapid and reliable methods for identification of cyanobacteria are a necessary prerequisite for such studies. This project proposes to develop rapid molecular technology for detection and identification of cyanobacteria. Rapid molecular methods based on plasmid typing, RAPD-PCR, PCR-amplification and RFLP analysis of the rDNA spacer, and rDNA sequencing will be developed and evaluated. Free living cyanobacteria will be isolated from water courses in Merseyside, Wales, Cumbria and China where cyanobacterial growth has been a problem in recent years. The organisms will be grouped by classical taxonomic procedures, such as those described by Rippka *et al.* (1979) and will be identified by the molecular techniques developed here. Comparisons will be made of molecular and traditional methods for identifying and classifying cyanobacteria.

The aims of this work are:

1. The use of morphological criteria for identification of free-living strains, which includes the isolation of other new organisms from local water courses and the determination of their status by morphological taxonomic methods.

2. To develop and select methods for recovery and characterisation of plasmids and total genomic DNA from cyanobacteria.

3. To develop and optimise rapid PCR-based methods: RAPD-PCR, RFLP-PCR and rDNA sequence to identify and classify different cyanobacteria.

4. To store and catalogue data from RAPDs, RFLP-PCR and rDNA sequencing and to analyse by using appropriate computer software. Dendograms will be constructed to indicate relationships among the laboratory strains and cyanobacterial isolates.

5. To compare molecular and traditional methods for identifying cyanobacteria and consider taxonomic implications.

6. To assess the possibility of designing cyano-specific PCR primer/probes derived from aligned 16S-23S rDNA spacer region sequences with a view to direct identification of cyanobacteria from water samples.

A flow diagram summarising the schedule of work is shown in Fig. 1.2

Figure 1.2. Flow diagram summarising the schedule of work:



Chapter 2

.

.

.

.

.

Materials and Methods

2.1 Materials

2.1.1. Chemicals

Unless otherwise stated, the chemicals used were obtained from either British Drug Houses Chemicals Ltd. (BDH, Poole, Dorset, U.K.) or the Sigma Chemical Company (Poole, Dorset, U.K.). The chemicals were of the highest analytical grade available. Tag DNA Polymerase was obtained from PE Applied Biosystems (Warrington, U.K.), Advanced Biotechnologies Ltd. (Surrey, U.K.), Promega Ltd. (Southampton, U.K.), Bioline (London, U.K.) and MBI Fermentas Molecular Biology (U.K.). DynaZyme II polymerase was obtained from Finnzymes OY (Espoo, Finland). Oligonucleotide primers for conserved rRNA sequences were obtained from PE Applied Biosystems (Warrington, U.K.), random primers were obtained from Advanced Biotechnologies Ltd (Kit1 1-20 and Kit A 1-20); dATP, dCTP, dGTP and dTTP were obtained from Bioline (U.K.). Restriction enzymes were obtain from Sigma (UK) or Boehringer Mannheim (Lewes, UK). Wizard PCR preps DNA and minipreps DNA purification system, and pGEM-T vector were obtained from Promega (UK). PCR II vector was obtained from Invitrogen (Oxford, UK). Sequenase 2.0 sequencing kit (USB) was obtained from Amersham (Little Chalfont, UK). ABI PRISM dye terminator cycle sequencing kit was obtained from PE Applied Biosystems (U.K.). Sequagels was obtained from National Diagnostics (U.K.). Molecular Markers were obtain from Advanced Biotechnologies Ltd. (1kb ladder -- 0.25, 0.5, 0.75, 1.0, 1.5, 2.0, 2.5, 3.0, 4.0, 5.0, 6.0, 8.0 and 10.0 Kb) or Sigma (123 bp ladder -- 123, 246, 369, 492, 612,bp) or Boehringer Mannheim (Lambda Hind III -- 23.1, 9.4, 6.6, 4.4, 2.3, 2.0, 0.6, 0.1 kb; pBR32 -- 4.36 kb).

2.1.2. Cyanobacterial Strains

Table 2.1 describes the cyanobacterial strains used.

2.1.3. Media

The composition of media used for growth of cyanobacteria is given in Table 2.2. Preparation of media is described in Appendix A.

The composition and preparation of buffers used for this study are given in Appendix B.

2.2. Methods

2.2.1. Growth and Maintenance of Cyanobacteria

Strains were cultured on agar medium (15ml medium per plate) or liquid medium (50, 125, 500ml medium in 100, 250, 1000ml flask, respectively), and incubated with shaking at 100 rpm in a GALLENKAMP rotary illuminated incubator (U.K.) which was continuously bubbled with CO_2 , at 20-30^oC (standard condition) autocycle under continuous light with two warm-white fluorescent (OSRAML, 30W, Germany) lights. Cultures were stored in 25% (v/v) glycerol appropriate liquid medium at -80^oC for long term maintenance. For routine use they were maintained on BG11 (Table 2.2) slopes at 10 ^oC-15 ^oC in the dark.

Strain	Source ^ª	Site	Reference	Growth condition	Medium ^b
Unicellular					
1. Chroococcidiopsis JM013	TIMU	Tai lake, Jiangsu, China	Lu (this study)	100µmol photon m ⁻² s ⁻¹ , 30°C	BG11, C
2. Chroococcus prescottii	BSM			100µmol photon m ⁻² s ⁻¹ , 30°C	BGII
3. Gloeocapsa sp.	BSM			60 μmol photon m ⁻² s ⁻¹ , 25°C	BG11
4. Microcystis aeruginosa	BSM			60 μmol photon m ⁻² s ⁻¹ , 25°C	BG11
5. Synechococcus PCC 7942	TIMU	California, U.S.A (Floyd 1973)	Grigorieva and	100µmol photon m ⁻² s ⁻¹ , 36°C	BG11
			Shestakor 19/0		
6. Synechococcus CCAP1479/7	LU			100μmol photon m ⁻² s ⁻¹ , 25°C	BG11
7. Synechococcus WH7803	ΓΩ			100μmol photon m ⁻² s ⁻¹ , 25°C	MN
8. Synechococcus JM 002	LJMU	Raby pool, Wirral, England	Lu (this study)	60μmol photon m ⁻² s ⁻¹ , 25°C	BG11
9. Synechococcus JM003	LIMU	Raby pool, Wirral, England	Lu (this study)	60µmol photon m ⁻² s ⁻¹ , 25°C	BG11
10. Synechococcus JM004	TIMU	Raby pool, Wirral, England	Lu (this study)	60μmol photon m ⁻² s ⁻¹ , 25°C	BG11
11. Synechococcus JM020	TIMU	Windmere, Cubria, England	Lu (this study)	60μmol photon m ⁻² s ⁻¹ , 25°C	BG11
12. Synechocystis PCC6803	NW	California, U.S.A	Stanier (1971)	100μmol photon m ⁻² s ⁻¹ , 30°C	BG11
Filamentous non heterocystous					
13. Leptolyngbya JM012	LIMU	Tai lake, Jiangsu, China	Lu (this study)	100µmol photon m ⁻² s ⁻¹ , 30°C	BG11, C
14. Leptolyngbya JM014	LJMU	Tai lake, Jiangsu, China	Lu (this study)	100μ mol photon m ⁻² s ⁻¹ , 30° C	BG11
15. Oscillatoria animalis	BSM			100μ mol photon m ⁻² s ⁻¹ , 30° C	BG11
16. Phormidium autumnale	BSM			100μ mol photon m ⁻² s ⁻¹ , 30° C	BG11, C
17. Phormidium JM017	LJMU	Bala lake, North Wales	Lu (this study)	100μ mol photon m ⁻² s ⁻¹ , 30° C	BG11
18.Pseudanabaena JM001	LJMU	Raby pool, Wirral, England	Lu (this study)	100μmol photon m ⁻² s ⁻¹ , 30°C	C, BG11
19.Pseudanabaena JM005	LJMU	Raby pool, Wirral, England	Lu (this study)	100 μ mol photon m ⁻² s ⁻¹ , 30°C	C, BG11
20.Pseudanabaena JM006	LJMU	Raby pool, Wirral, England	Lu (this study)	100μmol photon m ⁻² s ⁻¹ , 30°C	C, BG11
21. Pseudanabaena JM007	LIMU	Raby pool, Wirral, England	Lu (this study)	100μmol photon m ⁻² s ⁻¹ , 30°C	C, BG11
22.Pseudanabaena JM009	LJMU	Tai lake, Jiangsu, China	Lu (this study)	100μmol photon m ⁻² s ⁻¹ , 30°C	C, BG11

Table 2.1. Cyanobacterial strains used

Strain	Source	Site	Reference	Growth condition*	Medium
23.Pseudanabaena JM011	LJMU	Tai lake, Jiangsu, China	Lu (this study)	100µmol photon m ⁻² s ⁻¹ , 30°C	BGI1
24.Pseudanabaena JM016	LJMU	Bala lake, North Wales	Lu (this study)	100μmol photon m ⁻² s ⁻¹ , 30°C	BG11
25.Pseudanabaena JM019	LJMU	Windmere, Cubria, England	Lu (this study)	100μmol photon m ⁻² s ⁻¹ , 30°C	BG11
Filamentous heterocystous					
26. Anabaena PCC 7120	UW		Adolph & Haselkorn, 1971	100µmol photon m ⁻² s ⁻¹ , 30°C	BGI1
27. Anabaena ATCC27892 (PCC7118)	ΓΩ		Kratz & Myers, 1955	100μmol photon m ⁻² s ⁻¹ , 30°C	BG11
28. Anabaena ATCC27892 R 3.2	ГU			100µmol photon m ⁻² s ⁻¹ , 30°C	BG11
29. Anabaena cylindrica	UW	pond, Cambridge, England (Chu, 1939)	Fogg, 1942	100µmol photon m ⁻² s ⁻¹ , 30°C	BG11
30. Anabaena cylindrica AN 23	LJMU	pond, Cambridge, England	Lu (this study)	100µmol photon m ⁻² s ⁻¹ , 30°C	BG11
31. Anabaena flos-aquae D0848	an		isolated by Walsby, 1991	60 μmol photon m ⁻² s ⁻¹ , 25°C	AD P(1.0)Fc(0.4)
32. Anabaena sp. D0697	an	Aldabra cinq cases pool, Seychelles	Whitton (isolated), 1984	100μmol photon m ⁻² s ⁻¹ , 25°C	AD P(1.0)Fe(0.4), BG11
33. Anabaena sp. D0746	GU	Huntra dwr(pin gaew)field, Thailand	Catling (isolated), 1984	100μmol photon m ⁻² s ⁻¹ , 25°C	CHU10D-NBG11
34. Anabaena sp. D0752	an	Huntra dwr(pin gaew)stem, Thailand	Catling (isolated), 1983	100µmol photon m ⁻² s ⁻¹ , 25°C	CHU10D-N, BG11
35. Aphanizomenon gracile	BSM			60 μmol photon m ⁻² s ⁻¹ , 25°C	BG11
36. Chlorogloeopsis fritschii	LJMU	Indian soil	Mitra and Pandey, 1966	100µmol photon m ⁻² s ⁻¹ , 30°C	C, BG11
37. Fischerella musicola	BSM			100μmol photon m ⁻² s ⁻¹ , 25°C	BGI1
38. Gloeotrichia sp.	UW			100µmol photon m ⁻² s ⁻¹ , 30°C	BG11
39. Nostoc commune	BSM			100µmol photon m ⁻² s ⁻¹ , 25°C	BG11
40. Nostoc commune	ß	Scotland	isolated by Gibson	100μmol photon m ⁻² s ⁻¹ , 25°C	ACM, BG11
41. Nostoc ellipsospora	UW			100μmol photon m ⁻² s ⁻¹ , 30°C	BG11
42. Nostoc linckia D0734	đŋ	Huntra dwr(pin gaew)field, Thailand	Catling (isolated), 1983	100µmol photon m ⁻² s ⁻¹ , 25°C	CHU10D-N, BG11
43. Nostoc MAC PCC8009	UW		Bowyer & Skerman, 1968	100μmol photon m ⁻² s ⁻¹ , 25°C	BG11
44. Nostoc MAC R1	ГU			100μmol photon m ⁻² s ⁻¹ , 25°C	BG11
45. Nostoc muscorum	BSM			100μmol photon m ⁻² s ⁻¹ , 25°C	BG11
46. Nostoc piscinale D0201	đ	Aldabra atoll, Seychelles	Donaldson (isolated), 1972	100µmol photon m ⁻² s ⁻¹ , 25°C	AD P(1.0)Fe(0.4), BG11
47. Nostoc sp. D0766	QIJ	Sonargaon dwr field, Bangladesh	Kelly (isolated), 1985	100μmol photon m ⁻² s ⁻¹ , 25°C	CHU 10D-N, BG11

_
2
e)
F
્રંત્વ
L_
÷.
g
2

Strain	Source	Site	Reference	Growth condition*	Medium
48. Nostoc planctonicum	BSM			100µmol photon m ⁻² s ⁻¹ , 30°C	BG11
49. Nostoc sp. D0740	ß	Huntra dwr(pin gaew)roots, Thailand	Catling (isolated), 1984	100μmol photon m ⁻² s ⁻¹ , 25°C	CHU 10 D-N, BG11
50. Nostoc pelaeo	NCL			100µmol photon m ⁻² s ⁻¹ , 36°C	BG11
51. Tolypothrix distorta	BSM			100 μ mol photon m ⁻² s ⁻¹ , 30°C	BG11

a LJMU, UW, BSM, LU, UD,UB,UCL, Liverpool John Moores University, University of Warwick, Biobred Sciento Manchester, Lancaster University, University of Durham, University of Bristol, University of Central Lancashire, respectively.

b Media are described in Table 2.2

.

•

.

.

-

•

Component ^a (mM)				Medium	þ			
	ASM-1	BG11	BG11/10	С	CHU 10D	ACM A	AD P(1) Fe(0.4)	MM
NaCl						0.393	3.936	
NaNO ₃	2.00	17.65	1.765	ł	ł	1		1
KCI	·I			ł	1	1	0.256	1
KNO3	ł	ł		9.89	ł	4.945	ł	8.8
KH2PO₄	ļ	1	ł	ł	0.057	I	ł	ł
K ₂ HPO ₄	0.10	0.18	0.18	0.574	ļ	0.057	0.032	0.09
Na ₂ HPO ₄	0.10				ł	1	ł	
$MgSO_4$	0.20	0.30	0.30	1.01	0.101	0.985	0.811	0.15
MgCl ₂	0.20	ł	ł	l		1		I

Table 2.2. Media for growth of cyanobacteria

.

.

Component ^a (mM)				Medium	9			
	ASM-1	BG11	BG11/10	C	CHU 10D	ACM A	AD P(1) Fe(0.4)	WN
CaCl ₂	0.20	0.25	0.25		1	0.135	0.45	0.12
Ca(NO ₃) ₂	1		ł	0.11	0.244	ł	ł	ł
NaHCO ₃	ł	ł		i I	0.189	ļ	ł	ł
Na ₂ CO ₃	ł	0.19	0.19	I	ł	ł	1	0.19
Disodium EDTA	0.020	3		ł	0.009	0.0013	ł	I
Disodium-magnesium EDTA	ł	0.003	0.003	ł		ł	0.007	0.0013
Citric acid	ł	0.029	0.029					0.014
Trisodium citrate		ł	ł	0.561	ł	ł	1	ł
Ferric ammonium citrate	ł	0.03	0.03	ł	ł	•	ł	0.015
FeCl ₃	0.004	L L T		1	0.009	0.0045	0.0072	1
$Fe_2(SO_4)_3$	8	ł	1	0.008	ł	1	ł	ł

Component ^a (mM)				Medium	4			
	ASM-1	BG11	BG11/10	C	CHU 10D) ACM	AD P(1) Fe(0.4)	MN
Na ₂ SiO ₃ 5H ₂ O					0.013			
H ₃ BO ₃ (μM)	40.00	46.00	46.00	46.00	11.5	46.00	46.00	46.00
$MnCl_24H_2O$ (μM)	7.00	9.20	9.20	9.20	2.30	9.10	9.10	9.20
ZnSO ₄ 7H ₂ O (μM)	3.20	0.77	0.77	0.77	0.2	0.77	0.77	0.77
Na ₂ MoO₄2H ₂ O (μM)	1	1.60	1.60	1.60	0.03	1.60	1.60	1.60
$CuSO_45H_2O$ (µM)	0.0008	0.32	0.32	0.32	0.08	.032	0.32	0.32
Со(NO ₃) ₂ (µM)	0.080	0.17	0.17			0.16	0.16	ł
CoSO4 7H2O	ł	ł	ł	1	0.04	ł	l	1
NiSO4 7H2O	ł	-	;	ł	1	0.16	0.16	ł
Sea water	ł	ł	1	ł	E 1 3		ł	75% v/v
a Components are given as mM unless otherwise sta	ated. BG11 and BG	11/10 from 1	() [] of al (]	979). C fr.	m Carr (19	AD and	ACM from Whitton (herconal

ALINI LIVIII WILLION (DETSONAL 1/10 ITOM KIPPKa, et al. (19/9); C ITOM CAIT (1909); AD and כ Ā b ASM-1 from Gorham, et al. (1964); CHU, MN, BG11 and communication).

2.2.2. Description of the sampling sites

In recent years, cyanobacterial growth has been a problem in water courses in the United Kingdom and Peoples' Republic of China. Freshwater lakes in Merseyside, North Wales and Lancashire in U.K., and Jiangshu Province of China were chosen as sampling regions for this study.

(I) United Kingdom

a) <u>Merseyside</u>

Raby pool; Wirral (Plate 1) 53°19' N, 03° W

Sampling date: 19th June 1994

Sampling was undertaken at a freshwater pool near Raby Hall (Fig 2.1, 2.2), Wirral, Northwest England. (Details of sampling, see section 2.2.3, Table 2.3).

b) North Wales

Bala Lake; Southern Snowdonia (Plate 2) 52°54' N, 03°36' W

Sampling date: 19th August 1996

Sampling was undertaken at a freshwater lake near Bala (Fig.2.1, 2.3), Southern Snowdonia, North Wales. (Details of sampling, see section 2.2.3, Table 2.3).

c) <u>Cumbria</u>

Bowness, Windermere Lake (Plate 3) 54° 23' N, 02 ° 55' W



Figure 2.1. Map of the United Kingdom Showing collecting sites







Figure 2.4. Map of Windermere lake showing sampling location



Figure 2.5. Map of China showing sampling site



Figure 2.6. Map of Taihu lake showing sampling location



Plate 1 A view of the Raby pool which was one of the freshwater sampling sites in this study.



Plate 2 A view of the Bala lake which was selected for freshwater sampling in this study.



Plate 3 A view of the Bowness, Windermere Lake which was one of the freshwater sampling sites in this study.



Plate 4 A view of the Liyuan Garden, Taihu lake which was selected for freshwater sampling in this study.

Sampling date: 24th July 1996

Sampling was undertaken at the freshwater lake Windermere (Fig. 2.1, 2.4), Lake District, Northwest England. (Details of sampling, see section 2.2.3, Table 2.3).

(II) Peoples' Republic of China

Taihu Lake was chosen as sampling site in Jiangshu province of P.R. China.

Liyuan Garden, Wuxi, Jiangshu (Plate 4) 31°22' N, 120°30' E

Sampling date 1st October 1995

The sampling was undertaken at the freshwater lake Taihu (Fig. 2.5, 2.6), Wuxi, Jiangshu province of China. (Details of sampling, see section 2.2.3, Table 2.3).

2.2.3. Sampling

Twelve fresh-water samples (approximately 50ml each) were collected at random from the surface layer of water (i.e. the top 2-6cm), around each site within an approximate $500m^2$ area. Figures 2.2, 2.3, 2.4 and 2.6 show the sampling location of the lakes and the pool respectively, and Table 2.3 shows the location and type of water samples. Samples were transported to the laboratory in sterile glass bottles in the dark and analysed immediately. Samples were maintained in the dark at $15-20^{\circ}C$.

Sample	Location	Description of the sample
Number		
1	Raby pool, light-niche, stone-stair	Clear water, pH 6.8
2	Raby pool, dim-niche, wood-bottom	Clear water with rotten leaves,
		рН 6.9
3	Raby pool, dim-niche, wood-bottom	Clear water with rotten leaves,
		рН 6.9
4	Raby pool, light-niche, broken stone	Clear water, pH 6.8
	bottom	
5	Raby pool, light-niche, broken stone	Clear water, pH 6.8
	bottom	
6	Liyuan garden, light-niche, mud	Light green bloom water,
	bottom, stone bank	organic pollution, pH 7.2
7	Liyuan garden, light-niche, mud	Light green bloom water,
•	bottom, stone bank	organic pollution, pH 7.2
8.	Lake at Bowness on Windermere,	Clear water, pH 6.8
	dim-niche, mud bottom	
9	Lake at Bowness on Windermere,	Clear water, pH 6.8
	dim-niche, mud bottom	
10	Lake at Bowness on Windermere,	Clear water, pH 6.8
	light-niche, broken stone bottom	
11	Lake at Bowness on Windermere,	Clear water, pH 6.8
	light-niche, broken stone bottom	
12	Bala Lake near car park, light-niche,	Clear water, pH 7.0
	broken stone and mud bottom	

Table 2.3. Location and type of water sample

•

.

.

•

2.2.4 Isolation of cyanobacteria

(a) *Direct isolation* 0.1ml water from each sample was spread with a sterile glass spreader on to the following agar medium: C, BG11 and ASM-1.

(b) *Indirect isolation* 1.0 ml water samples from each of the sampling bottles were inoculated in to 50ml of sterile mineral liquid medium (C, BG11 and ASM-1) in a 100ml flask and incubated until growth was visible. The culture was then streaked on to the appropriate agar plates (as per liquid medium).

All the plates and flasks were incubated aerobically under two conditions: (1) In the laboratory, at 15-20^oC, under a daylight light-dark cycle. (2) In the rotary light incubator under standard conditions (see section 2.2.1).

Putative cyanobacterial colonies growing on the surface of the agar plates of appropriate medium were transferred to appropriate fresh liquid medium, and incubated in GALLENKAMP illuminated incubator under standard conditions (see section 2.21).

2.2.5 Recognition of cyanobacterial isolates

Initial identification of the cyanobacterial isolates was on basis of their characteristic colour. The UV-vis absorption spectra of putative cyanobacteria liquid cultures (see section 2.2.4) were measured (against a blank which was strain free medium) from 400nm-720nm using a KONTRON UVIKON-930 spectrophotometer. Absorption peaks were detected for chlorophylls and phycobilins. Samples were also examined under the light microscope (UNIL UX-11) for the absence of the nucleus. The cyanobacterial isolates were confirmed by light-induced O₂-evolution (see 2.2.6) and determination of pigment content (see 2.2.7). The cyanobacterial isolates were then transferred to appropriate agar medium, and culturing and maintenance were as described in section 2.2.1.

2.2.6 Measurement of O₂-evolution

A 2ml liquid culture of cyanobacteria, obtained as in section 2.2.5, was harvested by centrifugation at 6000g for 5min at 4°C in a MSE microcentrifuge. The cells were resuspended in 2ml (×1) distilled water and put in an O_2 evolution chamber in the dark (by covering with aluminium foil). After two minutes, the aluminium-foil was removed, the microscopy light switched on and O_2 evolution measured by using YS1 MODEL 5300 Biological OXYGEN MONITOR and SERUOGOR 120 channel recorder. Cyanobacteria and Prochloron evolve oxygen in the light during photosynthesis. Other bacteria will not evolve oxygen.

2.2.7 Determination of pigment content

Approximately 1g wet weight (see 2.2.13) of cyanobacterial cells were harvested by centrifugation at 6000g for 15 minutes at 4° C in a MSE Europa 24M centrifuge using 10×100ml rotor, washed with 5ml 5% (w/v) HEPES buffer (pH 7.8) and resuspended in 10ml 5% (w/v) HEPES buffer (pH 7.8). The cells were broken by sonication with ice chilled for 2 min. The sonicate was centrifuged at 35,000g for 60min at 10 °C in BECKMAN L-80 Ultracentrifuge using Ti70 rotor.
The uv-vis absorption spectrum of the supernatant was determined by using the KONTRON UVIKON-930 spectrophotometer. Absorbance peaks at 565nm and 620nm indicate the presence of phycoerythrin (PE) and phycocyanin (PC) respectively. The pellet was extracted with 5ml of methanol and twice with 5ml of methanol/diethyl ether (1:1,v/v) for 40 min. The combined extracts were centrifuged at 5,000g for 10min in a MSE Europa 24M centrifuge at 10 °C using 4×25ml rotor to remove unbroken cells and cell debris. The supernatant was examined spectrophotometrically by KONTRON UVIKON-930 spectrophotometer (380nm-720nm). The absorbance peaks at 380nm-470nm and 665nm indicate the presence of carotenoids and chlorophyll a (Danks *et al.*, 1983).

2.2.8 Purification of cyanobacteria

Single colonies of each cyanobacterial sample were sub-cultured repeatedly until pure cultures were obtained. Each colony isolated and restreaked in this way was given a reference number to facilitate comparison with the colonies obtained from either different isolation procedures or different water samples.

Purification of some filamentous cyanobacteria directly from the original growing plates proved difficult, accordingly Nystatin (0.02g/l) was added to BG11 medium to suppress the growth of competitive microorganisms during the purification (Jawad, 1982). The antibiotic treatment was carry out in BG11 medium which was supplemented with 0.05% casamino acids, 0.5% (w/v) glucose, and 1mg/ml ampicillin, and incubation in the dark for 16 h (Rippka, 1988).

2.2.9 Genus identification of cyanobacteria

Cyanobacterial isolates were identified to the generic level by using the method of Rippka *et al.* (1988) based on a bacteriological approach. Details of this method are given in Appendix C.

2.2.10 Determination of Nitrogenase Activity

A modification of the method of Rippka and Waterbury (1977) was employed for the measurement of nitrogen fixation by isolates as determined by nitrogenase activity. Mid-exponential phase cultures were harvested by centrifugation at 9000 g for 10 minutes at 15° C in a MSE Europa 24M centrifuge using 10×100ml rotor, and resuspended in 50ml of nitrate-free BG11 medium in 100ml flasks. Cultures were incubated for 48 h in a GALLENKAMP illuminated incubator under standard conditions with three warm-white fluorescent light, after which time they started to show signs of nitrogen depletion, the culture turning to a yellow colour. Flasks were bubbled with a mixture of nitrogen and carbon dioxide (97% and 3% v/v respectively), and incubated under reduced light intensity with one warm-white fluorescent light, at 100 rpm. Nitrogenase positive cultures are green or light blue-green.

2.2.11. Determination of ability to grow photoheterotrophically

A method of Rippka (1972) was employed for determination of photoheterotrophy. Photoheterotrophy was demonstrated by growth of strains in the light in the BG11 medium with 10^{-5} M dichlorophenyldimethylurea (DCMU),

supplemented with each of the following organic substrates alone: 0.5% (w/v) glucose, 0.5% (w/v) fructose, 0.5% (w/v) sucrose, 0.5% (w/v) ribose and 0.1% (w/v) glycerol. Stock solutions of 5% (w/v) substrate solution and 0.1 M dichlorophenyldimethylurea were sterilised by passing through 0.2um filters (Whatman, UK), and stored at -20 $^{\circ}$ C.

2.2.12 Electron microscopic and light microscopic examination

Cyanobacterial cells were fixed by a modification of the method of Rippka et al.(1974). Cells were prefixed at room temperature with 0.5% (w/v) glutaraldehyde in 0.1M sodium cacodylate buffer (pH 7.2) for 30 min. Cells were then fixed in 3.5% (w/v) glutaraldehyde for 2 h at room temperature and 15 h at 4°C, and then rinsed six times (15min for each rinse) in cacodylate buffer (pH 7.2). After fixation, the cells were enrobed in 3% (w/v) agar (in ×1 distilled water), which was cut into small pieces and washed four times in cacodylate buffer (pH 7.2). The cells were post-fixed in 1% (w/v) osmium tetroxide in cacodylate buffer for 1 h at room temperature and for another 4 h at 4 °C. Following three washes in cacodylate buffer (pH 7.2) and one in distilled water, the cells were stained in 1% (w/v) aqueous uranyl acetate for 2 h at room temperature, and washed three times with distilled water, dehydrated through an ethanol series (incubated with each 30% v/v, 50% v/v, 70% v/v, 90% v/v, 100% v/v ethanol for 10 minutes) and embedded in Taab spurr resin. Thin sections were obtained by using Ultra-cut microtomes. Glass knives were cut on an L.K.B. knifemaker type 7800. Sections exhibiting silver-gold interference colours were collected on formvar-coated copper grids of 200 mesh size. Thin sections were post-stained with both uranyl acetate (2% w/v aqueous) and Reynolds' lead citrate, and examined and photographed in a JEOL JEM-1200EX electron microscope.

Light microscopy was used to examine cyanobacterial cell division using specimens mounted on agar-coated slides (Appendix A), and photographed using OLYMPUS BH-2 NOMARSKI illumination. Colonies or trichome forms were directly photographed from agar plates with using Leitz LABORLUX 12 phase contrast illumination.

2.2.13 Determination of growth rates of cyanobacterial cultures

Growth was determined as a function of dry weight from optical density (OD) measurements at 720nm. Different sized inocula (0.5ml, 1.0ml, 1.5 ml; OD₇₂₀ 1.2-1.6) of cultures were transferred to 1 litre flasks containing 500ml of appropriate liquid medium, Cultures were incubated in a GALLENKAMP illuminated incubator which was continuously supplemented with CO_2 , at 20-30^oC (30-37^oC for *Synechococcus* PCC 7942) autocycle under continuous light with two warm-white fluorescent lights, at 100 rpm as described in section 2.2.1. Culture samples (2ml) were removed at various time intervals and measured at 720nm using the UV-930 spectrophotometer against a blank of appropriate uninoculated medium. OD₇₂₀ of 1.0 corresponds to approximately 1mg dry weight of cells per 1ml of culture (Lambert and Carr, 1982).

2.2.14 Preparation and purification of cyanobacterial DNA

A number of methods was used for the extraction of plasmid and genomic DNA from cyanobacteria.

2.2.14.1 Mini-methods for preparation of plasmid DNA

Method 1: Lysis by lysozyme, a modification of a method of Kado and Liu (1981) and Birnboim and Doly 1979. Approximately 1.5mg dry weight of cells were harvested by centrifugation at 6000g for 15 minutes at 4 $^{\circ}$ C in a MSE 24 M centrifuge using 4×25ml rotor and washed with 0.5 ml of TSE buffer (10mM Tris-HCl, 50mM NaCl, 50mM EDTA, pH 7.8). The pellet was resuspended in 0.5ml of Birnboim and Doly solution I (50mM glucose, 25mM Tris-HCl, 10mM EDTA, pH 8.0), and incubated with lysozyme (final concentration 5mg/ml) at 37 $^{\circ}$ C for 1 hour. RNase was then added to 20µg/ml and proteinase K to 100µg/ml (see Appendix B), and the mixture incubated at 55 $^{\circ}$ C for 2 hours. Two phenol-chloroform-isoamyl alcohol extractions, followed by two chloroform-isoamyl alcohol extractions were carried out (see 2.2.14.4). After extraction was complete, the plasmid DNA was recovered by ethanol precipitation (see 2.2.14.4) and the pellet resuspended in 50µl of TE buffer (10mM Tris-HCl, 1mM EDTA, pH 8.0) and stored at -20 $^{\circ}$ C.

Method 2: Lysis by Sodium Dodecyl Sulphate (SDS) based on the method of Van den Hondel *et al.* (1979). Approximately 1.5mg dry weight of cells were harvested by centrifugation, the pellet washed and resuspended in 0.5ml of sucrose buffer (50mM Tris-HCl, 50mM EDTA, 25% (w/v) sucrose, pH8.0), and incubated with lysozyme (final concentration 5mg/ml) at 37° C for one hour. 75μ l of 10% (w/v) SDS was then added and incubation continued for a further hour. 150μ l of 5M NaCl was added with gentle mixing and the mixture incubated over-night on ice. RNase was then added to 20μ g/ml and proteinase K to 100μ g/ml, and incubated at 37° C for 30 min. Two phenol-chloroform- isoamyl alcohol and then two chloroform-isoamyl alcohol extractions were carried out (see 2.2.14.4). After extraction was complete, the cleared lysate was precipitated with ethanol (see 2.2.14.4), and the pellet resuspended in 50 μ l TE buffer and stored at -20 $^{\circ}$ C.

Method 3: Alkaline lysis based on a method of Birnboim and Doly (1979). Approximately 1.5mg dry weight of cells were harvested by centrifugation, washed, and the pellet resuspended in 200 μ l of Birnboim and Doly solution I. 400 μ l of alkaline mixture (0.2M NaOH, 2% (w/v) SDS) was added and the tube inverted. After cooling on ice for 5 min, 300 μ l 3M sodium acetate (pH 5.2) was added and the mixture incubated on ice for 10 min. Two phenol-chloroform-isoamyl alcohol and then two chloroform-isoamyl alcohol extractions were carried out (see 2.2.14.4). After extraction was complete, the cleared lysate was precipitated with ethanol (see 2.2.14.4), the pellet was resuspended in 50 μ l TE buffer and stored at -20 °C.

Method 4: Boiling method of Holmes and Quigley (1981). Approximately 1.5mg dry weight of cells were harvested by centrifugation, the pellet washed and resuspended in 350μ l STET (50mM Tris-HCl, 50mM EDTA, 8% (w/v) sucrose, 5% (w/v) Triton X-100, pH8.0), and boiled with lysozyme (final concentration 2.5mg/ml) for 40 seconds. After holding on ice for 5 min, two phenol-chloroform

and two chloroform extractions were carried out (see 2.2.14.4). After centrifugation at 12,000g in a SANYO microcentrifuge for 5 min at room temperature, 40μ l of 3M sodium acetate (pH 5.2) was added to the supernatant and the DNA precipitated with ethanol (see 2.2.14.4), The pellet was resuspended in 50 μ l TE buffer and stored at -20 °C.

Method 5: Lysis by heat-alkali co-denaturation based on the methods of Birnboim and Doly (1979) and Holmes and Quigley (1981). Approximately 1.5mg dry weight of cells were harvested by centrifuge, the pellet washed and resuspended in 0.75ml of TES (25 mM Tris-HCl, 50mM EDTA, 10% (w/v) sucrose, pH 8.0), and incubated with lysozyme (final concentration 5mg/ml) on ice for 20 min. 1.5 ml of the denaturation solution (0.15M NaOH, 1% (w/v) SDS) was added and mixed quickly. The mixture was incubated at 80° C for 8 min with gently shaking. Then 1.5ml of the cold neutralising solution (3.5M KAc, pH5.8) was added with thorough mixing, and incubated on ice for 20 min. Two phenol-chloroform-isoamyl alcohol and then two chloroform-isoamyl alcohol extractions were carried out (see 2.2.14.4). After extraction was complete, a 0.4 volume of 2-isopropanol was added to the supernatant, and incubated at room temperature for 10 min. After centrifugation in a SANYO microcentrifuge at 12,000g, for 5 min at room temperature, the precipitate was resuspended in an appropriate volume of TE buffer and stored at -20 °C.

2.2.14.2 Large scale plasmid DNA preparation

Large scale plasmid preparation was carried out by caesium chlorideethidium bromide (CsCl-EtBr) density gradient centrifugation (van den Hondel et al., 1979). A 1L mid-exponential phase liquid culture (2×500ml) of cyanobacteria was harvested by centrifugation at 6000g for 15 min at 4°C in a MSE Europa 24M centrifuge using a 6×300ml rotor. The cells were washed and the pellet resuspended in 50ml of sucrose buffer (see 2.2.14.2 method 2), and incubated with lysozyme (final concentration 5mg/ml) at 37°C. After 30 min, 6.2ml of 10% (w/v) sodium dodecyl sulphate was added and mixed by inversion. After incubation at 37°C for a further hour, 15ml of 5M NaCl were added and gently mixed by inversion. The mixture was incubated on ice overnight and the centrifuged at 10,000g for 15 minutes at 4°C in a MSE 24 M centrifuge using 10×100ml rotor. The cleared lysate was precipitated with ethanol (see 2.2.14.4) and the precipitate was resuspended in a total of 16ml of TE buffer. Solid CsCl was then added to a final concentration of 1.05g/ml and 0.8ml of ethidium bromide solution (5mg/ml) was added. Plasmid DNA was separated from chromosomal DNA by caesium chloride-ethidium bromide density gradient centrifugation in a Beckman L-80 ultracentrifuge using a Ti-80 rotor at 65,000g for 40 hours at 18°C. The lower band of supercoiled plasmid DNA was collected through a #21 hypodermic needle inserted into the side of the tube. EtBr was extracted from the collected DNA fractions with an equal volume of 1-butanol saturated with 5M NaCl in TE buffer. The aqueous material was dialysed against TE buffer for several hours, with at least 3 changes of TE buffer, at room temperature. The sample was then ethanol precipitated, resuspended in an appropriate volume of TE buffer and stored at -20 °C.

2.2.14.3 Preparation of total genomic DNA

Method 1: Total genomic DNA was prepared by the method of Porter (1988). Approximately 1.5 mg dry weight of cells (mid-exponential phase) were harvested by centrifugation, the pellet washed and resuspended in 0.5ml of sucrose buffer (10-25% (w/v) sucrose, 50mM Tris-HCl, 100mM EDTA, pH 8.0) and incubated with lysozyme (final concentration 5mg/ml) at 37° C. After 30 minutes, SDS to1% (w/v) and proteinase K to 100μ g/ml were added. The mixture was incubated at 50° C for at least 2 h followed by two phenol-chloroform-isoamyl alcohol and then two chloroform-isoamyl alcohol extractions (see 2.2,14.4). 1/3 volume of 10.5 M ammonium acetate and 2 volumes of 2-propanol were added to the aqueous phase. The sample was incubated at -20° C for at least 15 min and then centrifuged for 5 min at room temperature in a microfuge at 13000g. The pellet was washed twice with 70% (v/v) ethanol. The pellet was dried under reduced pressure in a desiccator, resuspended in an appropriate volume of TE buffer and stored at -20° C.

Method 2: Total genomic DNA was prepared by the method of Neilan (1995). Approximately 1.5 mg dry weight of cells (mid-exponential phase) were harvested by centrifugation, the medium was decanted, and the pellet washed and resuspended in 0.5ml of 50mM Tris-HCl, 50mM EDTA, 50mM NaCl, pH 8.0. Lysozyme was added to a final concentration 1mg/ml, and the solution was incubated at 55° C for 30 min. After addition of 20 µl 1% (w/v) SDS and 10 µl of proteinase K (10mg/ml), the mixture was incubated at 55° C, until the solution cleared. The solution was chilled on ice and extracted twice with an equal volume of phenol-chloroform-isoamyl alcohol and then with chloroform-isoamyl alcohol

(see 2.2.14.4). An equal volume of 4 M ammonium acetate and 2 volumes of 2propanol were added to the aqueous phase. The sample was incubated at -20° C for at least 15 min and then centrifuged for 5 min at room temperature in a SANYO microfuge at 13000g. The pellet was washed twice with 70% (v/v) ethanol. The pellet was dried under reduced pressure in a desiccator and resuspended in an appropriate volume of TE buffer stored at -20 °C. Both methods provided template DNA for PCR (see 2.2.19 and 2.2.20).

Total genomic DNA as template for PCR was also obtained from cells directly. A 1ml aliquot of mid to late exponential phase culture was pelleted by centrifugation for 2 min at room temperature in a SANYO microfuge at 6000g, the cells were washed, resuspended in 0.1 ml of TE buffer or sterilise ×1 distilled water and used directly for PCR (section 2.2.20).

2.2.14.4 Phenol-chloroform extraction and ethanol precipitation of DNA

Proteins were routinely removed from DNA solutions by phenol-chloroform extraction (Sambrook *et al.*, 1989). DNA solutions were extracted with the same volume of phenol (saturated with 0.1M Tris-HCl, pH 8.0) : chloroform : isoamyl alcohol (25:24:1 (v/v)) (see Appendix B) by vortexing the two phases and centrifuging for 5 min at 12,000g at room temperature in a SANYO microfuge. The upper aqueous layer was then transferred to a fresh sterile microfuge-tube, followed by extraction with chloroform:isoamyl alcohol (24:1 (v/v)) twice. Two phases were separated by centrifuging for 5 min at 12,000g at room temperature in a SANYO microfuge. Ethanol precipitation was used to remove traces of organic solvent or to concentrate the DNA solution. DNA was precipitated at -20° C for 1 h after the addition of 1/10 volume 3M sodium acetate, pH4.8 and 2.2 volumes of cold (-20° C) 100% (v/v) ethanol. The precipitated DNA was collected by centrifugation for 5 min at 12,000g at room temperature in a SANYO microfuge. The DNA pellet was washed with 70% (v/v) ethanol, dried at room temperature, resuspended in an appropriate volume of TE (10mM Tris-HCl and 1mM EDTA, pH 8.0) buffer or sterile ×1 distilled water and stored at -20° C.

2.2.15 Agarose gel electrophoresis

Horizontal, 0.6% (w/v) to 1.5% (w/v) agarose gels (Boehringer Mannheim GmbH, multi purpose agarose (MP)) and 3% gels (2% (w/v) agarose NA (nucleic acids) + 1% (w/v) agarose MP, see section 2.2.21) in 1×TAE buffer (40mM Tris-Acetate, 1mM EDTA, pH 7.6) were cast in horizontal gel tanks (BRL, Horizon 58 and H5 or Flowgen). DNA (up to $0.5\mu g$) in loading buffer (100mM EDTA, 1% (w/v) SDS, 0.1% (w/v) bromophenol blue, 50% (v/v) glycerol) was applied to the gel slots together with appropriate molecular mass standard and electrophoresis was performed at 50-120 Volts by using a LKB BROMMA 2197 power pack, until the bromophenol blue tracking dye front reached the end of the gel. The DNA was stained by immersing the gel in ethidium bromide solution ($0.5\mu g/ml$ in ×1 distilled water) for 1 h at 4 °C in the dark. The gel was destained in distilled water for 20 min at room temperature in the dark. DNA was visualized by mid-range UV induced fluorescence on a transilluminator (LKB-2011 Macrovue Hi switch UV, Sweden)

and photographed with a Polaroid MP4 land camera equipped with an orange filter (Kodak Wratten 22,USA) using Polaroid 665 film (exposure 15 s).

2.2.16 Recovery of DNA from agarose gels

The DNA to be isolated was separated on a 1% (w/v) agarose gel. The band of interest was located on the gel using a long-wavelength UV lamp (LKB-2011 Macrovue Lo switch UV, Sweden) to minimise damage to the DNA. Using a sharp scalpel, a slice of agarose containing the band was cut out and placed in the fluidfilled (0.5 ×TAE buffer) pre-treated dialysis bag (see Appendix B). The gel slice was allowed to sink to the bottom of the bag, and most of the buffer removed, leaving just enough fluid to keep the gel slice in constant contact with the electrophoresis buffer. The bag was tied just above the gel slice, avoiding trapping air bubbles. The bag was immersed in a shallow layer of $0.5 \times TAE$ buffer in a horizontal electrophoresis tank (BRL, Horizon 58). Electric current was passed through the bag (usually 100V for 2-3 h) then the polarity of the current reversed for 2 min to release the DNA from the wall of the dialysis bag. The dialysis bag was opened and carefully all the buffer surrounding the gel slice was recovered. Two phenol-chloroform-isoamyl alcohol and then two chloroform-isoamyl alcohol extractions were carried out (see 2.2.14.4). After extraction was complete, the cleared lysate was precipitated with ethanol (see 2.2.14.4), the pellet was resuspended in 50µl TE buffer and stored at -20 °C.

2.2.17 Measurement of DNA concentration

The concentration and purity of DNA samples were determined by UVspectrophotometry (at 260 nm and 280 nm) using 5µl samples of DNA and 995µl TE buffer (Sambrook *et al.*, 1989) using a blank of TE buffer. DNA concentration may be calculated as follows:

concentration of DNA sample (μ g/ml) = 50 × (1000/5) × A₂₆₀

(where A_{260} = absorbance at 260nm)

The ratio of the absorbance readings at 260 nm and 280 nm (A_{260}/A_{280}) provided an estimate of the purity of the nucleic acid. Pure preparations of DNA have A_{260}/A_{280} values of 1.8 and above. If there was contamination with protein or phenol, the A_{260}/A_{280} was significantly less than this value, If there was contamination with RNA, the A_{260}/A_{280} was significantly more than this value, and accurate quantification of the amount of DNA was not possible.

Sometimes there was more than one plasmid DNA species in the preparation, or the DNA was heavily contaminated with other substances that absorb ultraviolet light and therefore impeded accurate analysis. 5μ l of such samples of DNA, mixed with 1μ l of gel-loading buffer, were subject to electrophoresis on an 0.8% (w/v) agarose horizontal gel (see 2.2.15) together with an appropriate DNA standard (e.g. pBR 322). Electrophoresis was carried out until the bromophenol blue had migrated 2 cm. The intensity of fluorescence (using the transilluminator in section 2.2.15) of the unknown DNA samples was compared with that of DNA standards of known amount and the quantity of DNA in the sample was estimated.

2.2.18 DNA restriction

To 13µl of DNA samples (0.5-2.0µg), was added 1.5µl of 10× restriction buffer (as specified by manufacturers) and 5-8 units of appropriate restriction enzyme (Boehringer Mannheim GmbH) in a total reaction mixture of 15µl and mixtures were normally incubated at 37° C for 2 h. Incubations were carried out in a heated block (Techne, Dri-Block OB-1, UK). Reactions were terminated by incubation at 65° C for 10 min. Restriction fragments were resolved by agarose gel electrophoresis using appropriate DNA standards (section 2.2.15).

2.2.19 Random Amplified Polymorphic DNA - Polymerase Chain Reaction (RAPD-PCR)

The polymerase chain reaction (PCR) using random primers was carried out essentially as described by Williams *et al.* (1990). The primers used were 10-base single oligonucleotides of arbitrary DNA sequence (see Table 2.4). Each PCR mix contained in a total reaction volume of 50μ l: 1×PCR amplification buffer, 2.5mM MgCl₂, 150 μ M each of dATP, dCTP, dGTP and dTTP, 0.2 μ M primer, 0.1 μ g of total DNA, and 1 unit of AmpliTaq polymerase, overlaid with 1 drop of mineral oil (Cat. No. M3516; Sigma). Amplifications were performed in a Perkin-Elmer Cetus DNA thermal cycler (UK) programmed for 35 cycles: Programme 1 (Williams et al., 1990):

Denaturing, 1 min at 94^oC;

Annealing, 1 min at 34^oC;

Extension, 2 min at 72°C,

using the fastest available transitions between each temperature.

Programme 2 (Bielawski et al., 1995):

Denaturing, 30 s at 94^oC;

Annealing, 30 s at 34^oC;

Extension, 2 min at 72°C,

using the fastest available transitions between each temperature.

Programme 3 (Yu and Pauls 1992):

Denaturing, 5s at 94^oC;

Annealing, 30s at 34^oC;

Extension, 60s at 72°C,

using the fastest available transitions between each temperature.

Programme 4 (this study):

Denaturing, 30 s at 94°C;

Annealing, 30 s at 34^oC;

Extension, 2 min at 72°C,

using the transition interval between each temperature, the rate of change of temperature being 0.45 $^{\circ}C/s$.

Programme 5 (this study):

Denaturing, 1 s at 94°C;

Code	Sequence	Code	Sequence
AB1-01	5'-GTTTCGCTCC-3'	AB1-11	5'-GTAGACCCGT-3'
AB1-02	5'-TGATCCCTGG-3'	AB1-12	5'-CCTTGACGCA-3'
AB1-03	5'-CATCCCCCTG-3'	AB1-13	5'-TTCCCCCGCT-3'
AB1-04	5'-GGACTGGAGT-3'	AB1-14	5'-TCCGCTCTGG-3'
AB1-05	5'-TGCGCCCTTC-3'	AB1-15	5'-GGAGGGTGTT-3'
AB1-06	5'-TGCTCTGCCC-3'	AB1-16	5'-TTTGCCCGGA-3'
AB1-07	5'-GGTGACGCAG-3'	AB1-17	5'-AGGGAACGAG-3'
AB1-08	5'-GTCCACACGG-3'	AB1-18	5'-CCACAGCAGT-3'
AB1-09	5'-TGGGGGGACTC-3'	AB1-19	5'-ACCCCCGAAG-3'
AB1-10	5'-CTGCTGGGAC-3'	AB1-20	5'-GGACCCTTAC-3'
ABA-01	5'-CAGGCCCTTC-3'	ABA-11	5'-CAATCGCCGT-3'
ABA-02	5'-TGCCGAGCTG-3'	ABA-12	5'-TCGGCGATAG-3'
ABA-03	5'-AGTCAGCCAC-3'	ABA-13	5'-CAGCACCCAC-3'
ABA-04	5'-AATCGGGCTG-3'	ABA-14	5'-TCTGTGCTGG-3'
ABA-05	5'-AGGGGTCTTG-3'	ABA-15	5'-TTCCGAACCC-3'
ABA-06	5'-GGTCCCTGAC-3'	ABA-16	5'-AGCCAGCGAA-3'
ABA-07	5'-GAAACGGGTG-3'	ABA-17	5'-GACCGCTTGT-3'
ABA-08	5'-GTGACGTAGG-3'	ABA-18	5'-AGGTGACCGT-3'
ABA-09	5'-GGGTAACGCC-3'	ABA-19	5'-CAAACGTCGG-3'
ABA-10	5'-GTGATCGCAG-3'	ABA-20	5'-GTTGCGATCC-3'

Table 2.4 10-base single random primers used in RAPD-PCR^a.

a, Primers were obtained from Advanced Biotechnologies, Surrey, UK

.

Annealing, 1 s at 34^oC;

Extension, 1 s at 72°C,

using the transition interval between each temperature, the rate of change of temperature being 0.244 $^{\circ}$ C/s from 72 $^{\circ}$ C to 94 $^{\circ}$ C, 0.400 $^{\circ}$ C/s from 94 $^{\circ}$ C to 34 $^{\circ}$ C and 0.211 $^{\circ}$ C/s from 34 $^{\circ}$ C to 72 $^{\circ}$ C.

DNA negative controls consisted of the standard PCR reaction mixture, but with the DNA template replaced by an equal volume of sterrile × 1 distilled water.

Amplification products were analysed by electrophoresis in 1.5% (w/v) agarose gels and detected by staining with ethidium bromide (see section 2.2.15), using appropriate molecular weight markers (see section 2.1.1).

2.2.20 PCR amplification of rDNA spacer

The PCR was used to amplify the internal transcribed spacer of rDNA (see Fig 1.1). Each 100 μ l reaction mixture for PCR contained 40 pmol of each primer, 2.5 units of *Taq* polymerase (Bioline) or DyNAZyme II (Finnzymes Oy), 0.2 mM deoxyribonucleoside triphosphates (dATP, dCTP, dGTP and dTTP), 2.5mM MgCl₂ and 0.1 μ g cyanobacterial DNA or 1 μ l TE buffered culture (see 2.2.14.3). The reaction mixture was overlaid with mineral oil, and thermal cycling was achieved in a Perkin-Elmer Cetus Thermal Cycler according to the following programme: initial denaturation at 94 °C for 3 min, 35 cycles of 94 °C for 1 min, 55°C for 1min and 72 °C for 3min. Primers 1, 14, 16, 17, 18 as described by Wilmotte, *et al.*(1993) (see Table 2.5), were

used. DNA negative controls consisted of the standard PCR reaction mixture, but with the DNA template replaced by an equal volume of sterile ×1 distilled water.

Primer	Sequence	Strand ^a	Position ^b		
1	5'-AGAGTTTGATCCTGGCTCAG-3'	R	8-27 (16S)		
14	5'-TGTACACACCGCCCGTC-3'	R	1334-1350 (16S)		
16	5'-AAGGAGGTGATCCAGCCGCA-3'	C	1486-1467 (16S)		
17	5'-ATTAGCTCAGGTGGTTAG-3'	R	6-23 (tRNA ^{ile})		
18	5'-TTTGCGGCCGCTCTGTGTGC	С	76-45 (238)		
	CTAGGTATCC-3'				

Table 2.5Primers used for PCR and sequence determination of rDNA regions

a, Primer sequences corresponding to the RNA sequence are indicated by R; primers complementary to the RNA sequence are indicated by C.

b, Nucleotide position in the corresponding rRNA sequence of *Synechococcus* sp. PCC6301.

2.2.21 Restriction digestion of PCR-amplified DNA

PCR-amplified DNA (10µl, approximately 0.5-0.8µg) was digested, without further purification, with various restriction enzymes (*Dde* I, *Taq* I, *Alu* I, *Sau* 3A, *Hae* III, *Pvu* I, *Xba* I, *Sal* I, *Sac* I, *Ava* I, *Bam* HI, *Hind* II, *Hind* III, *Pst* I, *Bgl* II, *Eco* RV (from Boehringer Mannheim, Germany) and *Hinf* I, *Hha* I, *Hpa* II, *Rsa* I (from Sigma, England) according to the manufacturer's instructions. The products were separated on gels containing 2% (w/v) agarose NA (nucleic acids) + 1% (w/v) agarose MP (multi purpose) in TAE (mM Tris-HCl, mM NaAC, mM EDTA, pH 8.0) buffer. The gels were stained with ethidium bromide and photographed (see 2.2.15). Appropriate molecular weight markers (123bp ladder, Sigma, England) were run on each gel.

2.2.22 Data handling by Binary Matrix and Similarity coefficients

Each band visualised on a gel was measured by using the Phoretix 1D software package developed by Phoretix International Ltd (Newcastle, UK). DNA band patterns were converted into a binary matrix: band present was indicated as 1, band absent was indicated as 0. Following band scoring similarity indices were calculated for further analysis. The method of Nei and Li (1979) was used as follows:

$$S = 2 Nxy/(Nx+Ny)$$

Where S is the similarity index, Nxy is the total number of bands shared between sample x and y, Nx is the total number of bands in sample x, Ny is the total number of bands in sample y. Identical patterns will therefore equal 1 and totally unrelated patterns equal 0.

2.2.23 Phylogenetic analysis of PCR-amplified products

Each band visualised on a gel was measured by using the Phoretix 1D software package developed by Phoretix International Ltd (Newcastle, UK). Phenetic analysis of both Random Amplified Polymorphic DNA (RAPD) (section 2.219) and Restriction Fragment Length Polymorphisms (RFLPs) (section 2.2.20) data was performed by calculating pairwise genetic distances from a binary matrix. Distances were calculated by using the 14 algorithms supplied by the RAPDistance package developed by Armstrong *et al.* (1994). The resulting trees were constructed by using the neighborjoining method and the NJTREE program.

Phenetic analysis of RFLP data was also performed by using the DOLLOP program in the PHYLIP package. The consensus tree was constructed by using the CONSENSE program in the PHYLIP package.

2.2.24 Purification of PCR-amplified product and TA-Cloning

PCR-amplified products using primer pair 17/18 were directly purified using the Wizard PCR preps DNA purification system (Promega) according to the manufacturer's instructions. The DNA concentration was adjusted to 30ng/ul (for direct automatic sequencing, see section 2.2.26 method 2). The purified PCR product was ligated into a pGEM-T vector (Promega) or a pCR II vector (Invitrogen) to which was added a 3'-terminal thymidine to both ends. Recombinant plasmids were used to transform *E. coli* JM109 high efficiency competent cells (provided in the Promega cloning kit) or One Shot competent cells (provided in the Invitrogen cloning kit). White colonies (recombinant clones) were selected from LB/ampicillin/IPTG/X-Gal agar (Appendix A) plates and each was transferred to 5ml LB/ampicillin medium (Appendix A) and incubated for 24 hours at 37°C with shaking at 100 rpm in a GALLENKAMP incubator.

2.2.25 Isolation and purification of recombinant plasmid DNA

Recombinant plasmid DNA was isolated and purified by using Wizard minipreps DNA purification system, according to the manufacturer's instructions. The plasmid DNA concentration was adjusted to 500ng/ul with sterile ×1 distilled water.

2.2.26 DNA sequencing

Method 1: Sequence determination of directly purifed PCR products (section 2.2.24) was carried out by using the ABI PRISM dye terminator cycle sequencing kit according to the manufacturer's instructions. Each 20ul reaction mixture for cycle sequencing contained 100ng PCR product, 3.2pmol primer and 8.0ul terminator ready reaction mixture. The thermal cycling was achieved in a Perkin-Elmer Cetus Thermal Cycler according to the following programme: 25 cycles of 96°C for 30 s, 50°C for 15 s and 60°C for 4 min. Primer 18 (see Table 2.5) was used. The extension products were purified using a spin-column (Perkin-Elmer). Then the products were run on 6% (w/v) polyacrylamide gels (29:1 acrylamide:bisacrylaminde, purified and degassed with 1% (w/v) purification resin and 0.2 μ m vacuum filter). The rDNA sequences were determined by using the automated DNA sequencer (ABI PRISM 377, UK).

Method 2: Sequence determination of cloned DNA (section 2.2.25) was carried out with the Sequenase 2.0 sequencing kit (USB, Amersham, UK) according to the manufacturer's instructions. The two primers (M13 forward (-21) and reverse (-26)) and 35 S-dATP were used. 6% (w/v) (19:1 acrylamide:bisacrylaminde) Sequagels (National Diagnostics, UK) were run in a vertical gel tank (Model S2, Life Technologies, UK) at 50W for 7h, by using a LKB BROMMA 2197 power pack. The gel was autoradiographed at -80 °C for 36 h using Fuji medical X-ray film. The sequence was manually determined from the developed film.

2.2.27 DNA sequence analysis

Primary DNA sequences from PCR products were aligned to the published rDNA sequences for *Arthrospira* sp. PCC7345 (Accession No. X75044, Nelissen *et al.*, 1994), *Spirulina* sp. PCC6313 (Accession No. X75045, Nelissen *et al.*, 1994), *Synechococcus* PCC6301 (Accession No. X00346; Tomioka and Sugiura, 1984), *Trichodesmium* NIBB1067 (Accession No. X72871; Wilmotte *et al.*, 1994), *E. coli* (Accession No. J01702, Young and Steitz, 1978), *Cyanophora paradoxa* cyanelle (Accession No. M19493, Janssen *et al.*, 1987) and *Chondrus crispus* (red alga) chloroplast (Accession No. Z29521, Leblanc *et al.*, 1995) from the GenBank database using the CLUSTAL W program version 1.6 (http://www.ibcp.fr/clustalw.html).

Estimates phylogenies by the parsimony method:

Bootstrapping (1, 100, 500 runs) of aligned sequences was undertaken by using SEQBOOT program in the PHYLIP package. The parsimony trees were constructed by using DNAPARS program in the PHYLIP package. A consensus tree was constructed by using CONSENSE program in the PHYLIP package.

Estimates phylogenies by the maximum likelihood method:

The maximum likelihood trees were constructed by using DNAML program in the PHYLIP package with directly input aligned sequences (from CLUSTAL W).

73

Estimates phylogenies by the neighbour joining method:

The distance values were calculated by using DNADIST program in the PHYLIP package with directly input aligned sequences (from CLUSTAL W). The distances then were used in the distance matrix program (neighbour joining). The neighbour joining trees were constructed by using NEIGHBOR program in the PHYLIP package.

2.2.28 Primer design

Sequences to be used in designing primers should be specific for the cyanobacterial target region. The specificity of proposed primer sequences were checked by using Genbank data (BLAST e-mail: blast@ncbi.nlm.nih.gov) and by alignment of cyanobacterial 16S-23S spacer sequences (from this study).

The dissociation temperature of the primer/template duplex (Tm) can be calculated from the primer sequence from the formula:

$$Tm = 4 (G+C) + 2 (A+T).$$

PCR primers that ensure Tm of 54 °C or higher provide the best chance for maintenance of specificity and efficiency (Dieffenbach et al., 1993).

General requirements for primer pairs are: 1) be similar in size between 18 to 25 bases in length, 2) 40%-60% GC content with a 3'-terminal "G/C clamp" (at least one or two 3' G's and /or C's), 3) be free of repetitive motifs, 4) similar Tm values, 5) complementary sequences should not be found between the primer pair (Roux, 1995).

Specific oligonucleotide sequences were synthesized (Perkin-Elmer) for use as primers SF/SR for PCR (see section 6.2.3).

Chapter 3

,

,

Isolation, Purification and Identification

of Cyanobacteria

.

.

.

•

3.1 Introduction

The isolation and purification of cyanobacteria from natural environments is important for the study of their taxonomy and characteristics. Methods for the isolation and purification of cyanobacteria have been reviewed by Walsby for planktonic species (1981), Castenholz (1981) for thermophiles, by Waterbury and Stanier (1981) for organisms from marine and hypersaline environments, and by Rippka *et al.*(1981) for strains of diverse origin (cited in Rippka, 1988). After preliminary experimentation the isolation method of Rippka (1988) was used.

Following successful isolation, cyanobacteria should be identified by a name, which serves as an indicator of the respective phenotypic properties and is therefore crucial for scientific communication. Unless the organisms have not been previously described, their names have to be chosen from an existing system of classification. Since cyanobacteria were first recognised more than 200 years ago, a bewildering array of genera and species has been created by botanists and ecologists. Classification was based either on the properties observable on samples collected from the natural habitat or on those extractable from dried herbarium specimens. Furthermore, many genera and species underwent repeated taxonomic revisions, leading to a large number of synonyms that only botanical experts are capable of unravelling.

Although the relationship between cyanobacteria (as blue-green algae) and the bacteria has been recognised for more than a century, the classification of natural cyanobacterial populations was established from a range of morphological

76

characteristics and these have been reviewed by Whitton (1992) and further details are given in Chapter 1.

Morphological evaluations by light microscopy enable a continuity in observation from the unaided eye, through the use of a dissecting microscope (10- $100\times$), to a magnification of a compound light microscope ($100-1000\times$). This scope has been extended, to include applications of greater magnifications (up to $1,000,000\times$) of external and internal structures, using scanning and transmission electron microscopes. Within the last two decades, ultrastructural data has accumulated, providing a new source of useful taxonomic information (e.g. Lang and Whitton 1973; Rippka 1974; Jensen 1985; Hoffman 1988; Anagnostidis and Komarek 1988).

At the level of taxonomic distinction between genera, the traditional systems of cyanobacteria placed a high value on cell division patterns, colony formation and relationship to extracellular envelopes and sheaths. Differences in cell shapes and dimensions were used largely to distinguish between species within each genus. this practice caused difficulties in classification by introducing a tendency to assign organisms with different cell organisation but similar cell arrangements to the same generic identify.

• .

A new and deeply reorganised taxonomic revision was published by Anagnostidis and Komarek (1985, 1988, 1990) and Komarek and Anagnostidis (1986, 1989). These treatises are valuable contributions towards a modern system of cyanobacterial classification, particularly since an attempt was made to take into account not only the knowledge gained by botanists over the last 30 years but also that of bacteriologists over a similar period. This has led to the creation not only of numerous new generic names, but also to the nomenclatural change of some traditional genera (Rippka and Herdman, 1992). The generic limits adopted tend to be quite narrow, leading to smaller and more homogeneous genera than in most other botanical revision. The authors also made an extensive review of the literature and integrated all the biochemical, ultrastructural and molecular characters available with their considerable taxonomic experience.

Stanier et al.(1978) proposed to place the nomenclature of cyanobacteria under the rules of the International Code of Nomenclature of Bacteria, under which a living type strain, allowing modern comparative analysis, designates both species and genus. The second, more immediate, solution for the identification of axenic cyanobacteria was the proposed "working classification" of Rippka et al.(1979), which certainly will have to be modified as more complete and stable data on fine structural, physiological, biochemical, and genetic characteristics become available. The system is based on axenic, clonal cultures and the use of characters from a range of disciplines (Stanier, 1971; Kenyon et al., 1972; Rippka et al. 1974; Waterbury, 1977,1978; Herdman et al. 1979), rather than morphology alone. In this system, cyanobacteria are divided into five major sections (or Orders, Castenholz and Waterbury, 1989). The criteria used to define Orders are based on morphology as botanical revision (see Table 1.2). At the level of taxonomic distinction between genera where some families have poor morphological differentiation, other characters like ultrastructure and physiology were also considered, the characters chosen are simple and defined as " present or absent". For example, Gloeobacter is defined as a unicellular organism lacking thylakoids, *Gloeothece* is defined as a unicellular organism with rod-shape, present sheath and thylakoids. The genera of each Order are described in Appendix C. At the level of species classification, a little more information is still required to describe all the species. Some part of this system was changed at genera level by Castenholz and Waterbury (1989), and Rippka and Herdman (1992). In section I, the genera Chamaesiphon (cells ovoid, reproduction by budding), *Gloeobacter* (cells rod-shaped, thylakoids absent, sheath present, reproduction by one plane binary fission) and *Gloeothece* (cells rod-shaped, thylakoids present, sheath present, aerobic nitrogen fixation) were retained as defined by Rippka et al. (1979). Waterbury and Rippka (1989) employ the term "group", rather than genus, for two large and heterogeneous assemblages. The Synechococcus group and the Synechocystis group, both of which clearly warrant further generic division; each group containing 5 clusters (Rippka and Herdman, 1992). A group Synechococcus (Rippka et al., 1979) with cell width >3 μ m and nitrogenase produced anaerobically has been recognised as a Cyanothece group by Waterbury and Rippka (1989) and genus Cyanothece (Komarek, 1976) by Rippka and Herdman (1992). The genus Gloeocapsa (Rippka et al., 1979) (Gloeocapsa group Waterbury and Rippka 1989) has been retained under a more restricted definition by Rippka and Herdman (1992). Genera Chroococcus and Microcystis were represented in the PCC by Rippka and Herdman (1992). In section two, all six genera of Rippka (1979) are same as described in the system of Waterbury (1989). The major change since the publications of Rippka et al (1979) and Waterbury (1989) is the recognition of the newly created genus Stanieria (Komarek and Anagnostidis, 1986; Rippka and Herdman, 1992), which replaces Dermocarpa (Rippka et al., 1979 and Waterbury, 1989). In recent botanical revision (Komarek and Anagnostidis, 1986), section I and section II were placed into one order Chroococcales, which contains seven families and eighty-four genera. The genera Chroococcidiopsis and Myxosarcina in Rippka et al (1979), Waterbury (1989) and Rippka and Herdman (1992) conform well to the botanical descriptions. In section III, significant change has been made since the publication of Rippka et al. (1979). The most significant change is the elimination of LPP groups A and B, made possible by the revisions of Anagnostidis and Komarek (1988). In the revisions of Castenholz (1989), all strains in LPP A and some strains in LPP B (Trichome immobile in persistent sheath) have been assigned to the genera Lyngbya (one trichome per sheath) and *Microcoleus* (two or more trichomes per sheath). Other LPP strains (trichome motile, no persistent sheath) have been transferred to Pseudanabaena (strong constriction), Oscillatoria and Trichodesmium (absent or feeble constriction). In the revisions of Rippka & Herdman (1992), the LPP A is renamed Lyngbya. Most of the strains previously assigned to the LPP group B have been assigned to the genera Geitlerinema and Leptolyngbya, and a few strains have been temporarily transferred to Microcoleus. The genera Pseudanabaena and Oscillatoria (Rippka et al., 1979) have been retained with minor changes. The genus Spirulina (Rippka et al., 1979) has been divided into two genera, Spirulina and Arthrospira in both revisions of Castenholz (1989) and Rippka & Herdman (1992). In section IV, The genus Anabaena has (Rippka et al., 1979) been divided into Anabaena Aphanizomenon and (more elongate), other genera Cylindrospermum, Nodularia, Nostoc, Scytonema and Calothrix were retained without changes by Castenholz (1989). In the revision of Rippka & Herdman (1992), the genera Cylindrospermum, Nodularia, Nostoc, Scytonema and Anabaena have been retained as previously defined (Rippka et al., 1979). The genus Calothrix Rippka et al., (1979) has been changed to Calothrix group because of high genetic diversity, and some members of Calothrix Rippka et al., (1979) have been transferred to *Tolypothrix* (a botanical genus that has been slightly redefined). Anabaenopsis, Aphanizomenon and Microchaete are genera newly represented in the PCC by Rippka and Herdman (1992). In section V, the genera *Chlorogloeopsis* was retained unchanged, the genus *Fischerella* was retained under a more restricted definition, and other seven genera were added in the revision of Castenholz (1989). In the revision of Rippka and Herdman (1992), the nomenclature of the strains assigned to section V is unchanged from that employed by Rippka *et al.* (1979).

In this study, the method used to identify cyanobacteria from natural sources is based on the results of pure culture studies (Rippka et al., 1979). Fig 3.1 shows a flow diagram of this system. Cyanobacteria are divided into five major sections (Orders) according to the morphologically characterised form of reproduction (single cell or chain), differentiation of cells, and number and regularity of planes of cell divisions. Section I was defined as unicellular cyanobacteria that reproduce by binary fission or budding. Cells are coccoid to rod-shaped. Division occurs in one, two or three successive planes at right angles to one another or in irregular planes, resulting in cells occurring singly or in aggregates of varying size. Section II was defined as cyanobacteria that reproduce by the formation of small, spherical cells (baeocytes) produced through multiple fission of a vegetative cell and released through rupture of the fibrous outer wall of the parental cell. Enlargement of vegetative cells is always accompanied by progressive thickening of the fibrous outer wall layer. Section III was defined as filamentous cyanobacteria that undergo binary fission in a single plane and that produce "vegetative" cells only. Heterocysts and akinetes do not occur. Section IV was defined as filamentous cyanobacteria dividing exclusively by binary fission in one plane only and having the potential to produce heterocysts. False branching may occur. Section V was defined as filamentous cyanobacteria dividing by binary fission in more than one plane and having the potential to produce heterocysts. True branching occurs. Characters used



to define genera are mainly based on morphology. In section I & II, for the genera with poor morphological differentiation, other characters like ultrastructure (present or absent thylakoids) and physiology (motility of spores) were also used. The purified isolates were assigned in Section I -- Synechococcus group (Unicellular coccoid to rod-shaped cyanobacteria that divide by binary fission in a single plane. The cells are $<3 \mu m$ in diameter, contain photosynthetic thylakoids located peripherally, and lack structured sheaths), Section II -- Chroococcidiopsis (Cyanobacteria that undergo repeated binary fission in three planes to produce more or less regular cubical cells in an aggregate which is followed by the massive release of immotile baeocytes. The baeocyte initiates growth by enlarging symmetrically into a spherical vegetive cell that, just before the onset of binary fission, attains a size that is characteristic and constant for any given strain), and Section III -- Pseudanabaena (Filamentous cyanobacteria which have conspicuous constrictions at the cross-walls. Cells are longer than broad to isodiametric and are often barrel-shaped. The diameter of trichome ranges from about 1 to 3 µm. Trichome are usually straight and motile. Most strains have polar gas vacuoles), Leptolyngbya (Filamentous cyanobacteria, cylindrical trichome slightly to intensely waved, slightly constricted at the cross walls. Ability to form sheaths around the trichome dependent on the environmental conditions, false branching lacking, cells \pm isodiametric or longer than wide, gas vesicles lacking, thylakoids peripherally arranged, filaments more or less involved into tangled flaky clusters) and Phormidium (Filamentous cyanobacteria, cylindrical trichome slightly to intensely waved, unconstricted or slightly constricted at the cross walls. Ability to form sheaths around the trichome dependent on the environmental conditions, false branching lacking, cells ± isodiametric or shorter or longer than wide, gas vesicles lacking, thylakoids perpendicular to the walls, thallus expanded, extensive, more or less fine). Some characters (cells length:width, special mucilaginous envelopes around cells, mucilage, colonial aggregates, shape of colonies, colour of colonies, thylakoids arrangement) of Anagnostidis and Kamarek (1986, 1988, 1989, and 1990) were also used.

.

.

3.2 Results

3.2.1 Isolation condition

Optimal media and conditions were chosen, with four different media and two incubation conditions used for isolation. Eight samples were isolated by using direct and indirect methods (see 2.2.4). Table 3.1a shows that organisms from Raby pool grew very quickly in BG11, as well as $BG11_{10}$. For growth in liquid media, BG11 appears slightly better than $BG11_{10}$, but for solid media, BG11₁₀ appears slightly better than BG11₁₀, but for solid media, BG11₁₀ appears slightly better than BG11. For medium C, the organisms always grew slower, appearing a few days later than in other media, but more cyanobacteria grew in liquid media, and cyanobacterial growth was faster than green algae. As shown in table 3.1b, the cultures growing in room conditions were very slow compared to those growing in the light cabinet.

3.2.2 Simple recognition of cyanobacteria

Recognition of cyanobacteria from media is greatly aided by their characteristic colour, and is mainly determined by the relative amounts of the major light-harvesting, water-soluble pigments phycocyanin and phycoerythrin (Glazer, 1988), minor colour contributions being due to chlorophyll a and carotenoids (Sandmann, 1988). Fig. 3.2 shows absorption spectra of phycobiliproteins of cyanobacteria (from Lazaroff, 1973; Anagnostidis and Komarek, 1988).

Media		Sample number *							
	Туре	L1	L2	L3	L4	L5	P6	P7	P8
BG11	Solid	3	10	4	2	3	2	2	2
	Liquid	2	8	5	3	5	3	3	3
BG11/10	S	2	7	4	2	3	2	2	2
	L	3	10	6	4	4	3	3	3
ASM-1	S	3			3	7	2	2	2
	L	4			7	4	3	3	3
С	S	4	15	5	2	4	2	2	2
	L	5	14	12	4	6	4	3	3

Table 3.1 Comparison of organism growth under different condition

a). light cabinet (at 25-30 °C, with CO₂ and two warm white fluorescence lights, shaking at 100 rpm)

b). room (at 15-20 °C, with indirect day light from window, no CO ₂ supply, shaking at 100

.

.

Media		Sample number *							
	Type		L2	L3	L4	L5	P6	P7	P8
BG11	S	6	20	18	10	10	6	6	3
	L	7	21	18	4	4	3	3	3
BG11/10	S	6	20	17	10	10	6	6	3
	L	7	27	21	5	5	4	4	4
ASM-1	S	7			10	10	8	7	4
	L	8			7	7	4	4	4
С	S	7	27	21	10	10	7	5	4
	L	7		18	7	7	4	4	4

* Result is the time (days) till medium turn to visible green colour, the large number indicate slow growing.

٠



Figure 3.2 Absorption spectra of phycobiliproteins from cyanobacteria.

A: Absorption spectra from whole cell cyanobacteria. C-- common spectrum of cyanobacteria; 1-- red *Phormdium ectocarpi*; 2-- Oscillatoria sp. 3-- Arthrospira maxima; 4-- Arthrospira africanum; 5-- *Planktothrix agardhii*; 6-- red *Planktothrix rubescens* (from Anagnostidis and Komarek, 1988).

B: Biliproteins of *Nostoc muscorum* A prepared from freeze-thaw extracts of cells, precipitated by ammonium sulphate and speparated on calcium phosphate gel. APC-- allophycocyanin; PC-- phycocyanin; PE I-- phycoerythrin I; PE II-- phycoerythrin II (from Lazaroff, 1973).
Samples from natural environments may contain cyanobacteria (oxygenic photosynthetic prokaryotes, containing carotenoids. chlorophyll а and phycobiliproteins), prochloron (oxygenic photosynthetic prokaryotes, containing carotenoids and chlorophll a, b, but do not contain phycobiliproteins), photosynthetic bacteria (prokaryotes, containing carotenoids, but do not contain chlorophyll a and phycobiliproteins) and eukaryote algae (non-prokaryotes, most of them not containing phycobiliproteins). A direct scan of whole cell samples by UVspectrophotometer (see 2.2.5 method), can differentiate the above organisms. For cyanobacteria, spectra were obtained with more than two peaks at 400-500nm, 615-650nm (and 560-580 nm) and 660-680nm which are carotenoids, phycobiliproteins and chlorophyll a. Respectively for prochloron, two peaks were obtained at 400-500nm for carotenoids and at 650-680nm for chlorophyll a, b. For green algae, two peaks were obtained at 400-500nm for carotenoids and at 650-680nm for chlorophyll a, b, similar to the prochloron result. Then critical examination of the samples under the microscope, ensure that they are prokaryotes. This method has been proved reliable by O₂-evolution measurements (for method see 2.2.6) and pigments (broken cell) test (for method see 2.2.7). The absorption spectra of with cyanobacteria phycocyanin, cyanobacteria with phycocyanin and phycoerythrin, and green-algae, measured on whole cells and a cell-free extract, are shown in Fig. 3.3 and 3.4. The 620nm absorbance maxima were obtained from eight isolates which were isolated from sample sites of Raby pool (see Fig. 2.2 Plate 1). Fig. 3.5 shows the relative composition of coloured isolates from different type of water sample.



Figure 3.3 Absorption spectra (of whole cells) of different organisms.

A, Synechococcus LJM 003; B, Pseudanabaena sp. LJM 007; C, green algae.



Figure 3.4 Absorption spectra (of cell-freeextracts) of different organisms extraction. A, Synechococcus LJM 003; B, Pseudanabaena sp. LJM 007; C, green algae.

.



Figure 3.5 The number of cyanobacterial isolates and the total photosynthetic isolates isolated from different types of water sample from Raby pool by usir direct isolation. The indicating of water samples was listed in table 2.3.

3. 2.3 Purification of isolates

Single colonies of each cyanobacterial sample were subcultured repeatedly until pure cultures were obtained. Purification of the cyanobacteria which produce sheath or aggregates proved the most difficult by repeat subculture. Antibiotics, such as Ampicillin (1mg/ml) was added to BG11 medium. A cyanobacterial culture from the mid-exponential phase of growth was incubated in such medium in dark for 16h. Then such culture was incubated in BG11 without Ampicillin to the midexponential phase of growth, and sub-cultured repeatedly on to BG11 agar plates which content Nystatin (0.02g/l) until pure cultures were obtained. The temperature was kept at room temperature.

3. 2.4 Classification of isolates from different sampling sites.

Identification to generic level was done following the method of Rippka *et al* (1979), Castenholz and Waterbury (1989) and Komarek and Anagnostidis (1986 and 1988). Specific names, following classical phycological texts are also included, because they have been used widely and could therefore aid the comparison between these strains and those reported elsewhere. Each isolate is described in the following text. Ultrastructure and micrograph of each isolates are given. The characters used to differentiate between various isolates are summarised in Table 3.2.

92

1. Synechococcus group

LJM 002

Unicellular short oval to short rod-shaped cyanobacteria that divide by binary fission in a single plane. The cells are $0.6--0.8 \ \mu m$ in diameter, cell length:width are 1.2--1.8, photosynthetic thylakoids present located peripherally, and lack structured sheaths and gas vesicle (see Fig. 3.6 a). Cells are motile with irregular type under light microscopy, do not contain PE or synthesise nitrogenase, and are obligately photoautotrophic. Colonies are pea green colour and spherical or irregular shape, and form soft colonial aggregates.

Figure 3.6. Micrograph of Synechococcus LJM 002. a) transmission electron micrograph of a thin section of strain LJM 002, bar, 0.5 μ m; b) micrograph of cells or colonies of strain LJM 002, bar, 20 μ m. T, thylakoid.





Unicellular short oval to short rod-shaped cyanobacteria that divide by binary fission in a single plane. The cells are 0.6--0.8 μ m in diameter, cell length:width are 1.2--2.0, photosynthetic thylakoids present located peripherally, and lack structured sheaths and gas vesicle (see Fig. 3.7 a). Cells are motile with irregular type under light microscopy, do not contain PE or synthesise nitrogenase, and are obligately photoautotrophic. Colonies are pea green colour and spherical or irregular shape, and form very soft colonial aggregates.

Figure 3.7. Micrograph of Synechococcus LJM 003. a) transmission electron micrograph of a thin section of strain LJM 003, bar, 0.5 μ m; b) micrograph of cells or colonies of strain LJM 003, bar, 10 μ m. T, thylakoid.





Unicellular short oval to short rod-shaped cyanobacteria that divide by binary fission in a single plane. The cells are 0.6--0.8 µm in diameter, cell length:width are 1.2--2.0, photosynthetic thylakoids present located peripherally, and lack structured sheaths and gas vesicle (see Fig. 3.8 a). Cells are motile with irregular type under light microscopy, do not contain PE or synthesise nitrogenase, and are obligately photoautotrophic. Colonies are light pea green colour and spherical or irregular shape, and form soft colonial aggregates.

Figure 3.8. Micrograph of Synechococcus LJM 004. a) transmission electron micrograph of a thin section of strain LJM 004, bar, 0.5 μ m; b) micrograph of cells or colonies of strain LJM 004, bar, 10 μ m. T, thylakoid.





Unicellular short oval to short rod-shaped cyanobacteria that divide by binary fission in a single plane. The cells are 0.6--0.8 μ m in diameter, cell length:width are 1.2--2.2, photosynthetic thylakoids present located peripherally, and lack structured sheaths and gas vesicle (see Fig. 3.9 a). Cells are motile with irregular type under light microscopy, do not contain PE or synthesise nitrogenase, and are obligately photoautotrophic. Colonies are pea green colour and spherical or irregular shape.

.

.

Figure 3.9. Micrograph of Synechococcus LJM 020. a) transmission electron micrograph of a thin section of strain LJM 020, bar, 0.25 μ m; b) micrograph of cells or colonies of strain LJM 020, bar, 20 μ m. T, thylakoid.





Cyanobacteria that divide by continual binary fission in three planes to produce more or less regular cubical cells of the aggregate followed by the massive release of immotile baeocytes. The baeocyte initiates growth by enlarging symmetrically into a spherical vegetive cell, just before the onset of binary fission. The baeocyte is 3--3.5 μ m in diameter, the vegetative cell is 4.5--5 μ m in diameter. Photosynthetic thylakoids present located irregularly (most) or peripherally, and strong structured sheaths and gas vesicles (see Fig. 3.10 a). Cells do not contain PE, but synthesise nitrogenase, and are photoheterotrophically positive for fructose. Colonies are dark pea green colour and of irregular shape.

Figure 3.10. Micrograph of *Chroococcidiopsis* LJM 013. a) transmission electron micrograph of a thin section of strain LJM 013, bar, $1.0 \mu m$; b) micrograph of cells or colonies of strain LJM 013, bar, $20 \mu m$. S, sheath; T, thylakoid.





3. Pseudanabaena

LJM 001

Non-heterocystous filamentous cyanobacteria which have conspicuous constrictions at the cross-walls. The diameter of trichome ranges from about 1.1 to 1.3 μ m, cell length:width are 2.2--3.0, photosynthetic thylakoids present located peripherally, and lack structured sheaths, strains have polar gas vacuoles (see Fig. 3.11 a). Trichomes are straight and slightly motile under light microscopy, do not contain PE or synthesise nitrogenase, and are obligately photoautotrophic, 4--10 cells in one trichome. Colonies are pea green colour and "creeping" mats type.

Figure 3.11. Micrograph of *Pseudanabaena* LJM 001. a) transmission electron micrograph of a thin section of strain LJM 001, bar, $1.0 \mu m$; b) micrograph of cells or colonies of strain LJM 001, bar, $10 \mu m$. G, gas vesicle; T, thylakoid.







Non-heterocystous filamentous cyanobacteria which have conspicuous constrictions at the cross-walls. The diameter of trichome ranges from about 1.1 to 1.3 μ m, cell length:width is 1.8--2.8, photosynthetic thylakoids present located peripherally, and lack structured sheaths, strains have polar gas vacuoles (see Fig. 3.12 a). Trichomes are straight and slightly motile under light microscopy, do not contain PE or synthesise nitrogenase, and are obligately photoautotrophic, 4--10 cells in one trichome. Colonies are pea green colour and "creeping" mats type.

Figure 3.12. Micrograph of *Pseudanabaena* LJM 005. a) transmission electron micrograph of a thin section of strain LJM 005, bar, 0.5 μ m; b) micrograph of cells or colonies of strain LJM 005, bar, 10 μ m. G, gas vesicle; T, thylakoid.







Non-heterocystous filamentous cyanobacteria which have conspicuous constrictions at the cross-walls. The diameter of trichome ranges from about 1.4 to 1.5 μ m, cell length:width is 1.2--2.0, photosynthetic thylakoids present located peripherally, and lack structured sheaths, strains have polar gas vacuoles (see Fig. 3.13 a). Trichome are straight and slightly motile under light microscopy, contain PE or synthesise nitrogenase, and are obligately photoautotrophic, 4--10 cells in one trichome. Colonies are black green colour and "creeping" mats type.

Figure 3.13. Micrograph of *Pseudanabaena* LJM 006. a) transmission electron micrograph of a thin section of strain LJM 006, bar, 0.5 μ m; b) micrograph of cells or colonies of strain LJM 006, bar, 10 μ m. G, gas vesicle; T, thylakoid.







Non-heterocystous filamentous cyanobacteria which have conspicuous constrictions at the cross-walls. The diameter of trichome ranges from about 1.3--1.5 μ m, cell length:width are 1.2--2.0, photosynthetic thylakoids present located peripherally, and lack structured sheaths, strains have polar gas vacuoles (see Fig. 3.14 a). Trichomes are straight and slightly motile under light microscopy, contain PE or synthesise of nitrogenase, and are obligately photoautotrophic, 4--10 cells in one trichome. Colonies are dark brown colour and "creeping" mats type.

Figure 3.14. Micrograph of *Pseudanabaena* LJM 007. a) transmission electron micrograph of a thin section of strain LJM 007, bar, 0.5 μ m; b) micrograph of cells or colonies of strain LJM 007, bar, 10 μ m. G, gas vesicle; T, thylakoid.







Non-heterocystous filamentous cyanobacteria which have conspicuous constrictions at the cross-walls. The diameter of trichome ranges from about 1.3 to 1.5 μ m, cell length:width are 1.3--2.5, photosynthetic thylakoids present located peripherally, and lack structured sheaths, strains have polar gas vacuoles (see Fig. 3.15 a). Trichomes are straight and slightly motile under light microscopy, do not contain PE or synthesise nitrogenase, and are obligately photoautotrophic, 3--4 cells in one trichome. Colonies are pea green colour and "creeping" mats type.

Figure 3.15. Micrograph of *Pseudanabaena* LJM 009. a) transmission electron micrograph of a thin section of strain LJM 009, bar, 0.5 μ m; b) micrograph of cells or colonies of strain LJM 009, bar, 40 μ m. G, gas vesicle; T, thylakoid.





Non-heterocystous filamentous cyanobacteria which have conspicuous constrictions at the cross-walls. The diameter of trichome ranges from about 1.2 to 1.4 μ m, cell length:width is 2.0--2.8, photosynthetic thylakoids present located peripherally, and lack structured sheaths, strains have polar gas vacuoles (see Fig. 3.16 a). Trichomes are straight and slightly motile under light microscopy, do not contain PE or synthesise nitrogenase, and are obligately photoautotrophic, 3--8 cells in one trichome. Colonies are pea green colour and "creeping" mats type.

Figure 3.16. Micrograph of *Pseudanabaena* LJM 010. a) transmission electron micrograph of a thin section of strain LJM 010, bar, 0.5 μ m; b) micrograph of cells or colonies of strain LJM 010, bar, 40 μ m. G, gas vesicle; T, thylakoid.





Non-heterocystous filamentous cyanobacteria which have conspicuous constrictions at the cross-walls. The diameter of trichome ranges from about 1.2 to 1.5 μ m, cell length:width is 1.3--2.2, photosynthetic thylakoids present located peripherally, and lack structured sheaths, strains have polar gas vacuoles (see Fig. 3.17 a). Trichomes are straight and slightly motile under light microscopy, do not contain PE or synthesise nitrogenase, and are obligately photoautotrophic, 3--4 cells in one trichome. Colonies are pea green colour and "creeping" mats type.

Figure 3.17. Micrograph of *Pseudanabaena* LJM 016. a) transmission electron micrograph of a thin section of strain LJM 016, bar, 0.5 μ m; b) micrograph of cells or colonies of strain LJM 016, bar, 40 μ m. G, gas vesicle; T, thylakoid.





Non-heterocystous filamentous cyanobacteria which have conspicuous constrictions at the cross-walls. The diameter of trichome ranges from about 1.0 to 1.45 μ m, cell length:width are 1.8--2.8, photosynthetic thylakoids present located peripherally, and lack structured sheaths, strains have polar gas vacuoles and (see Fig. 3.18 a). Trichomes are straight and slightly motile under light microscopy, do not contain PE or synthesise nitrogenase, and are obligately photoautotrophic, 4--6 cells in one trichome. Colonies are pea green colour and "creeping" mats type.

Figure 3.18. Micrograph of *Pseudanabaena* LJM 019. a) transmission electron micrograph of a thin section of strain LJM 019, bar, 0.5 μ m; b) micrograph of cells or colonies of strain LJM 019, bar, 40 μ m. G, gas vesicle; T, thylakoid.





4. Leptolyngbya

LJM 012

Non-heterocystous filamentous cyanobactertia, cylindrical trichome immotile, but slightly waved at end, slightly constricted at the cross walls. Form thin sheaths around the trichome in dependence on the culture conditions, false branching lacking, cells much longer than wide, gas vesicles lacking, thylakoids peripherally arranged (see Fig. 3.19 a). The diameter of trichome ranges from about 1.2 to 1.5 μ m, cell length:width are 2.0--2.8, Trichomes do not contain PE or synthesise nitrogenase, and are obligately photoautotrophic. Colonies are pea green colour and fascicles or mats type; filaments more or less involved into tangled flaky clusters

Figure 3.19. Micrograph of *Leptolyngbya* LJM 012. a) transmission electron micrograph of a thin section of strain LJM 012, bar, 0.5 μ m; b) micrograph of cells or colonies of strain LJM 012, bar, 20 μ m. G, gas vesicle; T, thylakoid.





Non-heterocystous filamentous cyanobacteria, cylindrical trichome immotile, with small deep constrictions at the cross walls. Thick sheaths around the trichome, false branching lacking, cells \pm isodiametric or shorter or longer than wide, lacking gas vesicles, thylakoids peripherally arranged (see Fig. 3.20 a). The diameter of trichome ranges from about 1.4 to 1.7 µm, cell length:width are 0.7--1.5, Trichomes contain PE or synthesise nitrogenase, and are obligately photoautotrophic. Colonies are black colour and fascicles or mats type. filaments more or less involved into tangled flaky clusters

Figure 3.20. Micrograph of *Leptolyngbya* LJM 014. a) transmission electron micrograph of a thin section of strain LJM 014, bar, 0.5 μ m; b) micrograph of cells or colonies of strain LJM 014, bar, 40 μ m. G, gas vesicle; S, sheath; T, thylakoid.





Non-heterocystous filamentous cyanobacteria, cylindrical trichomes waved at end, unconstricted at the cross walls. Ability to form thin sheaths around the trichome dependent on the culture conditions, lacking false branching, cells \pm isodiametric or shorter or longer than wide; lacking gas vesicles, thylakoids perpendicular to the walls, thallus, extensive, fine, and compact up to leather-like. The diameter of the trichome ranges from about 3.0 to 4.0 µm, cell length:width is 0.8--1.1, Trichomes do not contain PE, but synthesise nitrogenase, and are photoheterotrophic positive for sucrose. Colonies are pea green colour and mats type.

Figure 3.21. Micrograph of *Phormidium* LJM 017. a) transmission electron micrograph of a thin section of strain LJM 017, bar, 1.0 μ m; b) micrograph of cells or colonies of strain LJM 017, bar, 40 μ m. G, gas vesicle; S, sheath; T, thylakoid.


b.



a.

Table 3.2Properties of the cyanobacterial strains isolated from the lakes (Tai,Bala, and Windermere) and the Raby pool

	Unicellular							
		Synecho	coccus	Chr	oococcidiopsis			
Properties	LJM002	LJM003	LJM004	LJM020	LJM013			
Number of sample isolated from	1, 5	1, 4	4	10, 11	6,7			
Reproduction ^{*,b}	binary fission	binary fission	binary fission	binary fission	binary and multiple fission			
Division in one plane ^b	+	+	+	+				
Thylakoids*	+	+	+	+	+			
Sheath layers [*]			·		+			
Synthesis of PE								
Synthesisof nitrogenase					+			
Photoheterotroph								
Glucose								
Fructose					+			
Sucrose								
Ribose								
Glycerol								
VB ₁₂ requirement								
Cell width ^ь (µm)	0.6-0.8	0.6-0.8	0.6-0.8	0.6-0.8	4.0-4.5			

	Unicellular							
Properties		Synechoc	Chi	Chroococcidiopsis				
	LJM002	LJM003	LJM004	LJM020	LJM013			
Cells (lg:lt)	1.2-1.7	1.2-2.0	1.2-2.0	1.2-1.8	1.0-1.3			
Cell shape ^ь	shortly oval	shortly oval	shortly oval	shortly oval	spherical			
Cell motile ^b	++	++	++	++				
Gas vacuoles ^{a,b}								
Special mucilag- inous envelopes around the cells	(+)	(+)	(+)	(+)				
Mucilage (colonia	1)				+			
Colonial aggregates	+++	++	++	+	+++++			
Shape of colonies	spherical, irregular	spherical, irregular	spherical, irregular	spherical, irregular	irregular			
Colour of colonies	s pea green	pea green	light pea green	pea green	dark pea green			

cont. Table 3.2

	Filamentous						
	Pseudanabaena						
Properties	LJM001	LJM005	LJM006	LJM007	LJM010		
Number of sample isolated from	1	1	1	1	6, 7		
Reproduction ^{a,b}	trichome breakage	trichome breakage	trichome breakage	trichome breakage	trichome breakage		
Cells	vegetative	vegetative	vegetative	vegetative	vegetative		
Thylakoids*	peripheral concentric	peripheral concentric	peripheral concentric	peripheral concentric	peripheral concentric		
Sheath layers [*]							
Synthesis of C-PE		 ·	+	+			
Synthesisof nitrogenase							
Photoheterotroph							
Glucose							
Fructose							
Sucrose							
Ribose							
Glycerol							
VB ₁₂ requirement							
Cell width ^b (µm)	1-1.1.3	1.1-1.3	1.4-1.5	1.3-1.5	1.2-1.4		
No. of cells in a trichome	4-10	4-10	4-10	4-10	3-8		

	Filamentous								
	Pseudanabaena								
Properties	LJM001	LJM005	LJM006	LJM007	LJM010				
Cells (lg:lt)	2.2-3.0	1.8-2.8	1.2-2.0	1.2-2.0	2.0-2.8				
Cell shape⁵	cylindrical	l cylindrica	l cylindrica	l cylindrical	cylindrical				
Cell motile ^b	(+)	(+)	(+)	(+)	(+)				
Gas vacuoles ^{a,b}	+, polar	+, polar	+, polar	+, polar	+, polar				
Special mucilag- inous envelopes around the cells	(+)	(+)	(+)	(+)	(+)				
Mucilage (colonia	l)								
Colonial aggregates	+	+	+	+	+				
Type of colonies	"creeping mats	;" "creeping mats	" "creeping mats	g" "creeping" mats	"creeping" mats				
Colour of colonie	s pea green	pea green	black	dark brown	pea green				

cont. Table 3.2

	Filamentous						
	Pseudan	abaena	Leptolyn	Phormidium			
Properties	LJM009	LJM016	LJM012	LJM014	LJM017		
Number of sample isolated from	6, 7	12	6,7	6,7	12		
Reproduction ^{a,b}	trichome breakage	trichome breakage	trichome breakage	trichome breakage	trichome breakage		
Cells	vegetative	vegetative	vegetative	vegetative	vegetative		
Thylakoids*	peripheral concentric	peripheral concentric	peripheral concentric	peripheral concentric	perpendic to the walls		
Sheath layers*			+, thin	+++, thick	(+), thin		
Synthesis of PE				+			
Synthesisof nitrogenase				+	+		
Photoheterotroph							
Glucose							
Fructose							
Sucrose					+		
Ribose							
Glycerol							
VB ₁₂ requiremen	t						
Cell width ^b (µm)	1.3-1.5	1.2-1.5	1.2-1.5	1.4-1.7	3.0-4.0		
No. of cells in a trichome	3-4	3-4	no limite	d no limited	no limited		

	Filamentous Pseudanabaena							
Properties	LJM009	LJM016	LJM0012	LJM0014	LJM017			
Cells (lg:lt)	1.3-2.5	1.3-2.2	2.0-2.8	0.7-1.5	0.8-1.1			
Cell shape ^b	cylindrical	cylindrical	cylindrical	shortly cylindrical	shortly cylindrical			
Cell motile ^b	+	+	++	++	++			
Gas vacuoles ^{a,b}	+, polar	+, polar						
Special mucilag- inous envelopes around the cells				(+)	(+)			
Mucilage (colonial)				+			
Colonial aggregates			+ .	+	+			
Type of colonies	"creeping" mats	"creeping" mats	" mats	mats	mats			
Colour of colonies	pea green	pea green	pea green	black	pea green			

PE, phycoerythrin

a, result of electron microscopy

b, result of light microscopy

-



Figure 3.22. The taxonomic relationships of cyanobacterial isolates from different sampling sites. The relationship was listed by using classification methods of Rippka et al (1979), Castenholz and Waterbury (1989) and Anagnostidis and Komarek (1988). The taxonomic relationships of isolates from different sampling sites is given in figure 3.22.

3.3 Discussion

For isolation of cyanobacteria, some of the more popular media and those utilised in major cyanobacterial culture collections are compared by Rippka (1988). Four different media, two methods and two incubation conditions were chosen here. The four different media were introduced successfully for the isolation of freshwater cyanobacteria. In our experiment, cyanobacteria and green-algae always grew together, and the number of green-algae was always more than cyanobacteria. Media C, BG11 and BG11/10 have been successfully employed for the isolation of cyanobacteria from samples, but no good results were obtained from Medium ASM-1, and Medium C was proved more sensitive for some cyanobacterial (*Pseudanabaena*) isolation. In Medium C, the organisms' initial growth was slower than in other media, possibly due to the formation of H_2O_2 on autoclaving (Van Baalen 1967).

The temperature for isolation suggested by Rippka should also be varied in ranges 5 to 10°C apart. Although most cyanobacteria from moderately temperate habitats will grow well at 20-30°C, many do not tolerate temperatures above 20°C. In our studies, same isolates were obtained from samples at 15-20°C and 20-30°C. The Pseudanabaena grew well at 20-30°C, but incubation above 30°C should be avoided, the temperature of isolation being similar to sampling niche temperature.

Cyanobacteria, particularly in the hands of bacteriologists, are normally kept under a continuous light regime rather than under a light-dark cycle, since early investigations by Allen and Arnon (1955) indicated that such a treatment is not harmful to their growth. In our research, this opinion has been confirmed. For example, continuous light can increase the growth speed of *Pseudanabaena*, and prevents the formation of typical colonial aggregates under 30°C.

For purification, two different bactericidal compounds (ampicillin and Nystatin) have been successfully employed for aggregated and sheathed cyanobacteria. But the treatment must be undertaken at room temperature.

Recognition of cyanobacteria from media will only come from: (1) critical examination of the samples under the microscope, to ensure that they are prokaryotes; (2) a demonstration by growth requirements or O_2 -evolution measurements, to prove that they are not photosynthetic bacteria; (3) an analysis of their pigment content (cell free), to exclude the possibility that they may be prochloron-like organisms (Rippka, 1988). In our studies and review of previous work, we found one simple and direct method (the whole cell pigment test) to recognize cyanobacteria from axenic culture. Cyanobacteria, eucaryote algae and prochloron grew together in the media, when we used freshwater media to recover cyanobacteria from water courses. A number of workers have found that these organisms contain different pigments as previously stated (Rippka 1988, Lattimer 1957, Glazer 1988).

Previously some workers have used cell-free extracts and thylakoid membranes to test organisms for whether they contain phycobiliproteins and chlorophyll (Rippka, 1974, 1988; Glazer 1988). The absorbance maxima of the pure chlorophyll a was obtained at 430 and 663nm (Goedheer 1966). The maxima of the absorbance of the pure chlorophyll b was obtained at 455 and 645nm (Hoffman &Werner 1966); phycobiliproteins are present in the form of hexameric complexes with three different linker polypeptides of 27,000, 33,000Da in intact phycobilisomes, the absorbance maxima being obtained at 620nm (Glazer 1988). In this study the same result was obtained by measurement on a cell-free extract and whole cells. Absorbance maxima of phycerythrin at 560-570 nm and phycocyanin at 600-620nm in cell-free extracts were obtained. One absorbance maximum at 565-575nm and two absorbance maxima at 600-700nm were obtained in whole cell samples; two are phycobiliprotein (565-575 and 620-630nm), one is chlorophyll a (660-670) (see Fig. 6.3).

For other green isolates, only one absorbance maximum was obtained, which appeared to be composed of two components as a shoulder appears on the peak (see Fig. 6.3). This method is quiet simple and can be used for recognition of cyanobacteria.

Fewer cyanobacteria species than algae species were obtained during this work. This could be due to the limited amount of sampling used and the selection of sampling sites.

Cyanobacteria species were not isolated from two water samples (ie, 2 and 3). This was due to their sparseness or absence in these samples. It is likely that cyanobacteria are not suited to the local condition, because of low light intensities and phytotoxic compounds produced by rotten woods and leaves. Another reason

for the lack of cyanobacteria in these five water samples could be the antagonistic effect of other microbial types on cyanobacterial growth or some species present were not readily isolated in the standard media. In our studies, we were not particularly interested in obtaining isolates of the different microbial types from the different environment, but more interested in isolating more different species of cyanobacteria which could be obtained from water samples. So in further isolations, we will collect samples from rich-sun light environments, and employ different media and conditions in isolation.

It was necessary to purify all these cyanobacterial isolates in order to carry out the physiological and genetic studies, and it proved very difficult to obtain axenic cultures, although success was achieved with certain species by using antibiotic treatments. This purification problem was not a novel one, Echlin and Morris, (1965) reported that the difficulties might be due to the close interrelationships between cyanobacteria and the contaminant micro-organisms, making their physical separation difficult.

Differences in structure and development permit the classification among cyanobacteria of five large sections (Rippka *et al*, 1979). These five sections correspond to major taxa recognised by phycologists (see Table 1.2). From section to genera, the most part of these genera do not correspond precisely to major botanical taxa. In *Synechococcus*, cells occur singly, in pairs or in short chains and are devoid of sheaths. Nageli (1849) proposed another genus, *Aphanothece*, for organisms like *Synechococcus* that develop as irregular aggregates united by a common slime layer. Although many strains of *Synechococcus* produce considerable quantities of extracellular slime in culture, none develops in the form of aggregates. A new genus Cyanothece has been

proposed by Komarek (1976) to accommodate some species hitherto placed in *Synechococcus*. Its principal distinguishing character is that cells occur either singly or in pairs, but never in chains. Since the extent of chain formation in organisms of the *Synechococcus* type is strongly dependent on the conditions of culture, so Komarek's proposal was not accepted by Rippka (1979) and Castenholz and Waterbury (1989). In this study, four isolates were placed into genus *Synechococcus*, cluster *Cyanobium* or cluster 5 according to Rippka (1979, 1992) and Castenholz and Waterbury (1989). The first problem of identification for these *Synechococcus* is too small cell size and small length:width (see Table 3.2) to observe cell division in one plane or in two or three plane by light microscope. A further problem is the difficulty of distinguishing between these strains. These strains were roughly separated by slightly different length:width, colonies colour and extent of softness of colonies aggregates. This result was very easily confirmed by RAPDs data (see section 5.6).

The specificity of genus *Pseudanabaena* was mentioned by several authors (Anagnostidis 1961; Rippka *et al.* 1979; Guglielmi and Cohen-Bazire 1982, 1984a,b; Anagnostidis and Komarek 1988; and Castenholz and Waterbury 1989). In this study, a combined approach was used. Eight isolates were placed into genus *Pseudanabaena* according to Rippka (1979) and Castenholz and Waterbury (1989), and seven of them into *Pseudanabaena* sp. according to Anagnostidis and Komarek (1988). The similar strains LJM001 and LJM005; LJM009 and LJM016 were separated by RAPDs data (see section 5.6). The strain LJM009 isolated from Taihu lake of China is extremely similar to strain LJM016 isolated from Bala lake of North Wales. What relationship exists between these two strains, could be very interesting.

The LPP group has been changed a lot in the revisions of Castenholz (1989). All strains in LPP A and some strains in LPP B (Trichome immobile in persistent sheath) have been assigned to the genera Lyngbya (one trichome per sheath) and Microcoleus (two or more trichome per sheath). Other LPP strains (trichome motile, no persistent sheath) have been transferred to Pseudanabaena (strong constriction), Oscillatoria and Trichodesmium (absent or feeble constriction). In the revisions of Rippka & Herdman (1992), the LPP A is renamed Lyngbya. Most of the strains previously assigned to the LPP group B have been assigned to the genera Geitlerinema and Leptolyngbya, and a few strains (more than one trichome per sheath) have been temporarily transferred to Microcoleus. Three isolates were placed into LPP group, LJM 012 and LJM 014 were placed into genus Leptolyngbya, LJM 017 was placed into genus Phormidium according to Anagnostidis and Komarek (1988). The differences of strain LJM 012 and LJM 014 are trichome waved or not, slightly constricted or little deep constricted at the cross walls, form thin sheaths or thick sheaths around the trichome, cells much longer than wide or \pm isodiametric. The differences of Leptolyngbya LJM 012 and LJM 014 with Phormidium LJM 017 are thylakoids peripherally arranged or perpendic to the walls, slightly constricted to little deep constricted or unconstricted at the cross walls, ability to form sheaths around the trichome, and *Phormidium* cells are always shorter than wide. According to revisions of Castenholz (1989), all of these three strains would be placed into genus Lyngbya. It was not accepted in this study.

Chapter 4

.

.

Isolation of

DNA from Cyanobacteria

4.1 Introduction

Twenty years ago, little was known about the properties and organisation of the genome of the cyanobacteria, although a few studies had been published on the chromosomal (c) DNA of some unicellular cyanobacteria (Herdmann, 1976; Herdmann et al., 1979). The first report documenting the presence of covalently closed circular (ccc) DNA in cyanobacteria used Synechococcus 6301 (Asato and Ginoza, 1973). Subsequently, plasmids were investigated in Agmenellun quadruplicatum PR-6 (Roberts and Koths 1979), a number of other unicellular cyanobacteria (Lau and Doolittle, 1979; van den Hondel et al., 1979; Laudenbach et al., 1983; Anderson and Eiserling, 1985) and in filamentous species (Simon, 1978; Friedberg and Seikffers, 1979; Reaston et al., 1980). Lau and Doolittle (1979) indicated that different plasmids within the same strain and different plasmids within different strains might (but do not always) contain restricted regions of sequence homology. They suggested that these regions of homology may be analogous to the transposable genetic elements of bacterial plasmids. This together with indirect, but compelling, evidence for interspecific (or intergeneric) plasmid transfer, indicates that plasmids may play a role in the ecology and evolution of obligately autotrophic prokaryotes. Indeed plasmid profiles have proved useful in analysing genetic relatedness of various cyanobacteria (Lau and Doolittle, 1979; van den Hondel et al., 1979; Lau et al., 1980).

Early studies on cyanobacterial plasmids often involved the use of largescale isolation involving caesium chloride and, in many cases, sucrose gradients (van den Hondel *et al.*, 1979; Lau *et al.*, 1980), although Simon (1978) has used gentle lysis conditions, under which most chromosomal DNA remains membrane bound and can be removed by centrifugation.

To avoid such time-consuming steps, rapid mini-methods have been developed for resolving plasmid DNA (Friedberg and Seikffers, 1979; Lambert and Carr, 1982; Goyal, 1992). Chromosomal and plasmid DNA are often obtained from cells that are treated with lysozyme and lysed with a detergent. RNA and proteins are removed by RNase and protease treatments. The plasmid DNA is then released from the folded chromosomal complex by a shearing step or by RNase treatment (Kado and Liu, 1981). In this chapter some of the reported mini-methods have been modified and tested on laboratory strains and cyanobacterial isolates for the rapid isolation of plasmids and total DNA of suitable quality for restriction analysis and PCR. Initially methods were tested on laboratory cultures of *Synechococcus* PCC7942 (*Anacystis nidulans* R2), which is known to harbour two plasmids, pANI (large plasmid) and pANs (small plasmid) (van den Hondel *et al.*, 1979). In addition, plasmid profiles of selected cyanobacterial strains have been examined with a view to rapid identification of cyanobacteria.

4.2 Results

4.2.1 Comparison of methods for plasmid isolation

The presence of covalently closed circular DNA in Synechococcus PCC7942, was confirmed following purification of plasmid DNA from cleared lysates by CsCl-EtBr density-gradient centrifugation, as described in section 2.2.14.2. Two DNA bands were obtained, The lower band of plasmid DNA was analysed by agarose gel electrophoresis and found to contain two ccc plasmid DNAs, of molecular sizes 48.5 ± 1.5 kb (pANl) and 8 ± 0.3 kb (pANs), and the plasmid sizes were confirmed by restriction, using restriction endonuclease *Bam* HI. The sizes estimated from this study are comparable to the sizes reported by Laudenbach *et al.* (1983) (48.5kb, and 8kb), Lau and Doolittle (1979) (46.1±1.2 kb and 8.08±0.075 kb) and van den Hondel *et al.* (1979) (49.85±1.5 kb and 7.93±0.225).

p

Five rapid mini-methods (see section 2.2.14.2) were compared for plasmid recovery from Synechococcus PCC7942. The yield of plasmid DNA was measured by using the Phoretix ID package (version 2.51) based on the intensity of UV induced-fluorescence of DNA bands on agarose gels. Electrophoresis of the plasmid preparations from Synechococcus PCC7942 (Fig. 4.1), indicated that method 1 and method 2 were both sensitive procedures. A very high yield for pANI (large plasmid) and very low yield for pANs (small plasmid) were obtained by method 1. Both pANI (27% yield of method 1) and pANs were obtained by method 2. Chromosomal DNA and RNA were eliminated in method 1 by heat-treatment (55°C for 2h) and incubation with RNase. For method 2, the chromosomal DNA and RNA were removed by incubation with RNase and overnight precipitation by NaCl. Method 3, method 4 and method 5 were very rapid procedures, but only a low yield pANs was obtained from Synechococcus PCC7942 by method 3; and a low yield of plasmid pANI by using method 4. No plasmids were obtained by method 5. When the cell sample was subjected to freezing followed by thawing before extraction of the DNA, the ratio of the ccc form of plasmid DNA to oc form was decreased. The use of frozen and thawed cells produced 0%-40% less of the large plasmid (pANI) and a similar amount of the small plasmid (pANs), together with 30%-50% more RNA compared with the unfrozen sample (Fig. 4.2).



Figure 4.1. Agarose gel electrophoresis of plasmid DNA prepared by five rapid methods (lane m¹-m⁵) from Synechococcus PCC7942. Separation was achieved on 0.8% (w/v) agarose horizontal gel as described in materials and methods (section 2.2.15).



Figure 4.2. Agarose gel electrophoresis of plasmid DNA preparation. Plasmid prepared by two rapid methods: method 1 (lane m^1) and method 2 (lane m^2) from a fresh cell sample and from freeze-thawed cell sample (m^f) of *Synechococcus* PCC7942.

4.2.2 Analysis of extracted plasmid DNA

Electrophoresis of plasmids prepared from *Synechococcus* PCC7942 after heat-treatment showed the disappearance of the open circle (oc) and linear (l) DNA and simultaneous appearance of a faster-migrating band, presumably of single-stranded (ss) DNA, whereas no change occurred in the position of the ccc DNA bands (Fig. 4.3). The large plasmid, before heat treatment, comigrated with the chromosomal DNA. Upon heat treatment the plasmid remained at its original position, whereas the chromosomal DNA was completely denatured. Using plasmid DNA prepared by method 1, the pANI and pANs DNA bands did not change their position after heat treatment (pANs DNA band not clear in Fig. 4.3), suggesting that the plasmid DNA was in ccc form. For method 2, following this treatment the chromosomal DNA (c DNA) changed to ss DNA, and the plasmid DNAs were both in ccc form. For method 3, similarly some c DNA changed to ss DNA, whilst for method 4 heat treatment indicated that the large band was pANI in the oc form and pANs was ccc DNA.

4.2.3 Effect of culture age on plasmid isolation

Differences in the yield of plasmid DNA (on the basis of UV-induced fluorescence of DNA samples) were obtained with cultures of *Synechococcus* PCC7942 of different age (Fig.4.4). It can be seen, from five samples (1.0 mg) of different exponential phase cultures, that plasmid yield was a function of growth: In the early exponential phase (optical density at 720nm (OD₇₂₀) ~ 0.08-0.3), a high yield of pANl was obtained, but pANs was not found. For late exponential phase culture (OD₇₂₀ >1.6-2.0), pANl and pANs both were obtained in low yield. In the



Figure 4.3 Agarose gel electrophoresis of characterisation plasmid by heat treatment. heat-treated (lane m^h) and unheated (lane m) ethanol precipitated plasmid DNA prepared by four rapid methods m¹-m⁴ (Methods 1-4 as described in section 2.2.14.1) from *Synechococcus* PCC 7942. c, chromosomal DNA; ss, single-stranded DNA; oc/l, open circular/linear DNA; ccc, covalently closed circular DNA.



B.





Figure 4.4 Effect of culture age on plasmid isolation. (A). Agarose gel electrophoresis of plasmid DNA prepared by method 1 from 1 mg cultures of *Synechococcus* PCC7942 at different ages. Lane a or a', at 3rd day ($OD_{720}=0.206$ or 0.358); lane b or b', at 5th day ($OD_{720}=0.604$ or 0.923); lane c or c', at 7th day ($OD_{720}=1.262$ or 1.597); lane d or d', at 9th day ($OD_{720}=1.795$ or 1.924) and lane e or e', at 12th day ($OD_{720}=2.071$ or 2.121). Separation was achieved on 0.8% (w/v) agarose horizontal gel as described in section (2.2.15). (B). The different intensities of the small plasmid (pANs) isolated from a culture of *Synechococcus* PCC7942 at different ages. Each point is the mean \pm SD from three separate experiments. The intensities of DNA bands were measured by using Phoretix 1D software. (C). Growth curve (I) OD_{720} versus time ((day) and (II) Log OD_{720} versus time (day)) of *Synechococcus* PCC7942 from three separate experiments inoculated with 0.5%, 1.0% and 1.5% stock culture (1mg dry weight cells per ml). Axis OD_{720} of 1.0 corresponds to approximately 10mg dry weight of cells per 10ml of culture (Lambert *et al.*, 1982).

mid-exponential phase (OD₇₂₀ 0.3-1.6), the large and small plasmids were both routinely (5 replicates) obtained in high yield. Thus for further plasmid isolations, mid-exponential phase cultures were used.

4.2.4 Plasmid content of selected cyanobacteria

Different numbers and sizes of plasmid DNA were obtained from four laboratory strains and five isolates of cyanobacteria by using the rapid methods 1-5 (see Table 4.1). The sizes of the plasmids were confirmed by restriction with *Bam* HI, and using *Hind* III digested λ DNA as standards. For the isolates, *Synechococcus* sp.: LJM 002, 003 and 004, and *Pseudanabaena* sp.: LJM 001 and 005, no plasmids were obtained by any of the methods used. Plasmid DNA could be obtained from the unicellular laboratory strains by using methods 1, 2, 3, and 4. No plasmid DNA was obtained from the filamentous laboratory strains (*Nostoc* PCC7120 and *Nostoc palaeo*) tested by using methods 3, 4 and 5. For strain *Nostoc palaeo*, the smaller plasmid of 42 kb was not always obtained by any method used. It has previously been reported by Laudenbach (1983) that *Synechococcus* PCC 7942 contains two plasmids of 8.0, and 48.5 kb, and Simon (1978) reported that *Nostoc* PCC 7120 contained four plasmids of 3.3, 33, 42, 74 Megadaltons (MD). Two plasmids of similar size to those of 33MD (=49.5 kb) and 3.3MD (=4.95 kb) were obtained in this study.

			Number of plasmids, (molecular size, kb) ^c)°
Cyanobacterial strain ^a		Description ^b	Method 1	Method 2	Method 3 M	ethod4 N	1ethod5
Laboratory strain							·
Synechococcus	PCC 7942	U	2(8, 48.5)	2(8,48.5)	2(8,48.5)	1(8)	0
Synechocystis	PCC 6308	U	1(48)	1(48)	1(48)	0	0
Nostoc	PCC 7120	HF	1(45)	2(4.5,45)	0	0	0
Nostoc	Palaeo	HF	1(55)	2(42,55)	0	0	0
Isolates							
Synechococcus	LJM 002	U	0	0	0	ND	ND
Synechococcus	LJM 003	U	0	0	0	ND	ND
Synechococcus	LJM 004	U	0	0	0	ND	ND
Pseudanabaena	LJM 001	F	0	0	0	ND	ND
Pseudanabaena	LJM 005	F	0	0	0	ND	ND

Table 4.1 Plasmid content of selected cyanobacterial strains

a, Strains were grown in medium BG11 as described in materials and methods section 2.1.3.

b, U: unicellular F: filamentous HF: heterocystous filamentous.

c, Methods 1, 2, 3, 4, 5 are described in section 2.2.24.1. ND, not determined.

4.2.5 Isolation of total genomic DNA

Total genomic DNA was isolated from cyanobacteria by the mini-methods of Porter (1988) and Neilan (1995) (as described in section 2.2.14.3). Both methods proved to be rapid and reliable (Fig. 4.5 A). Amounts of DNA recovered were within the range 4.5µg-8.5µg per ml for mid-exponential phase cultures of *Synechococcus* PCC 7942 using either method. The method of Neilan (1995) was found to be more effective than the method of Porter (1988) for removal of protein from the lysate. The total genomic DNA obtained from either method was suitable for amplification by PCR and the PCR-amplified products, using the random primer ABA-10 (see Table 2.4), were identical in each case (Figure 4.5 B). Identical amplified products were also obtained using either freshly isolated DNA or DNA stored for one year at -20 °C from Porter's method (Figure 4.5 C), indicating that the isolated DNA was stable.

It should be noted that genomic DNA obtained simply by cell lysis could also be amplified by PCR (see Fig. 4.5 B, lane 2).

4. 2. 6. Restriction analysis of plasmid DNA and genomic DNA

Plasmid DNA obtained from *Synechococcus* PCC 7942 by method 2 and total genomic DNA obtained by method 1 (Porter, 1988) were restricted using *Bam* HI. Six distinct bands were obtained following restriction of the plasmid DNA, but only smears from the genomic DNA (Fig 4.6). This may due to the fact that too many bands were generated by restriction of the genomic DNA for effective resolution.



B.



A.



Figure 4.5 Agarose gel electrophoresis of total genomic DNA from Synechococcus PCC 7942. (A) Total genomic DNA isolated by different methods: Lane 1, Neilan's method (Neilan, 1995); 2, Porter's method (porter, 1988). (B). PCR-amplified products of total genomic DNA (obtaied from Porter's method or Neilan's method or direct cell lysis) by using random primer ABA-10 (as described in section 2.2.19). Lane 1, Porter's method; 2, direct cell lysis (section 2.2.14.3); 3, Neilan's method. (C). PCR-amplified products of total genomic DNA (isolated by Porter's method) by using random primer ABA-11, ABA-9 (section 2.2.19) to show the effect of storage. Lane 1 and 5, from freshly isolated DNA, 2 and 6 DNA, after one year storage (at -20 °C) from Nostoc paleo; Lane 3 from freshly isolated DNA from Synechococcus PCC 7942; 4, DNA after one year storage (at -20 °C) from Synechococcus PCC 7942.

C.



M 1 2 3 4 5 6

Figure 4.6 Agarose gel electrophoresis of *Bam* HI restricted plasmid and genomic DNA from *Synechococcus* PCC 7942. Lane 1, plasmid pANI; 2, *Bam* HI restricted plasmid; 3, plasmid pANI; 4, *Bam* HI restricted plasmid; 5, genomic DNA; 6, *Bam* HI restricted genomic DNA; M, λ *Hind* III DNA.

4.3 Discussion

When isolating DNA from cyanobacteria, a major difficulty was to achieve gentle but efficient cell lysis of the different strains used. One reason for this difficulty may be that some cyanobacteria possess a thick outer sheath, usually of undetermined composition (Potts, 1984), to interfere with lysis. Efficient cell lysis is crucial to the study of plasmids particularly where they are be present in very low copy number.

Method 2 of the rapid mini methods for isolation of plasmid DNA, was shown to be the most successful using either unicellular or filamentous cyanobacteria. This method, which is based on the method of van den Hondel and co-workers (1979), has been developed here for application to a range of cyanobacterial strains. Lambert and Carr (1982) also found a modification of the method of van den Hondel and co-workers (1979) to provide reproducible results. In our studies, incubation with RNase and overnight precipitation by using NaCl proved most effective for removal of chromosomal DNA and gave better results than the method of Lambert and Carr (1982), which involved 2 h precipitation by using NaCl, without RNase treatment.

Method 1 proved to be a better procedure than the remaining methods. For unicellular strains large yields of the plasmids were obtained, but it did not prove very effective for all the filamentous strains tested. The effectiveness of methods 3 and 4 varied depending on the strain tested, whilst method 5 did not yield plasmid DNA from any of the cyanobacterial strains tested. Differences in plasmid yield were obtained with cultures of the same strain, but at different stages of growth. Mid-exponential phase cultures (OD_{720} 0.3-1.6) produced optimal plasmid yields for all strains tested. Lambert and Carr (1982) also obtained maximal yields of plasmid DNA from exponential phase cultures. Two reasons for the lower yield from late-exponential phase ($OD_{720} > 1.6-2.0$) cultures, could be due to the cell wall or sheath layer becoming thicker, thereby leading to inefficient lysis. Alternatively, variations in plasmid copy number during different growth phases could account for differences in plasmid yield.

The plasmid DNA isolated from mid-exponential phase cultures of cyanobacterial strains by method 2 was of suitable quality for restriction analysis. Further studies on cyanobacterial plasmids therefore employed such methodology for DNA isolation.

Two of the four laboratory strains of cyanobacteria studied here, namely *Synechococcus* PCC 7942 and *Nostoc* PCC 7120 were all known to possess plasmids (van den Hondel *et al.*, 1979; Lambert and Carr, 1982), and the methods employed here, except method 5, confirmed this. It has previously been reported by Simon (1978) that *Nostoc* PCC 7120 contains four plasmids of 3.3, 33, 42, 74 MD. Lambert and Carr (1982) have reported that one plasmid (3.0 MD) was obtained from *Nostoc* PCC 7120 by using a modification of the method of van den Hondel and co-workers, (1979). Two plasmids of similar size to those of 33MD (=49.5 kb) and 3.3MD (=4.95 kb) were obtained in this study. These findings indicate that rapid methods of plasmid isolation are limited, such that it is difficult to recover all plasmids in a particular cyanobacterial strain every time.

Different strains harboured different numbers and sizes of plasmids, irrespective of the degree of relatedness. Table 4.2, shows that less than 50% of cyanobacterial strains tested here contained plasmids. Although this represents only a small sample size, it is in line with previous work on plasmid profiles in different cyanobacteria (see for e.g. Lau, *et al.*, 1980; Lambert and Carr, 1982; Kohl *et al.*, 1988 and Goyal, 1992). Plasmid profiles may be of limited use in determining the relationships between strains that contain plasmids acquired through horizontal gene transfer; however they would not be useful for establishing classification schemes.

In this study all the strains examined were grown in BG11 medium under the same environmental conditions. Whether changes in environmental regimes may affect plasmid profiles remains to be determined.

Total genomic DNA suitable for PCR (see Fig. 4.4 B) was consistently isolated from cyanobacterial strains by the methods of Porter (1988) (see 2.2.14.3 method 1) and of Neilan (1995) (see 2.2.14.3 method 2). Both methods were reliable and reproducible. The genomic DNA isolated by the method of Porter (1988) was found to remain stable following long term storage (one year) at -20 °C. So further studies with genomic DNA employed this method. Chapter 5

-

Identification of cyanobacteria by random amplified polymorphic DNA (RAPD) analysis

5.1 Introduction

Traditional techniques for identification and systematic of cyanobacteria have relied primarily on observed morphological characteristics. However, morphology may change depending on environmental conditions (Evans *et al.*, 1976) and the diversity of strains within a culture may be suppressed by selective culturing conditions (Doers and Parker, 1988). Photosynthetic pigment content, lipid composition, differentiated cell structures and other characters may also be subject to change because of the variable expression of cyanobacterial gene products in culture (Kenyon *et al.*, 1972; Klein *et al.*, 1973). The limitations of such methods have prompted the development of molecular biological techniques for identification of cyanobacteria.

The use of DNA markers of cyanobacteria in genetic mapping, genetic diagnostics, molecular taxonomy, and evolutionary studies has been well established (Wilmotte, *et al.*, 1984; Wood, *et al.*, 1990; Wilmotte, 1994; Neilan, 1995). The most commonly used DNA markers are restriction fragment length polymorphisms (RFLPs). Detection of RFLPs by DNA blot hybridisation is laborious and incompatible with applications requiring high throughput. Genetic tests based on the polymerase chain reaction (PCR) are rapid, sensitive and simple to perform. Previously required amounts of a specific gene could be determined by RFLPs or DNA-DNA hybridisation from cells, now even a single gene copy can be detected following nucleic acid amplification.

The polymerase chain reaction (PCR) is an *in vitro* method for amplifying selected nucleic acid (DNA or RNA) sequences (Mullis *et al.*, 1986; Mullis and Faloona, 1987; Mullis, 1990). The method consists of repetitive cycles of DNA

denaturation, primer annealing, and extension by DNA polymerase (Fig. 5.1). Two oligonucleotide primers flank the DNA segment to be amplified. They hybridise to opposite strands of the target sequence, such that synthesis proceeds across the region between the primers, replicating that DNA segment. The product of each PCR cycle is complementary to and capable of binding primers, and so the amount of DNA synthesised is doubled in each successive cycle (Saiki *et al.*, 1988).

Williams and co-workers (1990) and Welsh and colleagues (1990) have described a novel type of genetic marker that is based on DNA amplification by PCR, but requires no knowledge of target DNA sequence. Such markers, called RAPD (random amplified polymorphic DNA) markers, are generated by the amplification of random DNA segments with single primers of arbitrary nucleotide sequence. RAPD markers can be used for genetic mapping applications, as well as for genetic diagnostics. The assay is non-radioactive, requires only nanogram quantities of DNA, and is applicable to a broad range of species. In addition, the presence of a single RAPD band would be diagnostic for a sequence totalling 20bp in the target DNA (10 bases at each end of the DNA segment that is amplified). An average of five amplified bands per primer would mean that each primer is diagnostic for $5\times20=100$ bp in the template. This is in contrast to an RFLP, which is diagnostic for only 12bp per probe-enzyme combination (Williams, *et al.*, 1993).

To perform a RAPD assay, a single oligonucleotide of an arbitrary DNA sequence is mixed with genomic DNA in the presence of a thermostable DNA polymerase and a suitable buffer, and then is subjected to temperature cycling conditions typical of the polymerase chain reaction. The products of the reaction depend on the sequence and length of the oligonucleotide, as well as the reaction


Cycle 1: denaturing template DNA:

Figure 5.1 A diagrammatic representation of the PCR reaction. During the first cycle, template DNA is first denatured at 94 °C in the presence of primers. Lowering the temperature to 30-60 °C allows the primers to anneal to the template DNA. Extension occurs as the temperature is raised to 72 °C, to form the first two 'long products'. These long products have the PCR primer sequences at their 5' ends. During the second cycle, the two long DNA products made in the first cycle, now also become template DNA for primer binding and extension, to form two more long products, and two 'short products' that comprise only the DNA sequence between the two primers. After 30 cycles, well over 10⁶ molecules of the specific short product should have been amplified.

conditions. At an appropriate annealing temperature during the thermal cycle, the single primer binds to sites on opposite strands of the genomic DNA that are within an amplifiable distance of each other (within a few thousand nucleotides), and a discrete DNA segment is produced. The presence or absence of this specific product, although amplified with an arbitrary primer, will be diagnostic for the oligonucleotide-binding sites on the genomic DNA (Fig. 5.2). In practice, the DNA amplification reaction is repeated on a set of DNA samples with several different primers, under conditions that result in several amplified bands from each primer. Polymorphic bands are noted, for example, between parents of a cross, and the polymorphisms can be mapped in a segregating population. Often a single primer can be used to identify several polymorphisms, each of which maps to a different locus (Welsh *et al.*, 1990; Williams *et al.*, 1993).

Williams and co-workers (1990) used shorter, arbitrary primers, 9 or 10 nucleotides in length, and low stringency cycles to amplify DNA polymorphisms from several eukaryotic and prokaryotic organisms. When separated by agarose gel electrophoresis and stained with ethidium bromide, up to about 10 products were detected. In most studies, polymorphic products were used as genetic markers. Hamelin and colleagues (1993) used seven random amplified polymorphic (RAPD) markers to identify the ascomycete *Gremmeniella abietina* between the North American and European fales. McCutcheon and co-workers (1993) applied the RAPD method to assess genotypic diversity in populations of *Rhabdocline parkeri* from Douglas fir. RAPD analysis has also been used to distinguish among clinical isolates of *Helicobacter pylori*, a bacterium whose long term carriage is associated with gastritis, peptic ulcers and gastric carcinomas (Akopyanz *et al.*, 1992). Furthermore, Kaukas and colleagues (1994) produced RAPD profiles, by using four





oligonucleotide primers, with genomic DNA from 15 isolates of *Schistosome*. Both inter- and intra-specific variation were noted. The results confirm that RAPD profiles may be used for both strain and species differentiation and for the generation of phylogenetic trees. In addition, the method has been used to study the cyanobacteria symbionts of *Azolla-Anabaena* (Eskew *et al.*, 1993). Fingerprints that were unique to the symbionts from three different *Azolla* sp. were generated and the maternal transmission of one symbiont was indicated. Neilan (1995) demonstrated the use of multiplex random amplified polymorphic DNA markers (MRAPDs) to distinguish between the genera *Anabaena* and *Microcystis*. This approach uses multiple random primers for DNA amplification. It was possible to discriminate among all the toxigenic cyanobacteria studied to three taxonomic levels of genus, species and strain. Clearly RAPD analysis is proving very promising for identification of a wide range of organisms to the strain level.

In this study, the RAPD technique has been developed and optimised for cyanobacterial DNA, with a view to its use to identify specific cyanobacterial strains. The RAPD markers generated have been applied to a phylogenetic analysis of the strains to assess genetic relatedness.

5.2 Results

5.2.1 Reaction conditions for RAPD-PCR

Titration of the reaction constituents was carried out to determine optimum conditions for DNA amplification by PCR using different template DNA concentration, primer concentration, magnesium ion concentration, dNTP concentration and different annealing temperature. Template DNA concentrations in the range $0.55 \text{ng/}\mu l$ (lane 9) to $8.4 \text{ng/}\mu l$ (lane 5) gave clear, reproducible bands using Biolin-AmpliTaq polymerase (Fig. 5.3. A). Whilst for amplification using DyNAZyme II, template DNA concentrations in the range $17.2 \text{pg/}\mu l$ (lane 13) to $1.1 \text{ng/}\mu l$ of the reaction mixture (lane 10) gave clear, reproducible bands (Fig. 5.3. B). Template DNA concentration dependencies were not observed from $0.275 \text{ng/}\mu l$ (lane 11) to $1.1 \text{ng/}\mu l$, but they were observed between $17.2 \text{pg/}\mu l$ and $68.8 \text{pg/}50 \mu l$ (lane 12). High concentrations of genomic DNA (more than $8.4 \text{ng/}\mu l$) often resulted in smearing or in a lack of clearly defined bands (Fig. 5.3 A, lane 1, 2, 3, 4). At genomic DNA concentrations below $17.2 \text{pg/}\mu l$ of the reaction mixture (see Fig.5.3 B, lane 13), some bands were lost, and new bands appeared in the RAPD profile.

For reproducibility, it is important to note that both the magnesium ion concentration and the annealing temperature affected the relative intensity of amplified bands. PCR-amplified products were obtained with a magnesium chloride concentration between 1.5 and 5mM, but not with a concentration of 0.5mM. As the magnesium concentration increased, some DNA segments were amplified more efficiently whilst others were amplified less efficiently (Fig 5.4). Therefore, the same magnesium chloride concentration (2.5 mM), which gave clear bands, was used for all amplifications. The annealing temperature can also affect the relative amount of some amplified bands (Fig. 5.5). Similar amplified bands were obtained with the annealing temperature at 30°C or 34°C. However, different amounts of some bands were obtained with the annealing temperature over 38°C, as judged by band intensity. An annealing temperature of 42 °C was found to be too high to obtain reliable and reproducible amplification with many of the 10-mers tested.



Figure 5.3. Agarose gel electrophoresis showing the effect of DNA concentration on the RAPD assay. Different concentrations of *Nostoc mus* DNA were amplified by program 1 (see section 2.2.19) with primer A-9. Each sample was amplified in duplicate to assess the reproducibility of the reaction. A, DNA samples amplified by Bioline-AmpliTaq. Lane 1, 133.6 ng/µl of the reaction mixture; 2, 66.8 ng/µl; 3, 33.4 ng/µl; 4, 16.7 ng/µl; 5, 8.35 ng/µl; 6, 4.2 ng/µl; 7, 2.1 ng/µl; 8, 1.05 ng/µl; 9, 0.525 ng/µl reaction; B, DNA samples amplified by Finnzymes-DyNAZyme II. Lane 10, 1.1 ng/µl of the reaction mixture; 11, 275 pg/µl; 12, 68.75 pg/µl; 13, 17.2 pg/µl; 14, 4.3 pg/µl; 15, 1.07 pg/µl; 16, 0.268 pg/µl.



Figure 5.4. Agarose gel electrophoresis of RAPD products of Synechococcus PCC7942 DNA obtained with different magnesium concentrations. Lane 1-4: 0.5 mM, 1.5mM, 2.5mM and 5.0mM magnesium chloride in amplification reaction buffer. The products were amplified with the primer OPA-9 and OPA-10 under standard conditions with program 1 (see section 2.2.19).



Figure 5.5. Agarose gel electrophoresis of RAPD products using different annealing temperatures for the primers. *Synechococcus* PCC7942 DNA was amplified at four different annealing temperatures: A, 42°C; B, 38°C; C, 34°C; D, 30°C. Standard conditions were used (program 4, section 2.2.19). Three different primers were used: lane 1, ABA-10; lane2, ABA-11 and lane 3, primer pair ABA-10+11.

Increasing the annealing temperature above 38 °C increased the stringency and in some cases altered the RAPD pattern, with fewer amplification products generated.

No effect of deoxynucleotide triphosphate concentration on the relative intensities of the amplified bands was noted. The same RAPD profiles were generated with deoxynucleotide triphosphate concentrations of 50, 100, 150, 200, 250 μ M for each of the four bases (see Fig. 5.6). An average deoxynucleotide triphosphate concentration of 150 μ M for each of the four bases was chosen for generating RAPDs.

Primer concentrations within the range 0.11 to 0.95μ M were most effective (see Fig. 5.7). At primer concentrations of 0.95μ M or over, the intensity of the larger bands decreased and that of the smaller bands increased. Whilst at concentrations below 0.11μ M, smaller sized bands were not produced in a concentration dependent manner, and the profiles were not reproducible. A concentration of AmpliTaq of 1 units per 50 μ l reaction as recommended by the manufacturers gave consistent amplification products for every species studied.

The DNA polymerase used affected the out-come of the amplification reaction. Different patterns of amplified bands were obtained when using different DNA polymerases (AmpliTic and AmpliTaq), and when using the same DNA polymerase from different manufacturers with a given DNA template and primer. DNA polymerases from four companies were compared. In each case the most intense RAPD bands were of the same size, however faint, variable bands were also generated (Fig. 5.8).



Figure 5.6. Agarose gel electrophoresis showing the effect of deoxynucleotide triphosphate concentration on the RAPD assay. Nostoc mus DNA was amplified by program 1 (see section 2.2.19) using primer A-9. Each sample was amplified in duplicate to assess the reproducibility of the reaction. Different deoxynucleotide triphosphate concentrations were used. Lane 1, 50 μ M; 2, 100 μ M; 3, 150 μ M; 4, 200 μ M; 5, 250 μ M.



Figure 5.7. Agarose gel electrophoresis showing the effect of primer concentration on the RAPD assay. Nostoc mus DNA was amplified by program 1 with primer A-9 (see section 2.2.19). Each sample was amplified in duplicate to assess the reproducibility of the reaction. Different primer concentrations were used. Lane 1, 0.95 μ M; 2, 0.32 μ M; 3, 0.11 μ M; 4, 0.03 μ M; 5, 0.01 μ M.



Figure 5.8. Agarose gel electrophoresis of RAPDs generated using different DNA polymerases. DNA from *Nostoc mus* was amplified by program 1 (see section 2.2.19) with primer ABA-11, using different DNA polymerases. Lane 1, AB-AmpliTic; 2, MBI-AmpliTaq; 3, Bioline-AmpliTaq; 4, Promega- AmpliTaq. M, Molecular size 1 Kb markers (Kb) are as indicated.

Five PCR programs were tested. Variation of amplified bands was noticed by using the different programs (Fig. 5.9). Program 4, gave clear bands for all the strains tested with either primer ABA-7 or ABA-10. The results from programs 1 and 2 were similar, except for amplification of Anabaena cylindrica DNA with primer ABA-10. No PCR products were detected for Synechococcus LJM003 with either primer ABA-7 or ABA10 by using program 3. Program 5 always gave slightly different larger sized fragments. The amplification reaction was performed separately on a Perkin-Elmer Cetus DNA thermal cycler and a HyBAID OmniGene temperature cycler using PCR program 1 (see section 2.2.19). More bands were obtained from the reaction using the HyBAID OmniGene temperature cycler (Fig. 5.10 A). It was found that the transition interval time on automatic setting was different for the two PCR machines: 72 s for the Perkin-Elmer and 82 s for the HyBAID cycler between the denaturation (94 °C) and annealing (34 °C) temperature. After setting the transition interval between melting and annealing temperature to 135 s (program 4 and 5) for both machines, the differences between the amplified products, using either machine, were reduced (Fig. 5.10 B).

The RAPD technique was optimised for the various cyanobacterial strains. Each PCR mix contained 1×PCR amplification buffer, 2.5mM MgCl₂, 150 μ M each of dATP, dCTP, dGTP and dTTP, 0.2 μ M primer, 0.1 μ g of template DNA, and 1 unit of AmpliTaq polymerase, in a total reaction volume of 50 μ l. Amplifications were performed in a Perkin-Elmer Cetus DNA thermal cycler (UK) using program 4 for 35 cycles (section 2.2.19).



Figure 5.9. Agarose gel electrophoresis of RAPDs generated using different PCR programs. DNA samples from a, *Synechococcus* LJM003; b, *Pseudanabaena* LJM001; c, *Anabaena cylindrica* were amplified with primer (A) ABA-7 or (B) ABA-10, using DNA polymerase, Bioline-AmpliTaq (see section 2.2.19). Lane 1, program 1; 2, program 2; 3, program 3; 4, program 4; 5, program 5. M, 1 Kb marker.



Figure 5.10. Agarose gel electrophoresis of RAPDs generated using different PCR machines and different programs. DNA samples from *Synechococcus* PCC7942 were amplified with different primers. Lane 1, ABA-10; 2, ABA-11; 3, ABA-10+11, and DNA polymerase, Bioline-AmpliTaq, using (a) Perkin-Elmer Cetus DNA thermal cycler and (b) HyBAID OmniGene temperature cycler. Fig. A, amplification using program 1; Fig. B, amplification using program 4 (see section 2.2.19).

5.2.2 Primer specificity

The specificity of primer sequences for amplifying DNA is very important for reproducibility of the RAPD profile. Primer AB1-11 generated two strong bands with DNA from Nostoc mus. These bands were not changed when the magnesium ion concentration and annealing temperature were changed (Fig.5.11). In order to increase reproducibility of the RAPD profile, combinations of two 10mer oligonucleotides were tested on Synechococcus PCC7942 DNA at different magnesium ion concentrations (Fig. 5.12 and 5.13). In Figure 5.12, a similar DNA pattern was obtained with primer ABA-10 at magnesium chloride concentrations of 1.5, 2.5 and 5mM. A magnesium chloride concentration dependency was observed for DNA patterns amplified from primer ABA-9 and the combination ABA-9+10. In Figure 5.13, the combination of two 10-mer oligonucleotides did not increase the specificity of primers to generate a clear pattern. In the case of primers ABA-4 and 9 and AB1-1 and 17 none of the bands generated with the single primers remained when using primer pair ABA-4+9 or AB1-1+17. Furthermore, the pattern from the primer combinations was not as clear as the pattern from the single primers. With primer pair ABA-10/11, all the bands amplified from primer ABA-10 remained, and some new bands were generated.

The recommended primer for RAPDs under standard conditions, is the 10base oligonucleotide primer containing 60%-80% G+C (Williams *et al.*, 1993). The ability of forty 10-base oligonucleotide primers (Kit AB1 1-20; Kit ABA 1-20; 60%-70% G+C; see Table 2.4) to amplify genomic DNA from different laboratory strains was tested. Different primer sequences were required for different cyanobacteria. A few primers of Kit AB1 (AB1-1, 7, 10, 11, 15, 17, 18) could



Figure 5.11. Agarose gel elctrophoresis of RAPD products of Nostoc mus DNA using (a) different magnesium concentrations and (b) different annealing temperatures. Lane 1, 0.5 mM; 2, 1.5mM; 3, 2.5mM; 4, 5mM MgCl₂. Lane 5, 30^oC; 6, 34^oC; 7, 38^oC; 8, 42^oC. The products were amplified with the primer AB1-11. M, 1kb marker.



Figure 5.12. Agarose gel elctrophoresis of RAPD products of *Synechococcus* PCC7942 DNA with different magnesium concentrations and primer combinations. Lane 1, primer ABA-9; 2, primer ABA-10; 3, primer ABA-9+10. a, 0.5 mM; b, 1.5mM; c, 2.5mM and d, 5mM MgCl₂.



Figure 5.13. Agarose gel electrophoresis of RAPDs of Synechococcus PCC7942 DNA generated using different primers. Lane 1, primer ABA-4; 2, ABA-9; 3, ABA-4+9; 4, AB1-1; 5, AB1-17; 6, AB1-1+17; 7, ABA-10; 8, ABA-11 and 9, ABA-10+11, with DNA polymerase, Bioline-AmpliTaq, using the Perkin-Elmer Cetus DNA thermal cycler amplification, program 4 (see section 2.2.19).

amplify cyanobacterial DNA. Whilst most of the primers of Kit ABA (ABA-2, 3, 4, 5, 7, 8, 9, 10, 11, 12, 13, 14, 18, 19, 20) could amplify cyanobacterial DNA. Primers ABA-4, 7, 9, 10 and 14 always provided distinct bands for the different strains. Primer ABA-7 generated a more variable pattern, than the other primers, between closely related strains (Fig.5.14).

It was very difficult to choose a 10-base oligonucleotide primer specific to the wide diversity of cyanobacterial strains. All forty primers were compared, and primer ABA-7 and 10 were chosen for further study of RAPD profiles. Each amplification reaction was performed by using a single 10-base oligonucleotide primer, either ABA-7 or 10.

5. 2. 3. RAPDs of different cyanobacteria

Total genomic DNA of fifty different laboratory strains and isolates of cyanobacteria was amplified successfully by using primer ABA-7 and ABA-10. Reproducible genetic markers were produced that provided diagnostic fingerprints for each strain. In most cases a different RAPD pattern of amplified genomic DNA was generated (at least 2 bands) for each species tested (Fig 5.15). However, for strains of the same species, the same length of amplified product (or RAPD pattern) was obtained by using primer ABA-10. For example, the two *Nostoc MAC* strains exhibited an identical RAPD pattern, as did the two *Nostoc commune* strains. Different species and genera had different patterns using primer ABA-10, with the exception of *Nostoc ellipsospora, Anabaena* PCC 7120 and *Anabaena* ATCC 27892. An identical RAPD pattern was obtained for these cyanobacteria. Figure 5.15 also shows that all fresh water *Synechococcus* strains (PCC7942, LJM 002, 003, 004, 020) and the brackish water



Figure 5.14. Agarose gel electrophoresis of RAPDs generated using total genomic DNA from different cyanobacterial strains and different primers. Lane 1, Nostoc ellipsospora; 2, Anabaena PCC7120; 3, Anabaena PCC7118; 4, Anabaena PCC7118 R32. A, primer ABA-7; B, primer ABA-10 used in separate amplifications. The DNA was amplified with AmpliTaq (Bioline) using standard conditions (see section 2.2.19).

Figure 5.15. Agarose gel electrophoresis of RAPDs of different cyanobacteria. Total genomic DNA from fifty cyanobacteria (1, Synechococcus LJM020; 2, Synechococcus LJM003; 3, Synechococcus LJM004; 4, Synechococcus LJM002; 5, Synechococcus PCC 7942; 6, Synechococcus CCAP 1479/7; 7, Synechococcus WH 7803; 8, Pseudanabaena sp. LJM009 9, Pseudanabaena sp. LJM016; 10, Pseudanabaena sp. LJM011; 11, Pseudanabaena sp. LJM001; 12, Pseudanabaena sp. LJM005; 13, Pseudanabaena sp. LJM006; 14, Pseudanabaena sp. LJM007; 15, Pseudanabaena sp. LJM019; 16, Synechocystis PCC 6803; 17, Microsystis aeruginosa; 18, chroococcus prescottii; 19, Gloeocapsa sp.; 20, Chroococcidiopsis LJM013; 21, Leptolyngbya LJM014; 22, Oscillatoria animalis; 23, Phormidium autumnale; 24, Phormidium LJM017; 25, Leptolyngbya LJM012; 26, Nostoc palaeo; 27, Chlorogloeopsis fritschii; 28, Fisherella musicola. 29, Tolypothrix distorta; 30, Gloeotrichia sp.; 31, Aphanizomenon gracile; 32, Anabaena cylindrica; 33, Anabaena sp. D0697; 34, Anabaena flos-aquae D0848; 35, Anabaena sp. D0752; 36, Anabaena sp. D0746; 37, Anabaena PCC 7118; 38, Anabaena PCC 7118 R3.2; 39, Nostoc commune; 40, Nostoc commune D0800; 41, Nostoc MAC R1; 42, Nostoc MAC PCC8009; 43, Nostoc sp. D0740; 44, Nostoc linckia D0734; 45, Nostoc planctonicum; 46, Nostoc sp. D0766; 47, Nostoc piscinale D0201; 48, Nostoc muscorum; 49, Nostoc ellipsospora; 50, Anabaena PCC 7120; 51, Anabaena PCC 7118) and two primers (A, ABA-7; B, ABA-10) were used in separate amplifications. The DNA was amplified with AmpliTag (Bioline) using standard conditions (section 2.2.19).









B









strain *Synechococcus* CCAP 1479/7 share a band of identical length (580 bp) using primer ABA-10, which was not found in the marine *Synechococcus* tested. Using primer ABA-7, different RAPD patterns, but with high similarity, were obtained for strains of the same species.

Dendrograms (Fig. 5.16) were constructed by using the RAPDistance package (Armstrong *et al.*, 1994) on the basis of pair-wise genetic distances with the combined RAPD data from the two different primers. The cyanobacteria were analysed in the following morphological groups: A, *Synechococcus* group; B, *Pseudanabaena* group; C, *Nostoc* group; D, remaining heterocystous strains; E, remaining non-heterocystous strains. Dendrogram C indicates a high similarity between *Nostoc ellipsospora*, *Anabaena* PCC 7120 and *Anabaena* ATCC 27892. Dendrogram A reveals a low similarity between the freshwater *Synechococcus* PCC 7942 and the marine *Synechococcus* WH 7803, and shows that the marine *Synechococcus* WH 7803 is not closely related to brackish water *Synechococcus* CCAP 1479/7. The similarity indices between the cyanobacterial strains of group A-D using primers ABA-7 and ABA-10 are shown in Table 5.1. The similarity indices for the cyanobacterial strains of different species were not over 0.5.



В.





D.





Figure 5.16. Dendrograms indicating genetic distances among cyanobacterial strains. The genetic distance from the node, based on the RAPD profiles of cyanobacterial strains, is indicated above each branch. The less closely related the strains are the greater the genetic distance. A, *Synechococcus* group; B, *Pseudanabaena* group; C, *Nostoc* group; D, remaining heterocystous strains; E, remaining non-heterocystous strains.

Table 5.1 The similarity indices of RAPDs for cyanobacterial strains with primer ABA-7 and primer ABA-10. A, Synechococcus group; B, Pseudanabaena group; C, Nostoc group; D, Anabaena group.

A.			Sir	nilarity Index ^a			
Straun	S. PCC7942	S. LJM020	S. LJM003	S. LJM004	S. LJM002	S. CCAP1479/7	S. WH7803
Synechococcus PCC7942	1	0.40	0.30	0.41	0.26	0.24	0.15
Synechococcus LJM020		1	0.26	0.32	0.18	0.32	0.15
Synechococcus LJM003			1	0.53	0.38	6:0	0.22
Synechococcus LJM004				1	0.45	0.41	0.18
Synechococcus LJM002					1	0.28	0.21
Synechococcus CCAP1479/7						1	0.30
Synechococcus WH7803							1

B.				Similarity I	ıdex ^a				
Strain	P. LJM009	P. LJM016	P. LJM011	P. LJM001	P. LJM005	P. LJM006	P. LJM007	P. LJM019	
Pseudanabaena LJM009	1	0.52	0.26	0.32	0.42	0.24	0.18	0.27	
Pseudanabaena LJM016		1	0.24	0.32	0.50	0.07	0.13	0.15	
Pseudanabaena LJM011			1	0.32	0.15	0.12	0.19	0.28	
Pseudanabaena LJM001				1	0.5	0.3	0.17	0	
Pseudanabaena LJM005					1	0.18	0.12	0.07	
Pseudanabaena LJM006						1	0.5	0.12	
Pseudanabaena LJM007							1	0.18	
Pseudanabaena LJM019								1	-

IJ					Sii	milarity l	Index ^a						
Strain	N. commune	N. com. D0800	N. MAC RI	N. PCC8009	N. D0740	N. D0734	N. ellipsora	A. PCC 7118	A. PCC 7120	N. planctoicum	N. D0766	N. D0201	N. muscorum
Nostoc commune	1	0.81	0.12	0.12	0.30	0.15	0.27	0.26	0.27	0.22	0.14	0.37	0.27
Nostoc commune D0800		1	0.12	0.06	0.23	0.15	0.18	0.26	0.18	0.30	0.21	0.27	0.27
Nostoc MAC R1			1	0.80	0.21	0.15	0.25	0.20	0.20	0.06	0.13	0.12	0.06
Nostoc MAC PCC8009				1	0.15	0.22	0.27	0.20	0.27	0.06	0.07	0.13	0.06
Nostoc sp. D0740						0.27	0.23	0.26	0.27	0	0.16	0.22	0.16
Nastoc linckia D0734						-	0.33	0.32	0.25	0.27	0.24	0.16	0.17
Nostoc ellipsospora					 			0.8	0.92	0.08	0.22	0.2	0.14
Anabaena PCC 7118								-	0.76	0.15	0.22	0.14	0.14
Anabaena PCC 7120					 					0.15	0.14	0.14	0.2
Nostoc planctonicum	-					 				-	0.24	0.14	0.23
Nostoc sp. D0766					 						-	0.28	0.26
Nostoc piscinale D0201						 						-	0.27
Nostoc muscorum													

.

D.				Similarity Index	К ^а		
Strain	A. cylindrica	A. D0697	A. D0848	A. D0752	A. D0746	A. PCC7118	A. PCC7118 AN23
Anchaena cylindrica		0.22	0.34	0.14	0.12	0.37	0.30
Anabaena sp D0697		-1	0.31	0.17	0.21	0.20	0.20
Anabaena Jlos-aquae D0848			1	0.15	0.20	0.19	0.28
Anabaena sp. D0752				1	0.14	0.27	0.27
Anabaena sp. D0746						0.12	0.19
Anabaena PCC7118						1	0.43
Anabaena PCC7118 AN23							1

a, Similarity indices were determined from the combined RAPD data using primer ABA-7 and ABA-10. 1 = identical; 0 = unrelated.

5.3 Discussion

The RAPD-PCR using random oligonucleotide primers of 10 nucleotides in length, has been applied to rapid identification of animal, plant, fungal, algae and bacterial strains (Welsh and McClelland, 1990; Williams *et al.*, 1990; Akopyanz *et al.*, 1992; McCutcheon *et al.*, 1993). RAPD profiling of these genomes has been widely accepted as a valid taxonomic and phylogenetic tool. In this study, results indicate that both the heterogeneity and the homogeneity of the RAPD pattern can be useful for typing cyanobacteria.

The RAPD technique has been optimised for cyanobacterial DNA from the laboratory strains tested. It is important to note that reaction conditions, including the template DNA concentration, the magnesium ion concentration, annealing temperature, primer concentration and DNA polymerase, all affected the relative intensity and presence of the amplified bands. This is consistent with results of Williams and co-workers (1993) with the exception of the primer concentration, where no effect of primer concentration on the relative intensities of bands was found. Therefore, careful choice of conditions and of the thermostable enzyme used are required to obtain consistent results. For reproducibility of the RAPD technique, five PCR programs and two PCR thermal cyclers were tested. The results showed that the different programs and thermal cyclers, both affected the RAPD patterns. Clear and distinct patterns were obtained from program 4. Programs 3 and 5 were least reliable probably due to the holding time (5 s for program 3 and 1 s for program 5) at the denaturation temperature being too short. Programs 1 and 2 gave similar results suggesting that a shorter holding time of 30 s can be used without affecting the amplified DNA patterns. Setting an exact transition interval time (program 4) provided more reliable results when the two thermal cyclers were used. This supports the findings of Schweder and co-workers (1995) that the transition interval between the denaturation and annealing temperatures appears to have an important impact on RAPD analyses. In order to obtain reliable and consistent results, the conditions of the reaction must be reproduced exactly, which may require accurate recording of transition interval set points as well as temperature set points. Variable ramp characteristics of a thermal cycler during a reaction and between reactions, depending, for instance, on ambient conditions, may help explain why RAPD analyses are so precarious, particularly between different laboratories when analysing template DNA from a single source with identical primers (Penner *et al.*, 1993). So results from this analysis can only be compared with results of RAPDs produced in other laboratories if the same conditions are used.

The choice of primers has proven to be useful for optimising the RAPD reaction. Figure 5.11 shows that specific primers can alter the effect produced by changing magnesium ion concentration and annealing temperature. The primer sequence is very important for successful amplification. In this study, most of the 10-mer oligonucleotides from Kit ABA and a few from Kit AB1 gave successful amplification of the cyanobacterial DNAs tested. In some cases, the different genera of cyanobacteria required different primers for amplification. Other researchers, for example Eskew (1993) and Neilan (1995) also showed that some of these primer sequences could amplify several genera of cyanobacterial DNA. According to Neilan (1995) multiplex PCR increased reproducibility of the RAPD profile, primarily due to the use of pairs of random 10-mer oligonucleotide primers. The products of the multiplex RAPD method, under more stringent conditions, were reproducible across a range of varied parameters, such as enzyme preparation, type of thermal cycler, and quantity and quality of template DNA. In this study, experiments comparing single- and multiple-primer reactions indicated that most bands derived from the single priming reaction were retained, although some new products were additionally synthesised, as a result of multiplexing. However, no clear evidence was provided to show that multiple-primer reactions provided more reliable results than single-primer. Varying parameters such as enzyme preparation, type of thermal cycler, type of PCR-program, quality and concentration of template DNA, the magnesium ion concentration, annealing temperature, and primer concentration also affected the relative intensity of the amplified multiplex bands. Figure 5.12 indicated that the single primer, ABA-10, provided more reliable results than the combination of primers ABA-9+10. Thus selection of the correct primer is crucial for optimising the RAPD reaction, whether it is a single-primer or multiple-primer reaction. This contrasts with the results of Neilan (1995) where the combination of two primers in a single reaction has proven to be useful for optimising RAPD reaction as has 'the correct selection of primer'.

Primers ABA-7 and 10 were chosen here for generating RAPDs of cyanobacteria, because primer ABA-10 could generate patterns to separate different species in the same genus, whilst primer ABA-7 could generate patterns to distinguish between different strains of the same species. Furthermore, they could both amplify DNA from all the cyanobacteria tested and generate clear reproducible patterns.

Genomic DNA of fifty different laboratory strains and isolates of cyanobacteria was amplified successfully by using primer ABA-7 or ABA-10. For strains of the same species, the same RAPD pattern was obtained by using primer ABA-10. Different species and genera had different patterns, with the exception of *Nostoc ellipsospora, Anabaena* PCC 7120 and *Anabaena* ATCC 27892. The highly related branches in the distance tree (Fig. 5.16 C) and the similarity indices (Table 5.1) of 80%-90%, for *Nostoc ellipsospora, Anabaena* PCC 7120, and *Anabaena* PCC 7118 indicate that they may be different strains in the same species group. Furthermore, Figure 5.16 C shows that they are close to *Nostoc linckia* D0734. This supports the findings of Lachance (1981), based on DNA-DNA hybridisation data, that *Anabaena* PCC 7120, and *Anabaena* PCC 7118 are more closely related to *Nostoc strains* than to other *Anabaena* strains.

There were limitations to the application of the RAPD technique to the identification of cyanobacteria, where highly diverse strains were analysed. Particular strains of one genus may group together with some strains in other genera when they are present (Fig. 5.16 E). For strains in the same genus (Fig. 5.16 B) or particular strains (Fig. 5.16 D, *Chlorogloeopsis fritschii* and *Fisherella musicola*) from different genera in the same taxonomic order, RAPD-PCR can prove useful for the identification of cyanobacteria. However the results confirm the findings of Neilan (1995) that the use of a single, multiple-primer RAPD-PCR may not be suitable for inferring an accurate phylogeny for all taxonomic groups. In addition, the RAPD-PCR generated too many bands (2-15 bands per strain, per primer) to analyse the combined data from all the separate gels. The RAPDistance package is limited to use with less than 200 bands. So RAPD-PCR analysis can only be applied to about 20 cyanobacterial strains.

The application of RAPD analysis demands that the cultures are axenic, because the presence of any contaminating microorganisms would alter the RAPD
profile. Thus for the technique to be applicable to environmental isolates, the samples must first be purified by selective culturing or by using specific ultracentrifugation, in order to remove contaminating microorganisms. The RAPD technique has been applied to axenic cultures of newly isolated cyanobacteria. The genetic relatedness among the isolates of cyanobacteria, deduced from the RAPD profiles, was supported by the morphological data (see Fig. 5.16 A and B and compare with the morphological results of Fig. 3.22). In Figure 5.16 E, the larger distance between two *Leptolyngbya* strains suggests that they possibly should be separated into different genera, in line with their morphological differences in constriction at the cross walls and sheath formation, which are morphological characters for separation of simple filamentous cyanobacteria at the genus level.

Results from this study thus indicate that distinct RAPD profiles can be produced for a number cyanobacteria. The technique used was performed without the need for radioactive labelling of the RAPD-PCR products or previous knowledge of the cyanobacterial gene sequences. Therefore, RAPD analysis, which utilises the entire genome as the PCR template, provides a rapid, technically easy approach to the genetic typing of axenic cultures of cyanobacteria. Results are comparable with those achieved by more traditional methods. The finding that certain primers were specific to particular genera of cyanobacteria may be useful for identifying cyanobacteria at the level of genus. Thus RAPDs should provide suitable markers for the identification and classification of cyanobacteria, once a suite of RAPD profiles has been established for different cyanobacteria. Chapter 6

.

.

Identification of cyanobacteria by polymorphisms of PCR-amplified ribosomal DNA spacer region

6.1 Introduction

The limitations of traditional techniques have prompted the development of molecular biological techniques for identification of cyanobacteria. The random amplified polymorphic DNA (RAPD) technique, in conjunction with the polymerase chain reaction (PCR), has been employed to identify many organisms to the strain level of classification (Welsh and McClelland, 1990; 1991). Indeed the technique has been used to identify cyanobacteria in pure culture and cyanobacterial symbionts (Eskew *et al.*, 1993; Neilan, 1995; chapter 5, this study). The approach has enabled discrimination between different strains of cyanobacteria. However, the requirement for axenic cultures restricts the applicability of the RAPD technique.

The use of rRNA genes is well established for identifying evolutionary relationships between organisms, as rRNA genes show a high degree of functionally constancy, which assures relatively good clock-like behaviour (Woese, 1982, 1987). They are essential for the survival of all organisms and are highly conserved in the Bacteria, Archaea and Eukarya (Lane *et al.*, 1985; Woese, 1987). The rRNA operon is transcribed as a pre-rRNA transcript that contains the following components in the order (5' to 3'): 16S rRNA, spacer, tRNA, spacer, 23S rRNA, spacer and 5S rRNA gene sequences (Watson *et al.*, 1987; see Fig 1.1). Characterisation of the 16S rRNA gene has proved useful in the identification of species, genera and families of bacteria (Woese, 1987; Amann *et al.*, 1995) and cyanobacteria (Wilmotte, 1994). This is reflected in the large amount of sequence information that has been accumulated for the phylogenetic analysis of microorganisms using this gene (Olsen *et al.*, 1991). In contrast, relatively few 23S gene sequences have been reported, and neither this gene nor the 5S gene have been widely used for phylogenetic purposes (Wilmotte,

1994). Sequence information has also become available on the 16S-23S rRNA spacer region and suggests that considerable variation can occur between species in both the length and the sequence of this region (Gurtler and Stanisich, 1996). Such variation together with the fact that many bacteria possess multiple copies of the rRNA operon, provides scope for using this region to identify and type bacteria.

Systematic studies of spacer region variation would require amplification of the region by PCR (Mullis and Faloona, 1987) and hence the availability of conserved sequences in flanking 16S rRNA, tRNA and 23S rRNA genes to serve in primer recognition. In cyanobacteria a variety of primers has already been used for PCR amplification or sequencing purposes (Wilmotte et al., 1993), but the appropriateness of the choices, particularly for those based on the 23S rRNA gene, has been limited by the paucity of sequenced cynaobacterial 23S rRNA genes. Only two 23S rRNA genes from Synechococcus sp. strain PCC 6301 and Synechococystis strain PCC6803 (cited in Wilmotte, 1994) have been sequenced. The 16S-23S spacer region or internal transcribed spacer (ITS) region of cyanobacteria typically contains the following components in the order (5' to 3'): spacer, tRNA^{ile}, spacer, tRNA^{ala}, spacer (see Fig 6.1), although some cyanobacteria have only the tRNA^{ile} gene. For PCR amplification of the entire or partial ITS region, several primers are available (Wilmotte et al., 1993). Variation within this region may be detected by a number of techniques, including restriction fragment length polymorphisms (RFLPs) or DNA sequencing.





Figure 6.1. Schematic diagram of ribosomal DNA (rDNA) repeat showing internal transcribed spacer region (ITS). Transcription of the operon is from left to right. Some cyanobacteria have two transfer RNA (tRNA) genes as shown; others just have tRNA^{ile}. The arrowed bars show the location of primers used in this study.

SF/SR, cyano-specific forward/reverse primers developed in this study; tRNA^{ala}, alanine tRNA; tRNA^{ile}, isoleucine tRNA.

6.1.1 RFLP-PCR of rDNA spacer

PCR technology has been used in several recent studies as a rapid means of amplifying the 16S-23S spacer region for the specific purpose of detecting heterogeneity between and within species (Barry et al., 1991; Jensen et al., 1993). Such heterogeneity has been found in terms of both the number and the length of the spacer, and raises the possibility that this methodology has broad applicability as a rapid, possibly automatable, general method for bacterial identification (Gurtler and Stanisich, 1996). The polymorphic character of the PCR-amplified product can be revealed by with restriction enzymes. Resulting fragments digesting are resolved electrophoretically and patterns can be indicative of a particular species. Furthermore, size of the PCR- amplified products themselves can be diagnostic. In cyanobacteria, Neilan and co-workers (1995) developed primers to amplify the phycocyanin operon (cpc gene), and then used restriction fragment length polymorphisms of PCR products to infer the genetic relatedness and evolution of toxic and bloom-forming cyanobacteria. Genetic characterisation of strains of also been undertaken using restriction fragment length cyanobacteria has polymorphisms of the intergenic spacer of the genes encoding phycobilisome subunits (cpc BA) (Bolch et al., 1996).

Whilst RFLP analysis can be useful for rapid screening and for distinguishing between different genera and species, it only detects a small amount of the variation that may be present in the spacer region. More information can be obtained from DNA sequencing.

182

6.1.2 DNA Sequencing of rDNA spacer

Comparative studies of the nucleotide sequences of rRNA genes provide a means of analysing phylogenetic relationships over a wide range of taxonomic levels (e.g. Woese and Olsen 1986; Wilmotte, 1994; Wilmotte *et al.*, 1992,1993,1994; Neilan *et al.*, 1994; Neilissen *et al.*, 1994, 1996). Numerous sequences of rRNA genes have been obtained, primarily by isolating and sequencing individual cloned genes (Medlin *et al.*, 1988). Direct rRNA sequencing (Lane *et al.*, 1985) has also been used to obtain sequence data rapidly. However, this method requires relatively large amounts of RNA and is prone to errors, since only one strand is sequenced.

The polymerase chain reaction (PCR) and direct DNA sequencing provide several advantages over cloning and direct rRNA sequencing. These include: (1) the method utilises relatively crude preparations of total DNA, such as those from minipreps, or can be performed directly using whole cells, without further DNA preparation and purification (see section 2.2.14.3); (2) only small amounts of DNA are required, about 0.1 to 10 ng per amplification; (3) both strands of the gene can be sequenced, which reduces errors; (4) the method is compatible with automated DNA sequencing instruments that utilise fluorescently-labelled dideoxynucleotide triphosphates or primers (White *et al.*, 1990).

16S rRNA gene sequences are probably too conserved to investigate intraspecies variability (Ward *et al.*, 1992). The sequence of the ITS region (16S-23S rDNA spacer), may be more suitable for studying such variability. Gurtler and Stanisich (1996) reviewed and aligned 33 different 16S-23S spacer sequences from 13 species of bacteria. Despite the limited scope of the analysis and the occurrence of gaps in the DNA alignments, due to the variable length of the spacers and the relative positions of the tRNA genes, no highly conserved regions were found. The high inter-specific variability reported for the spacer makes it an appropriate candidate for restriction fragment length polymorphisms (RFLPs). These workers suggested that homology searches based on very short regions (20 bp or less) might reveal inter-species homologies; such regions would need to be avoided when designing species-specific primers. However, they could be of great use in the design of genera-specific primers.

Analysing phylogenetic relationships is a very complex and controversial issue and a review of the different methodologies is provided by Swofford and Olsen (1990). Distance methods and parsimony analysis are most often used to infer phylogenetic trees. Distance methods belong to the phenetic techniques and are based on the calculation of dissimilarities for each pair of organisms. These dissimilarity values are corrected for multiple mutations and entered in a matrix. A tree topology, in which the distances between organisms in the tree will be as close as possible to the matrix values, is constructed by an algorithm. Parsimony analysis is based on the cladistic approach. It considers that only the similarity due to the possession of new (derived) characters is phylogenetically informative. Thus, only the variable positions, where at least two organisms share the same derived nucleotide, are used in the calculation of the tree topology. Ancestral sequences are re-constructed at each node of the tree. The final tree topology is the one requiring the minimal number of evolutionary changes. A statistical evaluation of tree topologies can be carried out by bootstrap analysis (Felsenstein, 1985). New sequence alignments are generated by random sampling of the positions with replacement. For each re-sampled data set, a new tree is constructed. The bootstrap value for each node is the number of re-sampled trees in which the same organisms are

clustered together. Only the grouping of strains is considered, not the topology and lengths of the branches that diverge after the node.

This chapter reports the development of a PCR-based technique that requires only small quantities of relatively impure DNA, and the use of the variability of the length and sequence of the 16S-23S rDNA spacer region to distinguish between various genera and species of cyanobacteria. The approach used was to amplify the 16S-23S rDNA spacer from cyanobacteria by PCR, and test various restriction enzymes to find those producing RFLPs useful in identification. In addition, the extent of sequence similarity in the rDNA spacer of various cyanobacterial strains was established by direct DNA sequencing of PCR products and by cloning and plasmid sequencing, followed by sequence alignment. Oligonuleotide sequences selected from conserved regions were evaluated as cyano-specific primers, for amplification of cyanobacterial DNA from laboratory cultures and environmental samples.

6.2 Results

6.2.1 RFLP-PCR

A number of primer pairs was tested for amplification of the 16S-23S rDNA spacer with a view to generating products useful for discriminating between cyanobacterial strains. The priming sites were selected from regions of highly conserved sequences found in the 16S rDNA, tDNA and 23S rDNA regions adjacent to the intervening spacer region (see Fig. 6.1). The PCR procedure was optimised for the primers used and was either performed directly on cyanobacterial cultures or on DNA extracted from them (see section 2.2.20). In each case identical results were obtained.

The choice of primers was critical to the use of spacer variation for species identification. Primer pair 1/18 (which amplifies 16S rDNA, ITS region, tDNA^{ile} [or tDNA^{ile} and tDNA^{ala}] and 5'end 23S rDNA) and 14/18 (which amplifies 3'end of 16S rDNA, ITS region, tDNA^{ile} [or tDNA^{ile} and tDNA^{ala}] and 5'end of 23S rDNA) amplified the rDNA to give multiple DNA bands of different intensities, for a number of the species tested. The intense and reproducible bands have been classified as primary products, the weaker, variable bands as secondary products (Table 6.1 & Fig. 6.2 a, b). Weak bands were common in the heterocystous strains, especially in the genera Nostoc and Anabaena. The secondary amplification products could be heteroduplexes containing conserved homologous sequences at the 3' and 5' ends, separated by highly variable ITS sequences. These heteroduplexes would migrate more slowly in gels than the expected homoduplexes because of the single-stranded regions. In order to test for heteroduplex formation, susceptibility of amplification products to the single-strand specific endonulease, S1, was determined. Figure 6.3 shows that when amplified products (using primer pair 14/18) were digested with S1 nuclease, the most intense bands remained. Whilst the less intense and weak bands (normally >780 bases) were removed. A number of new, fainter fragments in the size range of 120-400bp could be seen. These smaller fragments are probably residual double-stranded components. remaining after digestion of single-stranded positions of the heteroduplex DNA structures. The undigested bands that remained are presumably the homoduplex 16S/23S spacer amplification products. The sizes of these products are also indicated in Table 6.1. The variation in heteroduplex and homoduplex 16S-23S spacer amplification products may be useful in identifying heterocyst-forming filamentous cyanobacteria, by using single-strand conformation polymorphisms (SSCP). For the unicellular and nonheterocyst-forming filamentous cyanobacteria a single amplified rDNA product was

Figure. 6.2 Agarose electrophoresis gels of amplified rDNA spacer products of selected cyanobacterial strains. (A) PCR-amplification using primer 1/18; (B) PCR-amplification using primer 14/18; (C) PCR-amplification using primer 17/18; M, 1kb DNA ladder; 1. Nostoc MAC R1; 2. Nostoc MAC PCC8009; 3. Nostoc commune; 4. Nostoc planctonicum; 5. Nostoc piscinale D0201; 6. Nostoc sp. D0740; 7. Nostoc linckia D0734; 8. Nostoc muscorum; 9. Nostoc ellipsospora; 10. Anabaena PCC 7120; 11. Anabaena PCC 7118; 12. Anabaena PCC 7118 R3.2; 13. Anabaena cylindrica; 14. Anabaena flos-aquae D0848; 15. Anabaena sp. D0697; 16. Synechococcus PCC 7942; 17. Synechococcus CCAP 1479/7; 18. Synechococcus WH 7803; 19. Synechocystis PCC 6803; 20. Microcystis aeruginosa; 21. Gloeocapsa sp.; 22. Chroococcus prescottii; 23. Pseudanabaena sp. LJM001; 24 Oscillatoria animalis; 25. Phormidium autumnale; 26. Gloeotrichia sp.; 27. Aphanizomenon gracile; 28. Tolypothrix distorta; 29. Chlorogloeopsis fritschii; 30. Fisherella musicola.













Table 6.1.	Cyanobacteria used and	the sizes of rDNA	fragments amplified	d by primer pairs	1/18,	14/18 and
17/1 8 .						

Strain	Source ^a	am	amplified rDNA (bp) ^b		
		primer 1/18	primer 14/18° pr	imer 17/18	
Unicellular	· ·				
Chroococcus prescottii	BSM	1865	555	310	
Gloeocapsa sp.	BSM	1910	560	322	
Microcystis aeruginosa	BSM	1905	585	320	
Synechococcus PCC7942	LJMU	2100	735	425	
Synechococcus CCAP1479/7	LU	2400	1010	665	
Synechococcus WH7803	LU	2300	930	625	
Synechocystis PCC6803	UW	1940	665	390	
Filamentous non-heterocystou	<u>.s</u>				
Oscillatoria animalis	BSM	2200	840	575	
Phormidium autumnale	BSM	2200	800	535	
Pseudanabaena liv JM 001	LJMU	2075	735	450	
Filamentous heterocystous					
Anabaena ATCC27892	LU	1875,2100(2300,3642)	<u>515,730(</u> 665, 8 57,960,1100) 435	
Anabaena ATCC27892 R3.2	LU	1875,2100(2300,3642)	<u>515,730(</u> 665, 8 57,960,1100) 435	
Anabaena cylindrica	UW	1890(2150,2230)	<u>575(723,767</u> ,948,1036,121	7) 485	
Anabaena cylindrica (AN 23)	LJMU	1890(2150,2230)	<u>575(723,767</u> ,948,1036)	485	
Anabaena flos-aquae D0848	UD	1836(2065)	<u>535(767</u> ,887,995)	430	
Anabaena PCC 7120	UW	1875,2100(2300,3642)	<u>515,730(</u> 665,857,960,1100) 435	
Anabaena sp. D0697	UD	1836(2065,2230)	<u>530(</u> 695 <u>,750</u> ,887,995)	430	
Anabaena sp. D0746	UD	1836(2165,2330)	<u>498,745(</u> 670,930,1180)	460	
Anabaena sp. D0752	UD	1870(2120,2285)	<u>515,740</u> (675,867,950)	478	
Aphanizomenon gracile	BSM	1780,2350	<u>450,858(329)</u>	440	

Filamentous heterocystous continued.....

Chlorogloeopsis(Chlorogloea) fritschii	LJMU	2018,2185(2300)	<u>593(730,</u> 980)	480
Fischerella musicola	BSM	2018(2105,2340)	<u>530</u> (730,930)	475
Gloeotrichia sp.	UW	1940,2300(2465)	<u>494,845</u> (1029,1170)	586
Nostoc commune	BSM	1970,2000,2190(2460,3943)) <u>560,790</u> (705,985,1230)	535
Nostoc commune D0800	UD	1970,2000,2190(2460,3943)) <u>560,790</u> (705,985,1230)	535
Nostoc ellipsospora	UW	1875,2100(2300,3642)	<u>515,730(</u> 665,857,960,1100)	435
Nostoc linckia D0734	UD	1930(2190,2400)	<u>530</u> ,686, <u>790</u> (930,1050,1182)	445
Nostoc MAC PCC8009	UW	2090 (1979, 2320)	<u>710</u> (630, 960)	495
Nostoc MAC R1	LU	2090 (1979,2320,1876)	<u>710 (</u> 630,960,540)	495
Nostoc muscorum	BSM	1875,2195(3642)	<u>780(510</u> ,675,857,960,1100)	435
Nostoc piscinale D0201	UD	1890,2190(2410,3743,4133)) <u>490,</u> 675, <u>760(</u> 880,985,1182)	490
Nostoc planctonicum	BSM	1875(2050)	<u>515(</u> 675,740,867)	520
Nostoc sp. D0740	UD	1890,(2190,2500)	<u>490,745</u> (675,930,1100)	445
Nostoc sp. D0766	UD	330,1975(1025)	<u>508,790(</u> 660,1010)	460
Tolypothrix distorta	BSM	2175(1821)	<u>682(</u> 602)	440

- LJMU, UW, BSM, LU, UD, Liverpool John Moores University, University of Warwick, Biobred
 Sciento Manchester, Lancaster University, University of Durham respectively.
- b, Data are the means of three determinations. Figures in brackets are secondary fragments (low intensity bands).
- c, Products not susceptible to endonuclease S1, presumed homoduplex products, are underlined, other secondary fragments susceptible to endonuclease S1 are heteroduplexes or single-stranded DNA.



Figure 6.3. Agarose gel electrophoresis of single-strand conformation polymorphisms (SSCP) patterns of selected cyanobacterial strains. 1 and 2, SSCP patterns of rDNA spacer amplified using primers 14 and 18 and digested with the single-strand specific endonuclease, S1, 40 units and 80 units respectively; a, *Anabaena* PCC 7120; b, *Nostoc muscorum; c. Anabaena cylindrica*; M, 1kb marker.

always obtained using either primer pair 14/18 or 1/18.

The pattern of amplified products with primer pair 14/18 enabled clearer discrimination between species than that with primer pair 1/18. Amplification with primer pair 1/18 generated larger fragments (containing 1.5 kb highly conserved 16S rDNA) making it more difficult to resolve the variability between amplified products, than with primer pair 14/18. The pattern and length heterogeneity of the amplified bands was generally sufficient to resolve the different genera and species with the exception of *Nostoc ellipsospora, Anabaena* PCC7120, *Anabaena* ATCC27892 and *Anabaena* ATCC27892 R.32 (which carries cyanophage AN23 DNA), which all shared the same sized fragments. However, a different RAPD DNA profile was generated in each case (Fig.5.14). The same primary fragment of 2090 bp (with primer 1/18) or 710 bp (with primer 14/18) was generated for both strains of *Nostoc MAC*, with two weak secondary fragments in each case. However, strain R1 showed a further weak fragment, using either primer pair. The same primary and secondary fragments were also generated from both strains of *Nostoc commune* using either primer pair.

Analysis of length polymorphism and RFLPs of the amplified rDNA spacer using primer pair 17/18 has generally permitted discrimination between the cyanobacteria tested to genus and species level. This primer pair (which amplifies tDNA^{ile} [or tDNA^{ile} and tDNA^{ala}], ITS region and 5'end of 23S rDNA) amplified rDNA from all the strains tested generating a single band of different sizes (Table 6.1 & Fig. 6.2 c). This amplified DNA was digested with twenty different restriction enzymes to identify RFLPs. Most of the enzymes did not cut the DNA, or only very small fragments were cut from the ends. However, *Hinfl*, *DdeI*, *AluI* and *TaqI* showed potentially useful discrimination among the different genera and species (Fig. 6.4 & Fig. 6.5).







i.





M 16 17 18 19 20 21 22 23 24 25 26 27 28 29 30



M 1 2 3 4 5 6 7 8 9 10 11 12 13 14 15 bp 492 -246 -123 -

D.



M 16 17 18 19 20 21 22 23 24 25 26 27 28 29 30

Figure 6.4. Agarose electrophoresis gels of RFLP patterns of selected cyanobacterial strains. (A) RFLP patterns of rDNA spacer amplified using primers 17 and 18 and digested with the restriction enzyme Hinf I; (B) RFLP patterns of rDNA spacer amplified using primers 17 and 18 and digested with the restriction enzyme Dde I; (C) RFLP patterns of rDNA spacer amplified using primers 17 and 18 and digested with the restriction enzyme Alu I; (D) RFLP patterns of rDNA spacer amplified using primers 17 and 18 and digested with the restriction enzyme Taq I; M, 123bp DNA ladder; 1. Nostoc MAC R1; 2. Nostoc MAC PCC8009; 3. Nostoc commune; 4. Nostoc planctonicum; 5. Nostoc piscinale D0201; 6. Nostoc sp. D0740; 7. Nostoc linckia D0734; 8. Nostoc muscorum; 9. Nostoc ellipsospora; 10. Anabaena PCC 7120; 11. Anabaena PCC 7118; 12. Anabaena PCC 7118 R3.2; 13. Anabaena cylindrica; 14. Anabaena flos-aquae D0848; 15. Anabaena sp. D0697; 16. Synechococcus PCC 7942; 17. Synechococcus CCAP 1479/7; 18. Synechococcus WH 7803; 19. Synechocystis PCC 6803; 20. Microcystis aeruginosa; 21. Gloeocapsa sp.; 22. Chroococcus prescottii; 23. Oscillatoria animalis; 24. Phormidium autumnale; 25. Pseudanabaena sp. LJM001; 26. Aphanizomenon gracile; 27. Gloeotrichia sp.; 28. Tolypothrix distorta; 29. Chlorogloeopsis fritschii; 30. Fisherella musicola.

Figure 6.5. Diagramatic representation of RFLP patterns of amplified rDNA products. RFLP patterns of amplified rDNA products (using primers 17 and 18) obtained from selected strains of cyanobacteria with the restriction enzymes *Hinfl*, *Ddel*, *Alul* and *TaqI*. The numbers below the tracks correspond to the RFLP patterns in Table 6.2.

,



In most cases, a different RFLP pattern of amplified 16S-23S rDNA spacer was generated (generally of 3 bands) for each species tested, with each restriction enzyme used. The organism could be assigned an RFLP pattern-type on the basis of the four enzymes used (Table 6.2). However, for strains of the same species the same length of amplified product and the same RFLP pattern-type were obtained. For example, the two Nostoc MAC strains exhibited an identical length (495 bp) and RFLP pattern-type (U), and the two Nostoc commune strains exhibited an identical length (535 bp) and RFLP pattern-type (S). Different species and genera had different pattern types, with the exception of Nostoc ellipsospora, Anabaena PCC 7120, Anabaena ATCC 27892 and Anabaena ATCC 27892 R3.2. An identical length (435 bp) and RFLP pattern-type (K) were obtained for these cyanobacteria, indicating that they may be in the same species group. This is consistent with results using primer pairs 1/18 and 14/18, where identical product profiles were obtained for these strains (Table 6.1). In addition, pattern-type K showed very high similarity to pattern-type V for Nostoc muscorum, and pattern-type M for Anabaena flos-aquae D0848 showed very high similarity to pattern-type N for Anabaena sp. D0697. This suggests that these strains are closely related. Furthermore, Figure 6.5 shows that all heterocystous cyanobacteria share a band of identical length (65bp) using the restriction enzymes Hinf I and Tag I, which is not found in the unicellular and non-heterocystous cyanobacteria tested, with the exception of Microcystis aeruginosa and Chroococcus prescottii. This indicates that these heterocystous strains share common recognition sites for the enzymes and therefore similar sequences.

Strain	Pattern-type ^a Hinf I		Dde I	Alu I	Taq I			
Unicellular								
Chroococcus prescottii	Α	H7	D7	A7	T 7			
Gloeocapsa sp.	В	H6	D6	A6	T6			
Microcystis aeruginosa	С	H5	D5	A5	T5			
Synechococcus PCC 7942	D	H1	Dl	A 1	T1			
Synechococcus CCAP 1479/7	Е	H2	D2	A2	T2			
Synechococcus WH7803	F	H3	D3	A3	T3			
Synechocystis PCC 6803	G	H4	D4	A4	T 4			
Filamentous non-heterocystous								
Oscillatoria animalis	Н	H8	D8	A8	Т8			
Phormidium autumnale	Ι	H9	D9	A9	Т9			
Pseudanabaena liv JM 001	J	H10	D10	A10	T10			
Filamentous heterocystous								
Anabaena ATCC27892	К	H17	D17	A18	T16			
Anabaena ATCC27892 R 3.2	К	H17	D17	A18	T16			
Anabaena PCC 7120	K	H17	D17	A18	T16			
Anabaena cylindrica	L	H19	D19	A20	T18			
Anabaena cylindrica (AN 23)	L	H19	D19	A20	T18			
Anabaena flos-aquae D0848	М	H20	D20	A22	T19			
Anabaena sp. D0697	N	H20	D20	A21	T19			
Anabaena sp. D0746	0	H21	D21	A23	T20			
Anabaena sp. D0752	Р	H22	D22	A24	T21			

Table 6.2. RFLP patterns and pattern-types obtained for selected cyanobacterial strains.

.

Filamentous heterocystous continued.....

Aphanizomenon gracile	Q	H23	D23	A25	T22
Chlorogloeopsis(Chlorogloea) fritschii	R	H26	D26	A28	T25
Fischerella musicola	S	H27	D27	A29	T26
Gloeotrichia sp.	Т				
Nostoc commune (BSM)	U	H12	D12	A12	T12
Nostoc commune D0800 (UD)	U	H12	D12	A12	T12
Nostoc ellipsospora	К	H17	D17	A18	T16
Nostoc linckia D0734	v	H14	D15	A15	T 14
Nostoc MAC PCC 8009	W	H11	D11	A11	T 11
Nostoc MAC R1	W	H11	D11	A11	T11
Nostoc muscorum	x	H16	D17	A17	T 16
Nostoc piscinale D0201	Y	H13	D13	A13	T13
Nostoc planctonicum	Z	H15	D16	A16	T15
Nostoc sp. D0740	AA	H14	D14	A14	T14
Nostoc sp. D0766	AB	H18	D18	A19	T17
Tolypothrix distorta	AC	H24	D24	A26	T23

a, Details of the patterns (H1-H24, D1-D24, A1-A26 and T1-T23) for amplified (primer pair 17/18) rDNA digests are given in Fig. 6.5. Pattern-type derives from the pattern combination with the four test restriction enzymes.

A dendrogram (Fig.6.6) constructed on the basis of pair-wise genetic distances using the combined RFLP data from the four restriction enzymes, further indicates the similarity between *Anabaena* and *Nostoc* strains. In addition, it reveals a low similarity between the freshwater *Synechococcus* PCC 7942 and the marine *Synechococcus* WH 7803, and shows the marine *Synechococcus* WH 7803 more related to brackish water *Synechococcus* CCAP 1479/7 than *Synechococcus* PCC 7942. It is noteworthy that all the heterocystous strains form a cluster excluding all the other strains.

Heterogeneity of length and RFLP pattern-type of the rDNA spacer can also be used to detect more than one cyanobacterial species in a mixed culture (Fig. 6.7 a). Both size and intensity of bands need to be considered for discrimination between the species (Fig. 6.7 b). However, the method would probably not be suitable for distinguishing between a large number of species, due to the multiplicity of bands generated.

In addition to amplifying the 16S-23S rDNA spacer, the 16S rDNA was amplified by using primer pair 1/16 (which in the case of *Synechococcus* PCC 6301, amplifies 16S rDNA from the 8th base at the 5' end to base 1486) for a number of strains. A similar sized amplified product of about 1.48 kbp was generated in each case. The amplified products were digested using the same restriction enzymes as those used for the spacer region (Fig. 6.8). Highly similar patterns were obtained for all strains tested. However, the sizes of the restriction fragments were not consistent with those expected from the size of the original amplified product.

Figure 6.6. Dendrogram indicating genetic distances among cyanobacterial strains. The genetic distance from the node, based on the RFLP profiles of 35 strains, is indicated above each branch. The less closely related the strains are the greater the genetic distance.



b 8 ٢ Γ T ٦ M 1 2 3 4 56 1 2 3 5 6 4 bp 492 246 123

B

A





Figure 6.7. Agarose electrophoresis gel of amplified rDNA spacer products and RFLP patterns of selected cyanobacterial strains. (A) a, PCR-amplification using primer 17/18. b, RFLP patterns of the amplified rDNA spacer digested with the restriction enzyme *Taq* I. 1. *Synechococcus* PCC 7942; 2. *Synechococcus* PCC 7942 + *Synechococcus* WH 7803; 3. *Synechococcus* WH 7803; 4. *Synechocystis* PCC 6803; 5. *Synechocystis* PCC 6803 + *Chroococcus* prescottii; 6. *Chroococcus* prescottii. M, 123bp DNA ladder. (B) Analysis of results of agarose electrophoresis gel by using the Phoretix ID (version 2.51).

Figure 6.8. Agarose electrophoresis gels of amplified partial 16S rDNA products and RFLP patterns of selected cyanobacterial strains. (A) PCR- amplification using primer 1/16; (B) RFLP patterns of 16S rDNA amplified using primers 1/16 and digested with the restriction enzymes *Alu* I or *Taq* I. M, 1kb DNA ladder; 1, *Aphanizomenon gracile*; 2, *Anabaena* sp. D0752; 3, *Nostoc linckia* D0734; 4, *Nostoc* sp. D0766; 5, *Nostoc commune*.

.



r

B



A

The high inter-specific variability found for the 16S-23S rDNA spacer, based on RFLP analysis, makes it a potentially useful region for revealing sequence heterogeneity that may be used for typing cyanobacteria. Sequences of the rDNA spacer of various cyanobacterial strains have therefore been determined and analysed. Moreover, by establishing the extent of sequence similarity, any highly conserved regions could be identified that may be useful in designing cyano-specific primers or probes.

PCR products of the partial 16S-23S rDNA spacer, amplified by using primer 17/18 and DyNAZyme II (see section 2.2.20), were purified by using spin columns (see section 2.2.24), and sequenced using both manual and automatic techniques. For manual sequencing, PCR products were cloned into the pGEM-T vector (Promega) or pCR II vector (Invitrogen) (see section 2.2.24), and recombinant plasmid DNA was isolated and purified (section 2.2.25), prior to sequencing. Results showed that 100% of the recombinant pCR II plasmid DNA (isolated from different white clones) contained the correct sized PCR products, compared with only 30-40% of recombinant pGEM-T plasmid DNA. Thus the pCR II vector was a more suitable choice for subsequent cloning experiments. For automated sequencing, amplified products from separate PCR-amplification reactions were sequenced directly.

The sequences of three strains, *Anabaena* PCC 7118, *Anabaena* PCC 7120 and *Nostoc ellipsospora*, each from three individual clones were determined by using both automated and manual sequencing (see section 2.2.26) to check the reliability of the two methods. The sequences were aligned, and were found to be almost identical for any particular species, using either method (Fig. 6.9). For manual sequencing, compression

Figure 6.9 Alignment of 16S-23S rDNA spacer sequences from Anabaena PCC 7118, Anabaena PCC 7120 and Nostoc ellipsospora starting from the 5' end of tRNA^{ile} to the beginning of 23S rRNA gene. Alignment was obtained using the CLUSTAL W program version 1.6 (http://www.ibcp.fr/clustalw.html). (M), manual sequencing data; (A), automated sequencing data. A-Adenine, T-Thymine, G-Guanine, C-Cytosine. Numbers represent base positions. Differences that appeared between upper and lower strands from manual sequencing data are underlined. Data are from three repeats. For manual sequencing both upper and lower strands were sequenced. The sequences of the upper strands are shown $(5^2 \rightarrow 3^2)$.

		10	20) 3	10 4	0 50	60
						1	
A.PCC7118	(M)	TATTAGCTCA	GGTGGTTAGA	AGCGCACCCC	TGATAAGG <u>G</u> T	GAGGTCCCTGG	TTCGAGTCCA
A. PCC7118	(A)	TATTAGCTCA	GGTGGTTAG	AGCGCACCCC	TGATAAGGGT	GAGGTCCCTGG	GTTCGAGTCCA
A.PCC7120	(M)	TATTAGCTCA	GGTGGTTAGA	AGCGCACCCC	TGATAAGGGT	GAGGTCCCTGG	STTCGAGTCCA
A. PCC/120	(A)	TATTAGCTCA	GGTGGTTAGA	AGCGCACCCC	TGATAAGGGT	GAGGTCCCTGG	TTCGAGTCCA
N.ellips.	(M)	TATTAGCTCA	GGTGGTTAGA	AGCGCACCCC	TGATAAGGGT	GAGGTCCCTGG	TTCGAGTCCA
N.ellips.	(A)	TATTAGCTCA	GGTGGTTAGA	AGCGCACCCC	TGATAAGGGT	GAGGTCCCTGG	TTCGAGTCCA
		70		• •	0 10	0 110	120
		, 0		, ,	1 10		120
	(14)			C			
A.PCC/118	(M)	GGATGGCCCA		GTCAAAAGT	TAAAAGACAA	AAGTCAAAAGT	AGTTATTTAC
A.PCC/118	(A)	GGATGGCCCA		GTCAAAAGT	ТААААGACAA	AAGTCAAAAGT	AGTTATTTAC
A.PCC/120	(14)	GGAIGGCCCA	CCTGAAGCAA	GICAMAAGI	TAAAAGACAA	AAGICAAAAGI	AGIIAIIIAC
N. elline	(M)	GGATGGCCCA	CCTGAAGCAA	GTCAAAAGT	TAAAGACAA	AGICAAAAGI.	
N.ellips.	(12)	GGAIGGCCCA	CCTGAAGCAA	GICARAAGI	TAAAAGACAA	AGICAAAAGI.	
N. errips.	(5)	GGAIGGCCCA	CCIGAAGCAA	GICANAAGI		-NGI CANANGI	AGITATIAC
		130	140	15	0 160) 170	180
	(14)	1					
A.PCC/118	(M)	TTTTGAATTT	TGGATTITGA	ATTTTGAAT	TGTATTTGGGG	GTTTAGCTCA	GTTGGTAGAG
A.PCC/118	(A) (M)	TTTTGAATTT	TGGATTTTGA	ATTTTGAAT	TGTATTTGGGG	COMMANCICA	STTGGTAGAG
A.PCC/120	(121)	TITIGAATTI TTTTCAATTI	TGGATTIGA	ATTIIGAAT	TGIAIIIGGGU	CTTTAGCICA	
N. Allina	(A) (M)	TTTTGAATTT		ATTIGAAT.		CTTTAGCICA	STIGGIAGAG
N.ellips.	(19)		TGGAITIGA				
N.ellips.	(A)	TTTTGAATTT	TGGATTTTGA	ATTTGAAT	IGIAIIIGGGG	GITIAGCICA	JIIGGIAGAG
		190	200	210	n 220) 230	240
		150	200	21	i	, 230	210
A PCC7118	(M)	CCCCTCCTTTT	CCDDCCDCCD	TETCACCCC	I PTCGAGTCCCC	ו גרמרריתמחיי	
A. PCC7118	(A)	CGCCTGCTTT	CCAAGCACGA CCAAGCACGA	TGTCAGCGG	TTCGAGTCCGC	IAACCICCACP IAACCTCCACP	TCAAAACCC
A. PCC7120	(M)	CGCCTGCTTT	GCAAGCAGGA	TGTCAGCGG	TTCGAGTCCGC	TAACCTCCAC	TGAAAAGGC
A. PCC7120	(A)	CGCCTGCTTT	GCAAGCAGGA	TGTCAGCGG	LTCGAGTCCGC	TAACCTCCAC	TGAAAAGGC
N.ellips.	(M)	CGCCTGCTTT	GCAAGCAGGA	TGTCAGCGG	TTCGAGTCCGC	TAACCTCCACA	TGAAAAGGC
N.ellips.	(A)	CGCCTGCTTT	GCAAGCAGGA	TGTCAGCGG	TCGAGTCCGC	TAACCTCCAC	TGAAAAGGC
	(/						
		250	260	270) 280	290	300
		1				1	
A.PCC7118	(M)	AAGCCAAGAA	AGAGGAAAAA	TTCAGCAACI	TAACGGGAGTT	AGACTGCTGGG	TAGTACCTA
A.PCC7118	(A)	AAGCCAAGAA	AGAGGAAAAA	TTCAGCAACI	TAACGGGAGTT	AGACTGCTGGG	TAGTACCTA
A.PCC7120	(M)	AAGCCAAGAA	AGAGGAAAAA	TTCAGCAACI	TAACGGGAGTT	AGACTGCTGGG	TAGTACCTA
A.PCC7120	(A)	AAGCCAAGAA	AGAGGAAAAA	TTCAGCAACI	AACGGGAGTT	AGACTGCTGGG	TAGTACCTA
N.ellips.	(M)	AAGCCAAG <u>AA</u>	AGAGGAAAAA	TTCAGCAACI	AACGGGAGTT	AGACTGCTGGG	TAGTACCTA
N.ellips.	(A)	AAGCCAAGAA	AGAGGAAAAA	ITCAGCAACI	AACGGGAGTT	AGACTGCTGGG	TAGTACCTA
		21.0	200				2.00
		310	320	330) 340	350	360
N D007110						I	
A. PCC/118	(M)	GCCAGAACCT"	I'GAAAACTGC	ATAAAAACGC	GATTAGATTA	GCAGGCAGACA	CAGACAAAC
A PCC7120	(A) (M)	GCCAGAACCT	I'GAAAACTGCA	ATAAAAACGC	GATTAGATTA	GCAGGCAGACA	
A PCC7120	(12)	GCCAGAACCT.	IGAAAACTGC/	ATAAAAACGC	GATTAGATTA	GCAGGCAGACA	
N elline	(A) (M)	CCCAGAACCI.		AIAAAAACGC		GCAGGCAGACA	
N ellips.	(11)	GCCAGAACCI.		AIAAAAACGC		GCGGGGCAGACA	
a.errips.	(5)	GCCAGAACCI.	GAMAACIGCA	ATAAAAACGC	GATTAGATTA	GCAGGCAGACA	CAGACAAAC
		370	380	390	400	410	
		I	Ĩ		Ĩ	Ĩ	
A.PCC7118	(M)	TGTGAATGCA	GATAAAACC	ו ATGGAATTG	TGGTCAAGCT	I AATAAGGGCTA	ATGGT
A.PCC7118	(A)	TGTGAATGCA	GATAAAACC	ATGGAATTG	TGGTCAAGCT	AATAAGGGCTA	ATGGT
A. PCC7120	(M)	TGTGAATGCAG	GATAAAACC	ATGGAATTG	TGGTCAAGCT	AATAAGGGCTA	ATGGT
A. PCC7120	(A)	TGTGAATGCA	GATAAAACC	ATGGAATTG	TGGTCAAGCT	AATAAGGGCTA	ATGGT
N.ellips.	(M)	TGTGAATGCAG	GATAAAACC	ATGGAATTG	TGGTCAAGCT	AATAAGGGCTA	ATGGT
N.ellips.	(A)	TGTGAATGCAG	GATAAAACCA	ATGGAATTG	TGGTCAAGCT.	AATAAGGGCTA	ATGGT

.

of the bands was sometimes found where there were runs of certain bases, and all nucleotides could occasionally be found at a particular position in the sequencing gel, making reading of some gels a little difficult. However, these technical discrepancies were eliminated by repeating the sequencing procedure and by sequencing both strands. For example, differences in sequence of the lower strand were found at base position 38, 249, 250, 332 and 343 by using the M13 forward primer in the sequencing reaction. At position 249 and 250, all four bands A, T, G, C were present at the same position in the gel, for *Anabaena* PCC 7120 and *Nostoc ellipsospora* DNA; whilst for *Anabaena* PCC 7118 DNA there was one very faint band T at position 249 and one strong band T at position 250 (Fig. 6.10). These errors were corrected by sequencing the upper strand by using the reverse M13 primer: a single band 'A' being found on the gel at both position 249 and 250 for all three strains. At position 343 A or G could be found using *Nostoc ellipsospora* DNA (Three cloned *Nostoc ellipsospora* DNA fragments were manually sequenced, two of them indicated as G in that position, one of them indicated as A, which may be due to microheterogeneities or misincorporations by the *Taq* polymerase).

Problems with resolving bands on the gels did not occur with the automated sequencing, where output was in a machine readable form, so that the sequence could be translated as a fluorogram trace. Representative auto-sequencing data from the 16S-23S spacer region of strain *Synechocystis* PCC 6803 and *Microcystis aeruginosa* are shown in machine output format in Fig. 6.11. The sequences can be automatically and easily determined from the peaks: A, T, G, C. Figure 6.12 shows the partial sequence of the 16S-23S rDNA spacer of strain *Synechocystis* PCC 6803 by automated sequencing, compared with the published sequence from Genbank data (Accession D 90916, upper strands sequenced). The sequences were identical. Table 6.3 compares manual and automated sequencing techniques for analysis of the 16S-23S rDNA spacer region.


Figure. 6.10 Sequencing electrophoresis gels of amplified products of partial 16S-23S rDNA spacer of selected cyanobacterial strains. The forward M13 primer, reverse M13 primer and the Sequenase 2.0 sequencing kit (see section 2.1.1) were used; 1. Anabaena PCC 7118; 2. Anabaena PCC 7120; 3. Nostoc ellipsospora. The positions of the arrow bars (----) correspond to base position 250 from Figure 6.9.

Figure 6.11. Fluorogram trace of automated sequencing data. The data from the 16S-23S spacer region of strain *Synechocystis* PCC 6803 and *Microcystis aeruginosa* in machine output format from the ABI PRISM 377 automated DNA sequencer.

.



0į1	- 2			Š		
Page 1	ບ 		T G C	S		
au	5	Z	C I G	X		
7 8:15 3:39 p ABI100		3	C A C	X		
127199 41997 3: 8.00	G T T 80		U U U	X	A A N 250	
MonJar ⁼riJan2 Spacinç	5 5 5 5	5	TTT	M	ATA	
2 H O)	L L	\sim	1 G T	- A		\sim
	A G T T 70			A A A A A A A A A A A A A A A A A A A	5 5 5	
	0 5		L I C	- Maria	C C T 24	\sim
89 2	C T A		C G C		V C C	Ş
97 C:15 Ier} e 1: 95	60 T	\leq			C T A	X
03 T:19 nyPrim 7 Bas	C T V		A T T		G C 7 230	- A
246 A:3 A Set-A ix 1403	T C A C	\sim	C T G		0 5 1	\longrightarrow
nal G:2 4%Ac{/ a . Matr	1 1 1 50	Ş	G T G		0 0 0	\sim
	0 ⊢ 0	\langle	A A G		C A G	
	T T G	~~~~~~~~~~~~~~~~~~~~~~~~~~~~~~~~~~~~~~~	V V D		T A T	
	T T T 40	\leq	3 T A			
ginosa	L C		T C A		C A C 210	
.1 stis aeru	1 1 1	\sim	V V V V		A C C	Ž
34•MA Microcy	T G C G 30 C G	<	A C T		9 9 9	
	V CC T					
	1 10	5	T C T	M	V V D	\leq
	20 20 20 20		T T C	Z	C L C	
	T A TT		V O U		V 9 9	
377 n 2.1.1	CCT 1	5	0000	S	0 0 1 1 0 0	A A A A A A A A A A A A A A A A A A A
Model Versio	T A G C		A G A		CAT	M
		<	C T C		S B B	
ABI		\leq	TITA	M	G T G	2

•

indicate the positions of each sequence. Identity is 100% for the 344bp sequences. Sequences are shown 5'-3' upper strand.

Criterion	Manual sequencing	Automatic sequencing
DNA requirement	Cloned PCR products by using TA vector	PCR products or cloned PCR products
Number of 500bp sequences resolved on one gel	1-2	9 (using dye- primer); 36 (using dye- terminator)
Sequence reading	Using autoradiogram (36h exposure), read manually or automatically by scanning into computer.	Directly read (real-time) from gel by computer
Number of bases read in one loading	150 - 200 bp	500 - 700 bp (dye - ddNTP); up to 900 bp (dye - primer)
Time taken to obtain results	5 days	2 days
Potential errors	Gaps or all bases A, T,G,C appearing at the same position or compression of bands or leakage of sample from next well leading to mis-reading.	Impure PCR products cause too much background or more than one peak shown in the same position. Also there can be leakage of sample from next well of gel.

Table 6.3 Comparison of manual and automated sequencing techniques.

A-Adenine, T-Thymine, G-Guanine, C-Cytosine.

Overall the sequencing results indicated that automated sequencing should provide a reliable method for rapidly surveying sequences of cyanobacterial DNA, and the technique has been used here to sequence selected strains. However, it should be noted that PCR products of the partial 16S-23S rDNA spacer, amplified by *Taq* polymerase from Bioline (U.K.), all failed to provide accurate data by automated sequencing. A strong background, in the form of non-specific peaks, appeared on the fluorogram trace when sequencing such amplified products. Clear sequencing data, without background, were provided from PCR products amplified using DyNAZyme II. The error frequency of DyNAZyme II is supposedly two-fold lower than that of *Taq* DNA polymerase (according to Finnzymes Oy instructions). This higher degree of specificity made DyNAZyme II the enzyme of choice for PCR-amplification of the cyanobacterial DNA for automated sequencing, in order to reduce background interference.

Partial cyanobacterial 16S-23S rDNA spacer sequences from amplified PCR products using primers 17/18 (which amplify tDNA^{ile}, spacer, tDNA^{ala}, spacer and 5' end of 23S rDNA) were recovered, except for an estimated 5 nucleotides at the 5' end of tDNA^{ile}, for selected cyanobacteria by automated sequencing. Conserved nucleotide positions were aligned using the CLUSTAL W program for all these laboratory strains, together with *Synechococcus* PCC6301 (Accession No. X00346), *Trichodesmium* NIBB1067 (Accession No. X72871), *E. coli* (Accession No. J01702) and *Chondrus crispus* (red alga) chloroplast (Accession No. Z29521), to determine the extent of sequence homology (Figure 6.13). The two tDNA sequences were found in all cyanobacterial strains, except *Microcystis aeruginosa*, which did not contain tDNA^{ala}. Two other conserved regions and one region of high sequence similarity were also present in the cyanobacteria: the two conserved regions are within the variable spacer

Figure 6.13 Alignment of 16S-23S rDNA spacer sequences from various cyanobacterial species. Sequence starting from the 5' end of tDNA^{ile} to the beginning of 23S rDNA. Alignment was obtained for sequences from 15 laboratory strains (this study) together with *Synechococcus* PCC6301 (Accession No. X00346), *Trichodesmium* NIBB1067 (Accession No. X72871), *E. coli* (Accession No. J01702) and *Chondrus crispus* (red alga) chloroplast (Accession No. Z29521) from Genbank, using the CLUSTAL W program version 1.6 (http://www.ibcp.fr/clustalw.html). Alignment across the tDNA^{ile} (position 1-70) and tDNA^{ala} (163-238) sequence was very good, but gaps were present within the spacer region. Some of these gaps were manually removed. A-Adenine, T-Thymine, G-Guanine, C-Cytosine. -, absence of nucleotide. Numbers above the sequences represent base positions. *, heterocystous cyanobacterial strains. The regions of high homology are boxed.

	10 	20 	30 	40 	50 	60
E.coli M.aeruginosa S.PCC6301 S.PCC7942 O.animalis T.NIBB1067 A.D0752 * A.D0746 * A.D0697 * A.Cylindrica * A.PCC7118 * A.PCC7120 * C.fritschii * F.musicola * G.sp * N. <i>ellipsospora</i> * N.muscorum * N.D0734 * C.c.chloroplast	TTGTAGCTCAGG TATTAGCTCAGG TATTAGCTCAGG TATTAGCTCAGG TATTAGCTCAGG TATTAGCTCAGG TATTAGCTCAGG TATTAGCTCAGG TATTAGCTCAGG TATTAGCTCAGG TATTAGCTCAGG TATTAGCTCAGG TATTAGCTCAGG TATTAGCTCAGG TATTAGCTCAGG TATTAGCTCAGG TATTAGCTCAGG	TGGTTAGAGC TGGTTAGAGC TGGTTAGAGC TGGTTAGAGC TGGTTAGAGC TGGTTAGAGC TGGTTAGAGC TGGTTAGAGC TGGTTAGAGC TGGTTAGAGC TGGTTAGAGC TGGTTAGAGC TGGTTAGAGC TGGTTAGAGC TGGTTAGAGC	GCACCCCTGAT GCACCCCTGAT GCACCCCTGAT GCACCCCTGAT GCACCCCTGAT GCACCCCTGAT GCACCCCTGAT GCACCCCTGAT GCACCCCTGAT GCACCCCTGAT CGCACCCCTGAT CGCACCCCTGAT CGCACCCCTGAT CGCACCCCTGAT	AAGGGTGAGGT AAGGGTGAGGT AAGGGTGAGGT AAGGGTGAGGT AAGGGTGAGGT AAGGGTGAGGT AAGGGTGAGGT TAAGGGTGAGGT TAAGGGTGAGGT TAAGGGTGAGG TAAGGGTGAGG TAAGGGTGAGG TAAGGGTGAGG TAAGGGTGAGG TAAGGGTGAGG TAAGGGTGAGG TAAGGGTGAGG TAAGGGTGAGG	CCGGTGGTTCA CCCTGGTTCA CCCTGGTTCA CCCTGGTTCA CCCTGGTTCA CCCTGGTTCG CCCTGGTTCG CCCTGGTTCG CCCTGGTTCG CCCTGGTTCG TCCCTGGTTCG TCCCTGGTTCG TCCCTGGTTCG TCCCTGGTTCG TCCCTGGTTCG	AGTCCA AGTCCA AGTCCA AGTCCA AGTCCA AGTCCA AGTCCA AGTCCA AGTCCA AGTCCA AGTCCA AGTCCA AGTCCA AGTCCA AGTCCA AGTCCA AGTCCA AGTCCA
E.coli M.aeruginosa S.PCC6301 S.PCC7942 O.animalis T.NIBB1067 A.D0752 * A.D0746 * A.D0697 *	70 CTCAGGCCTAC GGATGGCCCAC GGATGGCCCAC GAATGGCCCAC GGATGGCCCAC GGATGGCCCAC GGATGGCCCAC GGATGGCCCAC	80 CTAAAT CTAGC CTTAA CTTAAG CTGATAAGTA C-TGAAGA GGCCCACA	90 -TTGCACGGCA -TTTTTCATGA -TTTTTCATGA -TTAGAAGTTA -TTAGAAGTTA ATTTAGTAATTA ATTTTAGATTT	100 AATTTGAAGAQ GAGTGAAGAAQ AAAATCTCTAA AAAATCTCGTAA CTAGTTCGTAA TAAATTTTGGA TAGATTTTAGA	110 GGTTTTTAACTA GT GT GT GTTTGTAATCC AG ATTCGTAATGA ATTTTTAGATTG ATTTTAGATTA	120 TAAA AAAA TATTTTT AAAAAAT

.

A.cylindrica *	GGATGGCCCACCTGAAGCAAGTCAAAAGTTAGAAGTCAAAAGTCAAAAGTAATCAT
A.PCC7118 *	GGATGGCCCACCTGAAGCAAGTCAAAAGTTAAAAGACAAAAGTCAAAAGTAGTTATTT
A.PCC7120 *	GGATGGCCCACCTGAAGCAAGTCAAAAGTTAAAAGACAAAAGTCAAAAGTAGTTATTT
C.fritschii *	GGATGGCCCACTCGT
F.musicola *	GGATGGCCCACCTGA-~AGCAATTAGAAATGAAAATTTTAGAATTGAGAGTTAGACATTC
G.sp *	GGATGGCCCACCTGAA-CAAGTAAAAAGAGCAAAGTAAAAAGTAAAAAGAAAGTGTAA
N.ellipsospora*	GGATGGCCCACCTGAAGCAAGTCAAAAGTTAAAAGACAAAAGTCAAAAGTAGTTATTT
N.muscorum *	GGATGGCCCACCTGAATCAATGCAAAATTAAAAATTCAAAATTCAAAATTAAGAAATT
N.D0734 *	GGATGGCCCACCTGA-~AGCAATGAAAAATTCAAAATACAAAATTCAAAATGGTTTATTT
C.c.chloroplast	GGATAGCCCAAATGAAATAATAAT

	130 	140 	150	160	170 	180
E.coli			!	CATG	י TTATGGGGC	I TATAGCTCA
M.aeruginosa						CACA
S.PCC6301				G	ATGTGGGGG	TTTAGCTCA
S.PCC7942				G	ATGTGGGGG	TTTAGCTCA
O.animalis	TTTTTAA	TTTGTAATT	TTAATTTGTAA	TTTCCCAAA	GCGTGGGGG	TATAGCTCA
T.NIBB1067				G	GTAAGGGGG	TATAGCTCA
A.D0752 *	A-TTAGGAA	TTAGGAATT	AGAATTAAGAA	TTATTT	TGATGGGGG	TTTAGCTCA
A.D0746 *	AAATCTAAAA	TCGCAAATC	CAAAATCCAAAA	TTCAT	-GATGGGGG	TTTAGCTCA
A.D0697 *	AATCCAAAA	TCCGAAATC	CAAAATCCAAAA	TTGTG	-GATGGGGG	TTTAGCTCA
A.cvlindrica *	TTTTGA	TTTTGAGTT	TGAATTTTGAA	TTGTT	-TTTGGGGG	TTTAGCTCA
A.PCC7118 *	ACTTTTGA	TTTTGGATT	TGAATTTTGAA	TTGTA	-TTTGGGGG	TTTAGCTCA
A.PCC7120 *	ACTTTTGAA	TTTTGGATT	TGAATTTTGAA	TTGTA	-TTTGGGGG	TTTAGCTCA
C.fritschii *					GGGGGG	TATAGCTCA
F.musicola *	ΤΤΑΑ-ΤΤΤΤΤΑΖ	TTACTAANT	TTGAATTTTTAA	TTCTT	TAGCGGGGG	TTTAGCTCA
G.sp *	AT-TCTTTTGA	ATTTTTCCTT	TAAACTTTTACC	TTGTG	-GATGGGGG	TTTAGCTCA
N.ellipsospora*	ACTTTTGA	እጥጥጥጥGGAጥጥ'	ΓΓΓΑΔΤΈΓΓΓΑΔ	TTGTA	-TTTGGGGG	TTTAGCTCA
N.muscorum *	GTTTTTGA	ATTTTGCATT	TTGAATTTTGAA	TTGTG	TTGGGGGGGG	TTTAGCTCA
N.D0734 *	AATTTTGA	ATTTTGAATT'	TTTAATTTTGAA	TTGTG	-TGGGGGGGG	TTTAGCTCA
C.c.chloroplast				A	GTAGGGGGG	TATAGCTCA

	190	200	210) 220) 230	240
		1				ļ
E.coli	GCTGGGAGAG	GCCTGCTT	GCACGCAGG	AGGTCTGCGGT	TCGATCCCGC	ATAGCTCCAC
M.aeruginosa	GGTGGCAAAA-		ACAAGAGA	AGCGAGGAA	CAGCACC	TTATCT
S.PCC6301	GTTGGTAGAG	CGCCTGCTTI	GCAAGCAGG	ATGTCAGCGG	TCGAGTCCGC	TAACCTCCAC
S.PCC7942	GTTGGTAGAG	CGCCTGCTT	GCAAGCAGG	ATGTCAGCGG	FTCGAGTCCGC	TAACCTCCAC
O.animalis	GTTGGTAGAG	CGCTGCCTTI	GCAAGGCAG	ATGTCAGCGG?	FTCGAGTCCGC	TTACCTCCAC
T.NIBB1067	GTTGGTAGAG	CGCTGCCTTI	GCAAGGCAG	AGTCAGCGG	FTCGANTCCGC	CTTACCNCCAG
A.D0752 *	GTTGGTAGAG	CGCCTGCTT	GCAAGCAGG	ATGTCAGCGG	FTCGAGTCCGC	CTAACCTCCAC
A.D0746 *	GTTGGTAGAG	CGCCTGCTT	GCAAGCAGG	ATGTCAGCGG	TTCGAGTCCGC	CTAACCTCCAC
A.D0697 *	GTTGGTAGAG	CGCCTGCTT	rgcaagcagg	ATGTCAGCGG	TTCGAGTCCGC	CTAACCTCCAC
A.cylindrica *	GTTGGTAGAG	CGCCTGCTT	rgcaagcagg	ATGTCAGCGG'	TTCGAGTCCG	CTAACCTCCAC
A.PCC7118 *	GTTGGTAGAG	CGCCTGCTT	rgcaagcagg	ATGTCAGCGG'	TTCGAGTCCG	CTAACCTCCAC
A.PCC7120 *	GTTGGTAGAG	CGCCTGCTT	rgcaagcagg.	ATGTCAGCGG'	TTCGAGTCCG	CTAACCTCCAC
C.fritschii *	GTTGGTAGAG	CGCCTGCTT	rgcaagcagg.	ATGTCAGGAG	TTCGAGTCTC	CTTACCTCCAC
F.musicola *	GTTGGTAGAG	CGCCTGCTT	IGCAAGCAGG	ATGTCAGGAG	TTCGAGTCTC	CTAACCTCCAC
G.sp *	GTTGGTAGAG	CGCCTGCTT	IGCAAGCAGG	ATGTCAGCGG	TTCGAGTCCG	CTAACCTCCAC
N.ellipsospora*	GTTGGTAGAG	CGCCTGCTT	IGCAAGCAGG	ATGTCAGCGG	TTCGAGTCCG	CTAACCTCCAC
N.muscorum *	GTTGGTAGAG	CGCCTGCTT	IGCAAGCAGG	ATGTCAGCGG	TTCGAGTCCG	CTAACCTCCAC
N.D0734 *	GTTGGTAGAG	CGCCTGCTT	IGCAAGCAGG	ATGTCAGCGG	TTCGAGTCCG	CTAACCTCCAC
C.c.chloroplast	GCTGGTAGAG	CGCTGCCTT	TGCAAGGCAG	ATGTCAGCGG	TTCGAGTCCG	CTTATCTCCAC

	250 	260 	270 	280 	290 	300
E.coli						
M.aeruginosa						
S.PCC6301						
S.PCC7942						
O.animalis	TCTTACCCAACCA	GCCGAAAA				CAAAC
T.NIBB1067						
A.D0752 *						
A.D0746 *	CTGA					AG
A.D0697 *				-		
A.cylindrica *	ATTG					
A.PCC7118 *	ATGA					AA
A.PCC7120 *	ATGA					AA
C.fritschii *	CTGGATTTGCCAG	CAAAAGAAAI	AGAGAATGAC	CAAAAGCATC	TAGGTTTGTC	AATGAGC
F.musicola *	CTGT					A-
G.sp *	ATGATATCTGTTG	TGACAGCAGC	CAAAAGTGAAT	ATATCACAGO	TAAAAATTAG	CCAAAAG
N.ellipsospora*	ATGA					AA
N.muscorum *	CTTG				·	·TA
N.D0734 *	ATGA					AA
C.c.chloroplast						

r

	310	320	330	340	350	360
	1					
E.coli	-CATCTCTGTAGT		GGTT	AAAT-	AAAA	AATA
M.aeruginosa	TACTGAC	A				
S.PCC6301	-CAAAAGACTGCT		TAAA	ATTC-	АААА	AATC
S.PCC7942	-CAAAAGACTGCT		- TAAA	ATTC-	АААА	AATC
O.animalis	AGAAACAAAAGAC		CAAA	AACA-	АААА	AATA
T.NIBB1067	-GAATAAAAAAGG		TAGA	AATC-	AGCA	A
A.D0752 *				ATG-	GAAAA	GAC
A.D0746 *	CGGTAAGC		GAACAAA	CAAAA-	GGAAA	AAG
A.D0697 *	ATTGGC		TTACT			
A.cylindrica *					GAAAA	GAG
A.PCC7118 *	AGGCAAGC		CAAG	AAAGA-	GGAAA	AAT
A.PCC7120 *	AGGCAAGC		CAAG	AAAGA-	GGAAA	ААТ
C.fritschii *	AGACTAGATTGGG	ATTGGCTCI	TGTGCTGACAAA	TGTGGATGI	GATAAGATAG	AAAATTG
F.musicola *	TAGGT			AAATG-	GGTAA	AAT
G.sp *	TGATAGAGTGAGT	AGTTTTTGT	GTTCATAGCAGA	GAATGI	'GGTAAGATAA	GAGATTG
N.ellipsospora*	AGGCAAGC		CAAG	AAAGA-	GGAAA	AAT
N.muscorum *	GCGTCAGC		TATAAAT	AGAGG-	GATTA	AAT
N.D0734 *	AGAAGCTTCAAAC	AG	CTTCAAA	AAAAG-	AGAAA	AAA
C.c.chloroplast	-AAA				*********	

	370	380	390	400	410	420
			1	1		
E.coli	CTTCAGA	GTG		!	TACCTGCA	AAGGTTCACT
M.aeruginosa						TAGTAAG-
S.PCC6301	AGTTCAG	CA		FTTCGATTT	TCGTATCG	GAAGGTTA
S.PCC7942	AGTTCAG	CA	TCTTAAGT	rttcgatt <mark>t</mark> '	TCGTATCG	GAAGGTTA
O.animalis	AGTCAAA	AAACTGTTGC	GCTTTTAGCA	ATATGTGAT.	AGACTAGA	AAGAGTGACT
T.NIBB1067 -	CAAG	A			TAAAGO	AAAGCAA
A.D0752 *	ACAG	CAACTGG			GF	GATTTAAACC
A.D0746 *	TCAG	CAACTAATAC	SA		G1	AAGACTCGT-
A.D0697 *	TCAG	CACCTGAAA	CT		G]	GAAAATAGTA
A.cylindrica *	TCAG	CAACTGACG-				GATAGT-
A.PCC7118 *	TCAG	CAAC			1	AACGGGAGT-
A.PCC7120 *	TCAG	GCAAC			!	TAACGGGAGT-
C.fritschii *	TGAGTAATTTCAG	CAACTAAT			?	TAATAGTAGT-
F.musicola *	CTAG	GCACCTGC			·G.	raataaagaat
G.sp *	TGGGTAAATTCAG	SCAATTGACG	r		(CAAAAAGAGT-
N.ellipsospora*	TCA0	GCAAC			!	raacgggagt-
N.muscorum *	TCA0	GCAACA				IGGCTATAGC-
N.D0734 *	TCA0	GCAAC			'	FAATGATAGT-
C.c.chloroplast						GTTAA

.

.

	430	440	450	460	470	480
E.coli	G			CGAAG <u>TT</u>	<u></u>	-CTTTAA
M.aeruginosa				AGAGAATO	SCTOCTTCT	'G-AGTAA
S.PCC6301				GGAATO	GCTQA	-GTTGAA
S.PCC7942				GGAATO	GCTGA	GTTGAA
O.animalis	ATAAAAGCTCAG	GCACCTAGATTC	AGCTACTAGCI	TAATCGGAAT	зстфс	TCTTCC
T.NIBB1067				GTCATO	GCTQCTGGTA	ATTCTTAA
A.D0752 *				TAGAdTO	GCTCGG	TGA
A.D0746 *				TAGAdT(GCTCGG	TAA
A.D0697 *				CAGAAT	GCTCGA	TGA
A.cylindrica *				CAGAdT	GCTGGA	TTA
A.PCC7118 *				TAGAQT	GCTQGG	TAG
A.PCC7120 *				TAGA(T	GCTQGG	TAG
C.fritschii *				TAGA(T	GCTCGG	- -ATG
F.musicola *				CAGGQT	GCTCGG·	TGA
G.sp *				САААДТ	GCTGGA	AAT
N.ellipsospora*				TAGAdT	GCTGGG	TAG
N.muscorum *				CAGAdT	GCTGGA	TAG
N.D0734 *				TAGAd <u>T</u>	GCTGAG	GAA
C.c.chloroplast				GTTGT	ACT	TCTCAA

	490	500	510	520	530	540
			1			
E.coli	AAATCTGGATC	AAGCTG <u>AAA</u>	ATTGAAACAC	<u>TGAACAA</u> CGA	AAGTTGTTC-	-GTGAGTCTC
M.aeruginosa	AGAGTCC	:- A G-Adga <i>f</i>	ACCTTGAAAAC	TGCATAQAGC	TAG	-GTGAAAAAG
S.PCC6301	AGACTC	-AGCAAGAA	ACCTTGAAAAAC	TGCATAQ-AG	GATGAGAGTG-	TAGGTATC
S.PCC7942	AGACTC	-AGCAAGAA	ACCTTGAAAAA	TGCATAQ-AG	GATGAGAGTG-	TAGGTATC
O.animalis	AAGTCC	-AGCCAGA	ACCTTGAAAAA	CTGCATAQAAA	ATAAGTCATC-	AAAAAGTT
T.NIBB1067	ATA-CC	C-AGTAAGAA	ACCTTGAAAAA	TGCATACTCA	AAAAAAGTC-	AAAGTAAC
A.D0752 *	GATTCC	C-AGCCAGA	ACCTTGAAAAA	CTGCATAQAAA	ACGCGATAG	CAGGCAGT
A.D0746 *	TACC	TAGCCAGA	ACCTTGAAAAA	CTGCATAQAAA	ACGCGATTA	GAT-AGCAGG
A.D0697 *	AAGTCC	C-AGCCAGA	ACCTTGAAAAA	CTGCATACTA	ACGCGA	GGCAGT
A.cylindrica *	AA-ACO	C-AGCCAGA	ACCTTGAAAAA	CTGCATACTA	ACGCGAATT	'AGCAGGCAGA
A.PCC7118 *	TAC(CTAGCCAGA	ACCTTGAAAAA	CTGCATAAAAA	ACGCGATTA	GATTAGCAGG
A.PCC7120 *	TAC(CTAGCCAGA	ACCTTGAAAAA	стссаталал	ACGCGATTA	GATTAGCAGG
C.fritschii *	AT-TCC	C-AGCCAGA	ACCTTGAAAAA	CTGCATAQAGA	AGCGAAAAGI	AGGTAGTCAG
F.musicola *	GAGTC	C-AGCCAGA	ACCTTGAAAA	CTGCATAQAGA	AAGCGAATG	TAT-GGTAGT
G.sp *	A-GTC(C-AGCCAGA	ACCTTGAAAAA	CTGCATACTA	ACGCGAAAAAI	'AGCAGGCAGA
N.ellipsospora*	TAC	TAGCCAGA	ACCTTGAAAA	CTGCATAAAAA	ACGCGATTA	-GATTAGCAGG
N.muscorum *	TTATCO	C-AGCCAGA	ACCTTGAAAA	CTGCATAGGA	ATGCGATTT	·AAGCAGG
N.D0734 *	ACT0	C-AGCCAGA	ACCTTGAAAA	<u>CTGCATAG</u> AA/	ACGCGATTA	-GAGCAGG
C.c.chloroplast	ATA	A	AATTTTATTA	TTTGATAATAA	AAAAAAAGT	AAAGTAAG

	550	560	570	580	590	600
	1		1			
E.coli	TCAA				A'	TTTTCG
M.aeruginosa	CCAA					
S.PCC6301	ACAG				ACACCA	FCTTCA
S.PCC7942	ACAG				ACACCA	TCTTCA
O.animalis	TGAG				~AGACAT	TTACCAGG
T.NIBB1067	TGAA				CCAGAG	GTTAAG
A.D0752 *	CGAAAAA				GTCAAAA	AAAAGACG
A.D0746 *	CAGA				CACAG	ACAT-GTA
A.D0697 *	TCAG				TAATC	AAAA-GCG
A.cylindrica *	CGAAACTGAG	TACTGAGGAAA	AGAGTACCGAG	TAATGGGTGA	AGAACTCAT	TGCCGAGT
A.PCC7118 *	CAGA				CACA-	
A.PCC7120 *	CAGA				CACA-	
C.fritschii *	TGAACAGT				GAACAG	TAATCAGT
F.musicola *	TCA				GTCAAAA	GTGAGCAG
G.sp *	CGAAATTTCAAG	TAATGAGTGC:	FGAGTGCTGAG	TGCTAAGTGI	AAAACACAA	AGGCTCCC
N.ellipsospora*	CAGA				CACA-	
N.muscorum *	CAGA				CACA-	
N.D0734 *	CAGA				CACAA	GTATTGGA
C.c.chloroplast	Т					

	610	620	630	640	650	660
				4		
E.coli	CAACTC	TG	AAGTGAAACAT	rctt		CGGGTT
M.aeruginosa			AAAAGAAGAC	CGC		
S.PCC6301	TCATCAC-	TI	GATTGAGTCA	AGTGGGAGAA	ATGGAAGTC	AGAAAGAAG
S.PCC7942	TCATCAC-	TI	GATTGAGTCA	AGTGGGAGAA	ATGGAAGTC	AGAAAGAAG
O.animalis	AATTTATTCT-	AG	GTGAAAGTCA	AATCAAG	CCGACAATC	АААСААААА
T.NIBB1067	TAAAACC-	TI	AAAAAATCT	AGTTCAG	GTGTA	GAAAGAAAA
A.D0752 *	AAGAGACTGC-	AAGTTGGA	AAACCAATGA	AAAG	CATAATTAC	GAATTAAGT
A.D0746 *	ATAGATGG		-AAGTGTTTGC	AGTG	GTGAACACC	-AATGTGAT
A.D0697 *	AAAGCAGC		AGATAACTGA	AACG	AGAAAAACC	-AATGTAAT
A.cylindrica *	AGGAAAGAAA-	CAGGAAAGCA	GAAAAGTTTGC.	AA	GTCGAACAA	AATGTATT
A.PCC7118 *	GACAAA		-CTGTGAATGC	AGGA	ТА-АААСС-	-AATGGAAT
A.PCC7120 *	GACAAA		-CTGTGAATGC	AGGA	TA-AAACC-	-AATGGAAT
C.fritschii *	AAAAAAACT	GGTGA	CTGGTAACTGA	AAACTGAA	CAGACACCA	AAAGTTATT
F.musicola *	TGAACAGT	AATCAA	AAGAAAACTGA	GAAC	TGAAACAGA	AGACTGAGT
G.sp *	ATGAAAGTAGT	CAGGAAAGAG	AAAGAGTTTGC	AGAT	GTGAAACCA	-AATGTATT
N.ellipsospora*	GACAAA		-CTGTGAATGC	AGGA	TA-AAACC-	AATGGAAT
N.muscorum *	GACATT		-GGATGTGTAT	GCAA	TTGAAGCT	-AATTGAAT
N.D0734 *	AAGATATA		-AAGTGTTTGC	AGGA	TAGAAACC-	AATTGTGA
C.c.chloroplast						

		670	680	690	700	710
			1			
E.coli	GT		(GAGGTTAAC	<u>CGACTAAGC</u>	GTACACGGT
M.aeruginosa				AAGGTCAAG	CTAATAAGO	GCTAACGGT
S.PCC6301	TA			GTGGTCAAG	CTACGAAGO	GCTTACGGT
S.PCC7942	ТА			dtggtcaag	GCTACGAAGO	GCTTACGGT
O.animalis	AA			dtggtcaag	GCTACAAAGO	GCTGACGGT
T.NIBB1067	AA			GTGGTCAAG	GTTATAAAGO	GCTGACGGT
A.D0752 *	TTTCG-1	AATTAC	AATTATCAA	dtggtcaag	SCTAATAAG	GCTAATGGT
.A.D0746 *	A			dtggtcaag	SCTAATAAG	GCTAATGGT
A.D0697 *				dtggtcaad	SCTAATAAG	GCTAATGGT
A.cylindrica *	GA		A	dtggtcaag	GCTAATAAG	GCTAATGGT
A.PCC7118 *	T			diggicaad	GCTAATAAG	GCTAATGGT
A.PCC7120 *	т			diggicaad	JCTAATAAG	JGCTAATGGT
C.fritschii *	т			GTGGTCAAG	GCTAATAAG	GCTTGTGGT
F.musicola *	AACAGA	CACCAATO	STTGTGAGAA	dTGGTCAA	GCTAATAAG	GCTTGCGGT
G.sp *	G			TGGTCAA	GCTAATAAG	gctgatgt
N.ellipsospora*	т			diggicaa	GCTAATAAG	GCTAATGGT
N.muscorum *	т			GTGGTCAA	GCTAATAAG	GCTGACGGT
N.D0734 *	A			GTGGTCAA	GCGAATAAG	GCTAATGGT
C.c.chloroplast					AAG	AGCTTACGGT

•

between tDNA^{ala} and the 5' end of 23S rDNA, a very short conserved sequence (5 bp) from position 463 to 467 and a 20 bp sequence from position 498 to 517; the region of high sequence homology is from position 686 to 710, which is at the 5' end of 23S rDNA. High sequence similarities were exhibited in the spacer between tDNA^{ile} and tDNA^{ala}, from position 70 to 160 for the heterocystous cyanobacteria, except strain *Chlorogloeopsis fritschii*, which did not contain this spacer. Unicellular strains all possessed a short sequence in this spacer. Sequences of high similarity could also be found for heterocystous strains from position 370 to 376 and 518 to 526. Such similar sequences indicate that heterocystous strains are closely related.

The sequences for Anabaena PCC 7118, Anabaena PCC 7120 and Nostoc ellipsospora were not only similar to each other, but also to other heterocystous filamentous strains, in particular the Anabaena cylindrica, Nostoc muscorum and Nostoc linckia D0734. High sequence similarites were found for all these, except in those regions from positions 346 to 396 and 527 to 661, where Anabaena PCC 7118, Anabaena PCC 7120 and Nostoc ellipsospora only showed close homology to the Nostoc strains. The sequence of the unicellular strains, Synechococcus PCC6301 and Synechococcus PCC7942, were found to be identical.

Figure 6.14 shows the position of restriction endonuclease sites in the partial cyanobacterial 16S-23S rDNA spacer, deduced from the sequencing data, using the program of full restriction endonuclease sites analysis in GeneJockey II. Those enzymes that cut within the conserved tDNA regions are useful for RFLP analyses, because they cut the DNA from all cyanobacteria tested, and yield polymorphic markers from the variable length spacer. Digested fragments of the same size always exhibited a high

Figure 6.14. Diagrammatic representation of the position of restriction endonuclease sites in the cyanobacterial partial 16S-23S rDNA spacer. Complete restriction maps were obtained by using GeneJockey II package. , tDNA (tDNA^{ile} and tDNA^{ala}), — , variable spacer. The restriction endonuclease sites in the tDNAs were normally found for all the cyanobacteria. Sites may be present or absent in the variable spacer depending on the strain. Approximate primer annealing sites are shown for primers 17, 18, SF and SR by arrowed bars. Arrows indicate direction of extension.



degree of sequence similarity. However, the enzymes that only cut within the variable spacer region are not useful for RFLPs, because their recognition sequences are not likely to be found in all cyanobacteria. Most of the enzymes that cut within the conserved regions generated similar restriction profiles for all the cyanobacteria, in terms of number of bands, with either one band of the same size, and two of different sizes; or two bands of the same size, and one of different size, with the exception of Hinf I, Taq I and Hae II. From Figure 6.13, sequence differences were found between specific strains at the recognition sites for these three enzymes. At position 54 (in tDNA^{ile}), one base pair difference was found between heterocystous strains and nonheterocystous strains, except Microcystis aeruginosa, and this position is within the restriction site for Hinf I and Taq I. This difference proved useful in separating unicellular, non-heterocystous filamentous, and heterocystous filamentous cyanobacteria (see RFLP patterns Fig. 6.5). At position 194-196 (tDNA^{ala}), sequence differences were found for Oscillatoria animalis and Trichodesmium NIBB 1067, whilst all other strains appeared to have the same sequence. This is the position of the only restriction site for Hae II. Consequently the partial 16S-23S spacer of Oscillatoria animalis and Trichodesmium NIBB 1067 are not digested by Hae II. Thus Hae II would not be as good as Hinf I and Taq I, which both have one restriction site within each of the conserved regions for tDNA^{ile} and tDNA^{ala}, at discriminating between cyanobacteria. Some enzymes in Figure 6.14 only have single or double restriction sites in tDNA^{ile}. Such enzymes did not change the regional variation within the spacer.

Figure 6.15 shows three consensus trees constructed by using DNA parsimony algorithm with 1, 100, 500 bootstrapping runs, performed on the sequencing data from the partial 16S-23S spacer region of 15 laboratory cyanobacterial strains, together with *Synechococcus* PCC6301 (Accession No. X00346), *Trichodesmium* NIBB1067

Figure 6.15 Consensus trees derived from parsimony method based on partial 16S-23S spacer region sequence analysis. Data from 15 laboratory cyanobacterial strains, two published cyanobacterial sequences of *Synechococcus* PCC 6301 (Tomioka and Sugiura, 1984) and *Trichodesmium* NIBB 1067 (Wilmotte *et al.*, 1994); *E.coli* (Young and Steitz, 1978); and *Chondrus crispus* (red alga) chloroplast (Leblanc *et al.*, 1995) were analysed. The rooted tree was generated by the parsimony method and 500 bootstrapping runs were performed, using the DNAPARS, SEQBOOT and CONSENSE program in the software package PHYLIP. (A): The numbers at the forks indicate the number of times the group consisting of the species which are to the right of that fork occurred among the trees, out of 500 trees. (B): The numbers at the forks indicate the number of times the group consisting of the species which are to the right of that fork occurred among the trees, out of 100 trees. (C): The numbers at the forks indicate the number of times the group consisting of the species which are to the right of that fork occurred among the trees, out of 100 trees. (C): The numbers at the forks indicate the number of times the group consisting of the species which are to the right of that fork occurred among the trees, out of 100 trees. (C): The numbers at the forks indicate the number of times the group consisting of the species which are to the right of that fork occurred among the tree, out of 1 tree. The rDNA spacer was amplified with primer pair 17/18. Clusters A, B, C, D are assigned on each tree.







(Accession No. X72871), E. coli (Accession No. J01702) and Chondrus crispus (red alga) chloroplast (Accession No. Z29521). The three trees show identical results with the majority of branches showing bootstrapping values of 50% or more indicating that the trees are quite robust. Anabaena PCC7118 and Anabaena 7120 appear close to Nostoc ellipsospora, and are closer to other Nostoc strains (Nostoc linckia D0734 and Nostoc muscorum) than to other Anabaena strains. The sequence identity of 100% over 446 bases indicates that they are in the same species group (Table 6.4). Furthermore, Trichodesmium NIBB 1067 appears close to Oscillatoria animalis, but sequence homologies were not found in the aligned sequences, apart from the conserved regions described for all cvanobacteria (Fig. 6.13). The sequence identity of 56.87% indicates that they may not be closely related. Figure 6.16, shows a likelihood tree constructed by using the Maximum Likelihood method for the same strains as those of Figure 6.15. The parsimony tree and likelihood trees are similar, except for the relationship between Anabaena and Nostoc strains. In Figure 6.16, Anabaena PCC7118, Anabaena 7120 and Nostoc ellipsospora appear closer to Anabaena cylindrica than to Nostoc muscorum, and a little more close to Nostoc linckia D0734. This suggests that these strains may be located between Nostoc muscorum and Anabaena cylindrica, which contrasts with the relationship shown in figure 6.15. In both trees (Fig. 6.15 & 6.16), the cyanobacteria were separated into four clusters: (A) Synechococcus strains; (B) Oscillatoria animalis and Trichodesmium NIBB 1067 (together with Chondrus crispus chloroplast); (C) Microcystis aeruginosa; (D) Chlorogloeopsis fritschii, Fischerella musicola, Gloeotrichia sp. and all Nostoc and Anabaena strains.

Table 6.4 The si	imilarity of	partial 16	S-23S rDì	VA spacer	sednences	; from 16 (cyanobact	erial speci	Ś								
					}			Seg	luence iden	tity (%)*							
		S PCC 7942	T NIBB 1067	0. animalis	A. D0697	A. D0746	A. D0752	A. cylindrica	A. PCC7118	A. PCC7120 h	(, ellipsona	N. muscorum	N. D0734	G. sp	F. musicola	C. fritschii	<u>M. aeruginosa</u>
435	S. PCC 7942	100	53.3	53.83	52,35	54.81	48.14	51.36	51.85	51.85	47.16	50.12	53.09	52.35	50.12	50.37	57.89
394	T. NIBB 1067		100	56.87	52,47	54.12	5S.49	51.82	51.92	51.92	48.9	55.49	56.32	48.08	49.73	54.4	52.23
564	O. animalis			100	51,71	47.03	49.11	47.94	51.2	51.2	50.84	47.51	50.23	45.69	45.43	45.63	47.73
440	A D0697				ŝ	61.22	64.88	56.1	8	8	59.76	61.95	62.44	59.02	53.41	47.56	52.23
	A D0746					100	28'65	\$2.97	74.04	74.04	73.73	59.86	67.97	54.57	50.23	51.6	52.23
USA I	A D0752						100	58.44	62.41	62.41	62.23	58.67	60.14	56.22	59.11	47.33	50.61
491	A colindrica				}			100	69.15	69.23	67.52	56.29	61.98	68.76	51.3	44.68	53.04
446	A PCC7118								100	100	100	74.28	78.55	62.89	60.24	51.32	53.44
446	A PCC7120				}					10	100	74.28	78.55	62.89	60.27	51.32	53.44
446	N ellips										100	74.25	72.6	61.3	59.38	51.2	52.63
451	N muscorum											100	69.83	57.72	60.81	50.36	50.61
484	N. D0734												100	61.52	55.76	45.85	50.2
619	G.so				}									100	54.13	54.58	53.04
	E musicala														10	50.43	52.23
	Linestolla								}							100	52.63
777	M annunce													_			<u>6</u>
* Similarities were to using Genelockey I	ased on parti I.	ial 16S-23S	rDNA space	ar sequence:	s for 15 cyal	nobacterial I	aboratory s	trains this st	udy and the f	published se	quence for	Trichodesmi	um NIBB 1	067. Seque	nce identity	(%) was obt	ained

Figure 6.16 Maximum likelihood tree based on partial 16S-23S spacer region sequence analysis. Data from 15 various cyanobacterial and two published cyanobacterial sequences of *Synechococcus* PCC 6301 (Tomioka and Sugiura, 1984) and *Trichodesmium* NIBB 1067 (Wilmotte, *et al.*, 1994), *E.coli* (Young and Steitz, 1978) and *Chondrus crispus* (red alga) chloroplast (Leblanc *at al.*, 1995) was used to construct the likelihood tree by the Maximum Likelihood method using DNAML version 3.572c program in the software package PHYLIP. Total 680 trees were examined by program. The rDNA spacer was amplified with primer pair 17/18. Cluster A, B, C, D are assigned.



0.10

From the sequence alignment shown in figure 6.13 using 19 spacer sequences, two short regions of homology of about 20 bp were identified for all the cyanobacteria tested. One within the variable spacer, between tDNA^{ala} and 5' end of 23S rDNA, and one at the 5' end of 23S rDNA. These sequences were not found in the other organisms analysed, and thus showed potential for use in the design of cyano-specific primers. Accordingly two primers SF and SR were synthesised. These primers amplified a region between tDNA^{ala} and 5' end of 23S rDNA (see Fig. 6.1). SF derived from the conserved region from position 498 to 517 (see Fig. 6.13), which was found in all the cyanobacterial strains studied here, and in seven other cyanobacteria (see Wilmotte, 1994); but not in other bacteria as judged by searching the Genbank data base (see Table 6.5). Whilst SR was from the region of high homology found at position 686 to 704, which was also specific for cyanobacteria when compared with other bacteria.

The sequence and physical characteristics of each of these primers are presented in Table 6.6. The MgCl₂ concentration and the annealing temperature were tested to optimise the PCR reaction with these primers (Fig. 6.17). A MgCl₂ concentration of 2.5mM and annealing temperature of 50 °C generated clear, reproducible bands. Thus these conditions were used in the PCR. No primer-dimer formation occurred when primer pair SR/SF was used for the PCR, as judged by the absence of any band corresponding to amplified primers themselves on gels after amplification. Products could be detected using these primers with about 15pg purified cyanobacterial DNA in a 50µl reaction mixture, with 35 cycles of the PCR (Fig. 6.18). Template DNA concentration dependencies were observed from 15pg/50µl reaction to 150pg/50µl reaction, but they were not observed at 1.5ng/50µl reaction.

Table 6.5 Specificity of cyanobacterial primers

Accession No.	Sequences producing High-scoring Segment Pairs ^a	Identity (%)
(1). primer SF		
emb X70769 AS16S	Arthrospira sp. gene for 16S rRNA	100
emb X75044 ASRRNTRN	Arthrospira sp. genes for 16S rRNA, t	100
emb X75045 SSRRNTRN	Spirulina sp. genes for 16S rRNA, tRN	100
emblX72871 TSITS	Trichodesmium sp. gene for 23S rRNA	100
emblX00346 ANRN02	Anacystis nidulans 16S-23S spacer region	100
emblX00512IANRN03	Anacystis nidulans 23S rRNA gene and	100
dbilD64000 SYCSLRB	Synechocystis sp. PCC6803 complete gene	91
dbilD90916lD90916	Synechocystis sp. PCC6803 complete gene	91
gb U53123 GCU53123	Geminigera cryophila nucleomorph 18S rRNA	91
(2). primer SR		
emb X75044 ASRRNTRN	Arthrospira sp. genes for 16S rRNA, t	92
emb X00512 ANRN03	Anacystis nidulans 23S rRNA gene and	92
emb X75045 SSRRNTRN	Spirulina sp. genes for 16S rRNA. tRN	92
emb X72871 TSITS	Trichodesmium sp. gene for 23S rRNA	92
dbj D90916 D90916	Synechocystis sp. PCC6803 complete ge	92
(2) primer SP18		
(3), primer SKTO	Anonystic nidulane 225 rDNA cone and	100
	Sumochogystis and DCC6802 complete an	100
	Synechocysus sp. FCC0005 complete ge	100
	Ruman FLI 5/FLK2 ligand mKNA, complet	97
embjXUUU12jCHSURN1	Spirodela oligorniza chloroplast 5°	92
gb L35485 HUMIDS	Homo sapiens iduronate sulphate sulp	92
gb U66082 HSU66082	Human contig of seven cosmids from L	92
(4). primer 18		100
emb/X00512/ANRN03	Anacystis nidulans 23S rRNA gene and	100
ebiU04806iHSU04806	Human FLT3/FLK2 ligand mRNA, complet	97
emb[X00012]CHSORN1	Spirodela oligorhiza chloroplast 5'	91
gb L35485 HUMIDS	Homo sapiens iduronate sulphate sulp	92
gb U66082 HSU66082	Human contig of seven cosmids from L	92
gb M21402 SOYRDNASP	Soybean chloroplast 16S/23S ribosoma	91
gb M21453 SPICPRGDG	Spinach chloroplast 16S ribosomal RN	91
gb M75722 ALSCP23SA	Alnus incana chloroplast 23S ribosom	91
emb/X59768/CA23SRR	C.americana 23S rRNA gene	91
gb M37149 SOYCPRTG	Soybean chloroplast 16S rRNA (3' end	91
emb X01365 CHZM23S	Maize chloroplast DNA for 4.5S and 2	91
gbJ01446 TOBCPRG23	Tobacco chloroplast 23S and 4.5S rRN	91
emb X62099 CHEVIARRN	E.virginiana plastid ribosomal RNA o	91
emb X55033 CHPSRRNAO	P.sativum chloroplast DNA of the sin	91
gb M37430 PEACPDKAA	Pea Chloroplast 4.5S, 5S, 16S and 23	9 1
emb Z00028 CHZMRRNA	Z.mays chloroplast rRNA-operon	91
gb M81884 EPFCPCG	Epifagus virginiana chloroplast comp	91
dbj D17510 PINCPTRPG	Black pine chloroplast DNA, complete	91
emb X04465 CHMPXX	Liverwort Marchantia polymorpha chlo	91
emb Z00044 CHNTXX	Tobacco chloroplast genome DNA.	91
emblX86563 CHZMXX	Z.mays complete chloroplast genome	91
dbj D90916 D90916	Synechocystis sp. PCC6803 complete g	91

a, Sequences producing High-scoring Segment Pairs (>90 % identity) were obtained from 312,067 sequences in Genbank data (BLAST e-mail: blast@ncbi.nlm.nih.gov).

Table 6.6 Sequence and physical characteristics of primers SF, SR and SR-18.

		Primer ^a	
Property	SF	SR	SR18
Sequence	5'-GAACCTTGAAAACTGCATAG-3'	5'-AGCCCTTATT AGCTTGACC-3'	s'-TTTGCGGCCGCTCTGTGTGCC-3'
Length (bp)	20	19	21
Dissociation temperature (Tm)	56 °C	56 °C	70 °C
% G+C	40%	47.4%	66.7%

a, primer SF and SR developed in this study; SR18 was part of primer 18.



Figure 6.17 Agarose gel elctrophoresis of amplified partial 16S-23S rDNA spacer of Anabaena cylindrica at different magnesium chloride concentrations and different annealing temperatures. Lane 1, 5 mM; 2, 2.5mM; 3, 1.5mM; 4, 0.5mM MgCl₂. Lane 5, 60 °C; 6, 55°C; 7, 50°C; 8, 45°C with the magnesium concentration of 2.5mM for the primer SF/SR.



-

Figure 6.18 Agarose gel electrophoresis of amplified partial 16S-23S rDNA spacer of Anabaena cylindrica at different total genomic DNA concentration. Lane 1, 15ng; 2, 1.5ng; 3, 150pg; 4, 15pg; 5, 1.5pg; 6, 0.15pg in 50µl reaction with the magnesium concentration of 2.5mM for the primer SF/SR.

Specificity of the SF/SR primer pair was checked using DNA from 25 cvanobacteria, from a green algae (isolated from Raby pool) and from E.coli K12 (LJMU). A single band of 70-350 bp was generated for each cyanobacterium. No PCR products were obtained from the green algae or *E.coli* DNA (see Fig 6.19). Moreover, this pair of primers showed higher specificity for amplifying cyanobacterial DNA than either primer pair 17/18 or SF/18, which both produced a weak PCR product with green alga DNA and E.coli DNA. Whilst SF/SR18 (where SR18 was part (21 bp) of primer 18, see Table 6.5 & 6.6), which did not amplify green alga DNA and *E.coli* DNA, was also more specific than primer pair 17/18 or SF/18. In addition, SF/SR was tested directly on a mixed culture of isolates from a fresh water sample designated sample 1 (see Table 2.3 and section 3.2.4). This sample supposedly contained six strains of two cyanobacterial species, as determined from selective culturing (see Table 3.2) and RAPD data (see section 5.2.3). When the DNA of the sample was amplified using SF/SR, two clear bands were generated (Figure 6.19, lane 30). The larger band (350bp) corresponds to that of Synechococcus LJM003 (Fig. 6.19, lane 5) and the smaller band (90 bp) to Pseudanabaena LJM001 (Fig. 6.19, lane 9).

The validity of using the primer sequences of SF and SR for identifying cyanobacteria was tested by constructing parsimony consensus (Fig. 6.20) and Likelihood (Fig. 6.21) trees, based on the SF/SR amplified rDNA spacer sequences for cyanobacterial species. The parsimony consensus and likelihood trees are similar, except for the position of *Microcystis aeruginosa*. Furthermore, they are like the trees in figures 6.15 and 6.16 based on the 17/18 amplified spacer sequences, in terms of clusters, showing only one more cluster containing *Oscillatoria animalis*.



Figure 6.19 Agarose gel elctrophoresis of partial 16S-23S rDNA spacer amplified using primers SF and SR. Lane M, 123bp DNA ladder; 1, DNA negative control; 2, green alga; 3, E. coli; 4, Synechococcus PCC 7942; 5, Synechococcus LJM003; 6, Synechocystis PCC 6803; 7, Microsystis aeruginosa; 8, Oscillatoria animalis; 9, Pseudanabaena sp. LJM001; 10, Nostoc MAC PCC8009; 11, Anabaena cylindrica; 12, Aphanizomenon gracile; 13, Tolypothrix distorta; 14, Chlorogloeopsis fritschii; 15, Fisherella musicola; 16, DNA negative control; 17, Gloeotrichia sp.; 18, Anabaena sp. D0752; 19, Nostoc sp. D0766; 20, Anabaena flos-aquae D0848; 21, Anabaena sp. D0746; 22, Nostoc sp. D0740; 23, Nostoc linckia D0734; 24, Nostoc muscorum; 25, Nostoc commune; 26, Anabaena PCC 7120; 27, Anabaena sp. D0697; 28, Leptolyngbya JM014; 29, Nostoc paleo; 30, fresh water sample 1.

Figure 6.20. Consensus trees derived from parsimony method based on analysis of the partial 16S-23S spacer region sequence amplified using primers SF and SR. Data from 15 laboratory cyanobacterial strains, two published cyanobacterial sequences of *Synechococcus* PCC 6301 (Tomioka and Sugiura, 1984) and *Trichodesmium* NIBB 1067 (Wilmotte, *et al.*, 1994), *E.coli* (Young and Steitz, 1978) and *Chondrus crispus* (red alga) chloroplast (Leblanc *at al.*, 1995) were analysed. The rooted tree was generated by the parsimony method and 500 bootstrapping runs were performed, using the DNAPARS, SEQBOOT and CONSENSE program in the software package PHYLIP. (A): The numbers at the forks indicate the number of times the group consisting of the species which are to the right of that fork occurred among the trees, out of 500 trees. (B): The numbers at the forks indicate the number of times the group consisting of the species which are to the right of that fork occurred among the trees, out of 500 trees. (B): The numbers at the forks indicate the number of times the group consisting of the species which are to the right of that fork occurred among the trees, out of 500 trees. (B): The numbers at the forks indicate the number of times the group consisting of the species which are to the right of that fork occurred among the trees, out of 1 tree. Clusters A, B, C, D are assigned.

h --





B.



0.10

Figure 6.21. Maximum likelihood tree based on SF/SR amplified partial 16S-23S spacer region sequences. Date from 15 various cyanobacterial and two published cyanobacterial sequences of *Synechococcus* PCC 6301 (Tomioka and Sugiura, 1984) and *Trichodesmium* NIBB 1067 (Wilmotte, *et al.*, 1994), *E.coli* (Young and Steitz, 1978) and *Chondrus crispus* (red alga) chloroplast (Leblanc *at al.*, 1995) were used to construct the likelihood tree by Maximum Likelihood method using DNAML version 3.572c program in the software package PHYLIP. Total 751 trees were examined by program.

6.3 Discussion

The results indicate that heterogeneity in both the length and the sequence of the 16S-23S rDNA spacer (or ITS region) is useful for typing cyanobacteria. Clearly PCR analysis of the 16S-23S rDNA spacer region may open the way for a rapid, possibly automatable, general method for cyanobacterial identification and taxonomy. The priming sites for PCR amplification were selected from regions of highly conserved sequences found for example in the 16S rDNA, tDNA and 23S rDNA regions adjacent to the intervening spacer region (see Fig. 6.1). The choice of primers was crucial to the use of spacer variation for species identification. Three primer pairs 1/18, 14/18 and 17/18 (Wilmotte et al., 1993) tested for amplification of the 16S-23S rDNA spacer generated products useful for discriminating between cyanobacterial strains. Multiple DNA bands of different intensities were obtained by using primer pairs 1/18 and 14/18. The intense and reproducible bands have been classified as primary products, the weaker bands as secondary products (see Table 6.1 & Fig. 6.2 a, b). Such classification of product profiles has been reported in the amplification of the 16S-23S rDNA spacer of other bacteria (Jensen et al., 1993). Weak bands were common in the heterocystous strains, which may reflect rDNA heterogeneity for heterocyst differentiation. The pattern of amplified products with primer pair 14/18 enabled clearer distinction between species than that with primer pair 1/18. This suggests that primers closer to the 16S-23S spacer were more discriminatory than those further away. The effect of S1 endonuclease on banding patterns indicated that most secondary products were heteroduplex DNA, containing substantial single-stranded regions and single-stranded DNA (see Table 6.1). These heteroduplexes would migrate more slowly in gels than homoduplexes, because of the single-stranded regions. The heteroduplex structures presumably formed by partial primer sequence matches or cross-hybridisation between similar, but not
identical, spacer amplification products, whilst the formation of single-stranded DNA structures could result from asymmetry in either amplification efficiency or consumption of the primers (Jennies and Stares, 1993). The formation of heteroduplex DNA could be extremely useful in providing additional pattern information by single-strand conformation polymorphisms (SSCP) (Gurtler and Stanisich, 1996), which helps in the differentiation of heterocystous cyanobacteria. Two homoduplex 16S-23S rDNA spacer amplification products were always obtained from heterocystous strains. The variability in products generated for different cyanobacteria with these primer pairs may be due to differential specificity of the primers, variation in copy number and microheterogeneities of the 16S-23S rDNA spacer regions among cyanobacteria.

Only a single band was obtained by using primer pair 17/18, so that speciesspecific differences were usually detected by digestion of the PCR product with a restriction endonuclease, prior to electrophoresis, and RFLP analysis; or by DNA sequencing. The reasons for the recovery of only a single band with this primer pair may be that: (1) the primers may anneal with only one of the multiple 16S-23S rDNA spacer regions being amplified; (2) only one copy of the 16S-23S rDNA spacer region is actually present per genome of the organism being studied; (3) different PCR products with similar sequence and G+C content generate the same sized band on the gel.

Analysis of length polymorphism and RFLPs of the amplified rDNA spacer using primer pair 17/18 generally permitted discrimination between the cyanobacteria tested to genus and species level. For strains of the same species the same RFLP patterntype was obtained. Different species and genera had different pattern types, with the exception of *Nostoc ellipsospora*, *Anabaena* PCC 7120, and *Anabaena* PCC 7118. An identical length and RFLP pattern-type, together with high sequence similarity of the amplified partial 16S-23S rDNA for these cyanobacteria suggests that they are closely related and may be in the same species group. Caudales and Wells (1992) pointed out that strains of the genus Nostoc are closely related to, and often confused with those of the genus Anabaena, particularly when morphological characteristics are used for identification. By contrast, on the basis of DNA-DNA hybridisation data and hybridisation patterns with highly repetitive (STRR) DNA sequences the strain Anabaena PCC 7120 has been renamed Nostoc sp. PCC7120 (Lachance, 1981; Mazel et al., 1990). In addition, the low similarity obtained between the fresh water Synechococcus PCC7942 and the marine Synechococcus WH7803, on the basis of both length and pattern-type of the spacer, suggests that these strains are not closely related, in line with the hybridisation data (Wood and Townsend, 1990). A comparison of the RFLPs of the 16S rDNA and the 16S-23S rDNA spacer indicates that the 16S-23S rDNA spacer is better for discrimination among the different genera and species of cyanobacteria. This is consistent with the view that the spacer region is generally considered to be under minimal selective pressure (Barry et al., 1991). Its evolutionary rate is ten times greater than that of 16S rDNA (Leblond-Bourger et al., 1996), making it possible to distinguish closely related bacterial species.

Restriction endonuclease sites within the partial cyanobacterial 16S-23S spacer region were identified from the sequencing data (Fig. 6.14). Some of the enzymes cut in the conserved region, so proving useful for RFLP analysis, whilst others cut only in the variable spacer and so were not as useful, because the recognition sequence was not found in all cyanobacteria. The useful enzymes could be placed into three groups: (1) Enzymes that have only a single restriction site in tDNA^{ala}: *Dde* I, *Alu* I, *Nsp*B II, *Tth*111 II and *Hae* II. (2) Enzymes that have a single restriction site in tDNA^{ala}, and a single restriction site in tDNA^{ile}, generating one band of the same size (within tDNA^{ile}), and

two or more of different sizes, from all cyanobacteria. They are Mnl I, Fok I, HinP I or *Hha* I. The restriction profiles using an enzyme from group 1 and group 2 should be similar, except for individual differences in the spacer. (3) Enzymes that have a single restriction site in tDNA^{ala}, and in heterocystous strains and *Microcystis aeruginosa* a single restriction site in tDNA^{ile}, generating further polymorphisms between nonheterocystous strains (excepting Microcystis aeruginosa) and heterocystous strains. They are Taq I, Hinf I and Ple I. Thus for analysis of length polymorphism and RFLPs of the amplified rDNA spacer using primer pair 17/18, the sequencing data indicates that restriction enzymes should be chosen from these groups, preferably two or three enzymes from group 1 and/or 2, and one or two from group 3. Such choices are in line with those that were made for RFLP-PCR (section 6.2.1), on the basis of digesting PCRamplified products. Four restriction enzymes were chosen from twenty tested: two (Dde I and Alu I) are in group 1, and two (Hinf I and Taq I) are in group 3. In this study, the restriction map of the 16S-23S rDNA spacer deduced from the sequencing data thus confirms that the restriction enzymes used for analysis of RFLPs of the amplified spacer using primer pair 17/18 were very suitable.

The trees based on RFLPs (Fig. 6.6) and sequencing data (Fig. 6.15 and 6.16), all show clustering of the unicellular, rod-shaped cyanobacterial strains (genera *Synechococcus*); the non-heterocystous, filamentous strains (genera Oscillatoria, *Phormidium* and *Trichodesmium*); the unicellular, spherical-shaped strains (genera *Microcystis, Chroococcus, Gloeocapsa* and *Synechocystis*) and the heterocystous strains (genera *Anabaena, Nostoc, Gloeotrichia, Fischerella* and *Chlorogloeopsis*), but the trees are not identical. This is largely due to the limitation of the RFLP analysis. To generate RFLPs only a small portion of the rDNA spacer is scanned. Small sequence variations affecting restriction enzyme sites can lead to changes in RFLP profiles for sequences that are otherwise very similar. This may lead to re-grouping of closely related species. For example, *Anabaena cylindrica*, *Anabaena* PCC 7120 and *Nostoc muscorum* contain very similar sequences between the two tDNAs, but a four base deletion appears at position 119-122 in *Anabaena cylindrica*, which is not found in these other strains (Fig. 6.13). This four base difference produces one band of a different size in RFLP profiles among these strains. As a consequence *Anabaena cylindrica* is positioned further away from *Anabaena* PCC 7120 and *Nostoc muscorum* in the distance tree based on RFLP data than in that based on sequencing data. Thus RFLP pattern type does not necessarily correlate fully with the extent of sequence similarity. RFLP analysis of the amplified rDNA spacer thus provides a useful method for rapid identification and typing of cyanobacteria, but there are drawbacks to its use in establishing taxonomic relationships.The clustering of cyanobacterial strains shown from the trees is consistent with that of the 16S rRNA sequence trees (Wilmotte, 1994).

From sequence alignment of the cyanobacterial 16S-23S spacer regions, the sequences of the unicellular strains, *Synechococcus* PCC6301 and *Synechococcus* PCC7942, were identical. This is consistent with the results of RFLP analysis of genomic DNA from strains PCC6301 and PCC7942, which indicated that the two genomes were identical (Golden *et al.*, 1989; Wood and Townsend, 1990) and suggests that these two strains are possibly of the same species or sub-species. The sequence identity of 100%, over 446 bases, for *Nostoc ellipsospora, Anabaena* PCC 7120, and *Anabaena* PCC 7118 (Table 6.4) indicates that they may be in same species group. This supports the findings of Lachance (1981) that *Anabaena* PCC 7120 and *Anabaena* PCC 7118 were more closely related to *Nostoc* strains than to other *Anabaena* strains, on the basis of DNA-DNA hybridisation data. *Trichodesmium* NIBB 1067 appeared close to *Oscillatoria animalis* on the basis of the sequence trees (Fig. 6.15 and 6.16), but

sequence homologies were not found from the aligned sequences, apart from the conserved regions, described for all the cyanobacteria (Fig. 6.13). The sequence identity of 56.87% indicated that they may not be closely related. From sequences of the 16S rDNA, Wilmotte and colleagues (1994) found that *Trichodesmium* NIBB 1067 and *Oscillatoria* PCC7515 showed 94.9% sequence similarity. Nevertheless, the genotypic heterogeneity exhibited by a number of strains assigned to the genus *Oscillatoria* hinders a taxonomic decision on the separate existence of the genera *Trichodesmium* and *Oscillatoria*.

The high sequence similarities found for heterocystous strains suggest that these strains are closely related. In the `spacer between tDNA^{ile} and tDNA^{ala}, high sequence similarity was found for all heterocystous cyanobacteria, excepting Chlorogloeopsis (Chlorogloea) fritschii which lacks this spacer. This species produces 'Gloeocapsa' like aggregates (Rippka, 1988), which is a characteristic of unicellular spiracleshaped strains. Furthermore, the unicellular spiracle-shaped strains of Microcystis also lacked this spacer, and other unicellular strains exhibited a shorter sequence in this spacer. In the other spacer, gaps are found in the sequences of the unicellular strains when aligned with those of the heterocystous strains. These gaps presumably reflect the genetic distance between the different strains: the more gaps there are in the sequence of one strain relative to another, the greater distance between the strains. As a consequence sequence alignment becomes increasingly more difficult and less meaningful as sequence diversity between strains increases. However, this can be compensated for by provision of sequence information for strains that are intermediate to the unrelated strains. Wilmotte (1994) concluded that no meaningful alignment of the rDNA spacer sequences of their tested strains was possible. However, in this study, useful information on the extent of sequence similarity between selected cyanobacterial strains has been

obtained based on the alignment of the partial 16S-23S rDNA spacer sequences. Furthermore, since the spacer sequence is short, and of variable length (300-750 bp), it is easy to sequence and analyse, and the variation shown for different species can be exploited in the identification and taxonomy of cyanobacteria.

Apart from the two tRNA gene sequences that were conserved in all the cyanobacterial strains, except Microcystis aeruginosa (which did not have tDNA^{ala}), three other conserved regions were present in all the cyanobacteria tested. Two conserved regions were found in the middle of the variable length spacer between tDNA^{ala} and 23S rDNA. The third conserved region was at the 5' end of the 23S rDNA. By searching the Genbank data base the two conserved regions of about 20bp appeared more specific for cyanobacteria than other bacteria, making them candidates for cyanospecific primers. The pair of primers, SF and SR, developed from these regions, successfully amplified cyanobacterial DNA generating a product of between 70bp and 350bp, but did not amplify DNA from the green alga or *E.coli*. In order to identify isolates, the SF/SR amplified product could be sequenced and data correlated with parsimony consensus and maximum likelihood trees (Fig. 6.20 and 6.21); such trees were found to be similar to those based on the 17/18 amplified sequences (Fig. 6.15 and 6.16). Given the specificity and other properties of the SF/SR primers, the relatively short size of the SF/SR amplified product and the reliability of automated sequencing (as demonstrated here, see Table 6.3), a rapid PCR-based approach to identifying cyanobacteria directly from DNA recovered from environmental samples could be developed (see section 8.2). It should be noted that for direct automated sequencing of PCR products a high degree of specificity of DNA polymerase must be used for amplification of template DNA to reduce errors caused by normal Tag polymerase. An example of such a polymerase is DyNAZyme II, which was used in this study.

Thus PCR amplification and RFLP analysis of the 16S-23S rDNA spacer provide a potentially useful approach to the identification of cyanobacteria. With more data it should be possible to generate a database of RFLP-patterns, in order to identify unknown isolates by the matching of pattern-type and determining similarity to laboratory strains from phenetic analysis. Furthermore, it is possible to detect more than one cyanobacterial strain in a mixed culture from the variability in length and RFLP pattern-type of the rDNA spacer. In addition, sequence variation within the spacer shows promise for use in the identification and phylogeny of cyanobacteria. Moreover, from sequence alignment cyano-specific primer sequences have been designed, opening up possibilities for their use in the rapid identification of cyanobacteria directly from water samples. Exploitation of spacer-region variation should thus provide a rapid and reliable method of identifying cyanobacteria. Chapter 7

Morphological and genetic methods in identification and taxonomy of cyanobacteria

7.1 Introduction

Cyanobacteria taxonomy has been approached in different ways, all with one major goal: defining those phenotypic properties which are taxonomically useful. Because the phenotype is the product of genetic and environmental factors, the stability of certain characteristics is a function of the properties of the environment. Cyanobacteria exhibit a wide range of specific environmental requirements. Conversely, their growth and metabolic activities leave specific impacts on the environment they occupy. The diversity in form and function of cyanobacteria appears to be the result of a long evolutionary process. Fossils similar to cyanobacteria have been reported from about 3.5 billion years (Ga) old sedimentary rocks (Schopf and Walter, 1983), while diversified assemblages of taxa that are morphologically comparable to modern cyanobacterial genera and species existed in rocks 1.0 to 2.0 Ga of age (Knoll and Golubic, 1991). The proper ages of some cyanobacterial genera from fossil evidence are listed in Figure 7.1 (Schopf, *et al.* 1987; Knoll and Golubic, 1991; Woese, 1994).

Diversity of cyanobacteria is expressed by their morphological, biochemical, and physiological properties, which enable them to settle and persist in a wide range of habitats. As in all other organism groups, various cyanobacterial properties are phenotypic expressions of particular genotypes. Traditionally, the task of systematic practice was to analyse, evaluate, and order phenotypes, and to provide a practical frame of reference for purposes of comparing, identifying and quantifying organisms. Implicit in this effort was the ambition to re-construct the phylogenetic history of the observed diversity and, in the process, to assess the relatedness between organisms. This indirect approach met several objective limitations. One of them is, for example, the difficulty to



Billions of years ago

Figure. 7.1 Time line for the planet Earth and the age of cyanobacteria from fossil evidence (Schopf, *et al.* 1987; Knoll and Golubic, 1991; Woese, 1994).

recognise and separate inherited traits from the environmentally induced ones. Another one is the difficulty to decide whether similar phenotypes reflect a close taxonomic relation (homology), or are the result of convergent evolution (analogy) (Wilmotte and Golubic, 1991).

As a result of the morphological simplicity of most prokaryotes, their classification was based largely on physiological properties, as expressed in pure cultures. Only when applied with relatively large and morphologically complex prokaryotes, such as cyanobacteria could a systematic practice rely on morphological criteria. Similar ecological functions of cyanobacteria and eukaryotic algae have traditionally prompted a similar systematic treatment of their natural populations (Geitler, 1925, 1932). While field studies relied mostly on morphological analyses of natural populations, laboratory studies concentrated on culture characterisations. Since the early 1970's the latter approach was applied to those cyanobacteria that were transferred and maintained in culture collections (Rippka et al., 1979; Rippka, 1988), although the differences between the two taxonomic treatments have yet to be reconciled. Both treatments, however, rely largely on characterisation of phenotypes. The limitations discussed above apply to all studies of phenotypic expression, irrespective of whether natural populations or pure cultures are studied, and whether morphological or physiological criteria are applied. However, morphology may change depending on environmental conditions (Evans et al., 1976) and the diversity of strains within a culture may be suppressed by selective culturing conditions (Doers and Parker, 1988). Photosynthetic pigment content, lipid composition, differentiated cell structures and other characters may also be subject to change because of the variable expression of cyanobacterial gene products in culture (Kenyon et al., 1972; Klein et al., 1973). The limitations of morphological and physiological approaches have prompted the development of molecular biological techniques for identification of cyanobacteria.

Edelman, Stanier, Leach and Herdman (1973) suggested that increasing knowledge concerning genotypic relationships would lead to a more satisfactory classification of the cyanobacteria. The idea that evolutionary changes were recorded within the molecular architecture of organisms, and that they could be studied by comparing the sequences of marcromolecular elements, such as nucleic acid bases and amino acids, was proposed 30 years ago by Zuckerkandl and Pauling (1965). In the years that followed, molecular biology and genetics achieved significant methodological advancements and break-throughs. These methods opened the way towards a more direct evaluation of genetic information. However it soon became evident that none of them proved to be the panacea to modern taxonomy, and that each methodological improvement carried a price in terms of newly introduced biases and artefacts. In the use of rRNA nucleotide sequences to study animal phylogeny, Nielsen (1989) warned: "It is important to point out that the molecular data do not provide unequivocal phylogenetic trees and must be treated with just as much criticism, care, and tact as the traditional morphological characters". A certain practical experience with the use of rRNA sequences (whether partial or complete) for the taxonomy of different kinds of organisms has meanwhile been accumulated, and is reflected in the current literature. Thus, it becomes possible to evaluate the advantages and shortcomings of these new methods and to define more realistically what can be expected from them for the cyanobacteria in the future.

In this study, several rapid molecular methods (developed in chapters 5 and 6) were compared with the morphological method, and comment on conflicts and

T.

consensus between morphological data and the positions of the cyanobacterial strains in the genetic data of this study. Comparisons were also made between the molecular data in this study and the 16S rRNA sequencing phylogenetic tree (Wilmotte, 1994).

7.2 Results

An analysis of the 16S-23S rDNA spacer region of cyanobacterial isolates was carried out by using both RFLP-PCR and DNA sequencing. RFLP profiles of the amplified 16S-23S rDNA spacer were generated by using restriction enzyme *Hinfl*, *DdeI*, *AluI* and *TaqI* which were used in section 6.2.1. The RFLP pattern and pattern-type of isolates are shown in Figure 7.2 and Table 7.1. They showed potentially useful discrimination among the different genera and species for isolates. A dendrogram (Fig.7.3) was constructed on the basis of pair-wise genetic distances, using the combined RFLP data from the four restriction enzymes for all the laboratory strains and isolates used. In the tree (Fig. 7.3), the cyanobacteria were separated into seven clusters: (A) *Synechococcus* strains; (B) *Pseudanabaena* strains; (C) *Leptolyngbya* strains; (D) *Nostoc palaeo*; (E) *Oscillatoria* and *Phormidium* strains; (F) *Chroococcidiopsis* LJM013, *Chroococcus prescottii, Gloeocapsa* sp., *Microcystis aeruginosa*, and *Synechocystis* PCC6803; (G) *Aphanizomenon gracile*, *Chlorogloeopsis fritschii, Fischerella musicola*, *Gloeotrichia* sp., *Tolypothrix distorta* and all *Nostoc* and *Anabaena* strains.

Figure 7.4 shows the consensus tree constructed by using DNA parsimony algorithm with 500 bootstrapping runs performed on the sequencing data from the partial 16S-23S spacer region (between primer 17 and 18) of 20 laboratory cyanobacterial strain and 13 isolates, together with *Arthrospira* sp. PCC7345 (Accession

Figure 7.2. Diagramatic representation of RFLP patterns of amplified rDNA products (using primers 17 and 18) from selected cyanobacterial isolates and laboratory strains with the restriction enzymes *Hinfl*, *Ddel*, *AluI* and *TaqI*. The numbers below the tracks correspond to the RFLP patterns in Table 6.2 and 7.1.

.

•



R. 47

251

Isolates ^a	Pattern-type ^b	Hinf I	Dde I	Alu I	Taq I
Unicellular					
Chroococcidiopsis LJM013	AD	H31	D30	A32	T28
Synechococcus LJM002	AE	H30	D29	A31	T27
Synechococcus LJM003	AF	H29	D28	A30	T27
Synechococcus LJM004	AG	H28	D28	A30	T27
Filamentous non-heterocystous					
Leptolyngbya LJM 012	AH	H35	D34	A36	T32
Leptolyngbya LJM 014	AI	H34	D33	A35	T31
Phormidium LJM 017	AK	H32	D31	A33	T29
Pseudanabaena JM 001	J	H10	D10	A10	T10
Pseudanabaena JM 005	J	H10	D10	A10	T10
Pseudanabaena JM 006	J	H10	D10	A10	T10
Pseudanabaena JM 007	J	H10	D10	A10	T10
Pseudanabaena JM 009	AL	H10	D35	A10	T10
Pseudanabaena JM 011	AM	H36	D36	A37	T33
Pseudanabaena JM 016	AL	H10	D35	A10	T10
Pseudanabaena JM 019	AN	H37	D37	A38	T34

Table 7.1. RFLP patterns and pattern-types obtained for cyanobacterial isolates.

a, Isolates identified to genus level on basis of morphological criteria (see section 3.2.4 and Table 3.2).

b, Details of the patterns (H28-H37, D28-D37, A30-A38 and T27-T34) for amplified (primer pair 17/18) rDNA digests are given in Fig. 7.2. Pattern-type derives from the pattern combination with the four test restriction enzymes.

Figure 7.3. Dendrogram indicating genetic distance among cyanobacterial laboratory strains and isolates. The genetic distance from the node, based on the RFLP profiles of 48 strains, is indicated above each branch. The clusters A, B, C, D, E, and F are designated according to branch forming and genetic distance. The less closely related the strains are the greater the genetic distance.

Cyanobacterial isolates are shown in bold.

•



Figure 7.4. Consensus trees derived from parsimony method based on partial 16S-23S spacer region sequence analysis. Sequencing data from 21 laboratory strains and 12 isolates (this study), together with *Arthrospira* sp. PCC7345 (Nelissen *et al.*, 1994), Spirulina sp. PCC6313 (Nelissen *et al.*, 1994), *Synechococcus* PCC6301 (Tomioka and Sugiura, 1984), *Trichodesmium* NIBB1067 (Wilmotte, 1994), *E. coli* (Young and Steitz, 1978), *Cyanophora paradoxa* cyanelle (Janssen *et al.*, 1987) and *Chondrus crispus* (red alga) chloroplast (Leblanc *et al.*, 1995) from the Genbank were used to constract the tree. The rooted tree was generated by the parsimony method and 500 bootstrapping runs were performed, using the DNAPARS, SEQBOOT and CONSENSE program in the software package PHYLIP. The numbers at the forks indicate the number of times the group consisting of the species which are to the right of that fork occurred among the trees, out of 500 trees. Clusters A, B, C, D, E, F, G are indicated. The cyanobacterial isolates are shown in bold.

,



No. X75044), Spirulina sp. PCC6313 (Accession No. X75045), Synechococcus PCC6301 (Accession No. X00346), Trichodesmium NIBB1067 (Accession No. X72871), E. coli (Accession No. J01702), Cyanophora paradoxa cyanelle (Accession No. M19493) and Chondrus crispus (red alga) chloroplast (Accession No. Z29521). The majority of branches show bootstrapping values of 50% or more indicating that the trees are quite robust. Figure 7.5 and Figure 7.6 shows the DNA distance trees constructed by using the Maximum Likelihood method and the Neighbour joining method based on the same sequences as in Figure 7.4. The DNA parsimony tree (Fig. 7.4) and the DNA distance tree (Fig. 7.5) are identical. The DNA distance tree with the Neighbour joining method (Fig. 7.6) is largely congruent with other two trees (Fig. 7.4 and 7.5), except the group of Synechococcus and group of Pseudanabaena exchanged position, and Chlorogloeopsis fritschii returned into the heterocystous strain cluster. In the three trees (Fig. 7.4, 7.5 and 7.6), the strains used were separated into seven clusters: (A) E. coli; (B) Nostoc palaeo; (C) Arthrospira sp. PCC7345, Oscillatoria animalis, Phormidium LJM017 and Trichodesmium NIBB 1067 (together with Chondrus crispus chloroplast and Cyanophora paradoxa cyanelle); (D) Synechococcus strains; (E) Pseudanabaena strains; (F) Chroococcidiopsis LJM013, Chroococcus prescottii, Synechocystis PCC6803, Microcystis aeruginosa, Spirulina sp. PCC6313 and Leptolyngbya LJM012; (G) Chlorogloeopsis fritschii, Fischerella musicola, Gloeoetrichia sp. and all Nostoc and Anabaena strains. All the clusters of cyanobacteria shows great distance to cluster A--E.coli. Observations of the branches, presented below, gives interesting information on the genotypic relationships among the cyanobacteria within each cluster and morphological features are presented below:

 Branches A from RFLP dendrogram and branches D from sequences trees (Synechococcus cluster)

Figure 7.5. Maximum likelihood tree based on partial 16S-23S spacer region sequence analysis. Sequencing data from 21 laboratory strains and 12 isolates (this study) together with *Arthrospira* sp. PCC7345 (Nelissen *et al.*, 1994), *Spirulina* sp. PCC6313 (Nelissen *et al.*, 1994), *Synechococcus* PCC6301 (Tomioka and Sugiura, 1984), *Trichodesmium* NIBB1067 (Wilmotte, 1994), *E. coli* (Young and Steitz, 1978), *Cyanophora paradoxa* cyanelle (Janssen *et al.*, 1987) and *Chondrus crispus* (red alga) chloroplast (Leblanc *et al.*, 1995) from the Genbank, were used to construct the tree by the Maximum Likelihood method using DNAML version 3.572c program in the software package PHYLIP. Total 4168 trees were examined by program. Clusters A, B, C, D, E, F, G are indicated. The cyanobacterial isolates are shown in bold.



Figure 7.6. Distance tree derived from neighbour-joining method based on partial 16S-23S spacer region sequence analysis. Sequencing data from 21 laboratory strains and 12 isolates (this study) together with *Arthrospira* sp. PCC7345 (Nelissen *et al.*, 1994), *Spirulina* sp. PCC6313 (Nelissen *et al.*, 1994), *Synechococcus* PCC6301 (Tomioka and Sugiura, 1984), *Trichodesmium* NIBB1067 (Wilmotte, 1994), *E. coli* (Young and Steitz, 1978), *Cyanophora paradoxa* cyanelle (Janssen *et al.*, 1987) and *Chondrus crispus* (red alga) chloroplast (Leblanc *et al.*, 1995) from the Genbank, were used to construct the tree by Neighbour-joining method using Neighbour-joining/UPGMA version 3.572c program in the software package PHYLIP. Clusters A, B, C, D, E, F, G are indicated. The cyanobacterial isolates are shown in bold.



The trees of Figures. 7.3, 7.4, 7.5 and 7.6 show that all six *Synechococcus* strains form three sub-clusters, with a distance between each sub-cluster (Figs. 7.3, 7.5 and 7.6). Fresh water *Synechococcus* strains PCC6301 and PCC7942, which contain identical sequences of the rDNA spacer, were indicated at same position in one sub-cluster. The other three fresh water *Synechococcus* strains LJM002, 003 and 004 form a tight sub-cluster, the DNA distance between strain LJM003 and LJM004 is less than from any one of them to LJM002. The strain *Synechococcus* CCAP1479/7 and WH7803 form another sub-cluster, but the distance between the two strains is large.

All these strains are simple, rod-shaped, unicellular forms, with distinguished by the division of elongated to cylindrical cells by binary fission in one plane, the presence of thylakoids, and the absence of a thick sheath. The differences between these strains relate to cell size, pigment content and habitat. Synechococcus strains PCC7942, LJM002, LJM003, LJM004 and LJM020 were isolated from fresh water, and can only grow in low salt tolerance media. Synechococcus CCAP1479/7 isolated from brackish water, can grow in both freshwater and sea water media. Synechococcus WH7803 is a marine strain, and can only grown in sea water media. Synechococcus WH7803 is different from other strains used, only containing PE (the culture colour is pink); whilst the other strains do not contain PE. Figure 7.7 shows TEM microphotographs of strains Synechococcus CCAP1479/7 and WH7803. Morphological change depending on selective culturing conditions was found with Synechococcus PC7942. Figure 7.8 C shows the microscopy photographs of strain Synechococcus PCC7942 forming long 'snake' shaped cells under high light intensity or lower CO_2 supplement than normal conditions. Figure 7.8 B shows that the 'snake' is a single unhealthy cell, compared with a normal cell (Figure 7.8 A). The other Synechococcus strains only showed slight changes when culturing conditions were changed.

Figure 7.7. Transmission electron micrograph of a thin section of A, Synechococcus CCAP1479/7 and B, Synechococcus WH 7803. bar, 0.5 μ m. T, thylakoid.

•





B.



Figure 7.8. Micrograph of Synechococcus PCC7942. A) transmission electron micrograph of a thin section of healthy cell of strain PCC7942, bar, 0.5 μ m; B) transmission electron micrograph of a thin section of unhealthy cell of strain PCC7942, bar, 0.5 μ m; C) micrograph of unhealthy cells of strain PCC7942, bar, 10 μ m. T, thylakoid.

.

.



B.

A.



C.



2 Branches B from RFLP dendrogram and branches E from sequences trees (*Pseudanabaena* cluster)

The tree of Figure 7.3 includes eight *Pseudanabaena* strains, the tree of Figures 7.4, 7.5 and 7.6 only include six of them. All the *Pseudanabaena* strains in each tree form a tight cluster. The four trees show the same four sub-clusters; *Pseudanabaena* sp. LJM019 in one sub-cluster with a large distance to the other three sub-clusters; *Pseudanabaena* sp. LJM011 in another sub-cluster with a slight smaller sub-cluster distance to the other sub-clusters than strain LJM019; *Pseudanabaena* LJM009 and LJM016 in another sub-cluster; *Pseudanabaena* LJM001, 005, 006 and 007 in the fourth sub-cluster. The sub-cluster distances between the *Pseudanabaena* clusters are about ten times lower than the sub-cluster distances in the *Synechococcus* cluster.

Branches E from RFLP dendrogram and branches C from sequences trees (Other non-heterocystous filamentous strains)

This cluster contain different strains from the RFLP and sequence trees. In the RFLP tree (Fig. 7.3), this cluster contain three strains, *Oscillatoria animalis*, *Phormidium autumnale* and *Phormidium* LJM017, each strain being separated by a great distance. In sequence trees (Fig. 7.4, 7.5 and 7.6), this complex branch contains exclusively simple filamentous strains, the cyanelle of *Cyanophora paradoxa*, and the chloroplast of *Chondrus crispus*. *Arthrospira* sp. PCC7345 and *Oscillatoria animalis* form one sub-cluster. *Phormidium* LJM017 forms a sub-cluster, and *Trichodesmium* NIBB1067 together with *Chondrus crispus* chloroplast and *Cyanophora paradoxa* cyanelle form a sub-cluster. The great distance was also found between each sub-cluster.

4 Branches D from RFLP dendrogram and branches B from sequences trees

This cluster only contains one strain *Nostoc palaeo*. The RFLP dendrogram shows that this strain is located at a non-heterocystous filamentous strains cluster. The sequence trees show that this strain locates at the early branch far away from other cyanobacterial clusters. This strain which was not listed in Table 2.1 was a gift with no reference from University of Central Lancashire. The morphology of the strain was re-examined as follows: filamentous cyanobacteria with no heterocysts and akinetes, producing cylindrical trichome waved, unconstricted or little constricted at the cross walls; ability to form thin sheaths around the trichome dependence on the culture conditions, lacking false branching, cells \pm isodiametric or longer than wide, polar gas vacuoles, thylakoids present located peripherally and perpendicular to the walls. The diameter of trichome ranges from about 2.0 to 2.5 µm, cell length: width is 1.0–1.5, Trichomes do not contain PE, but synthesise nitrogenase, and are photoheterotrophic positive for sucrose. Colonies are pea green in colour and mats type. Figure 7.9 shows the TEM and microscopy photographs of *Nostoc palaeo*.

5 Branches F from RFLP dendrogram and sequences trees

Branches F from RFLP dendrogram contain five strains grouping into three subclusters, *Chroococcidiopsis* LJM013 in one sub-cluster, *Chroococcus prescottii* and *Microcystis aeruginosa* in one sub-cluster, *Gloeocapsa* sp. and *Synechocystis* PCC6803 in one sub-cluster. The great distances are not only shown between sub-clusters, but also between strains in same sub-cluster. Branches F from sequences trees are heterogeneous branches containing unicellular and filamentous strains. *Leptolyngbya* LJM012 locates at the early branch cluster F in the sequence tree, but is in cluster C in the RFLP dendrogram close to other simple filamentous strains. The morphological result was shown in chapter 3.

262



B.

A.



Figure 7.9. Micrograph of Nostoc palaeo. A) transmission electron micrograph of a thin section of strain Nostoc palaeo, bar, 1.0 μ m; B) micrograph of trichome of strain Nostoc palaeo, bar, 20 μ m. T, thylakoid. G, gas vesicle.

6 Branches G from RFLP dendrogram sequences trees

(Heterocystous filamentous strains)

This cluster contain all the heterocystous filamentous strains. In sequences trees, this cluster divided into six sub-clusters: Chlorogloeopsis fritschii and Fisherella musicola in one sub-cluster; Anabaena sp. D0752 in one sub-cluster; Anabaena sp. D0697 and D0746 in one sub-cluster; Gloeoetrichia sp. and Anabaena cylindrica in one sub-cluster; Nostoc muscorum and MAC PCC8009 in one sub-cluster; Nostoc linckia D0734, Nostoc ellipsospora, Anabaena PCC7118 and Anabaena PCC7120 in one subcluster. In the RFLP dendrogram, this cluster divided into four sub-clusters: Chlorogloeopsis fritschii, Fisherella musicola and Tolypothrix distorta in one subcluster; Nostoc MAC R1, MAC PCC8009, Nostoc sp. Do740, Nostoc linckia D0734, Nostoc piscinale D0201 and Nostoc sp. D0766 in one sub-cluster; Nostoc commune, Nostoc commune D0800, Nostoc planctonicum, Nostoc muscorum, Nostoc ellipsospora, Anabaena PCC7118 and Anabaena PCC7120 in one sub-cluster; Aphanizomenon gracile, Gloeoetrichia sp., Anabaena cylindrica, Anabaena sp. D0752, D0697 and D0746 in one sub-cluster. Figure 7.10 shows a micrograph of N.ellipsospora, Anabaena PCC7118 and Anabaena PCC7120. N.ellipsospora and Anabaena PCC7120 produce heterocysts, Anabaena PCC7118 do not produce heterocysts, and trichome shorter than Anabaena PCC7120 and N.ellipsospora. All three strains perform small life cycle (step 3-14-15-16-17 of Komarek and Anagnostidis 1989).

The parsimony consensus (Fig. 7.11), Likelihood distance (Fig. 7.12) and Neighbour joining distance (Fig. 7.13) trees were also constructed based on the SF/SR amplified rDNA spacer sequence. The parsimony consensus tree and likelihood tree are similar, except for the position of *Synechococcus* PCC7942 and PCC6301, *Phormidium* LJM017 and *Chondrus crispus* chloroplast. Furthermore they are similar in terms of



Figure 7.10. Micrographs of trichome or clony of some heterocystous strain. A) Anabaena PCC7118; B) Anabaena PCC7120; C) Nostoc ellipsospora.

Figure 7.11. Consensus trees derived from parsimony method based on partial 16S 23S spacer region (between two specific primers) sequence analysis. Sequencing data from 21 laboratory strains and 12 isolates (this study) together with *Arthrospira* sp. PCC7345 (Nelissen *et al.*, 1994), *Spirulina* sp. PCC6313 (Nelissen *et al.*, 1994), *Synechococcus* PCC6301 (Tomioka and Sugiura, 1984), *Trichodesmium* NIBB1067 (Wilmotte, 1994), *E. coli* (Young and Steitz, 1978), *Cyanophora paradoxa* cyanelle (Janssen *et al.*, 1987) and *Chondrus crispus* (red alga) chloroplast (Leblanc *et al.*, 1995) from the Genbank were used to constract the tree. The rooted tree was generated by the parsimony method and 500 bootstrapping runs were performed, using the DNAPARS, SEQBOOT and CONSENSE program in the software package PHYLIP. The numbers at the forks indicate the number of times the group consisting of the species which are to the right of that fork occurred among the trees, out of 500 trees. Clusters A, B, C, D, E, F, G are indicated. The cyanobacterial isolates are shown in bold.


Figure 7.12 Maximum likelihood tree based on partial 16S 23S spacer region (between two specific primers) sequence analysis. Sequencing data from 21 laboratory strains and 12 isolates (this study) together with *Arthrospira* sp. PCC7345 (Nelissen *et al.*, 1994), *Spirulina* sp. PCC6313 (Nelissen *et al.*, 1994), *Synechococcus* PCC6301 (Tomioka and Sugiura, 1984), *Trichodesmium* NIBB1067 (Wilmotte, 1994), *E. coli* (Young and Steitz, 1978), Cyanophora paradoxa cyanelle (Janssen *et al.*, 1987) and *Chondrus crispus* (red alga) chloroplast (Leblanc *et al.*, 1995) from theGenbank, were used to constract the tree by Maximum Likelihood method using DNAML version 3.572c program in the software package PHYLIP. Total 4882 trees were examined by program. Clusters A, B, C, D, E, F, G are indicated. The cyanobacterial isolates are shown in bold.



Figure 7.13. Distance tree derived from neighbour-joining method based on partial 16S-23S spacer region (between two specific primers) sequence analysis. Sequencing data from 21 laboratory strains and 12 isolates (this study) together with Arthrospira sp. PCC7345 (Nelissen *et al.*, 1994), Spirulina sp. PCC6313 (Nelissen *et al.*, 1994), Synechococcus PCC6301 (Tomioka and Sugiura, 1984), Trichodesmium NIBB1067 (Wilmotte, 1994), E. coli (Young and Steitz, 1978), Cyanophora paradoxa cyanelle (Janssen *et al.*, 1987) and Chondrus crispus (red alga) chloroplast (Leblanc *et al.*, 1995) from the Genbank were used to constract the tree by Neighbor-joining method using DNADIST and Neighbor-joining/UPGMA version 3.572c program in the software package PHYLIP. Clusters A, B, C, D, E, F, G are indicated. The cyanobacterial isolates are shown in bold.



clusters to those in Figure 7.4 and 7.5 based on the 17/18 amplified spacer sequence except cluster C arrangment. In Figure 7.11, the majority of branches of clusters D and E showing bootstrapping values of 50% or more indicating that the branches are quite robust. The majority of branches of other high diverse strains show bootstrapping values of below 50%, except the strain *Arthrospira* sp. PCC7345 with *Oscillatoria animalis, Anabaena cylindrica* with *Gloeotrichia* sp., *Anabaena* PCC7118 and PCC7120 with *Nostoc ellipsospora, Nostoc palaeo,* and *E. coli.* In the Neighbour joining tree (Fig. 7.13), it is confusing that the positions of *Synechococcus* PCC7942 and PCC6301, *Trichodesmium* NIBB 1067 and *Spirulina* sp. PCC6313 are moved into heterocystous strain cluster.

Partial cyanobacterial 16S-23S rDNA spacer sequences from amplified PCR products using primers 17/18 (which amplify tDNA^{ile}, spacer, tDNA^{ala}, spacer and 5' end of 23S rDNA) were recovered, except for an estimated 5 nucleotides at the 5' end of tDNA^{ile} for cyanobacteria by automated sequencing. Conserved nucleotide positions were aligned using the CLUSTAL W program for 20 laboratory cyanobacterial strain and 13 isolates, together with Arthrospira sp. *PCC7345* (Nelissen *et al.*, 1994), Spirulina sp. *PCC6313* (Nelissen *et al.*, 1994), *Synechococcus* PCC6301 (Accession No. X00346), *Trichodesmium* NIBB1067 (Accession No. X72871), *E. coli* (Accession No. J01702), Cyanophora paradoxa cyanelle (Janssen *et al.*, 1987) and *Chondrus crispus* (red alga) chloroplast (Accession No. Z29521), to determine the extent of sequence homlogy (Figure 7.14). It was noticed that *Chroococcidiopsis* LJM013, *Chroococcus prescottii*, *Synechocystis* PCC6803, *Microcystis aeruginosa*, *Spirulina* sp. PCC6313 and *Leptolyngbya* LJM012 do not contain conserved region tDNA^{ala}. Except the conserved regions described in chapter 6, more high homology regions were found present in the cyanobacterial strains within same species or same genera: Three conserved regions are

Figure 7.14. Alignment of 16S-23S rDNA spacer sequences from various cyanobacterial species. Sequences start from the 5' end of tDNA^{ile} to the beginning of 23S rDNA. Alignment was obtained for sequences from 21 laboratory strains and 12 isolates (this study) together with *Arthrospira* sp. PCC7345 (Accession No. X75044), *Spirulina* sp. PCC6313 (Accession No. X75045), *Synechococcus* PCC6301 (Accession No. X00346), *Trichodesmium* NIBB1067 (Accession No. X72871), *E. coli* (Accession No. J01702) *Cyanophora paradoxa* cyanelle (Accession No. M19493) and *Chondrus crispus* (red alga) chloroplast (Accession No. Z29521) from the Genbank, using the CLUSTAL W program version 1.6 (http://www.ibcp.fr/clustalw.html). Alignment across the tDNA^{ile} (position 1-70) and tDNA^{ala} (163-238) sequence was very good, but gaps were present within the spacer region. Some of these gaps were manually removed. A-Adenine, T-Thymine, G-Guanine, C-Cytosine. - , absence of nucleotide. Numbers above the sequences represent base positions. *, heterocystous strains. The regions of high homology were boxed.

	10	20	30	40	50	60
			<u> </u>			
E.COli	TTGTAGCTCAGG	IGGTTAGAG	GCGCACCCCT	GATAAGGGTG	AGGTCGGTGG	TTCAAGTCCA
M.aeruginosa	TATTAGCTCAGG	TGGTTAGAG	GCGCACCCCT	GATAAGGGTG	AGGTCCCTGG	TTCGAGTCCA
C.prescottii	TATTAGCTCAGG	TGGTTAGAG	GCGCACCCCT	GATAAGGGTG	AGGTCCCTGG	TTCGAGTCCA
S. PCC6803	TATTAGCTCAGG	TGGTTAGAC	GCGCACCCCT	GATAAGGGTG	AGGTCTCTGG	TTCAAGTCCA
C.LJM013	TATTAGCTCAGG	TGGTTAGAC	GCGCACCCCT	GATAAGGGTG	AGGTCCCTGG	TTCGAGTCCA
S.PCC6301	TATTAGCTCAGG	TGGTTAGAO	GCGCACCCCT	GATAAGGGTG	AGGTCCCTGG	TTCAAGTCCA
S.PCC7942	TATTAGCTCAGG	TGGTTAGAO	GCGCACCCCT	'GATAAGGGTG	AGGTCCCTGG	TTCAAGTCCA
S.WH1803	TATTAGCTCAGG	TGGTTAGAC	GCGCACCCCI	GATAAGGGTG	AGGTCCCTGG	TTCAAGTCCA
S.CCAP1479	TATTAGCTCAGG	TGGTTAGA	GCGCACCCCI	CATAAGGGTG	AGTTCCCTGC	STTCAAGTCCA
S.LJM003	TATTAGCTCAGG	TGGTTAGA	GCGCACCCC1	GATAAGGGTG	AGTTCCCTGC	STTCAAGTCCA
S.LJM002	TATTAGCTCAGG	TGGTTAGA	GCGCACCCC1	rgataagggtg	AGGTCCCTGO	GTTCAAGTCCA
S.LJM004	TATTAGCTCAGG	TGGTTAGA	GCGCACCCC1	GATAAGGGTG	AGTTCCCTGC	GTTCAAGTCCA
P.LJM001	TATTAGCTCAGG	TGGTTAGA	GCGCACCCC1	rgataagggtg	AGGTCACTGO	STTCGTGTCCA
P.LJM006	TATTAGCTCAGG	TGGTTAGA	GCGCACCCC	IGATAAGGGTG	SAGGTCACTG	STTCGTGTCCA
P.LJM009	TATTAGCTCAGG	TGGTTAGA	GCGCACCCC	FGATAAGGGTG	SAGGTCACTG	STTCGTGTCCA
P.LJM011	TATTAGCTCAGO	STGGTTAGA	GCGCACCCC	IGATAAGGGTO	GAGGTCACTG	GTTCGTGTCCA
P.LJM016	TATTAGCTCAGO	STGGTTAGA	GCGCACCCC	rgataagggte	GAGGTCACTG	GTTCGTGTCCA
P.LJM019	TATTAGCTCAGO	STGGTTAGA	GCGCACCCC'	TGATAAGGGTO	GAGGTCTCTG	GTTCGTGTCCA
L.LJM012	TATTAGCTCAGO	STGGTTAGA	GCGCACCCC	IGATAAGGGT	SAGGTCCCTG	GTTCGAGTCCA
P.LJM017	TATTAGCTCAGO	STGGTTAGA	GCGCACCCC'	TGATAAGGGT	GAGGTCCCTG	GTTCAAATCCA
O.animalis	TATTAGCTCAG	STGGTTAGA	GCGCACCCC	TGATAAGGGT(GAGGTCTCTG	GTTCAAATCCA
T.NIBB1067	TATTAGCTCAG	STGGTTAGA	GCGCACCCC	TGATAAGGGT	GAGGTCCCTG	GTTCAAGTCCA
T.paleo	TATTAGCTCAG	GTGGTTAGA	GCGCACCCC	TGATAAGGGT	GAGGTCCCTG	GTTCAAGTCCA
Spir.PCC6313	TATTAGCTCAG	GTGGTTAGA	GCGCACCCC	TGATAAGGGT	GAGGTCCCTG	GTTCAAGTCCA
Arth.PCC7345	TATTAGCTCAG	GTGGTTAGA	GCGCACCCC	TGATAAGGGT	GAGGTCCCTG	GTTCAAGTCCA
A.D0752 *	TATTAGCTCAG	GTGGTTAGA	GCGCACCCC	TGATAAGGGT	GAGGTCCCTG	GTTCGAGTCCA
A.D0746 *	TATTAGCTCAG	GTGGTTAGA	AGCGCACCCC	TGATAAGGGT	GAGGTCCCTG	GTTCGAGTCCA
A.D0697 *	TATTAGCTCAG	GTGGTTAGA	AGCGCACCCC	TGATAAGGGT	GAGGTCCCTG	GTTCGAGTCCA
A.cylindrica *	TATTAGCTCAG	GTGGTTAG <i>I</i>	AGCGCACCCC	TGATAAGGGT	GAGGTCCCTC	GTTCGAGTCCA
A.PCC7118 *	TATTAGCTCAG	GTGGTTAG <i>I</i>	AGCGCACCCC	TGATAAGGGT	GAGGTCCCTC	GTTCGAGTCCA
A.PCC7120 *	TATTAGCTCAG	GTGGTTAGZ	AGCGCACCCC	TGATAAGGGT	GAGGTCCCTC	GTTCGAGTCCA
C.fritschii *	TATTAGCTCAG	GTGGTTAG	AGCGCACCCC	TGATAAGGGT	GAGGTCCCTC	GTTCGAGTCCA
F.musicola *	TATTAGCTCAG	GTGGTTAG/	AGCGCACCCC	TGATAAGGGT	GAGGTCCCT	GTTCGAGTCCA
G.sp *	TATTAGCTCAG	GTGGTTAG	AGCGCACCCC	TGATAAGGGT	GAGGTCCCTC	GTTCGAGTCCA
N.ellipsospora [,]	* TATTAGCTCAG	GTGGTTAG/	AGCGCACCCC	TGATAAGGGT	GAGGTCCCT	GTTCGAGTCCA
N.MAC *	TATTAGCTCAG	GTGGTTAG	AGCGCACCC	CTGATAAGGGT	GAGGTCCCT	GTTCGAGTCCA
N.muscorum *	TATTAGCTCAG	GTGGTTAG	AGCGCACCCC	TGATAAGGGT	GAGGTCCCT	GTTCGAGTCCA
N.D0734 *	TATTAGCTCAG	GTGGTTAG	AGCGCACCC	TGATAAGGGT	GAGGTCCCT	GTTCGAGTCCA
C.c.chloroplast	TATTAGCTCAG	ATGGTTAG	AGCGCACCCC	TGATAAGGGT	GAAGTCCCT	ADDITCAATTCCA
C.p.cyanelle	TATTAGCTCAG	TTGGTTAG	AGCGCACCC	TGATAAGGGT	GAGGCCCCT	GTTCGAATCCA

70	80	90	100	110	120
		1		l l	I
CTCAGGCCTACC	-AAATTT	-GCACGGCAAA	T		
GGATGGCCCACCT	GCACAG	-GTGGCAAAAA	CA	AGAGAI	AGCGAGG
GGGTGGCCCA		GGT-AAA	T	AAA	
GAATGGCCCACCT	AACCAA	-AAAAGCCAAA	GA	GAAGG	ATAAACC
GGATGGCCCACCT	CAAA	GAACTGAGA	A	T'	TTCCAGT
GGATGGCCCACCT	AGCTTT			TTCATGAG	AGTGAAG
GGATGGCCCACCT	AGCTTT			TTCATGAG	AGTGAAG
GGATGGCCCATT-					
GGATGGCCCATT-					
GGATGGCCCATTT					
GGATGGCCCATTT					
GGATGGCCCATTT					
GTATGGCCCACT-	TGAG	AAGCTAGC	AA		С
GTATGGCCCACT-	TGAG	AAGCTAAC	AA		C
GTATGGCCCACT-	TGAG	AAGCTAGA	AA		C
GTATGGCCCACT-	TGAG	AAGCTAAA	AG		C
GTATGGCCCACT-	TGAG	AAGCTAGA	AA		C
GAATGGCCCACT-	-TGAG				T
GGATGGCCCACCT	TCAA	-AACGCTCAA	GAG	AGAGA	ATACCTGT
GGATGGCCCACCT	GGAAA	-AAAGTGATG	AAGTCAG	FAATGAGTAAT	GGATAAT
GAATGGCCCACCT	TAATTAG	AAGTTAA	AAA	TCTCTAAT	TAGAAAT
GGATAGCCCACCT	TAAG				
GGATGGCCCACC-	TGAACC	CCAAAGCAA	AAA	ACGCTTT	CCCTGCC
GGATGGCCCACA	rTTAACCGAG	CAGAACCAGA	AAG	ACAG	PAGCACCT
GGATGGCCCACAT	CA				C
GGATGGCCCACC-	-TGATAA	GTAATTAGT	AA	TTACTAG	TCGTAAT
GGATGGCCCACC-	-TGAA	GAATTTTA	GA		TTTTGGAT
GGATGGCCCACG	GCCCA	CAATTTTG	GA	TTTTAGA	TTTTAGAT
GGATGGCCCACC-	-TGAA	-GCAAGTCAA	AA	GTTAGAA	GTCAAAAG
GGATGGCCCACC	-TGAA	GCAAGTCAA	AA	GTTAAAA	GACAAAAG
GGATGGCCCACC		GCAAGTCAA	AA	GTTAAAA	GACAAAAG
GGATGGCCCACT					
GGATGGCCCACC	-7622	GCAATTAGA	AA		тттасаат
GGATGGCCCACC			AA	GAGCAAA	GTAAAAAG
GGATGGCCCACC	-TGAA	GCAAGTCAA	AA	GTTAAAA	GACAAAAG
GGATGGCCCCACC	-TCAAC		AA	TGCAAAA	ТТАААААТ
GGATGGCCCACC	-TCAA		 AAA	TTAAAAA	ТТСААААТ
CONTROCCORCO			NAA	~~~~~~~~~~~~~~~~~~~~~~~~~~~~~~~~~~~~~~	ТАСААААТ
CCATAGOCCCACC					
CCATAGUCUAAA					
GONIGOCCCHOC	,				

130	140	150	160	170	180
1	 		1	1	1
TTGAAG		AGGTTTT	TAACTACAT	GTTA	
		AATCAG	GCACCTTAT	CTTACTG	
		TT(GTTCCCGGC1	AGGA	
		CTTCAG	GCATCCTGT	CTGA	
TCGAGATT-CAAA		AAAATT(GAAACAAGC!	AATTGGT	
AAGTGATG					
AAGTGATG					
CGGTGT					
CGTGTT					
TAGAAGCCAAAA		GCTGAAAG	TAATAGTTA	ACTC	
TAGAAGCTAAAA		GCTGAAAG	TGATAGTTA	ACTC	
TGAAGGCGACGA		GCTAAAAG	TGATAGTTA	ACTC	
TAGAGGCAAAAAA-		GCCAAAAG	CAATAGTTA	ACTC	
TGAAGGCGACGA		GCTAAAAG	TGATAGTTA	ACTC	
TAGAGGCACCGAA-		AGGGATAG	CTAAGAAAA	ACTC	
-TGAGGTA		G-CTTT	GT-TCTAAA	AACAGACTC	
CAAGGATTAATTCA	GGATTAAT	TCAGGACTCA	TACAATCAA	AATCAGAACTTA	CAAG
CCTAAATTTTTAAT		TTGTAATTTT	TAATTTGTA	ATTTCCCAAAGC	
-AAGGGTA					
CAACAATTGAAGTG		CGGGTGA	CAAGAAAGC	GTCGATTGCTGA	ACG-AAG
ATGAGGTAACAGC-		АААСТА	GTGTCAAGA	CATAAGTTTTCA	CTGTGAA
CCCAAACT					
TCGTAATGAAAAA-	AT	TAGGAATTAG	GAATTAAGA	ATTAAGAATTAI	'- - -TTT
TTTAGATTGTAT	-TTTTAAA	TCTAAAATCG	CAAATCCAA	ААТССААААТТС	AT
TTTAGATTAAAA	-АААТАА-	TCCAAAATCC	GAAATCCAA	AATCCAAAATTG	TG
TCAAAAGTAATCAT	T	TTTGAATTTT	GAGTTTTGA	ATTTTGAATT	GTT
TCAAAAGTAGTT	-ATTTACT	TTTGAATTTT	GGATTTTGA	ATTTTGAATTGI	AT
TCAAAAGTAGTT	-ATTTACT	TTTGAATTTT	GGATTTTGA	ATTTTGAATTG1	"AT
CG					
TGAGAGTTAGACAT	-TCTTAAT	TTTTAATTAC	TAA-TTTGA	ATTTTTAATT-C	:- -TTT
TAAAAAGAAAGTGT	-AAATTCT	TTTGAATTTT	TCCTTTAAA	CTTTTACCTT	GTG
TCAAAAGTAGTT	-ATTTACT	TTTGAATTTT	GGATTTTGA	ATTTTGAATTGI	AT
GAAGAGTTTCT	TGGAT	TTTGAATTTT	TAATTTCTA	ATTTTTAATTG-	AT
TCAAAATTAAGA	-AATTGTT	TTTGAATTTT	GCATTTTGA	ATTTTGAATTG	'GT
TCAAAATGGTTT	-ATTTAAT	TTTGAATTTT	GAATTTTTA	ATTTTGAATTGI	:GT
-AATAGTA					

190	200	210	220	230	240
	1	1			1
TGGGGCTATAG	CTCAGCTGGGA	GAGCGCCTG	CTTTGCA	CGCAGGAGGT	CTGCGGTTCG
			AC	ATAGTAAGAG	AGA
			AA	AGAATGGGGG	GAAT
			TG	AGAGTCAGAG	GGA
CAAGTCAAA	TCCAG	TGGGATG	GTGATAGA	CTAGTCAAGT	TGAG
TGGGGGTTTAG	CTCAGTTGGTA	GAGCGCCTO	CTTTGC	AGCAGGATGT	CAGCGGTTCG
TGGGGGTTTAG	CTCAGTTGGTA	GAGCGCCTC	SCTTTGC/	AGCAGGATG	CAGCGGTTCG
TGGGGGTTTAG	CTCAGTTGGTA	GAGCGCCTC	SCTTTGC/	AGCAGGATGI	CAGGAGTTCG
TGGGGGTTTAG	CTCAGTTGGTA	GAGCGCCTC	GCTTTGCI	AGCAGGATG	CAGCGGTTCG
TGGGGGTTTAG	CTCAGTTGGTA	GAGCGCCTC	CTTTGC#	AGCAGGATGI	AAGCGGTTCG
TGGGGGTTTAG	CTCAGTTGGTA	GAGCGCCTC	GCTTTGC	AGCAGGATGI	AAGCGGTTGG
TGGGGGTTTAG	CTCAGTTGGTA	AGAGCGCCTC	CTTTGC#	AGCAGGATGI	AAGCGGTTCG
TGGGGATATAG	CTCAGTTGGTA	AGAGCGCCTC	GCTTTGC	LCGCAGGAAGI	CAGGAGTTCG
TGGGGATATAG	CTCAGTTGGTA	AGAGCGCCTC	GCTTTGC/	ACGCAGGAAGI	CAGGAGTTCG
TGGGGATATAG	CTCAGTTGGTA	AGAGCGCCTC	GCTTTGCA	CGCAGGAAG	CAGGAGTTCG
TGGGGATATAG	CTCAGTTGGT	AGAGCGCCTC	GCTTTGC?	ACGCAGGAAGI	CAGGAGTTCG
TGGGGATATAG	CTCAGTTGGT	AGAGCGCCTC	GCTTTGC	ACGCAGGAAGT	CAGGAGTTCG
TGGGGATATAG	CTCAGTTGGT	AGAGCGCCTC	SCTTTGC/	ACGCAGGAAGT	CAGGAGTTCG
-CTGCATCAGGAA	GTCAAAA	GTTTTC	SCCTTCAP	AGAAGAAAA	ACAAC
TGTGGGGGGTATAG	CTCAGTTGGT/	AGAGCGCCTC	SCCTTGC	ACGCAGGAAGI	TAGCGGTTCG
-GTGGGGGTATAG	CTCAGTTGGT/	AGAGCGCTGC	CTTTGC	AGGCAGATG	CAGCGGTTCG
AGGGGGTATAG	CTCAGTTGGT!	AGAGCGCTGC	CTTTGC	AGGCAGAAG	CAGCGGTTCG
AGCGGGGGGTATAG	CTCAGTTGGT!	AGAGCGCCTC	GCTTTGC	AGCAGGATG	CAGCGGTTCG
GCTGCATGATATC	TCTAAGAGTG/	ATCAGAACO	GCTGTCG	AGAAAAAAGI	raagg
GGGGGTATAG	CTCAGTTGGT	AGAGCGCTG	CCTTTGC	ACGGCAGAAGI	FCAGCGGTTCG
GATGGGGGTTTAG	CTCAGTTGGT/	AGAGCGCCTC	GCTTTGC!	AGCAGGATG	CAGCGGTTCG
GATGGGGGTTTAG	CTCAGTTGGT/	AGAGCGCCTC	GCTTTGC	AGCAGGATG	CAGCGGTTCG
GATGGGGGTTTAG	CTCAGTTGGT	AGAGCGCCTC	GCTTTGC?	AGCAGGATG	CAGCGGTTCG
TTTGGGGGGTTTAG	CTCAGTTGGT/	AGAGCGCCTC	GCTTTGC2	AGCAGGATG	rcagcggtt-g
-TTGGGGGTTTAG	CTCAGTTGGT	AGAGCGCCT	GCTTTGC/	AGCAGGATG	FCAGCGGTTCG
-TTGGGGGTTTAG	CTCAGTTGGT/	AGAGCGCCTC	GCTTTGC/	AAGCAGGATG	PCAGCGGTTCG
TGGGGGTATAG	CTCAGTTGGT!	AGAGCGCCTC	GCTTTGCI	AGCAGGATG	FCAGGAGTTCG
AGCGGGGGTTTAG	CTCAGTTGGT/	AGAGCGCCT	GCTTTGCI	AGCAGGATG	FCAGGAGTT-G
GATGGGGGTTTAG	CTCAGTTGGT	AGAGCGCCTC	GCTTTGC/	AAGCAGGATG	FCAGCGGTTCG
-TTGGGGGTTTAG	CTCAGTTGGT/	AGAGCGCCT	GCTTTGC	AAGCAGGATG	PCAGCGGTTCG
TGGGGGGGTTTAG	CTCAGTTGGT	AGAGCGCCTC	GCTTTGC	AAGCAGGATG	FCAGCGGTTCG
TGGGGGGGGTTTAG	CTCAGTTGGT/	AGAGCGCCT	GCTTTGC/	AAGCAGGATG	ICAGCGGTTCG
-GGGGGGGGTTTAG	CTCAGTTGGT/	AGAGCGCCTO	GCTTTGCI	AAGCAGGATG	rcageggtteg
GGGGGGTATAG	CTCAGCTGGT	AGAGCGCTG	CCTTTGCI	AAGGCAGATG	FCAGCGGTTCG
AGGGGGTATAG	CTCAGTTGGT/	AGAGCGCTG	CCTTTGC	AAGGCAGATG	FCAGCGGTTCG

-

2	250	260 I	270	280	290	300
ATCCCGCF	I ATAGCTCC	I ACCATCTC	ا CTGTAG	 TGGT		ן TT
AGTCCGC	FAACCTCC	ACCAAA	AGACTGC	TT		A
						A
AGTUTUU		ACTGA-CC	CAACIGIAI			A
AGTCCGI.		ACIGA-CC	AACCCGC(CGGAICGCG-		A
AGTUUGU:	TAACTICC	ATCCAGCI		CACAACATG-		
AGTUUGU:		ATCAAGCI		CACAACATG-		
AGTUUGU		ATCAAGCI		CACAACATG-		14
AATUICC	TIAICIUC		IACTAALA	10AUUAAU		AA
AATUTCU	TATUTUU		TACTAACA	HGALLAAL		GA
AATCTCC	TTATCTCC		ractaaca	AGACCAAC		GA
AATCTCC	TTATCTCC	ACCAAAC		AGACCAAC		AA
AATUTUU	TTATCTCC	ACCAAAC	ractaaca	AGACCAAC		GA
AATCTCC				CA		
AGTCCGC	TTACCTCC	CACTCTC-A	ACTAAC	-TATTACCAC-		CGCC
AGTCCGC	TTACCTCC	CACTCTTA	CCCAACCAGC	CGAAAACAAAC	AGAAACA	ААА
A-TCCGC	TTACC-CO	CAGGAATA	AAAAAGGTA-			G
AGTCCGC	TTACCTCO	CACCA				
AGTCCGC	TTACCTC	CACTCTCC	TTTG	TGATGGTG		
AGTCCGC	TAACCTC	CAC				
AGTCCGC	TAACCTC	CACCTG	AAGCGG	TAAGC		GAAC
AGTCCGC	TAACCTC	CAC	A	TTGGC		TTAC
AGTCCGC	TAACCTC	CACATTG-				
AGTCCGC	TAACCTC	CACATGA-	AAAGG	CAAGC		С
AGTCCGC	CTAACCTC	CACATGA-	AAAGG	CAAGC		C
	᠂ᡎᡎ᠗᠅ᡣᡎᢕ	ГАССТССА	TTTGCCAGCA	AAAGAAATAG		A
AGTCTCC	LIACCIC	0110010011	1110001001			
AGTCTCC AGTCTCC	TAACCTC	CACCTGT-				AT
AGTCTCC AGTCTCC AGTCCGC	TAACCTC TAACCTC	CACCTGT- CACATGAT	ATCTGTT	GTGACAGCAG	 CAAAAGTGAAT	АТ АТАТСАС
AGTCTCC AGTCTCC AGTCCGC AGTCCGC	CTAACCTC CTAACCTC CTAACCTC	CACCTGT- CACATGAT CACATGA-	ATCTGTT	GTGACAGCAG	CAAAAGTGAAT	AT ATATCAC C
AGTCTCC AGTCTCC AGTCCGC AGTCCGC AGTCCGC	CTAACCTC CTAACCTC CTAACCTC CTAACCTC	CACCTGT- CACATGAT CACATGA- CACCTGT-	 ATCTGTT AAAGG ATGTAG	GTGACAGCAGC CAAGC	 CAAAAGTGAAT 	AT ATATCAC C TAA
AGTCTCC AGTCTCC AGTCCGC AGTCCGC AGTCCGC	CTAACCTC CTAACCTC CTAACCTC CTAACCTC CTAACCTC	CACCTGT- CACATGAT CACATGA- CACCTGT- CACCT	ATCTGTT AAAGG ATGTAG	GTGACAGCAG CAAGC -CGCCAGC -CGTCAGC	CAAAAGTGAAT	АТ АТАТСАС С ТАА ТАТ
AGTCTCC AGTCTCC AGTCCGC AGTCCGC AGTCCGC AGTCCGC	CTAACCTC CTAACCTC CTAACCTC CTAACCTC CTAACCTC CTAACCTC	CACCTGT- CACATGAT CACATGA- CACCTGT- CACCT CACCT CACATGA-	ATCTGTT AAAGG ATGTAG -TGTAG AAAGAAG	GTGACAGCAG CAAGC -CGCCAGC -CGTCAGC CTTCAAACAG	CAAAAGTGAAT	AT ATATCAC C TAA TAT CTTC
AGTCTCC AGTCTCC AGTCCGC AGTCCGC AGTCCGC AGTCCGC AGTCCGC	CTAACCTC CTAACCTC CTAACCTC CTAACCTC CTAACCTC CTAACCTC CTAACCTC CTAACCTC	CACCTGT- CACATGAT CACATGA- CACCTGT- CACCTGT CACCTGA- CACATGA- CACAAAGT	ATCTGTT AAAGG ATGTAG -TGTAG AAAGAAG TAAGTTGTA	GTGACAGCAGC CAAGC -CGCCAGC -CGTCAGC -CTTCAAACAG	CAAAAGTGAAT	AT ATATCAC C TAA TAT CTTC

I

31	.0 	320 	330	340	350 	360
АААТАА					۱ 	ا
AAATTCAAA	A		·			A-
AAATTCAAA	A					A-
AGATGTTGG	GA					G-
AAGCAGAGG	<u>G</u>					A-
AGCTGCATC	A					·
AGCTGCATC	2 4					·
AGCTGCATC	<u>A</u>					·{d-
AAGCAAAA-						· -
AAGCAAA						
AAGCAAAA-						
AAGCAAGA-						
AAGCAAAA-						
-GACAAAA-						
ААААААААСС			A			-AAAAA-
GACCAAAAA	C					A-
AAATCAGCA	A					C-
-AAGAG						
			A			
CTAGTTO	G					G-
						ATG-
AAACAAA						AGG-
T						
AAGAAAG						AGG-
GAATCACCA	A					
AGGTAA						DTG-
AGGTAAAAA	TTAGCCAA	ААСТСАТАСА	GTGAGTAGTT	᠃ͲͲͲႺͲႺͲͳ	ATAGCAGA	GAATGT
AAGAAAG						AGG-
GAGAATA						TGG-
AAATAGA	--					GGG-
ААААААА						GAG-

370	380	390	400	410	420
			1	- i	
AAAAT-					ACTTC
TAAAAT-					TCAGC ·
ATCAGTTC	:A			- -GC-AT	CTTAAGT
ATCAGTTC	:A			GC-AT	CTTAAGT
TTGAGCTC	GATTGGAGT	GTGATTTAGA'	IGTGTCCGCT	GGAAGA	CCCCAGC
GAGAGCTT	GTGGTGG	TGATTTTAGA	TGATTGTACT	<u>GAGAAAAGGA</u>	CTCTAGC
GAGGAATT	'CTGATGG	GTGAAGTGAC	CGGACATGTT	CAGCTGAGGA	DDADDTD
GAGGAATT	'CTGATGG	GTGAAGTGAC	CGGACATGTT	CAGCTGAGGA	CTCCAGQ
	CTGATGG	GTGAAGTGAC	CGGACATGTT	CAGCTGAGGA	CTCCAGO
GCAAGAG					TTTAGC
GCAAGAG					TTTAGC
GCGAGAG-					TTTAGC
GCAAGAG					TTTAGC
GCAAGAG					TTTAGC
					TTTAGO
TCAG-T-					TCAGC
AATAGAAA-					TCAGC
	AGTCAAAAA	ACTGTT	-GC	GC	TTTTAGC
AAGACCT-				GP	ATAAAGC
AAAAT-					TCAGC
TAAGAT-					TCAGC
GTGAGAT(GAGATGA			G-	GTTA-C
GAAAAGAC					ACAGC
AAAAAG					TCAGC
					TCAGC
GAAAAGAG					TCAGC
AAAAAT					-~-TCAGC
AAAAAT					-~-TCAGC
AGCATCTA	AGGTTTGTCA	ATGAGCAGAC	CT	AGi	ATTGGGAT
GGTAAAAT					CTAGC
GGTAAGATAAGAG	ATT '			GTGGGTA	AATTCAGC
AAAAAT					TCAGC
AGTAAGAT					TCAGC
ATTAA-AT					TCAGC
AAAAAAA					TCAGC

	430	440	450	460	470	480
		1	1			
AGA			GTGTAC			
20000						
ACIGI-						
			GATTTTC	GTAIC		
TICCIG			AGTAAAAACT			CAMACCC
	ICALLII IIIIII IIIIIIIIIIIIIIIIIIIIIIIII			TGAIGCGAGI		
CICCII			ATTCATCGTA	TCGGATGGG1	TTCAT	momen
		JUGI IGGATG	MATCIACITIA	AACGGTGAGA	.111A1	ICIGA
UTUUTI	TCATTCC	UAAAAGAATG	SATTCATTGTA	TAAGATGAGT	TICAT	TITGI
AACTCA	TC	†TA	STAGGTAA	-AACCTA		
AACTCA	TO	†TA0	GTCAGGTAA	-AACCTA		
AACTCA	\TC	+TA	STAAGGAAA	-ATCCTA		
AACTCA	\TC	†TA	GTICATGTCACA	TAAGACA		
AACTC	\TC	†TA	GT¦GAGGAAA	-ATCCTA		
AACTCA	AIC	tIG	GECA			
AACTT-			CTAACGA			
ACCCA	GAC	ТА	ACTTGGTCA		-	
ААТА	T G	IGATAGACTA	GAAAGAGTGAC	TATAAAA		
200						
ACC			GGCTGT			
ACTGC-			TCGAAAG			
0	TC	TGATAGAT~-	AGATAATI	TATCACT		
AAC			TG-GGA			
AACTA			ATAGAGTAA			
ACCTG			AAACTGTGA			
AAC			TGACG			
AAC			TAACG			
AAC			TAACG			
TGGCT	CTTGTGC	TGACAAAT	-GTGGATGTG#	ATAAGATA		
ACC			TG-CGT			
AAT			TGACGT			
AAC			TAACG			
AACA-			TGACAT			
AACA-			TGGC			
AAC			TAATG			

490	500	510	520	530	540
 רדקכאא-	 	 	۔ا	<u> </u>	ا
TGGAGA	\				
GGP	AGGTTAGG				
GGP	AGGTTAGG				
TTTGGTTGATAAGA	ATGCTGGGCT	CAAATGAG	TTG	A'	TCAAGC-
CTTGATTGAAAAGA	ACGCTGGACT	CCAGCAAG		A	TCAGCT-
GATGGTTGAAAGG	AGCTGGACT	CGCTGATCAT	CATTTTGCTT	GCTCATGAGA	TCATGTT
GAAGGTTGAAAGG	AGCTGGACT	CGCTGATCAC	CCATTT		CCATGTT
GAAGGTTGAAAGG	AAGCTGGATT	CGCTGATCAT	CATTTTGCTT	TCTCATGAGA	TCATGTT
AACAAC	ГАG				
AACAAC	ГАG				
AGAGAC	ľAG				
AAAAAC'	ГАG				
AGAGAC	TAG				
AAGAC	TAG				
TTCAG	A				
ACAAAG	AAG				
GCTCAG	CACCTAGAT-	-TCAGCTAC	PAGCT		
AAAGC	AAGTCATGCT	·			
CAGCAG					
	ርሞርርሞልልልሞር	יידידיים	PACTCT		
	AAC				
	ית				
BACICG	ጠ				
	TA				
GP					
GGAG	,				
CDDDDDDDDDDDDDDDDDDDDDD					
GAAAATTGTGAGT	AATTTCAGCA	ААСТААТТАА	T		
AATAAA	AGAA				
CAAAA/	AGAG				
GGA0	G				
- -ATATA(GATG				
 TATA(3				
ATA(<u> </u>				

-

•77

278

550	560	570	580	590	600
			1	1	
	AGGTTCACTGCO	AAGTTTT	GCTCTTT-	AAAA	ATCTGGAT
	ATGCTGG	TTCTGA-	-GTAAAGAGI	CCAGA	GGAACCTTGA
	GAAATTAGI	T	-GTTCAGTTA	AAAAA	AGCACCTTGA
	ATGCTGG	T	-GTAAGTO	CAGTA	AGGACCTTGA
	-CAACTGCTGG	AATTC	TCCAGCG-		AGCACCTTGA
A-ATGC	TGAGTTGAAAGA	ACTCAGC-	A-		AGAACCTTGA
A-ATGC	TGAGTTGAAAG	ACTCAGC-	A-		AGAACCTTGA
AATTGA	TCAATTGATTTC	GAAGTTCT	-AGC-		AGAACCTTGA
GAAAGG	CTGATGGTTGC	GTGTTC-	-AGT-	- 	AGAACCTTGA
CAAGGGAAATGA	TGCATCGGTGT	AGCCAGC-	AA-		AGAACCTTGA
AATCGGAGTGGG	TGAGGCAGTGT	AGCCAGC-	AA-		AGAACCTTGA
CAAGTGAAATGA	TGAATCAGTGC	AGCCAGC-	AA		AGAACCTTGA
Æ	GAGACTGCTAA	ATTCTAGC	GAAAGQGA-(GAATTTAGAA	AGAACCTTGA
P	GAGACTGCTAA	ATTCTAGC	GAAAGOGA-(GAATTTAGAA	AGAACCTTGA
P	GAGACTGCTAA	ATTCTAGC	GAAAGCAA-(GAATTTAGAA	AGAACCTTGA
P	AGAGACTGCTAA	ATTCTAGC	GAGAG¢GA-(GAATTTAGAA	AGAACCTTGA
ip	GAGACTGCTAA	ATTCTAGC	GAAAGCAA-(GAATTTAGAA	AGAACCTTGA
II	GAGACTGCTAA	ATTCTAGC	GAAAGCGAA	GAATTTAGAA	AGAACCTTGA
	AGCCTGCTGA	ACTAAT	-GTTCAGCC		AGAACCTTGA
]	CGGAATGCTGG	AGTCTAGA	ACTCTAGTC		AGAACCTTGA
AA7	CGGAATGCTGG	-CTTCCAA	-G-CCAGCC		AGAACCTTGA
(SCTGGTATTCTT	AAATACC-	AGTA		-AGAACCTTGA
	-CAGACTGCTGG	AGAAATTC	TCCAGTC	AA/	AGAACCTTGA
	TAAATGC-GG	ATTTCAA-	-GTCCAGEC	C	-AGAACCTTGA
GAGAT	TTGGATAGCTGG.	ACATCTGT	-T-CCAGTC		-AGAACCTTGA
(CTAGACTGCTGG	GTGAGAT-	TCCAGCC		-AGAACCTTGA
	-TAGACTGCTGG	GTAATA	CCTAGCC		-AGAACCTTGA
	-CAGAATGCTGG	ATGAAAG-	TCCAGTC		-AGAACCTTGA
?	ICAGACTGCTGG	АТ-ТААА-	ACCAGCC	_	-AGAACCTTGA
?	FTAGACTGCTGG	GTAGTA	CCTAGCC		-AGAACCTTGA
!	FTAGACTGCTGG	GTAGTA	CCTAGCC		-AGAACCTTGA
AGTAG	TTAGACTGCTGG	GATGATTC	C-CAGCC		-AGAACCTTGA
?	FCAGGCTGCTGG	GTGAGAG-	TCCAGCC		-AGAACCTTGA
!	FCAAACTGCTGG	AA-ATAG-	TCCAGCC		-AGAACCTTGA
!	FTAGACTGCTGG	GTAGTF	ACCTAGCC		-AGAACCTTGA
!	FCAGACTGCTGG	GTGATTAA	ATCCAGCC		-AGAACCTTGA
(CCAGACTGCTGG	ATAGTTA-	TCCAGCC		-AGAACCTTGA
	TTAGACTGCTGA	GGAAP	ACTCAGCC		-AGAACCTTGA
	-СТТСТСАААТА	AAATTTT-			
	-TATAGCAA				

610	620	630	640	650	660
1	<u> </u>	i i	<u> </u>		Ī
CAAGCTGAAAATT	GAAACACTGAA		GAAAGTT	GTTCGTGAGT	
AAACTGCATAGA	GCTAGGTG		GCCAA	-AAAAGA	
AAACTGCATGAA	GAATAG	AGAAA	GCAAA	-GAAAGT	
AAACTGCATAAA	GAAAAG	AGAAA	GCAGG	-GAAAATCTTG	
AAACTGCATAGA	GATAAGAG	AAATA	GTCAGGT	AGAAAACC	
AAACTGCATAGA	GATGAGAGT	GTAG	GTATCAC	AGACACCATC	
AAACTGCATAGA	GATGAGAGT	GTAG	GTATCAC	AGACACCATC	
CAACTGCATAGG	TAAGTCTGGAAA	GAATAAA	-GCATCTC	ATGGATGATGC	
CAACTGCATAGG	TAAGTCTGTA	-AATAA	-GCATCTC	ATGGATGCTTCA-	
CAACTGCATAGG	TGAGTCTGGAA-	-AAGAAA	-GCATCTO	ATGGATGCCCAG-	
CAACTGCATAGG	TGAGTCTGGAA-	-AAGAAA	-GCATCTO	CATGGATGCCCAG-	
AAACTGCATAGG	TGAGTCTGGAA-	-AAGAAA	-GCATCTO	CATGGATGCCCAG-	
AAACTGCATAGT	ААТАСТАА				
AAACTGCATAGT	AATAGTAA				
AAACTGCATAGT	'AATAGTAA				
AAACTGCATAGT	'AATAGTAA				
AAACTGCATAGT	'AATAGTAA				
AAACTGCATAGI	AATAGTAA				
AAACTGCATAGO	CAACAAAAG	CCAAA	-GCAAG-		
AAACTGCATAGA	AATTAAGTCA		-GCGTGT	CAGTCATC	
AAACTGCATAGA	AATAAG-CA	CAAAAA		GAGAGACATTTA-	
AAACTGCATAG	rcaa		GTCA		
AAACTG-ATAG	rcataa-gtcag	GCAA	-GAAGGA	GATTAGTAAAT	
AAACTGCATAG	rcaagaatc-	AAAAAGI	'G	TAGAGAGTC	
AAACTGCATAG	AGAAAAG	САТААЗ	GGTG	TT	
AAACTGCATAG	AAACGCGATA		-GCA-GG	CAGTCGAAAAA	GTCAA
AAACTGCATAG	AAACGCGATTA-	GAT-A	-GCA-GG	CAGACACAG	
AAACTGCATAG	TAACGCGA	G	GCA-GI	TCAGTAATC	
AAACTGCATAG	TAACGCGAAT	TA	GCA-GG	CAGACGAAACT	-GAGTACT
AAACTGCATAA	AAACGCGATTA-	GATTA	GCA-GG	GCAGACACA	
AAACTGCATAA	AAACGCGATTA-	GATTA	GCA-GG	GCAGACACA	
AAACTGCATAG	AGAAGCGAA-	AAGTAG	GTAGTO	AGTGAACAGT	
AAACTGCATAG	AGAAGCGAAT		GTATGO	STAGTTCA	GTCAA
AAACTGCATAG	TAACGCGAAAA-		GCA-GO	GCAGACGAAATTT(CAAGTAAT
AAACTGCATAA	AAACGCGATTA-	GATTA-	GCG-G(GCAGACACA	
AAACTGCATAG	TAACGCGAAAA-	TA	GCA-G(GCAGACACAG	
AAACTGCATAG	GAATGCGATTT-	AA	GCA-G	GCAGACACAG	
AAACTGCATAG	AAACGCGATTA-	GA	GCA-G	GCAGACACAA	

670 	680 	690 CTCTCAAATT1	700 !T	710 	720
C-T	·	CTTCCTAACCO		GATGTTA	
TCA ACG ACG ACG	GCAGT GACGT GACGT GTCGT	GAT-GATTCT CGAT-GATTCT CGAGGGGATCCT CGACAGATCCT CGACGGATCCT	GGTGGCTTT GGTAACCCC GGTAACCCC GGTAACCCC	TTTCTGTGA GGCTCCTGGGA GGGTGAAGGAT AGGTGAAGGAG AGGTGAAGGAG	TC TCAACG GCGGC- ACGGC- CCGGC-
				ATCGGAG	
		-CTAATCAACT	 FCACA	GACAAAA	C
AAAAAAGACGAAG	AGACTGCAAG	GTTGGAAAACC -ACATGTAATA -AAAAGCGAAA GGTGAAGAACT	CATTGCCGA	GATGAAA GATGGA- GCAGCA- GTAGGAAAGAAA	AAGCA
AAGTGAGCAG GAGTGCTGAGTGC	TGAACAGTA	ATCAAAAGAAA AGTGTAAAACA	CAAAGGCTC	GACAAAC GACAAAC ACTGAGA CCATGAAAGTAG	ACTG
		-ACATTCAAGT -ACATTGGA -GTATTGGAAA		GACAAA(AATGAAT GATATA4	} !

•

730	740	750	760	770	780
1		1	1		
	CGCAACT-CT	GAAGTGAAAC	ATCTTCGG	·	
		AGACCGCAAG	G- 		
	AACTTTTAAG	TAATTAAGAG	C		
ACCACAA	GGCTTTTGGC	TGAGTGTGAG	ACTGAAGG		
A	AGTAT	TTGCGAA	ATAACT		
TTC	CATCATCACTT	GATTGAG	TCAAG		
TTC	CATCATCACTI	GATTGAG	TCAAG		
CGF	AAGGATGATG	GAGAA-GGAG	AGTTGC	AAATTCTATC	GAGCAAG
TTTGTTTTTGC	GAGCGATTCTA	GAGTA-GGAG	CCGAGACTCC	ACATTGCTCC'	TGGTTCT
TGACTGO	GGA-GATTCTT	GAGTTTAGAG	GCCGAGAGCTT	CCAGCGTCTT	TTCGCTC
TGACTG	GGAAGATTCTT	GAGTTTAGAG	GCCGAGAGCTT	CCAGCGTCTT	ттссстф
TGACTG	GGAAGATTCT	<u>rgagtttaga</u>	SCCGAGAGCTT	CCAGCGTCTT	TTCGCTC
T	AAGAAGCA-AG	GTAGATCCAAA	GTGGAAAC		
T	AAGAAGCA-AG	STAGATCCAA A	AGTGGAAAC		
T	AAGAAGCA-A	GTAGATCCAA	AGTGGA¦AAC−-		
T	AAGAAGCA-A	GTAGATCCAA	AGTAGAGAT		
T	AAGAAGCA-A	GTAGATCCAA	AGTGGAAAC		
'n	AAGAAGCA-A	GTAGATCCAA	AGTGGAAAC		
	GTAG	TCGAGA	GTCAACAG		
TATCT	GACCAGCG-A	CCACCATGAA	ААААТСАА		
CCA	GGAATTTATT	CTAGGTGAAA	G-CAAA		
	AAGTAAC	TGAACCAGAG	GTTAAG		
AAAGTCAAG	ACCAAGCC-A	AAAAGCAAAA	GATCTTCTA-		
	-CTGTTTAGCA	GGAATCTTAA	AACAAGTT		
P	GGAAAACGTC	GTAAA	GACAAT		
TAATTACGAATT	TAAGTTTTC-G	TAATTACGAA	T-TATCAA		
AGTG1	TTGCAGTG~G	TGAACACCAA	TGTGATA		
GATA/	ACTGAAACG-A	GAAAAACCAA	TGTAAT		
AAGCAGAAAAG	TTTGCAA~0	TCGAACAAAA	TGTATTGAA-		
TGTGI	AATGCAGGA	TA-AAACCAA	TGGAA-TT		
TGTG	AATGCAGG <i>I</i>	TA-AAACCAA	TGGAA-TT		
	-GAACAGTAA1	CAGTAAAAAA	ACTGG		
AAACAGAAGAC	TGAGTAACA-(GACACCAATGI	TGTGAGAA		
AAGAGAAAGAG	TTTGCAGAT-(GTGAAACCAAA	ATGTATTG		
TGTG	AATGCAGG1	ата-ааассая	ATGGAA-TT		
TGTG	TTTGCAGG	FTTTACACCA	ATGTAA		
- TGTG	TATGCAA	-TTGAAGCTG	AATTGAATT		
AGTG	TTTGCAGG	ATAGAAACCA	ATTGTG-AA		

790 	800 	810 	820 	830 	840
AGCCGAGACT	CCATCAC AATCGG	GTTCTTT GTTTGAT	CA CT	ATTGTGTCG TGTTTTTAG	AGCAATTGCG AGTCAATGCG
CGCCGAGGAG CGCCGAGGAG CGCCGAGGAG	GGAAGAGTCAG GGAAGAGTCAG GGAAGAGTCAG	ATTCTGCATT ATTCTGCCTT ATTCTGCCTT	ITCAAACCGCT ITCAAACCGCT ITCAAACCGCT	TGCGTTTGA) TGAGTTTGA(TGGGTTGGA)	atggggcgag Stgagacgag Aggggacaag
TAAAA					
TGA					
A					

-

283

	850 	860 	870 	880 	890 1	900 1
				GTTGT		
				GCGGTGA-		A
				TGGGAG-		
·				TGGGAG-		
AGAT	CAATTGAAGA	AATTCA	GTTGATCAG	GCGTTGAGGA-		-TAAAG
AGAC	CGCGACGGGC	AACCGCAGCG	GTGGATGGT	ITCATGAAAAG	CGAAAGCAGO	TGAAAG
AGAC	CGCGACGGGC	AACCGCAGCG	GTGAATTGT'	FGCATGAAAGG	CGAAAG¢AGG	TGAAAG
				ACA		
				ACA		
				ACA		
				ACA		
				GT		
				TCAA		
				GT		
				TC		
				GT		
				GT		
				GT		
				GT		
			CTGG1	TAACTGAAAA		
				GT		
				T		
				GT		
				GT		
				GT		
				TTA		
				T		

	910	920	930	940	950	960
	L.	1	1	1	1	1
						I ATTAGO
					TCAAG	СТААТА
						СТААТА
GGTAGG	GGGAAAG	ATGATCCTAGT	тассааааа	GAAGATCTTA	AAAAGGTCAAG	CTACAA
					GGTCAA	SCGA-AA
		АААТССАА	GTCAG	AAAGAAG	PAGTGGTCAA	CTACGA
		AAATGGAA	GTCAG	AAAGAAG'	PAGTGGTCAA	GCTACGA
GAG	AATCTGT	TT-CAACGACA	GCAAGCGTG	T-TGGAGATT	ATATGGTCAA	SCTACAA
CAGGAG	AATCTGC	GTACAACCGCA	GTAAGGTGA	TGTGGAGATA	AATTGGTCAA	GCTACAA
CAGGAG	AATGGGT	CT-CAACAACC	CTGAACGT-	TGGAAGTGTT	AATTGGTCAA	GCTACAA
CAGGAG	AATGGGT	CT-CAACAACC	CTGAACGT-	TGGGAGTGTT	AATTGGTCAA	GCTACAA
CAGGAG	AATGGGT	CT-CAACAACC	CTGAACGC-	TGGGAGTGTT	AATTGGTCAA	GCTACAA
					GAGGTCAA	GCTACAA
					GAGGTCAA	GCTACAA
					GAGGTCAA	GCTACAA
					GAGGTCAA	GCTACAA
					GAGGTCAA	GCTACAA
					GAGGTCAA	GCTACAA
					CGATCAA	GCTACAA
					GGTCAA	GTTACAA
		GCCGAC	A-CAA	ACAAAAA	AAGTGGTCAA	GCTACAA
		AATCTAG	STTCAGGTG1	AGAAAGAAAA	AAGTGGTCAA	GTTATAA
					GGTCAA	GCTAGAA
					AGGTCAA	GTTACAA
				CAATGT	AGGTCAA	GCTACAA
					GGTCAA	GCTAGAA
					GGTCAA	GCTAATA
					GGTCAA	GCTAATA
					GGTCAA	GCTAATA
					GGTCAA	GCTAATA
					GGTCAA	GCTAATA
		-CTGAACAGAC	ACCAA	AAGTTAT	TTGTGGTCAA	GCTAATA
					GGTCAA	GCTAATA
					GGTCAA	GCTAATA
					GGTCAA	GCTAATA
					GGTCAA	GCTAATA
					GGTCAA	GCTAATA
					GGTCAP	GCGAATA
		-TT		-TGATAATAAA	AAAAAGTAAA	GTAAGTA
			~	GCAAACATA	AAGTGGTCAA	GTGACCA

970	980	990	1000
1	1	1	1
AGCGTACACGGTGGA	TGCCCTGG	CAGTCAGAG	GCGATG-AAG
AGGGCTAACGGTGGA	TACCTAGG	CACACAGAG	CGGCCGCAAA
AGGGCTAACGGTGGA	TACCTAGG	CACACAGAG	CGGCCGCAAA
AGGGCTAACGGTGGA	TACCTTGG	CACACAGAG	CGGCCGCAAA
ACAGCTGACGGTGGA	TACCTAGG	CACACAGAG	CGGCCGCAAA
AGGGCTTACGGTGGA	TACCTAGG	CACACAGAG	CGGCCGCAAA
AGGGCTTACGGTGGA	TACCTAGG	CACACAGAG	CGGCCGCAAA
AGAGCTCACGGTGGA	TACCTAGG	CACACAGAG	CGGCCGCAAA
AGGGCTCACGGTGGA	TACCTAGG	CACACAGAG	CGGCCGCAAA
AGGGCTCACGGTGGA	TACCTAGG	CACACAGAG	CGGCCGCAAA
AGGGCTCACGGTGGA	TACCTAGG	CACACAGAG	CGGCCGCAAA
AGGGCTCACGGTGGA	TACCTAGG	CACACAGAG	CGGCCGCAAA
AGGGCTCACGGTGGA	TACCTAGG	CACACAGAG	CGGCCGCAAA
AGGGCTCACGGTGGA	TACCTAGG	CACACAGAG	CGGCCGCAAA
AGGGCTCACGGTGGA	TACCTAGG	CACACAGAG	CGGCCGCAAA
AGGGCTTACGGTGGA	TACCTAGG	CACACAGAG	CGGCCGCAAA
AGGGCTCACGGTGGA	TACCTAGG	CACACAGAG	CGGCCGCAAA
AGGGCTCACGGTGGA	TACCTAGG	CACACAGAG	CGGCCGCAAA
AGGGCTCACGGTGGA	TACCTAGG	CACACAGAG	CGGCCGCAAA
AGGGCTAACGGTGGA	TACCTAGG	CACACAGAG	CGGCCGCAAA
AGGGCTGACGGTGGA	TACCTAGG	CACACAGAG	CGGCCGCAAA
AGGGCTGACGGTGGA	TACCTAGG	CACACAGAG	CGGCCGCAAA
AGGCTTTACGGTGGA	TACCTAGG	CACACAGAG	CGGCCGCAAA
AGAGCTAACGGTGGA	TACCTAGG	CACACAGAG	CGGCCGCAAA
AGGGCTAACGGTGGA	TACCTAGG	CACACAGAG	CGGCCGCAAA
AGGGCTAATGGTGGA	TACCTAGG	CACACAGAG	CGGCCGCAAA
AGAGCTAATGGTGGA	TACCTAGG	CACACAGAG	CGGCCGCAAA
AGGGCTAATGGTGGA	TACCTAGG	CACACAGAG	CGGCCGCAAA
AGGGCTAATGGTGGA	TACCTAGG	CACACAGAG	CGGCCGCAAA
AGGGCTAATGGTGGA	TACCTAGG	CACACAGAG	CGGCCGCAAA
AGGGCTAATGGTGGA	TACCTAGG	CACACAGAG	CGGCCGCAAA
AGGGCTTGTGGTGGA	TACCTAGG	CACACAGAG	CGGCCGCAAA
AGGGCTTGCGGTGGA	TACCTAGG	CACACAGAG	CGGCCGCAAA
AGGGCTGATGGTGGA	TACCTAGG	CACACAGAG	CGGCCGCAAA
AGGGCTAATGGTGGA	TACCTAGG	CACACAGAG	CGGCCGCAAA
AGGGCTCATGGTGGA	TACCTAGG	CACACAGAG	CGGCCGCAAA
AGGGCTGACGGTGGA	TACCTAGG	CACACAGAG	CGGCCGCAAA
AGAGCTAATGGTGGA	TACCTAGG	CACACAGAG	CGGCCGCAAA
AGAGCTTACGGTGGA	TACCTTGG	CATTTAGAA	GCGATG-AAG
AGGGCTTACGGTGGA	TACCTAGG	CATTCAGAA	GCGATG-AAG

.

within the variable spacer between tDNA^{ala} and the 5' end of 23S rDNA among *Synechococcus* LJM002, 003 and 004, two very long conserved sequences (about 90 bp) from position 261 to 434 and from position 727 to 823, and a 42 bp sequence from position 892 to 936. Three conserved regions are within the variable spacer between tDNA^{ala} and the 5' end of 23S rDNA among *Pseudanabaena* LJM001, 005, 006, 007, 009, 011, 016 and 019, a 25 bp sequence from position 367 to 444, a 21 bp sequence from position 552 to 580 and another 25 bp sequence from position 731 to 757.

7.3 Discussion

By aligning the sequence, more conserved regions in the variable length spacer were found between the strains assessed in same species or genus. These are the best candidates for which to design cyanobacterial genera or species-specific primers or probes. This will be very useful to detect presence of particular cyanobacterial species or genus in environmental samples more rapidly, such as toxigenic species. Thus the 16S-23S rDNA spacer sequence does provide a potentially useful molecule for the identification of cyanobacteria.

The phylogenetic significance of trees using each genetic method were compared with morphological phenotypes. The morphological taxonomic groups (see Table 1.1) appear largely consistent with the phylogenetic information obtained to date. In Neighbour-joining (sequence) trees, all the cyanobacterial strains were separated into three large group, (1) Simple filamentous strains (*Oscillatoriales* or section III); 2 Unicellular strains (*Chroococcales & Pleurocapsales* or section I & II); (3 Heterocystous strains (*Nostocales & Stigonematales* or section IV & V). In Parsimony and Maximum Likelihood sequence trees and the RFLP dendrogram, the unicellular group was separated by a simple filamentous group into two clusters; unicellular rodshaped strains with division in one plane and unicellular spherical shaped strains with division in more than one plane.

These four isolates and three laboratory strains, are simple rod-shaped unicellular forms, traditionally classified within the Synechococcus. In phylogenetic studies, all these strains grouped into one large cluster with three far related sub-clusters (Fig. 7.3, 7.4, 7.5 and 7.6). This clearly indicates further generic division in this morphologically simplest genera. In contrast with current the taxonomic system, Rippka and Herdman (1992) have changed genera Synechococcus to Synechococcus group which contain five clusters, and Komarek and Anagnostidis (1986) put them into two families containing thirteen genera. The isolates, Synechococcus strains LJM002, LJM003, LJM004 and LJM020 are morphological very similar; Synechococcus strains LJM003 and LJM004 are a little more close than others (Fig. 3.22). This were confirmed by phylogenetic study (Fig. 5.16, 7.3, 7.4, 7.5 and 7.6). Marine strain Synechococcus WH7803 shows a far distance to both groups of freshwater strains, even if it is morphologically very similar to the isolates (Fig. 3.6A, 3.7, 3.8, 3.9 and 7.7). This marine strain was proved unrelated to other freshwater strains, in line with DNA-DNA hybridisation data (Wood and Townsend, 1990). Synechococcus CCAP1479/7 and WH7803 were grouped in one sub-cluster, but distant, in line with morphological data, a little similar in habitat and a little different in cell structure and pigment content. It suggests that they may be in two closer groups, than with other two. In the tree topologies, the Synechococcus group did not separate and form clusters with other simple filamentous strains. This provide a better result than the 16S rRNA-sequence data reviewed by Wilmotte (1994).

The *Pseudanabaena* cluster contain eight freshwater isolates. They are far away from other simple filamentous genera. This supports the finding of Geitler (1982) and Guglielmi and Cohen-Bazire (1984) that the genera *Pseudanabaena* is rather different from other Oscillatoriace. All eight strains form tight clusters (Fig. 3.16B, 7.3, 7.4, 7.5 and 7.6). This suggested that these strains should belong to the same genus. Four sub-clusters were drawn based on the distance: *Pseudanabaena* sp. LJM019; *Pseudanabaena* sp. LJM011; *Pseudanabaena* LJM009 and LJM016; *Pseudanabaena* LJM001, 005, 006 and 007. This suggests that these four sub-clusters probably are four species. This appear largely consistent with morphological data (chapter 3), except for strains LJM001, LJM005, LJM006 and LJM007. They were separated into two groups because strains LJM001 and LJM005 do not contain PE, and because of slight morphological differences.

The heterogeneous branch containing unicellular spherical-shaped and filamentous strains is only found in sequences trees. This branch unifies the remaining four of the set of eleven unicellular strains studied. Four strains are from four different genera based on morphological classification. It is in agreement with the tree topologies of Figures 7.5 and 7.6. The isolate, *Chroococcidiopsis* LJM013 shows in one sub-cluster without other strains, but the distance is not far. In morphological classification, *Chroococcidiopsis* is placed in order *Pleurocapsales* (section II), different from *Chroococcales* (section I). This strain also shows high morphological similarity with the other three strains, for example, repeated binary fission in three planes to produce a more or less regular cubical cell in an aggregate. In line with cluster G of the 16S rRNA tree (Wilmotte, 1994), unicellular strains with binary fission in more than one plane (in section I) and baeocyte-forming strains (section II) form one cluster. This

suggests that these groups are probably in one Section or Order. The presence of *Spirulina* in this lineage is surprising, because the morphology of this truly multicellular form is rather complex: it consists of an helical trichome with cross-walls visible only by electron microscopy. However, the reason that *Spirulina* PCC6313 and *Leptolyngbya* LJM012 are placed in this cluster may be that they both lack tDNA^{ala} as do other unicellular strains in this cluster. In the RFLP dendrogram, *Leptolyngbya* LJM012 located at the early branch of simple filamentous strains with *Leptolyngbya* LJM014 is at a great distance. This suggests that these strains probably belong to different genera.

In sequence trees, other simple filamentous strains form a complex branch with the cyanelle of *Cyanophora paradoxa* and the chloroplast of *Chondrus crispus* (red alga), in line with branch F of the 16S rRNA sequence tree(Wilmotte, 1994). Four cyanobacterial filamentous strains form a loose group. This indicates that four strains may be from four different genera. *Arthrospira* sp. PCC7345 is close to *Oscillatoria animalis* and not to *Spirulina* sp. PCC6313, although it has helical trichomes like the latter strain (see Fig. 17, Rippka et al., 1979). The genus *Arthrospira* and *Spirulina* were fused in the same genera, *Spirulina*, by Geitler (1932) and Rippka et al., (1979). Later, Rippka and Herdman (1992) recognised the separation of the two genera, already proposed by Anagnostidis and Komarek (1988) and Castenholz (1989). Ultrastructural studies (Guglielmi and Cohn-Bazire, 1982) supported this separation and indicated the presence of one circle of pores in the peptidoglycan layer near the cross-walls in *Arthrospira* sp. PCC7345 and several *Oscillatoria* sp. strains sensu Rippka et al. (1979). The presence of one circle of pores was not tested on *Oscillatoria animalis*.

A early branch for cyanobacteria in all sequence trees is *Nostoc palaeo*. This suggests that this strain probably is very ancient (viz. the name palaeo). There is no

evidence that this strain could belong to any group of cyanobacteria, except in the Neighbour-joining tree. It was shown to have a little close relatedness to simple filamentous strains, but not closely related. In the RFLP Dendrogram, *Nostoc palaeo* located in the simple filamentous group (cluster D) with a large distance from other strains. No evidence from morphology results indicated that *Nostoc palaeo* is a heterocystous strain, as no heterocysts and akinetes production was found in this study. Other morphological characters (Fig. 7.9), such as cylindrical trichome waved, unconstricted or little constricted at the cross walls; ability to form thin sheaths around the trichome, false branching lacking, cells \pm isodiametric or longer than wide; polar gas vacuoles, shows that it may be belong to a gas vacuole-producing genus of the *Phormidiaceae* familia (Anagnostidis and Komarek, 1988). To rename this strain will need more phenotypic and phylogenetic information.

All heterocystous strains are situated in the same cluster, in line with branch H of the 16S rRNA sequence tree(Wilmotte, 1994). The genotypic unity of the two strains from section V, or the Order Stigonematales, are located at the early branches of this cluster, and form a loose cluster away from other heterocystous strains. The distance between two strains clearly shows that two strains belong to different genera, and *Fisherella musicola* appears a little close to other heterocystous strains. *Chlorogloeopsis fritschii*, in contrast with the morphology of two strains. *Chlorogloeopsis fritschii* (same strain with PCC 6718) which contains both unicellular and heterocystous filamentous morphology, forms cell aggregates, short filaments and heterocysts (see Fig. 79-84 in Rippka et al., 1979). *Fisherella musicola* in which cell division occurs in more than one plane, forms filaments with true branches and heterocysts. *Anabaena* and *Nostoc* strains in this cluster appear mixed together. By contrast, Caudales and Wells (1992) pointed out that strains of the genus *Nostoc* are

closely related to, and often confused with those of the genus *Anabaena*, particularly when morphological characteristics are used for identification. *Nostoc ellipsospora*, *Anabaena* PCC 7120, and *Anabaena* PCC 7118 have been discussed in chapter 6, and placed in the same species group of *Nostoc*. From this study, morphologies of *N.ellipsospora* and *Anabaena* PCC7120 were very similar (Fig. 7.10), they both produce heterocysts, but Anabaena PCC7118 does not produce heterocysts. All three strains perform a small life cycle (step 3-14-15-16-17 of Komarek and Anabaena PCC 7120 and *Anabaena* PCC 7120 and *Anabaena* PCC 7118, and placed them in the Nostoc cluster 3. Komarek and Anagnostidis (1989) moved *Nostoc linckia, Nostoc ellipsospora* and *Anabaena variabilis* to a new genus *Trichormus*. By contrast, the few strains Nostoc linckia D0734, *Nostoc ellipsospora*, Anabaena PCC7118 and *Anabaena* PCC7120 four strains form a sub-cluster in the sequence tree.

The sequence trees were also successfully constructed based on the primers SF/SR amplified rDNA spacer sequence. The trees based on the primers SF/SR amplified rDNA spacer sequence and the trees based on the primers 17/18 amplified rDNA spacer sequence are similar, except the Neighbour joining tree. It is a little confusing that the positions of *Synechococcus* PCC7942 and PCC6301, *Trichodesmium* NIBB 1067 and *Spirulina* sp. PCC6313 are moved into the heterocystous strain cluster in the Neighbour-joining tree based on the SF/SR amplified sequence. The majority of branches of highly diverse strains which show great distance in the distance tree, show very low bootstrapping values in parsimony tree, except the groups with clear relationships. For example, the Synechococcus group. It is not surprising that the distance and parsimony methods give different branch topologies for the basic nodes that are not supported by the bootstrap analysis. Otherwise, using the single spacer

292

sequence (SF/SR) without tDNA will increase the variation (gaps) of alignment of sequences. This will affect on the particular packages used which are not designed for too many gaps per sequence. In the heterocystous cluster, the *Nostoc* strains and Anabaena strains are clearly separated into two sub-clusters. This proved again the SF/SR primers potential usefulness for identification of cyanobacteria, and suggests that it is not a very suitable analysis for highly diverse strains.

To involve more strains from the same morphological species or genera may be useful in confirming or contradicting the present scheme. The assessment may be useful in pointing out positive as well as negative correlations that are emerging when different methods are applied to the same organisms. This approach may help in formulating questions for further research, and determining the best candidate for future phylogenetic analyses.

The evolution information (Fig. 7.1) based on fossil record available to date, which is by necessity limited to morphological assessment, is largely consistent with the results of molecular phylogeny discussed in here: (a) Both, coccoid and filamentous forms are found very early, preserved in sedimentary rocks of Archean age (3500 Ma). (b) Endolithic forms that reproduce with baeocytes were found preserved in midproterozoic rocks (1500 Ma). (c) helically wound filaments were found in the strata (700 Ma), while heterocystous forms, and forms with true branching have not been convincingly documented in the Proterozoic fossil record of cyanobacteria.

Hillis (1987) pointed out that morphological and molecular systematic techniques each have distinct advantages for phylogenetic reconstruction ... studies that combine the two approaches can thereby maximise both information content and usefulness. Interestingly, the traditional system of cyanobacterial classification based on morphology proved to be more consistent with the emerging phylogenetic information, than the traditional systems of other phototrophic bacteria, which were based on functional rather than structural phenotypes. In this study, some morphological criteria proved very useful, such as cell formation or style (unicellular, filamentous or heterocystous), cell division patterns, colony formation, cell structure; some morphological criteria needed to be modify for particular strains, such as cell length and shape change depending on culture condition, thylakoids located irregularly or peripherally in same strain. The pigment composition can only be used to separate strains in same species.

In theory it would be desirable that taxonomy be based on phylogeny, because the evolutionary processes are the ones responsible for the diversification of a group of organisms. However, the possibility cannot be excluded that convergent evolution has taken place too often at the morphological level, making the use of morphological characters difficult. There is already an indication that this may be the case with morphologically simple organisms. In some cases the acuity of observation may come to good use by including additional characters, which correlate well with phylogenetic positioning. Chapter 8

-

I

General Discussion

-

8.1. General Discussion

Traditional methods for identification and taxonomy of cyanobacteria have been based primarily on morphological and biochemical characteristics. However, such methods are limited particularly when applied to newly isolated strains. For example morphology may be altered when environmental isolates are cultured in the laboratory. At present, the names of cyanobacteria, which are based on cultures, are often inapplicable to field material because of the loss of certain morphological characteristics under common culture conditions. There is reason for optimism, however, since it is becoming a more common practice to vary culture conditions to the extent where these characteristics are often exhibited (e.g. Evans *et al.*, 1976; Stulp, 1982; Stulp and Stam, 1984). Analysis of photosynthetic pigments, isozyme variation and differentiated cell structures may be variable depending upon levels of expression of cyanobacterial gene products. Furthermore, organisms may be viable but non culturable. Thus other methods that do not rely on gene expression and/or culturing should be used in addition to traditional methods to identify cyanobacteria.

Two approaches were used in this study for identification of cyanobacteria: one based on morphological characters, the other on molecular characters. Both approaches generally provided similar conclusions (see chapters 5, 6, and 7).

Plasmid profile is a limited genetic marker, it can be used to research the relationship of some strains which contain plasmids, but is not suitable for the establishment of classification schemes.

296

PCR-based techniques, including RAPD-PCR (chapter 5), PCR-RFLP analysis of the rDNA spacer, and rDNA sequencing (chapter 6), have been developed for identification of cyanobacteria and show promise for the genetic typing of cyanobacteria. Such technology has been assessed along side traditional techniques for rapid identification and for phylogenetic analysis. The applicability of the RAPD technique was restricted by the requirement for axenic cultures and the limitation on number of samples processed, but it can recognise more diversity among the cyanobacteria than PCR-RFLP analysis of the rDNA spacer. The method can be employed to identify rapidly cyanobacteria to the species and strain level of classification in pure culture. It is a good method to combine with morphological methods for identifying cyanobacterial isolates. New isolates could be identified to genus level by using morphological methods before purification, and then identified to strain level by RAPD analysis after purification. This would involve matching the RAPD pattern (see chapter 5).

PCR amplification and RFLP analysis of the 16S-23S rDNA spacer provides a potentially useful approach to the identification of cyanobacteria. Analysis of length polymorphism and RFLPs of the amplified rDNA spacer, using primer pair 17/18, generally permitted discrimination between the cyanobacteria tested to genus and species level without the requirement for axenic cultures. With more data, it should be possible to generate a database of RFLP-patterns, in order to identify unknown isolates by the matching of pattern-type and determining similarity to laboratory strains from phenetic analysis. rDNA sequencing provided a more accurate approach than PCR-RFLP analysis of the rDNA spacer, and identified greater diversity among the cyanobacteria. The rDNA sequencing data can be used to generate a database more easily than the RFLP-patterns and it would be applicable world wide. Furthermore, the

sequencing methodology is suitable for application to a large number strains. There was generally a good correlation between molecular and morphological methods for identifying cyanobacteria, except for a few individuals (chapters 6, 7).

Heterogeneity of the sequence of the 16S-23S rDNA spacer proved useful not only for typing cyanobacteria, but also for developing specific primers. Such specificity opens up possibilities for their use in the rapid identification of cyanobacteria directly from water samples (chapter 6). Furthermore toxigenic species- (or genera-) specific PCR primers/ probes might be developed by the sequencing and analysis of particular toxigenic strains. It should then be possible to detect the present of toxigenic cyanobacteria in water samples more rapidly.

8.2. Recommendations for Further work

The cyano-specific primers developed here, and derivatives there of, could be used to amplify the rDNA spacer of cyanobacteria using DNA extracted from water samples. This would circumvent potential problems associated with culturing of the organisms from water samples. A nested PCR approach (Zhang and Ehrlich, 1994; Zimmermann *et al.*, 1994) could be used: first using primer pair 17/18 to enrich for cyanobacterial DNA, then using SF/SR (or derivatives). The amplified fragments would be cloned and sequenced, following by matching to the sequence database (Fig. 8.1).

By modifying this approach and developing primers specific for toxigenic species/genera of cyanobacteria, could be identified directly from water samples. Furthermore, by tagging specific PCR primers/ probes with fluorophores, it should
Figure 8.1 Flow diagram summarising the schedule of further work based on this study.



be possible to detect the present of toxigenic cyanobacteria rapidly in water samples by assessing fluorescence, using PCR and hybridisation. Such a study could provide a more rapid and reliable method for identifying toxigenic cyanobacteria in particular. This could prove useful to the water industry in terms of assessing water quality and selecting appropriate water sites. The procedures of this further study are outlined in Figure 8.2. Figure 8.2 Flow diagram summarising the schedule of further work of detection of the presence of toxigenic cyanobacteria rapidly in water samples.



Summary

- 51 axenic cyanobacterial strains were obtained from several culture collections or purified from freshwater courses, and grown in the laboratory for genetic study. The genus and species investigated were selected to cover a range of taxonomic groups and to reflect different environmental habitats.
- Several medium and culturing condition were used to isolate cyanobacteria from freshwater courses. It was noticed that some species from water samples were visible but non culturable in the medium used here. 16 isolates were purified and identified by using traditional methods. It was found that the traditional methods used for identification and taxonomy of cyanobacteria based on morphological and biochemical characteristics were slow and difficult, particularly at genera, species and strain level (see chapter 3).
- A range of methods for the extraction of plasmid and genomic DNA from cyanobacteria were investigated. It was found that plasmid isolation was effected by isolation method, culture stage and culturing condition. Methods requiring rapid plasmid isolation are limited, as it is difficult to recovery all plasmids in the cyanobacterial strain every time. The results show that plasmid profile is a limited genetic marker and can be used to research the relationship of some strains which contain plasmids, but not for establishment of classification schemes (see chapter 4).

- The randomly amplified polymorphic DNA (RAPD) technique in conjunction with the polymerase chain reaction (PCR) for identification of cyanobacteria was established and optimised. The choice of primers was found useful for optimising the RAPD reaction. Two 10 nucleotides arbitrary primers, ABA-7 and 10 were chosen for generating RAPDs profile. Genomic DNA of fifty different cyanobacterial laboratory strains and isolates was amplified successfully by using primer ABA-7 or ABA-10. The pattern of amplified products was analysed. The results shows that RAPDs profile can be employed to identify rapidly cyanobacteria to the species and strain level of classification in pure culture, but was difficult at genera and order level. It is opening possibilities to identify rapidly cyanobacterial isolates by combining RAPDs technique with morphological method (see chapter 5).
- The 16S-23S rDNA spacer region of selected cyanobacteria was amplified using primers to conserved flanking regions. Single or multiple amplification products were generated depending on the strain and primer pair. Analysis of length polymorphism and RFLPs of products amplified using primer pair 17/18, generally permitted discrimination between the cyanobacteria to genus and species level without requirement of axenic cultures. Four restriction enzymes (*Hinf I, Dde I, Alu I* and *Taq I*) generated distinguishing restriction profiles. With more data, it should be possible to generate a database of RFLP-patterns, in order to identify unknown isolates by the matching of pattern-type and determining similarity to laboratory strains from phenetic analysis (see chapter 6 and 7).
- Sequences of part of the 16S-23S rDNA spacer of selected cyanobacteria were determined by PCR amplification and direct sequencing of products amplified using primers 17 and 18. Significant variation was generally found in composition and

length of the spacer sequence between cyanobacterial species. A restriction map of the rDNA spacer region was constructed from the sequencing data. Restriction endonuclease sites were found both within the conserved tDNAs and the variable spacers of the region. The map revealed polymorphisms useful in distinguishing between genera and species of cyanobacteria and confirmed the appropriateness of the four enzymes used here in RFLP analysis. A number of phylogenetic trees were generated by using software package RAPDistance and PHYLIP for RFLP profiles and sequences analysis. The clustering of cyanobacterial strains shown from the trees is consistent with that of the 16S rRNA sequence trees (Wilmotte, 1994) and morphological section (see chapter 6 and 7).

- The heterogeneity in the sequence of the 16S-23S rDNA spacer is useful for both typing cyanobacteria and developing specific primers. From rDNA sequence alignment, three short (5-26bp) regions of high sequence homology were found, in addition to the conserved tDNAs in the spacer. Oligonucleotides derived from two of these regions showed promise for use as cyano-specific PCR primers, opening up possibilities for the amplification of cyanobacterial DNA directly from water samples, for rapid species identification. (see chapter 6, 7 and 8).
- Comparing molecular and morphological methods for identifying cyanobacteria, it prove that morphology and genetic of cyanobacteria are related. The morphologically differentiated and complex taxonomic groups appear consistent with phylogenetic information obtained except some individuals.

References

Abdul-Latif Muhammed Jawad (1982) Interactions between cyanobacteria and other micro-organisms. Thesis, The University of Liverpool. p, 17

Akopyanz N., Bukanov N. O., Westblom T. U., Kresovich S. and Berg D. E. (1992) DNA diversity among clinical isolates of *Helicobacter pylori* detected by PCR-based RAPD fingerprinting. Nucleic Acids Res. 20 (19): 5137-5142.

Allen, M. B. and Arnon, D. I. (1955) Studies on nitrogen-fixing blue-green algae 1. Growth and nitrogen fixation by *Anabaena* cylindrica lemm. Plant Physiol. 30: 366-372

Amann, R. I., Ludwig, W. and Schleifer, K. H. (1995) Phylogenetic identification and *in situ* detection of individual microbial cells without cultivation. Microbiol. Rev. 59: 143-169

Anagnostidis, K. and Komarek, J. (1985) Modern approach to the classification system of cyanophytes, 1-Introduction, Arch. Hydrobiol. Suppl. 71, Algological studies 38/39: 291-302

Anagnostidis, K. and Komarek, J. (1988) Modern approach to the classification system of cyanophytes. 3-Oscillatoriales. Arch. Hydrobiol. Suppl. 80, Algological Studies 50/53: 327-472

Anagnostidis, K. and Komarek, J. (1990) Modern approach to the classification system of cyanophytes. 5-Stigonematales. Arch. Hydrobiol. Suppl. 86, Algological Studies 59: 1-73

Anderson, L. K. and Eiseling, F. A. (1985) Plasmids of the cyanobacterium Synechocystis 6701. FEMS Microbiol. Lett. 29: 193-195

Armstrong, J.A., Gibbs, A., Peakall, R. and Weiller, G. (1994) The RAPDistance package. ftp://life.anu.edu.au (150.203.38.74) pub/molecular-biology/rapdpack.exe

Asato, Y. and Ginoza, H.S. (1973) Separation of small circular DNA molecules from the blue-green alga *Anacystis nidulans*. Nature New Biol. 244;132-133

Barry, T., Colleran, G., Glennon, M., Dunican, L. K. and Gannon, F. (1991) The 16S/23S ribosomal spacer region as a target for DNA probes to identify eubacteria. PCR Methods Appl. 1:51-56

Bauld, J. (1981) Occurrence of benthic microbial mates in saline lakes, Hydrobiologia 81: 87-111

Bielawski, J.P., K. Noack and D.E. Pumo (1995) Reproducible Amplification of RAPD Markers from Vertebrate DNA. BioTechiques, Vol. 18 No. 5: 856-900.

Birnboim, H. C. and Doly, J. (1979) A rapid alkaline extraction procedure for screening recombinant plasmid DNA. Nucleic Acids Research 7:1513-1523

Britschgi T. B. and Giovannoni S. J. (1991) Phylogenetic analysis of a natural marine bacterioplankton population by rRNA gene cloning and sequencing. Appl. Environm. Microbiol. 57: 1707-1713

Bolch C.J.S., Blackburn S.I., Neilan B.A. and Grewe, P.M. (1996) Genetic characterization of strains of cyanobacteria using PCR-RFLP of the *cpaBA* intergenic spacer and flanking regions J.Phycol. 32:445-451

Carmichael, W.W. (1989) In "Natural Toxins", Oxford, Pergamon, P3-16

Carmichael, W.W. (1992) Cyanobacterial secondary metabolites-cyanotoxins. J. Appl. Bacteriol. 72: 445-459

Castenholz, R. W. (1989) Oxygenic Photosynthetic Bacteria (sect. 19), Group I. Cyanobacteria. In: Staley JT, Bryant MP, Pfenning N and HoltJg (eds) Bergey's Manual of Systematic Bacteriology, Vol3, pp 1770-1799. Williams and Wllkins Co, Baltimore

Castenholz, R. W. and Waterbury, J.B. (1989) Oxygenic Photosynthetic Bacteria (sect. 19), Group I. Cyanobacteria. In: Staley JT, Bryant MP, Pfenning N and HoltJg (eds) Bergey's Manual of Systematic Bacteriology, Vol3, pp 1710-1728.Williams and Wilkins Co, Baltimore

Caudales, R., Wells, J. M. and Antoine, A. D. (1992) Cellular fatty acid composition of symbiotic cyanobacteria isolated from the aquatic ferm *Azolla*. J. Gen. Microbiol. 138: 1489-1494

Cohen, Y., Jorgensen, B.B, Revsbech, N. P. and Poplawski, R. (1986) Adaptation to hydrogen sulfide of oxygenic and anoxygenic photosynthesis among cyanobacteria, Appl. Environ. Microbiol. 51: 398-407

Danks, S.M., Evans, E.H. and Whittaker, P.A.(1983) Photosynthetic systems : structure, function and assembly, John Wiley & Sons, Chichester.

Desikachary, T. V. (1959) Cyanophyta, Indian Council of Agricultural Research, New Delhi, India, pp. 1-686

Dieffenbach, C.W., T. M. J. Lowe and G.S. Dveksler (1993) General concepts for PCR primer design. PCR meth. and Appli. 3:S30-S37

Doers, M.P. and Parker, D.L. (1988) Properties of *Microcystis aeruginosa* and *M. flos-aquae* (cyanophyta) in culture: taxonomic implications. J. Phycol. 24: 502-508

Douglas, S. E. and Carr, N. G. (1988) Examination of genetic relaredness of marine *Synechococcus* spp. by using restriction fragment length polymorphisms. App1. Environ. Microbiol. 54: 3071-3078

Drouet, F. (1968) Revision of the classification of the Oscillatoriaceae. Monogr. Arcid. Nat. Sci. 67: 1-341 **Drouet, F.** (1973) Revision of the Nostocaceae with cylindrical trichomes, Hafner Press, New York

Drouet, F. (1978) Revision of the Nostocaceae with constricted trichomes, Beih. Nova Hedwigia 57: 1-258

Drouet, F. (1981) Revision of the Stigonemataceae with a summary of the classification of the blue-green algae, J. Cramer, Vaduz

Drouet, F. and Daily, W. A. (1956) Revision of the coccid Myxophyceae. Butler Univ, Bot, Stud, 10: 1-218

Echlin, P. and Morris, I. (1965) The relationship between blue- green algae and bacteria. Biol. Rev. 40, 143-187

Eskew, D. L., Caetano-Anolles, G., Bassam, B. J. and Gresshoff, P. M. (1993) DNA amplification fingerprinting of the symbiosis.Plant Mol Biol 21: 363-373

Evans, E. H., Foulds, I. and Carr, N. G. (1976) Environmental conditions and morphological variation in the blue-green aglae *chloroglora fritschii*, J. Gen. Microbiol. 92: 147-155

Fay, P. (1983) The Blue-Greens, The Camelot Press Southampton

Felsenstein, J. (1985) Confidence limits on phylogenies—an approach using the bootstrap. Evolution, 39 (4): 783-791

Fogg, G. E., Stewart, W. D. P., Fay, P., and Walsby, A. E., (1973) The Blue-Green Algae, Academic Press, London and New York.

Franche, C. and Cohen-Bazire, G. (1987) Evolutionary divergence in the *nifHDK* gene region among nine symbiotic *Anabaena azollae* and between *Anabaena* azollae and some free-living heterocystous cyanobacteria. Symbiosis 3: 159-178

Friedberg, D., and Seikffers, J. (1979) Plasmids in two cyanobacterial strains, FEBS Lett. 107: 165-168

Fritsch, F. E, (1945) The Structure and Reproduction of Algae, Vol II, Cambridge University Press, Cambridge

Gebhatdt, J. S. and Nietzwicki-Bauer, S. (1991) Identification of a common cyanobacterial symbiont associated with *Azolla* spp. through molecular and morphological characterization of free-living and symbiotic cyanobacteria. Appl. Environ. Microbiol. 57: 2141-2146

Geitler, L. (1925) Cyanophyceae. In: Pascher's subwasserflora 12: 1-450 Fischerverlng, Jena.

Geitler, L. (1932) Cyanophyceae. Rabenhorst's Kryptogamenflora von Deutschland, osterreich und der Schweiz 14: 1-1196. Akad Verslagsges, Leipzig. Reprinted 1971, Johnson, New York

Geitler, L. (1982) Zurkenntnis der Oscillatoriacee Pseudanabaena galeata (cyanobacteria). Plant Syst. Evol. 141:169-175

Gibson, C. E. and Smith, R.V. (1982) Fresh water plankton. In: The Biology of cyanobacteria (ed. Carr, N.G. and Whitton, B.A.). Blackwell Scientific, Oxford, p,353-367

Giovannoni SJ, Turner S, Olsen GJ, Barns S, Lane DJ and Pace NR (1988) Evolutionary relationships among cyanobacteria and green chloroplasts. J Bacteriol 170:3584-3592.

Giovannoni SJ, Britschgi TB, Moyer CL and Field KG (1990) Genetic diversity in sargasso sea bacterioplankton. Nature 345: 60-63.

Glazer, A. N. (1988) Phycobiliproteins. In Methods in enzymology. (ed. Packer, L., Glazer, A. N. and Board, E.A.), Academic Press, New York. Vol. 167 (291-312).

Glover, H. E. (1986) The physiology and ecology of marine cyanobacteria, *Synechococcus* spp, Adv. Aquat. Microbiol. 3: 49-107

Goedheer, J.C. (1966) Visible absorption and fluorescence of chlorophyll and its aggregates in solution. In The Chlorophylls, Academic Press New York. P, 147-181.

Golden, S. S., Nalty, M. S. and Cho, D-SC (1989) Genetic relationship of two highly studied *Synechococcus* strains designated *Anacystis nidulans*. J. Bacteriol. 171: 24-29

Goyal, D. (1992) A simplified method for screening and characterization of plasmid DNA in cyanobacteria. J. Microbiol. Methods 15: 7-15

Guglielmi, G. and Cohen-Bazire, G. (1982) Structure et distribution des pores et des perforations de I; enveloppe de peptidoglycane chez quelques cyanobacteries. Protistologica 18: 151-165.

Guglielmi, G. and Cohen-Bazire, G. (1984) Etude taxonomique dur genre de cyanobacterie Oscillatoriacee: le genre *Pseudanabaena* Lauteborn. II. Analyse de la composition moleculaire de la structure des phycobilisomes. Protistologica 20: 383-413

Gurtler, V. and Stanisich, V.A. (1996) New approaches to typing and identification of bacteria using the 16S-23S rDNA spacer region. Microbiology 142: 3-16

Gutell, R.R., Weiser, B., Woese, C. R. and Noller, H. F. (1985) Comparative anatomy of 16S-like ribosomal RNA. Pro. Nucl. Acid. Res. Mol. Biol. 32: 155-216

Gutell, R.R., Larsen, N. and Woese, C.R. (1994) Lessons from an evolving rRNA: 16S and 23S rRNA structures from a comparative perspective. Microbiol. Rev. 58, No.1: 10-26

Hamelin, R.C., Ouellette, G.B., and Bernier, L., (1993) Identification of *Gremmeniella abietina* Races with Random Amplified Polymorphic DNA Markers. Appl. Environ. Microbiol. 59, No. 6: 1752-1755

Herdman, M. (1976) The evolution of cyanobacterial genomes, Abstr. 2nd Int. Symp photosynthetic prekaryotes, August 1976, pp. 229-231, University of Dundee, Scotland.

Herdman, M., Janvier, M., Waterbury, J. B., Rippka, R., Stanier, R. Y. and Mandel, M. (1979a) Deoxyribonucleic acid base composition of cyanobacteria. J. Gen. Microbiol. 111: 63-71

Herdman, M., Janvier, M., Rippka, R. and Stanier, R. Y. (1979b) Genome size of cyanobacteria. J. Gen. Microbiol. 111: 73-85

Hillis, D. M. (1987) Molecular versus morphological approaches to systematics. Ann. Rev. Ecol. Syst. 18: 23-42

Hoffmann, L., (1988) Criteria for the classification of blue-green algae (cyanobacteria) at the genus and at the species level, Arch. Hydrobiol. 80, 131-139

Hoffman, P. and Werner, O. (1966) Spectrophotometric chlorophyll determination having special regard to varions types of equipment, Jena. Rev. 7: 300-303.

Holmes, D. S. and Quigltey, M. (1981) A rapid boiling method for the preparation of bacterial plasmids. Analyt. Biochem. 114: 193-197

Holton, R. W. (1981) Some possible chemotaxonomic approaches to understanding cyanobacterial taxonomy and phylogeny. Ann. New York Acad. Sci. 361: 397-408

Holton, R. W., Stam, W. T. and Boele-Bos, S. A. (1990) DNA-DNA reassociation studies with DNA from *Prochloron*(prochlorophyta) samples of Indo-West Pacific origin. J. Phycol. 26: 358-361

Jannes, G., Vaneechoutte, M., Gillis, M., Vancanneyt, M., Vandamme, P., Verschraegen, G., Van Heuverswyn, H. and Rossau, R. (1993) Polyphsic taxonomy leading to the proposal of *Moraxella canis* sp. nov. for *Moraxella catarrbalis*-like strains. Int. J. Syst. Bacteriol. 43:438-449.

Janssen, I., Mucke, H., Loeffelhardt, W. and Bohnert, H. J. (1987) The central part of the cyanelle rDNA unit of *Cyanophora paradoxa*: Sequence comparison with chloroplasts and cyanobacteria. Plant Mol. Biol. 9: 479-484.

Jensen, T. E. (1985) Cell inclusion in the cyanobacteria. Arch. Hydrobiol./Suppl. 71; Algological studies 38/39: 33-73

Jensen, M.A., Webster, J.A. and Straus, N. (1993) Rapid identification of cyanobacteria on the basis of polymerase chain reaction-amplified ribsomal DNA spacer Polymorphisms. Appl. Environ. Microbiol. 59: 945-952

Jensen, M.A. and Straus, N. (1993) Effect of PCR conditions on the formation of heteroduplex and single-stranded DNA products in the amplification of bacterial ribosomal DNA spacer regions. PCR Methods and Appl. 3: 186-194.

Kado, C.I. and Liu, S.T. (1981) Rapid procedure for detection and isolation of large and small plasmids. J. Bacteriol. 145, No.3 P. 1365-1373

Kann, E. (1972) Zur Systematik und Okologie der Gattung Chamaesiphon Arch. Hydrobiol. 41: 117-171, 243-282

Kann, E. and Komarek, J. (1970) Systematish-Okologishe Berekungen zu den Arten des Formenkreis phormidium autumnale, schweiz Z. Hydrol. 32:495-518

Kaukas, A., Neto, E. D., Simpson, A. J. G., Southgate, V. R. and Rollinson, D. (1994) A phylogenetic analysis of *Schistosoma haematobium* group species based on randomly amplified polymorphic DNA. Inter. J. Parasitol. 24; No. 2: 285-290

Kenyon, C. N., Rippka, R. and Stanier, R.Y. (1972) Fatty acid composition and physiological properties of some filamentous blue-green algae. Arch Mikrobiol 83: 216-236

Klein, S., Chapman, D. J. and Garber, E. D. (1973) Chemotaxonomy of the Oscillatoria-Phormidium complex. Biochem. Syst. 1: 173-177

Knoll, A. H. (1985) The distribution and evolution of microbial life in the late Proterozoic era, Ann. Rev. Microbiol. 39: 391-417

Knoll, A. H. and Golubic, S. (1991) Modern and ancient cyanobacteria. In Schidlowski, M. (eds.): Early Organic Evolution: Implications for Mineral and Energy Resources. Springer Verlag, Heidelberg-Berlin-New York.

Kohl, JG., Borner, T., Henning, M., Schwabe, W. and Weihe, A. (1988) Plasmid content and differences in ecologically important characteristics of different strains of *Microcystis aeruginosa*. Arch. Hydrobiol. Suppl. 80: 195-201

Komarek, J. (1976) Taxonomic review of the genera Synechocystis Sauv. 1892, Synechococcus Nag. 1849, and Cyanothece gen. nov. (Cyanophyceae.). - Arch. Protistenk. 118: 119-179

Komarek, J. (1970) Generic identity of the *Anacystis nidulans* strain Kratz-Allen/Bloom 625 with *Synechococcus* Nag. 1849. Arch Protistenk 112: 343-364

Komarek, J. and Anagnostidis, K. (1986) Modern approach to the classification system of cyanophytes-2 Arch. Hydrobiol. 73:157-226

Komarek, J. and Anagnostidis, K. (1989) Modern approach to the classification system of cyanophytes-4, Arch. Hydrobiol. 82: 247-345.

Komarek, J. and Kann, E. (1973) Zur Taxonomoc und Okologie der Gattung Homoethrix, Arch. Protistenk. 115: 173-233 Lachance, M-A (1981) Genetic relatedness of heterocystous cyanobacteria by deoxyribonucleic acid-deoxyribonucleic acid reassociation. Int. J. Syst. Bacteriol. 31: 139-147

Ladha, J. K. and Watanabe, I. (1982) Antigenic similarity among Anabaena azollae separated from different species of Azolla. Biochem. Biophys. Res. Comm. 109: 675-682

Lambert, G. R. and Carr N. G. (1982) Rapid small-scale plasmid isolation by several methods from filamentous cyanobacteria. Arch. Microbiol. 133: 122-125

Lane DJ, Pace B, Olsen GJ, Stahl DA, Sogin ML and Pace NR (1985) Rapid determination of 16S ribosomal RNA sequences for phylogenetic analyses. Proc. Natl. Acad. Sci. USA 82: 6955-6959

Lang, N. J. and Whitton, B. A. (1973) Arrangement and structure of thylakoides. In: The Biology of Blue- Green Algae (ed. Carr, N. and Whitton, B. A.), Botanical Monographs 9: 66-79

Lau, R.H. and Doolittle, W. F. (1979) Covalently closed circular DNAs in clodely related unicellular cyanobacteria. J. Bacteriol. 137 (1): 648-652

Lau, R.H., Sapienza, C. and Doolittle, W. F. (1980) Cyanobacterial plasmids: their widespread occurrence and the existence of regions of homology between plasmids in the same and different species. Mol. Gen. Genet. 178: 203-211

Laudenbach, D.E., Straus, N.A., Gendel, S. and Willington, J.P. (1983) The large endogenous plasmid of *Anacystis nidulans*: mapping, cloning and localization of the origin of replication. Mol. Gen. Genet. 192: 402-407

Lawton, L.A. and Codd, G.A. (1991) Cyanobacterial (Blue- Green Algal) Toxins and their Significance in UK and European waters. J. IWEM. 460-465

Lazaeoff, N. (1973) Photomorphogenesis and Nostocacean Development. In: The Biology of Blue- Green Algae (ed. Carr, N. and Whitton, B. A.), Botanical Monographs, p,279-319

Leblanc, C., Boyen, C. and Loiseaux-de Goer, S. (1995) Sequence and phylogeny of the plastidial 16S rRNA gene from the red alga *Chondrus crispus*. Eur. J. Phycol. 30: 133-140.

Leblond-Bourget, N., Philippe, H. Mangin, I. And Decaris, B. (1996) 16S rRNA and 16S to 23S internal transcribed spacer sequence analyses reveal inter-and intraspecific *Bifidobacterium* phylogeny. Int. J. Syst. Bacteriol. 46: 102-111.

Ligon PJB, Meyer KG, Martin JA and Curtis SE (1991). Nucleotide sequence of a 16S rRNA gene from *Anabaena* sp. strain PCC 7120. Nucl Acids Res 19:4553

Mazel, D., Houmard, J., Castets, A. M. and Tandeau de Marsac, N. (1990) Highly repetitive DNA sequences in cyanobacterial genomes. J. Bacteriol. 172: 2755-2761

Mazel, D., Bernard, C., Schwarz, R., Castets, A. M., Houmard, J. and Tandeau de Marsac, N. (1991) Characterization of two insertion sequences, IS701 and IS702, from the cyanobacterium *Calothrix* species PCC 7601. Mol. Microbiol. 5: 2165-2170

McCuccheor, T.L. and Carroll, G.C. (1993) Genotypic diversity in populations of a fungal endophyte from Douglas Fir, Mycologia. 85 (2): 180-186

Medlin, L., H. J. Elwood, S. Stickel and M. L. Sogin (1988) The characterization of enzymatically amplified eukaryotic 16S-like rRNA-coding regions. Gene 71: 491-499

Meeks, J.C., Joseph, C.M. and Haselkorn, R. (1988) Organization of the *nif* genes in cyanobacteria in symbiotic association with *Azolla* and *Anthoceros*. Arch. Microbiol. 150: 61-71

Mullis, K.B., (1990) The unusual origin of the polymerase chain reaction, Sci. Am., 262: 56.

Mullis, K.B. and Faloona, F.A., (1987) Specific synthesis of DNA in vitro via a polymerase catalysed chain reaction, Meth. Enzymol., 155: 335

Mullis, K.B., Faloona, F.A., Scharf,S., Horn,G., and Erlich, H., (1986) Specific enzymatic amplification of DNA in vitro: the polymerase chain reaction, Cold spring harbor symp. Quant. Biol., 51:263.

Nageli, C. (1849) Gattungen einzelligen Algen, physiologisch und systematisch bearbeitet. Neue Denkschr. allg. schweiz. ges Naturw. 10. 139.

Nei, M. and Li, W.H. (1979) Mathematical model for studying genetic variation in terms of restriction endonucleases Proc. Natl. Acad. Sci. 76 No. 10 : 5269-5273

Neillan, B.A., Cox, D. T., Hawkins, P. R. and Goodman, A. E. (1994) 16S ribosomal RNA gene sequence and phylogeny of toxic Microcystis spp. (Cyanobacteria). DNA Sequence- The Journal of Sequencing and Mapping. Vol. 4: 333-337

Neillan, B.A.(1995) Identification and Phylogenetic Analysis of Toxigenic Cyanobacteria by Multiplex Randomly Amplified Polymorphic DNA PCR. Appl. Environ. Micribiol. 61: No 6: 2286-2291

Neilan, B.A., Jacobs, D. and Goodman A.E. (1995) Genetic diversity and phylogeny of toxic cyanobacteria determined by DNA polymorphisms within the phycocyanin locus. Appl. Environ. Microbiol 61: No 11: 3875-3883

Nelissen B, Wilmotte A, De Baere R, Haes F, Van de pear Y, Neefs JM and Dewachter, R (1992) Phylogenetic study of cyanobacteria on the basis of 16S ribosomal RNA sequences. Belg. J. Bot. 125:210-213

Nelissen B, Wilmotte A, Neefs JM and Dewachter, R (1994) Phylogenetic relationships among filamentous helical cyanobacteria investigated on the basis of 16S ribosomal RNA gene sequence analysis. Syst. Appl. Microbiol. 17: 206-210

Nelissen B, De Baere R, Wilmotte A, and De wachter, R. (1996) Phylogenetic relationships of nonaxenic filamentous cyanobacterial strains based on 16S rRNA sequence analysis. J. Mol. Evol. 42: 194-200

Nielsen, C. (1989) Phylogeny and molecular data- Science 243: 548

Noller, H. F. (1984) Structure of ribosomal RNA. Annu. Rev. Biochem. 53: 119-162

Olsen, G. J., Larsen, N. and Woese, C. R. (1991) The ribosomal RNA database project. Nucleic Acids Res. 19 (suppl.): 2017-2021

Padan, E. and Cohen, Y. (1982) An oxygenic photosynthesis, In: The Biology of cyanobacteria (ed. Carr, N.G. and Whitton, B.A.). Blackwell Scientific, Oxford. p, 215-235.

Paer. H W, (1990) physiological ecology and regulation of N₂ fixation in natural waters, Adv. Microb. Ecol. 11: 305-344

Pearson, J. E. and Kingsbury, J. M. (1966) Culturally induced variation in four morphologically diverse blue-green algae. Am. J. Bot. 53: 192-200

Penner, G. A., Bush, R. Wise, W. Kim, L. Domier K. Kasha, A. Laroche, G. Scoles, S.J. Molnar and G. Fedak (1993) Reproducibility of random amplified polymorphic DNA (RAPD) analysis among laboratories. PCR Methods Appl. 2: 341-345

Pierce, J. and Omata, T. (1988) Uptake and utilization of inorganic carbon by Cyanobacteria, Photosynth. Res. 16:141-154

Plazinski, J. Zheng, Q. Taylor, R. Croft, L. Rolfe, B.G. and Gunning, B.E.S. (1990) DNA probes show genetic variation in cyanobacterial symbionts of the *Azolla* fern and a closer relationship to free-living *Nostoc* strains than to free-living *Anabaena* strains. Appl. Environ. Microbiol. 56: 1263-1270

Plazinski, J., Croft, L., Taylor, R., Zheng, Q., Rolfe, B.G. and Gunning, B.E.S. (1991). Indigenous plasmids in *Anabaena azollae:* their taxonomic distribution and existence of regions of homology with symbiotic genes of *Rhizobium*, Can. J. Microbiol. 37: 171-181

Porter, R.D. (1988) DNA Transformation, Methods in Enzymology (ed. Packer, L., Glazer, A. N. and Board, E.A.), Academic Press, New York, Vol 167, Cyanobacteria, p: 703-714.

Potts, M. (1984) Distribution of plasmids in cyanobacteria of the LPP group, FEMS Micro. Lett. 24 (2-3): 351-354.

Radloff, R., Bauer, W. and Vinograd, J. (1967). A dye-bouyant-density method for the detection and isolation of closed circular duplex DNA: the close cicular DNA in Hela cell. Proc. Natl. Acad. Sci. U.S.A. 57: 1514-1512.

Reaston, J., Van den Hondel, C.A.M.J.J., Ende A van der, Arkel GA van, Stewart WDP, and Herdman M. (1980) Comparison of plasmids from the cyanobacteriam, *Nostoc* PCC 7542 with two mutant strains unable to form heterocysts, FEMS Micro. Lett. 9:185-188

Rippka, R. (1972) Photoheterotrophy and chemoheterotrophy among unicellular blue-green algae. Archiv fur Mikrobiologie 87: 93-98

Rippka, R. (1988) Isolation, Identification and Culturing, Methods in Enzymology, (ed. Packer, L., Glazer, A. N. and Board, E.A.), Academic Press, New York, Vol 167, Cyanobacteria, 1-67.

Rippka, R., Deruelles, J., Waterbury, J.B., Herdman, M. and Stanier, R.Y. (1979) Generic assignments, strain histories and properties of pure cultures of cyanobacteria. J. Gen. Microbiol. 111: 1-61

Rippka, R. and Herdman, M (1992) Pasteur culture collection of cyanobacteria. Catalogue and Taxonomic Handbook. Vol. I: Catalogue of strains, Institut Pasteur Paris.

Rippka, R., Waterbury, J.B. and Cohen-Bazire, G. (1974) A cyanobacterium which lacks thylakoids. Arch. Microbiol. 100, 419-436

Rippka, R. and Waterbury, J.B. (1977) The synthesis of nitrogenase by nonheterocystous cyanobacteria. FEMS Microbiol. Letters 2: 83-86

Robert, T.M., and Koth, K.E. (1979) The blue-green algae Agmenellum quadruplicatum contains covalently closed DNA circles, Cell 9: 551-557

Rossau, R., Michielsen, A., Jannes, G., Duhamel, M., Kersters K and van Heuverswijn H. (1992) DNA probes for *Bordetella* species and a colorimetric reverse hybridization assay for the detection of *Bordetella pertussis*. Mol. Cell Probes 6: 281-289

Rowan, K.S., (1989) Photosynthetic pigments of algae, Cambridge University Press, Cambridge.

Saiki, R. K., Gelfand, D.H., Stoffel, S., Scharf, S. J., Higuchi, R., Horn, G.T., Mullis, K.B., and Erlich, H.A., (1988) Primer directed enzymatic amplification of DNA with a thermostable DNA polymerase, Science, 239: 487

Sambrook, J., Fritsch, E.F. and Maniatis, T. (1989) Molecular cloning, Cold Spring Harbor Laboratory Press. Sandmann, G. (1988) In vitro carotenoid biosynthesis in Aphanocapsa. In Methods in Enzymology. (ed. Packer and Glazer), Academic Press, New York, Vol. 167: 329-336

Schleifer, K.H. and Stackebrandt, E. (1983) Molecular systematics of prokaryotes. Ann. Rev. Microbiol. 37: 143-187

Schopf, J. W. and Packer, B. M. (1987) Early archean (3.3-billion to 3.5- billionyears-old) microfossils from warsanoona group. Australia Science 237: 70-73

Schopf, J. W. and Walter, M.R (1982) Origin and early evolution of cyanobacteria: the geological evidence. In: Carr, N.G. and Whitton, B.A. (eds) The Biology of Cyanobacteria, pp, 543-564, Blackwell Scientific, Oxford

Schopf, J. W. and Walter, M.R (1983) Archean microfossils: new evidence of ancient microbes. In: Schopf, J.W. (ed): Earth's Earliest Biosphere, its Origin and Evolution, P 214-239, Princeton University Press, Princeton, NJ

Schweder, M.E., Shatters, R.G., West, S.H. and Smith, R.L. (1995) Effect of transition interval between melting and annealing temperatures on RAPD analyses. BioTechniques Vol19, No. 1: 38-42

Schwimmer, D. and Schwimmer, M. (1964) Algae and medicine, In: Jachson D. F (Ed) Algae and Man pp 434. plenum Press, New York, 1811

Simon, R.D., (1978) Survey of extrachromosomal DNA found in the filamentous cyanobacteria, J. Bateriol. 136: 414-418

Stam, W. T. (1978) A taxonomic study of a number of blue-green algal strains (cyanophyceae) based on morphology, growth and deoxyribonucleic acid homologies Proefschr. Rijksuniv. Groningen. P 1-116

Stam, W. T (1980) Relationships between a number of filamentous blue-green algal strains(cyanophyceae) revealed by DNA-DNA hybridization. Arch. Hydrobiol. Suppl. 56: 351-374

Stam, W.T. (1980) Relationships between a number of filamentous blue-green algal strains (Cyanophyceae) revaled by DNA-DNA hybridization. Arch. Hydrobiol. Suppl. 56, Algological Studies 25:351-374

Stam, W.T., Boele-Bos, S.A. and Stulp, B.K. (1985) Genotypic relationships between *Prochloron* samples from different localities and hosts as determined by DNA-DNA reassociations. Arch. Microbiol. 142: 340-341

Stam. W. T and Holleman, H. C. (1975) The influence of different salinities on growth and morphological variability of a number of *Phormidium* strains in culture. Acta. Bot. Neerl. 24: 379-390

Stam, W. T. and Holleman, H. C. (1979) Cultures of *Phormidium*, *Plrctonema*, *Lyngbya* and *Synechococcus* under different conditions, their growth and morphological variability. Acta. Bot. Neerl. 28: 45-66

Stam, W. T. and Stulp, B. K. (1988) New taxonomic methods: DNA/DNA hybridization. In Methods in Enzymology. (ed. Packer and Glazer), Academic Press, New York, Vol. 167: 125-132

Stam, W.T. and Venema, G. (1975) The use of DNA-DNA hybridization for determination of the relationship between some blue-green algae(cyanophyceae). Acta. Bot. Neerl. 26: 327-342

Stam, W. T. and Venema, G. (1977) The use of DNA-DNA hybridization for determination of the relationship between some blue-green algae. Acta. Bot. Neerl. 26:327-342

Stanier, R.Y. (1970) Some aspects of the biology of cells and their possible evolutionary significance. Symp. Soc. Gen.Microbiol. 20, 1-38.

Stanier, R. Y., Kunisawa, R., Mandel, M. and Cohen-Bazire, G (1971) Purification and properties of unicellular blue-green algae (Order Chroococcales). Bacteriological Reviews 35: 171-205

Stanier, R. Y., Sistrom, W.R., Hansen, T.A., Whitton, B.A., Castenholz, R.W., Pfenning, N., Gorlenko, V.N., Kondratieva, E.N., Eimhjellen, K.E., Whittenbury, R., Gherna, R.L. and Truper, H.G. (1978) Proposal to place the nomenclature of the cyanobacteria under the rules of the International Code of Nomenclature of bacteria. Int. J. Syst. Bacteriol 28: 335-336

Stanier, R.Y. and Cohen-Bazire, G. (1977) Phototrophic prokaryotes: the cyanobacteria, Ann. Rev. Microbiol. 31: 225-274

Starmach, K. (1966) Cyanophyta-sinice, Glaucophyta-glaukofity. Flora slodkow polski 2:1-808.

Stulp, B. K. (1982) Morphological variabillity of Anabaena strains under different culture conditions. Arch. Hydrobiol. Suppl. 63:165-176

Stulp, B. K and Stam, W. T. (1982) General morphology and akinete germinotion of *Anabaena* Strains in culture, Arch , Hydribiol. Suppl. 63: 35-52

Stulp, B.K. and Stam, W.T. (1984) Genotypic relationships between strains of *Anabaena* (Cyanophyceae) and their correlation with morphological affinities. Br. Phycol. J. 19: 287-301

Swofford, D. L. and Olsen, G. J. (1990) Phylogeny reconstruction. In: Hillis D.M and Moritz, C. (eds) Molecular Systematics 411-501

Tilden, J. E. (1910) Minnesota Algae. I. The Myxophyceae of North America. Report . Survey, Bot, Set, 8, Minneapolis **Tomioka N and Sugiura M** (1983) The complete nucleotide sequence of a 16S ribosomal RNA gene from a blue-green alga, *Anacystis nidulans*. Mol. Gen. Genet. 191:46-50

Tomioka N and Sugiura M (1984) Nucleotide sequence of the 16S-23S spacer region in the rrnA operon from a blue-green alga, *Anacystis nidulans*. Mol. Gen. Genet. 193: 427-430

Turner S, Burger-Wiersma T, Giovannoni SJ, Mur LR and Pace NR (1989) The relationship of a prochlorophyte, *Prochlorothrix hollandica*, to green chloroplasts. Nature 337:380-382

Urbach E., Robertson D.L. and Chisholm S.W. (1992) Multiple evolutionary origins of prochlorophytes within the cyanobacterial radiation. Nature 355: 267-270

Van Baalen, C. (1967) Further observations on growth of single cells of coccoid blue-green algae J. Phycol. 3, 154

Van den Hondel, C.A.M.J.J., Keegstra, W., Borrias, W.E. and van Arkel, G.A. (1979) Homology of plasmids in strains of unicellular cyanobacteria. Plasmid 2: 323-333

Vaneechoutte M, Rossau R, De Vos P, Gillis M, Janssens D, Paepe N, De Rouck A, Fiers T, Claeys G and Kersters K (1992) Rapid identification of bacteria of the comamonadaceae with amplified ribosomal DNA-restriction analysis (ARDRA). FEMS Microbiol. Lett 93: 227-234

Van Liere, L. and Walsby, A. E. (1982) Interactions of cyanobacteria with light, in: The Biology of Cyanobacteria (ed. Carr, N.G. and Whitton, B.A.), Blackwell Scientific, Oxford, p, 9-45

Versalovic, J., Koeuth, T. and Lupski, J.R. (1991) Distribution of repetitive DNA sequences in eubacteria and application to fingerprinting of bacterial genomes. Nucl. Acids Res. 19: 6823-6831

Voss. E.G (Editor) (1983) International Code of Botanical Nomenelature. In Bohn. Scheltema and Holkema Regnum Vegetabile, Vol. III, Utrecht. (QK96. V96) eubacteria

Walsby, A. E. (1981) Cyanobacteria: Planktonic Gas-Vacuolate Forms. The Prokaryotes, Vol. 1, 224-235

Ward D.E., Bateson M.M., Weller R. and Ruff-Roberts A.L. (1992) Ribosomal RNA analysis of microorganisms as they occur in nature. Adv. Microb. Ecol. 12:219-286

Ward, E. and Akrofi, A.Y. (1994) Identification of fungi in the Gaeumannimyces-Phialophora complex by RFLPs of PCR-amplified ribosomal DNAs. Mycol. Res. 98 (2): 219-224

Waterbury, J.B. (1989) Subsection II, order Pleurocapsales Geitler 1925, emend. Waterbury and Stanier 1978. In: Stakey JT, Bryant MP, Pfennig N and Holt JG(eds) Bergey's Manual of Systematic Bacteriology, Vol3, pp 1746-1770. Williams and Wilkins Co, Baltimore

Waterbury JB and Rippka R (1989) Subsection I, order Chroococcales Wettstein 1924, emend. Rippka., Deruelles, J., Waterbury, J.B., Herdman, M. and Stanier, R.Y. 1979. In: Staley JT, Bryant MP, Pfennig N and Holt JG (eds) Bergey's Manual of Systematic Bacteriology, Vol3, pp 1728-1746. Williams and Wilkins Co, Baltimore

Waterbury, J.B. and Stanier RY., (1977) Two unicellular cyanobacteria which reproduce by budding. Arch. Microbiol. 115: 249.

Waterbury, J.B. and Stanier, R.Y. (1978) Pattems of growth and development in plearocapsalean cyanobacteria. Microbiol. Rev. 42: 2-44

Watson, J. D. and Crick, F.H.C. (1953) Molecular structure of nucleic acids: a structure for deoxyribose nucleic acid. Nature 171: 737-738

Watson, J. D. Hopkins, N.H., Roberts, J.W., Steitz, J.A. and Weiner, A.M. (1987) Molecular Biology of the Gene. Menlo. Park, CA: The Benjamin/ Cummings Publishing Company

Wayne, L.G., Brenner, D.J., Colwell, R.R., Grimont, P.A.D., Kandler, O., Krichevsky, M.I., Moore, L.H., Moore, W.E.C., Murray, R.G.E., Stackebrandt, E., Starr, M.P. and Truper, H.G. (1987) Report of the ad hoc committee on reconciliation of approaches to bacterial systematic. Int. J. Syst. Bacteriol. 37 : 463-464

Weller R, Walsh Weller J and Ward DM (1991). 16S rRNA sequences of uncultivated hot springs cyanobacterial mat inhabitants retrieved as randomly primerd cDNA. Appl. Environm. Microbiol. 57: 1146-1151

Weller R, Bateson MM, Heimbuch BK, Kopczynski ED and Ward DM (1992) Uncultivated cyanobacteria, *Chloroflexus*- like inhabitants, and spirochete- like inhabitants of a hot spring microbiol mat. Appl. Environm. Microbiol. 58:3964-3969

Welsh, J. and McClelland, M. (1990) Fingerprinting genomes using PCR with arbitrary primers. Nucleic Acids Res. 18: 7213-7218

White, T. J., Bruns T., Lee, S. and Taylor, J. (1990) Amplification and direct sequencing of fungal ribosomal RNA genes for phylogenetics. PCR protocals : A Guide to Methods and Application. 38: 315-322

Whitton, B. A. (1987) Survival and dormancy of blue-green algae, in: Survival and Dormancy of Microorganisms (eds, Y. Henis) Wiley, New York, pp, 109-167

Whitton, B. A. (1992) Diversity, Ecology and Taxonomy of cyanobacteria, in: Photosynthetic Prokaryotes (ed. Mann, N.H. and Carr, N.G.), Coventry, England, pp 1-37

Willmotte, A. (1994) Evolutionary affiliation of the marine nitrogen-fixing cyanobacterium *Trichodesmium* sp. strain NIBB 1067, derived by 16S ribosomal RNA sequence analysis. Microbiology 140 (8) : 2159-2164.

Willmotte, A. (1994) Molecular Approach to cyanobacterial Evolution and Taxonomy. In: The Molecular Biology of the Cyanobacteria (eds. Bryant, D.A.). Kluwer Academic, USA, p. 1-25

Wilmotte, A. and Golubic, S. (1991) Morphological and genetic criteria in the taxonomy of cyanophyta/ cyanobacteria. Algological. Studies 64: 1-24

Willmotte, A., Nelissen, B. and De Wachter, R. (1994) The molecular evolution of the cyanobacteria. Proc. VII. Int. Congr. Photosynthetic Prokaryotes., Urbino Italy, P47.

Willmotte, A. and Stam, W.T. (1984) Genetic relationships among cyanobacterial strains originally designated as `*Anacystis nidulans*` and some other *Synechococcus* strains. J. Gen. Microbiol. 130: 2737-2740

Wilmotte A, Turner S, Van de Peer Y and Pace NR (1992) Taxonomic study of marine oscillatoriacean strains with narrow trichomes. II. Nucleotide sequence analysis of the 16S ribosomal RNA. J. Phycol. 28: 828-838

Wilmotte A, Van Der Auwera G and De Wachter R (1993) Structure of the 16S ribosomal RNA of the thermophilic cyanobacterium *Chlorogloeopsis* HTF strain PCC7518, and phylogentic analysis. FEBS Lett 317:96-100

Williams, J.G.K., A.R.Kubelik, K.J.Livak, J.A.Rafalski and S.V.Tingey (1990) DNA polymorphisms amplified by arbitrary primers are useful as genetic markers. Nucleic Acids Res. 18: 6531-6535 Williams, J.G.K., Hanafey, M.K., Rafalski, J.A. and Tingey, S.V (1993) Genetic Analysis Using RAPD Markers. In : Method in Enzymology (ed. Packer, L., Glazer, A. N. and Board, E.A.), Academic Press, New York, Vol 218: 705-740

Woese, C.R. (1982) Archaebacteria and cellular origins: an overview. Zentralbl. Bakteriol. Parasitenkd. Infetionskr. Hyg. Abt. 1 Orig.Reihe C 3: 1-17

Woese, C.R. (1987) Bacterial evolution. Microbiol. Rev. 51: 221-271

Woese, C.R. (1994) There must be a prokaryote somewhere: microbiology's search for itself. Microbiol. Rev. 58: No. 1:1-9

Woese, C.R. and Olsen, G.J. (1986) Archaebacterial phylogeny: perspectives on the urkingdoms. Syst. Appl. Microbiol. 7: 161-177

Wood, A.M. and Townsend, D. (1990) DNA polymorphism within the WH7803 serogroup of marine *Synechococcus* spp. (cyanobacteria). J. Phycol. 26: 575-585

Young, R.A. and Steitz, J.A. (1979) Complementary sequences 1700 nucleotides apart form a ribonuclease III cleavage site in *Escherichia coli* ribosomal precursor RNA. Proc. Natl. Acad. Sci. U.S.A. 75: 3593-3597.

Yu, Y. and K.P. Pauls. (1992) Optimization of the PCR program for RAPD analysis. Nucleic Acids Res. Vol. 20, No. 10: 2606

Zhang, X.Y. and M. Ehrlich (1994) Detection and quantitation of low numbers of chromosomes containing *bcl*-2 oncogene translocations using semi-nested PCR Bio Techniques 16: 502-507

Zimmerman, W.J. and Rosen, B.H. (1992) Cyanobiont diversity within and among cycads of one field site. Can. J. Microbiol. 38: 1324-1328

Zimmermann, K., K. Pischinger, and J.W. Mannhalter (1994) Nested primer PCR detection limits of HIV-1 in a background of increasing numbers of lysed cells. Bio Techniques 17: 18-20

Zuckerkandl, E. and Pauling, L. (1965) Molecules as documents of evolutionary history. J. Theor. Biol. 8: 357-366

.

Appendix A: Preparation of Media

Medium BG11

Stock solution	g/100ml	ml(stock solution)/L(media)
K ₂ HPO ₄	4.00	1
MgSO ₄ 7H ₂ O	7.50	1
CaCl ₂ 6H ₂ O	5.34	1
Citric Acid	0,60	1
Ferric Ammonium citrate	0.60	1
Disodium-magnesium EDT	A 0.10	1
Na ₂ CO ₃	2.00	1
Trace solution		1
H ₃ BO ₃	0.286	-
MnCl ₂ 4H ₂ O	0.181	-
ZnSO ₄ 7H ₂ O	0.022	-
Na₂MoO₄2H₂O	0.039	-
CuSO ₄ 5H ₂ O	0.0079	-
Co(NO ₃) ₂ 6H ₂ O	0.0049	-
Solid	-	-
NaNO,	-	1.5g/L

For media BG11/10 is BG11 with decrease of the NaNO₃ concentration (as much as 10-fold).

ŧ

Medium C

Stock solutions	g/100ml	ml/L(media)
$Fe_2(SO_4)_39H_2O$	0.404	1
Trisodium citrate	16.46	1
Ca(NO ₃) ₂ 4H ₂ O	2.5	1
K₂HPO₄	127.56	1
MgSO₄7H₂O	25.25	1
Trace solution	-	1
H ₃ BO ₃	0.286	-
MnCl ₂ 4H ₂ O	0.181	-
ZnSO ₄ 7H ₂ O	0.022	-
Na ₂ MoO ₄ 2H ₂	0.0024	-
CuSO ₄ 5H ₂ O	0.079	-
Solid	-	-
KNO ₃	-	1g/L

Medium ASM-1

Stock solution	g/100ml	ml/L(media)
K₂HPO₄	1.7418	1
Na₂HPO₄	1.4196	1
MgSO₄7H₂O	4.9296	1
MgCl ₂ 6H ₂ O	4.0662	1
CaCl ₂	2.2198	1
Disodium EDTA	0.7445	1
FeCl ₃ 6H ₂ O	0.1081	1
trace solution	-	1
H,BO,	0.2487	-
MnCl ₂ 4H ₂ O	0.1377	-
ZnSO ₄ 4H ₂ O	0.0914	-
CuSO ₄ 5H ₂ O		-
Co(NO ₃) ₂ 6H ₂ O	0.0023	-
Solid		
NaNO ₃		0.17g/L(media)

+

CHU 10 D Medium

Stock solution	g/100ml	ml/L(media)
KH₂PO₄	0.780	1
MgSO ₄ 7H ₂ O	2.500	1
Ca(NO ₃) ₂ 4H ₂ O	5.760	1
NaHCO ₃	1.585	1
Na2SiO3 5H2O	1.087	1
Disodium EDTA	1.335	0.25
FeCl ₃ 6H ₂ O	0.970	0.25
trace solution	-	1
H ₃ BO ₃	0.0715	-
MnCl ₂ 4H ₂ O	0.0453	-
ZnSO ₄ 4H ₂ O	0.0055	-
CuSO ₄ 5H ₂ O	0.0019	-
CoSO ₄ 7H ₂ O	0.0010	-
Na2MoO4 2H2O	0.0007	-
BUFFER 2.5mM HEPES pH 7.6 (0.6 g/L)		1000ml

.

ACM medium

Stock solution	g/100ml	ml/L(media)
K ₂ HPO ₄ 3H ₂ O	1.310	1
MgSO₄7H₂O	5.000	5.0
CaCl ₂ 2H ₂ O	1.986	1
NaCl	2.300	1
KNO3	10.00	5
Disodium EDTA	0.25	0.2
FeCl ₃ 6H ₂ O	0.605	0.2
trace solution	-	1
H ₃ BO ₃	0.2860	-
MnCl ₂ 4H ₂ O	0.1810	-
ZnSO ₄ 7H ₂ O	0.0220	-
Na₂MoO₄2H₂O	0.0390	-
CuSO ₄ 5H ₂ O	0.0079	-
Co(NO ₃) ₂ 6H ₂ O	0.0049	-
NiSO4 7H,O	0.0048	-

BUFFER 2.5mM HEPES pH 7.6 (0.6 g/L)

1000ml

.

.

AD P(1) Fe(0.4) medium

Stock solution	g/100ml	ml/L(media)
K₂HPO₄3H₂O	0.736	1
MgSO₄7H₂O	5.000	4.0
CaCl ₂ 2H ₂ O	6.620	1
NaCl	4.600	5
KCl	1.91	1
Disodium EDTA	1.335	0.2
FeCl ₃ 6H ₂ O	0.970	0.2
trace solution	-	1
H ₃ BO ₃	0.2860	-
MnCl ₂ 4H ₂ O	0.1810	-
ZnSO ₄ 7H ₂ O	0.0220	-
Na ₂ MoO ₄ 2H ₂ O	0.0390	-
CuSO ₄ 5H ₂ O	0.0079	-
Co(NO ₃) ₂ 6H ₂ O	0.0049	-
NiSO ₄ 7H ₂ O	0.0048	-

BUFFER 2.5mM HEPES pH 7.6 (0.6 g/L)

1000ml
MN medium

Stock solution	g/100ml	ml(stock solution)/L(media)
K₂HPO₄	2.00	1
MgSO₄7H₂O	3.80	1
CaCl ₂ 6H ₂ O	2.67	1
Citric Acid	0.30	1
Ferric Ammonium citrate	0.30	1
Disodium-magnesium EDT	A 0.05	1
Na ₂ CO ₃	2.00	1
Trace solution		1
H ₃ BO ₃	0.286	-
MnCl ₂ 4H ₂ O	0.181	-
ZnSO₄7H₂O	0.022	-
Na ₂ MoO ₄ 2H ₂ O	0.039	-
CuSO ₄ 5H ₂ O	0.0079	-
Co(NO ₃) ₂ 6H ₂ O	0.0049	-
Solid	-	-
NaNO ₃	-	0.75g/L
Sea water		750ml
distilled water		250ml

.

These above media were sterilized by autoclaving at 121°C, 15p.s.i for 20 minutes. After autoclaving VB_{12} , Nystatin and Ampicillin were added if the media was to be used for isolation and purification (section 2.6). For preparation of solid media, 1.5% Difco Bacto-agar should be added in the media. Before sterilization, the mineral medium and agar should be separated (as double strength) and mix them only after cooling to about 50 °C.

LB medium:

10g/L Bacto-tryptone 5g/L Bacto-yeast extract 5g/L NaCl

Adjust pH to 7.0 with NaOH.

LB plates with ampicillin:

Add 15g agar to 1 L of LB medium. Allow the medium to cool to 50 °C before adding ampicillin to a final concentration of 100μ g/ml. Pour 30ml of medium into 85mm petri dishes. Let the agar harden. Store at 4 °C for up to one month or at room temperature for up to one week.

LB plates with ampicillin/IPTG/X-Gal:

100 μ l of 100mM IPTG and 20 μ l of 50mg/ml X-Gal (Appendix B) were spread over the surface of the plate an LB-ampicillin plate and allowed to absorb for 30 minutes at 37 °C prior to use.

Preparation of slide culture for light microscopic examination

A drop of BG11 medium containing Difco agar (1%, w/v) was spread in a thin film over an area of 2×2 cm on a sterile slide and allowed to solidify. A small drop of a liquid culture and medium were placed on the surface of the agar, and a sterile coverslip (2.4×2.4 cm) gently lowered on to the agar surface, the edges then being sealed with sterile vaseline + paraffin (1:1). The slide was examined microscopically, and suitable field for observation selected.

Appendix B: Preparation of Reagents and Buffers

Equilibration of phenol

1. Liquified phenol should be stored at -20C. As needed, melt it at 68C. An equal volume of buffer (usually 0.5M Tris·Cl, PH 8.0 at room temperature) add to the melted phenol. Stir the mixture for 15minutes, when the two phases have separated, aspirate as much as possible of the upper phase using glass pipette.

2. Add an equal volume of 0.1M Tris·Cl (PH 8.0) to the phenol. Stir the mixture for 15minutes. Remove the upper aqueous phase as step1. Repeat the extractions until the pH of the phenolic phase is > 7.8 (with pH paper).

3. After the phenol is equilibrated and the final aqueous phase has been removed, add 0.1 volume of 0.1M Tris·Cl (pH 8.0) containing 0.2% b-mercaptoethanol. The phenol solution may be stored in this form under 100mM Tris·Cl (pH 8.0) in a light-tight bottle at 4C for 1 month.

Phenol:Chloroform:Isoamyl Alcohol (25:24:1)

A mixture consist equal parts of equilibrated phenol and chloroform: isoamyl alcohol (24:1). Neither chloroform nor isoamyl alcohol requires treatment before use. The mixture may be stored under 100mM Tris·Cl (pH 8.0) in a light-tight bottle at 4C for 1 month.

Lysozyme

Prepare a stock solution of lysozyme at a concentration of 50 mg/ml in water. Dispense into aliquots and store at -20°C. Discard each aliquot after use.

Proteinase K

Prepare a stock solution of proteinase K at a concentration of 10 mg/ml in water. Dispense into aliquots and store at -20°C. Discard each aliquot after use.

DNAase Free RNAase

Dissolve pancreatic RNAase (RNAase A) at a concentration of 10 mg/ml in 0.01 M sodium acetate(pH 5.2). Heat to 100° C for 15 minutes. Allow to cool slowly to room temperature. Adjust the pH by adding 0.1 volumes of 1 M Tris-Cl (pH7.4). Dispense into aliquots and store at -20°C.

RNAase precipitates when concentrated solutions are heated to 100°C at neutral pH.

Ligase 10× buffer:

300mM Tris-HCL, pH 7.5
100mM MgCl₂
100mM DTT
10mM ATP

Store in aliquots at -20 °C. Avoid multiple freeze/thaw cycles.

IPTG stock solution (100mM):

1.2g IPTG, add H₂O to 50ml final volume. Filter-sterilized and store at 4 °C.

X-Gal (2ml):

100mg 5-bromo-4-chloro-3-indolyl- β -D-galactoside, dissolve in 2ml N,N' dimethyl-formamide.

Appendix C: The genera classification of cyanobacteria (after Rippka 1988)

Order Chroococcales (Section I):

Unicellular cyanobacteria, that occur singly or form colonial aggregates held together by additional cell wall layer (sheaths); they reproduce by binary fission or by budding.

- I. Reproduction by equal binary fission
 - A. Thylakoids present
 - 1. Division in only one plane, crosswise, cells rod shaped
 - a. Sheath layers absent Synechococcus Nageli 1849
 - b. Sheath layers present Gloeothece Nageli 1849
 - 2. Division in two or more planes, cells coccoid
 - a. Sheath layers absent Synechocystis Sauvageau 1892
 - b. Sheath layers present Gloeocapsa Kutzing 1843
 - B. Thylakoids absent

Division in only one plane, crosswise, cells rod shaped, sheath layers present

Gloeobacter Rippka,

Waterbury and Cohen-Bazire

1974

II. Reproduction by repeated budding at the apical end of the cells

Thylakoids present

Division in only one plane, producing small spherical daughter cells from ellipsoidal, cylindrical, club-, or pear-shaped mother cells

Chamaesiphon Braun and

Grunow

1865 emend. Geitler 1925 (sensu Waterbury and Stanier 1977)

Order Pleurocapsales (section II)

Unicellular cyanobacteria that occur singly or form colonial (often pseudofilamentous) aggregates; mature cells are always enclosed by an additional fibrous outer cell wall layer; they reproduce by multiple fission only or by binary fission followed by multiple fission; the baeocytes, at the moment of their release, are either motile and lack the fibrous outer cell wall layer or possess the latter structure and are immotile.

I. Reproduction uniquely by multiple fission

A. Vegetative growth leads to the formation of spherical cells of varying size

1. Baeocytes motile Dermocarpa Crouan and Crouan

1858

(sensu Waterbury and Stanier 1978)

B. Growth and binary fission lead to cubical aggregates

1. Multiple fission yields motile baeocytesMyxosarcina Printz 1921
(sensu Waterbury and Stanier
1978)

2. Multiple fission yields immotile baeocytes Chroococcidiopsis Geitler 1933

(sensu Waterbury and Stanier 1978)

C. Growth and binary fission lead to irregular, pseudofilamentous cell aggregates

1. Multiple fission yields motile baeocytes	Pleurocapsa group
a. Baeocytes enlarge symmetrically	Type I
b. Baeocytes develop early polarrity	Type II

Order Oscillatoriales (Section III)

Filamentous cyanobacteria; trichomes composed uniquely of vegetative cells, even in the absence of combined nitrrogen; reproduction by intercalary cell division in only one plane and random transcellular trichome breakage; hormogonia that are structurally distinguishable from parental trichomes are never produced; nonsheathed (or thinly sheathed?) members generally exhibit gliding motility, which may or may not involve rotation around the long axis of the trichomes; firmly ensheathed immotile filaments may release transiently motile trichomes that lack heavy sheaths.

I. Trichome helically coiled

A. Shape of cells not fixed; little or no constellitions between adjacent cells; sheat formation never pronounced; probably transcellular trichome breakage; motility involves rotation around the long axis of the trichomes

1. Cross walls invisible by light microscopy

Spirulina Turpin 1827

2. Cross walls visible by light microscopyArthrospira Stitzenberger1852

II. Trichome straight

A. Cells that compose trichome are disk shaparated by shallow constrictuins (if any); end cells different in shape to intercalary cells; transcellular trichome breakage

1. Trichome motile, not heavily ensheathed; motility involves rotation around thelong axis of the trichomeOscillatoria Vaucher 1803

(sensu Rippka et al. 1979)

2. Trichome immotile and heavily ensheathed; motility restricted to sheathless (or thinly sheathed?) trichomal fragments released from the ends of the immotile ensheathed parental filaments; whether motility involves rotation around the long axis of the trichome is unknown LPP group A

("Lyngbya aestuarii" type, this proposal)

B.Cells that compose trichome are isodiametric, cylindrical, or barrel shaped; the degree of constriction between adjacent cells varies; inter- or transcellular trichome breakage; motility(if exhibited) may or maynot involve rotation around the long axis of the trichome; sheath production and false branching variable

1. Trichome motile, not ensheathed; cells contain polar gas vacuoles, are cylindrical or barrel shaped, and are generally separated from one another by pronounced constrictions; intercellular trichome breakage' motility not involving rotation around the long axis of the trichome *Pseudanabaena* Lauterborn 1905

(sensu Rippka *et al.* 1979) LPP group B

2. Properties not as above

a. Trichome composed of isodiameteic(or somewhat shorter) cells(2- 3μ m wide) that are not separated by deep constrictions; probably transcellular trichome breakage; some members produce thin sheaths and may exhibit false branching; motility is never pronounced and does not involve rotation around the long axis of the trichome; thylakoids resteicted to the peripheral regions of the cells; generally hosts to cyanophage LPP-1

LPP group B: type I

("Plectonema boryanum"

type, yhis proposal)

b. Properties different to above (this assemblage harbors members that vary greatly with respect to cell morphology, degree of consteluctions between cells, sheath peoperties, motility, and genetic complexity; it is probably representative of several different genera)

LPP group B: type X

Order Nostocales (Section IV)

Filamentous cyanobacteria that divide in only one plane; in the absence of combined nitrogen, a certain percentage (~10%) of the vegetative cell populstion differentiates into heterocysts, the specific cellular sites of aerobic N₂ fixation; akinetes (resting cells), resistant to desiccation and cold, are produced by some members; reproduction occurs by random trichome breakage, by germination of akinetes(if produced) or, in addition, by formation of hormogonia (sersu Rippka *et a.* 1979).

I. Hormogonia not produced

A. Trichome never exhibits basal-apical polarity

1. Heterocysts are differentiated from terminal and intercalary cells; the position of akinetes (if produced) is variable

a. Vegetative cells are cylindrical, spherical, or ovoid

Anabaena Bory de St. Vincent

1822

b. Vegetative cells are disk shaped Nodularia Mertens 1822

2.Heterocysts are differentiated exclusively from the terminal cells at both ends of the trichome; akinetes are located adiacent to heterocysts

Cylindrospermum Kutzing 1843

II. Hormogonia produced

A. Trichome never exhibita basal-apical polarity

1. Hormogonia give rise to young trichomes that differentiate terminal heterocysts at both ends of the cellular chain

a. Mature trichomes are composed of barrel-shaped, spherical, or ovoid vegetative cells and differentiate heterocysts in both intercalary and terminal

342

positions; they are immotile and may be enclosed by sheath material of variable consistency; some adult trichomes take on a coiled configuration; akinetes (if produced) differentiate away from the heterocysts and are often formed in chains; hormogonia are composed of cells that are generally smaller in size, different in shape (often cylindrical or isodiametric), and may contain polar or irregularly distributed gas vacuoles; they are sheathless and often exhibit active gliding motility, which (as gas vacuolation) is lost with their maturation toadult trichomes *Nostoc* Vaucher 1803

(sensu Rippka et al. 1979)

2. Hormogonia give rise to young trichomes that differentiate heterocysts from the terminal cells at only one end of the cellular chains

a. Mature trichomes are composed of disk-shaped, isodiameteic, or cylindrical cells and differentiate heterocysts predominantly in an intercalacy position; they are immotile, heavily ensheathed, and exhibit frequent false branching. (leading on solid media to aerial growth); akinetes not produced; hormogoniaare composed of cells that are not markedly different in size or shape (although they are generally somewhat shorter) than those of the mature filaments and do not contain gas vacuoles; active gliding motility of the sheathlless (or thinly sheathed?) hormogonia is never pronounced or is absent(?)

Scytonema Agardh 1824

(sensu Rippka et al. 1979)

B. Mature trichome exhibits basal-apical polarity

1. Hormogonia give rise to young trichomes that differentiate a single terminal cell into a heterocyst

a. Mature trichome composed of cells that are disk shaped., isodiametric, or cy-lindrical, and whose size decreases from the base to the apex; young trichomes bear exclusively terminal heterocysts at the basal end; mature trichomes are immotile and may contain intercalary heterocysts; differentiation of akinetes (if produced) is initiated adjacent to basal heterocysts, and they may be formed in chains; sheath production variable; false branching frequent in ensheathed members; hormogonia are sheathless (or thinly sheathed), generally motile, and composed of cells that are smaller in size and different in shape; some hormogonial cells contain gas vacuoles.

Calothrix Agardh 1824

(sensu Rippka et al. 1979)

i. Degree of tapering little pronounced, even after growth in the absence of combined nitrogen; hormogonia (normally motile) are composed of isodiametric cells that contain phycoerythrin and adapt chromatically

Calothrix: type Iii. Properties not as aboveCalothrix: type X

Order Stigonematales (Section V)

Filamentous cyanobacteria capable of divisions in more than one plane; in the absence of combined nitrogen, heterocysts are formed that are in terminal, intercalary, or lateral positions; reproduction by random trichome breakage, germination of akinetes (if produced), or hormogonia production.

I. Hormogonia produced

A. Trichom never exhibits basal-apical polarity

1. Hormogonia give rise to young filaments that differentiate terminal or intercalary heterocysts

Cells of mature trichomes are spherical and divide in more than one plane; subsequent detachment of resulting multicellular groups (derived from the individual cells that constituted the primary trichome) leads to a mode of growth resembling that of chroococcalean cyanobacteria (i.e., *Gloeocapsa*, see Section I); heterocysts in the multicellular aggregates are terminal or lateral (with respect to the long axis of the original trichome); rapid division of some cells in only one plane leads to the production of uniseriate, motile hormogonia, composed of small cylindrical cells; gas vacuoles not produced

Chlorogloeopsis Mitra and Pandey 1966

2. Hormogonia give rise to young trichomes in which heterocyst differentiation occurs exclusively in an intercalary position

Cells of mature trichomes are barrel shaped to spherical and may divide in more than one plane; concomitant division of some cells in only one plane leads to multiseriate primary trichomes with lateral uniseriate branches; heterocysts in the multiseriate trichomes are predominantly terminal or lateral, those of the branches are uniquely in an intercalary position ; hormogonia , composed of small cylindrical cells, are produced by rapid division from the ends of the primary trichomes or from the lateral branches and are generally motile; gas vacuoles not produced; akinetes, formed in some members, occur in chains

Fischerella Gomont 1895 (sensu Rippka et al. 1979)

MOLUCULAR COMPARISONS OF CYANOBACTERIAL FREE LIVING AND LABORATORY STRAINS.

Lu W, V.A. Saunders, E.H. Bvans and <u>S.M.McColl</u>.

School of Biomolecular Sciences, Liverpool John Moores University, Byrom Street, Liverpool, L3 3AF, U.K.

Taxonomic studies on cyanobacteria have traditionally used methods such as morphological characteristics and DNA G/C content. Recently molecular studies based on the use of polymerase chain reaction (PCR) techniques and RFLP (restriction fragment length polymorphism) analysis have become widely used. Comparisons of the techniques and their efficacy in strain differentiation is needed, in particular to enable analysis of the strains involved in the recent increase in free living, cyanobacterial growth.

We have isolated and purified a number of free living cyanobacteria from local ponds. These ponds varied in size and aspect but did not appear to be associated with high nutrient levels. From these ponds both unicellular and filamentous strains were obtained and a number of techniques are being used to characterise these stains. Comparisons with laboratory strains are also being performed. So far the isolated strains have mainly been characterised by morphological characteristics whilst both genomic and plasmid DNA has been isolated and analysed from the laboratory strains. RAPD (randomly amplified polymorphic DNA) analysis of the genomic DNA gave very distinct banding patterns using both the normal 10 mer primers as well as 17 mer primers for PCR amplification. This anallysis is now being applied to the isolates. However, it must be noted that careful choice of conditions and thermostable enzyme used was required for consistent results. Results from our analysis so far performed will be presented and discussed along with the associated difficulties in comparing molecular analysis results of the RAPD type from different laboratories.

P-13-018 ALTERATIONS OF THE PHOTOSYNTHETIC APPARATUS IN A SYNECHOCYSTIS SP. PCC 6803 ORF184 MUTANT

Hendrik Schubert¹, Annegret Wilde², Heiko Hartel² & Thomas Börner² ¹FB Biologie, Universität Rostock, 18055 Rostock; ²Institut für Biologie, Humboldt-Universität, 10115 Berlin, Germany

We cloned and sequenced a gene of the unicellular cyanobacterium Synechocystis 6803 that is homologous to the conserved chloroplast open reading frame (ORF) 184. The function of the putative gene product remained unknown till present. Synechocystis orf184 mutants were constructed by insertional inactivation and deletion. We have investigated the effects of this mutation on kinetics and composition of components of the linear electron transport chain. Besides extensive changes in photosystem (PS)I- to PSII ratio, antenna size and quantum yield, we found also alterations in kinetic parameters as light dependent slope of PSII fluorescence yield and PSI donor and acceptor limitation, respectively. Further analyses concerning effects of photosynthetic and respiratory inhibitors as well as excitation energy transfer efficiency elucidated the role of the putative ORF184 protein in photosynthesis.

P-13-019 SIGNAL TRANSDUCTION DURING CHROMATIC ADAPTATION

David M. Kehoe and Arthur R. Grossman, Carnegie Institution of Washington, 290 Panama Street, Stanford, CA 94305 USA

We have used a combination of site directed mutagenesis (SDM) and complementation of novel mutants to extend our understanding of the mechanisms controlling complementary chromatic adaptation (CCA). Found in a number of cyanobacteria, CCA is an adjustment of the phycobilisomes' composition in response to changes in ambient light quality. Previous work demonstrated that one of the components controlling CCA, RcaC, shows strong sequence similarity to response regulator proteins of two component regulatory systems. The activity of such proteins is controlled by the reversible phosphorylation of an aspartate (D) residue found within a domain of highly conserved amino acids. RcaC contains two such putative D phosphorylation domains, at the amino- and carboxy-terminal ends. SDM of the D within each of these domains demonstrated that the amino-terminal D is primarily responsible for the control of CCA and established the relationship between the phosphorylation state of this D and the physiological state of the cells. We have also gained insights into photoperception and signal transduction during CCA by complementing a number of new mutants.

P-13-020 CLONING AND CHARACTERIZATION OF SEC AND FFH GENES FROM THE CYANOBACTERIUM Synechococcus PCC7942

Masato Nakai, Tetsuya Nohara, Daigo Sugita, and Toshiya Endo Department of Chemistry, Faculty of Science, Nagoya University, Chikusa-ku, Nagoya 464-01, Japan

Cyanobacteria have an internal thylakoid membrane network in addition to the cell envelopes that consist of the outer and the inner cytoplasmic membranes. Distinct protein compositions between the cytoplasmic and the thylakoid membranes suggest the presence of a mechanism of protein sorting between the two membranes in cyanobacterial cells. We have cloned secY and secA gene homologs from the genome of cyanobacterium Synechococcus PCC7942; SecA, a peripheral membrane protein, and SecY, an integral membrane protein, are known to play essential roles in the protein translocation across the cytoplasmic membrane in *E. coli*. The cyanobacterial *sec* genes are likely single-copy genes and their gene products are localized in both the cytoplasmic and thylakoid membranes, suggesting that the same Sec proteins probably mediate protein translocation across both the cytoplasmic and thylakoid membranes in the cytoplasmic subject the cytoplasmic and thylakoid membranes in the cyanobacterium. Furthermore, we have cloned a cyanobacterial *ffn* gene which encodes a component of putative bacterial homolog of signal recognition particle (SRP). Intracellular localization of the Ffn protein and its possible involvement in protein transport in the cyanobacterial tells are under investigation.

P-13-021

Several Methods for Isolation and Characterisation of Plasmid DNA in Cyanobacteria

Weigun Lu, E. Hilary Evans, S. McColl and V. Sounders School of Biomolecular Science, Liverpool John Moores University, Byrom street, Liverpool L3 3AF. U.K.

Plasmid distribution among four different cyanobacteria was studied using six simplified method which involved direct agarose gel electrophoresis of heat-treated, ethanol-precipitated. plasmid preparations from the cleared lysates without requiring ultracentrifugation and plasmid screening from different growth period culture. Two of these methods, based on lysis by lysozyme and lysis by SDS. Were successful. The latter method proved more sensitive. All of strains examined possessed one or two plasmids, and mid-exponential phase culture was optimal for all plasmids and large plasmid yield obtained. These procedures permitted the selective isolation of plasmids DNA that can be used directly in electrophoretic analysis and restriction endonuclease analysis.

P-13-022 GENERATION AND CHARACTERIZATION OF A PHOTOHETE-ROTROPHIC MUTANT OF Synechocystis sp. PCC 6803

S.H. Kim¹, J.S. Choi¹, Y.H. Chung¹, L. Bogorad² & Y.M. Park¹ Biomolecule anal group, Korea Basic Science Institute, Taejeon, Korea; ²Dept of Cell Development Biol, Harvard University, Cambridge, USA

Cyanobacterium Synechocystis sp. PCC 6803 performs oxygenic photosyn-thesis analogue to that of higher plants. The reaction center of photosys-tem I (PSI) is composed of a heterodimer of homologous polypeptides, PsaA and PsaB. In order to investigate the structure and function of the PSI, carridge mutagenesis technique was used to inactivate the psaBgene which encodes subunit Ib (PsaB protein) of PSI. The psaB mutants gene which encodes subunit 1b (PsaB protein) of PSI. The psaB mutants were generated by transforming wild type cells with the psaB gene of Synechococcus 7002 which was interrupted by a chloramphenicol acetyl-transferase gene. Unexpectidly, many kinds of mutants having different color were obtained. One of the mutants, B2, was selected for further dissection. The B2 mutant would not grow photoautotrophically, which is the common characteristics of photosystem mutants. The amount of chlo-rophyll of the mutant was gravity generative down number of the rophyll of the mutant was greatly reduced. Oxygen uptake rate of the mutant was ca. 60% of wild type. Immunodetection analysis indicated that the mutan has reduced amount of PsaA/PsaB, but a normal range of PsaC and PsaD. This mutant will be used to study structure/function relationship and biogenesis of PSI.

P-13-023

SOME STUDIES WITH MUTANT STRAINS IN WHICH SPECIFIC GENES ENCODING SUBUNITS OF PHOTOSYSTEM I WERE INACTIVATED

<u>Hitoshi Nakamoto</u> Dept. of Molecular Biology, Saitama University, Urawa 338, JAPAN

In order to clarify functions of subunits of photosystem I complex (PSI), we have conducted biochemical analysis of PsaD-, E-, F-, and I-less

biochemical analysis of PsaD-, E-, F-, and I-less mutant strains of <u>Synechocystis</u> sp. PCC6803. We observed slow photoautotrophic growth of PsaD-less mutant at all temperatures examined and of PsaE-less mutant under 22° C. Western blot analysis showed the absence PsaC in thylakoid membranes from PsaD less mutant. Thus, PsaC is unstable in PSI in the absence of PsaD. Photo-autotrophic growth of PsaD- and PsaE-less mutants were severely inhibited at high light intensity (1200 uE/m²/s). This photoinhibition was further increased under low temperature. Possible functions of PsaD page in relation to photoincreased under low temperature. Possibl fuctions of PsaD and PsaE in relation to photoinhibition will be discussed.



FEMS Microbiology Letters 153 (1997) 141 149



Identification of cyanobacteria by polymorphisms of PCR-amplified ribosomal DNA spacer region

Weiqun Lu, E. Hilary Evans, Suzzanne M. McColl, Venetia A. Saunders *

S Loc of Biomolecular Science Liverpool John Moores University Byrom Street Liverpool L3 3AF UK Received 3 March 1997, revised 25 May 1997; accepted 25 May 1997

Abstract

The 16S-23S ribosomal DNA spacer region of selected cyanobacterial strains was amplified by the polymerase chain reaction using primers to conserved flanking sequences. Single or multiple rDNA amplification products were generated depending on the strain and primer pair Species could generally be distinguished on the basis of size heterogeneity of the products. Analysis of restriction digests of the amplified rDNAs indicated polymorphisms useful in identification. Four enzymes (*Hinfl, DdeI, A ul, Taql*) generated restriction fragment length patterns that could discriminate between the cyanobacteria to the taxonomic levels of genus and species. This approach should prove useful in the rapid identification of cyanobacteria.

A ds Ribosomal DNA, Spacer region, Cyanobacteria, Polymorphism

1. Introduction

The cyanobacteria are one of the largest subgroups of Gram-negative photosynthetic prokaryotes, containing a number of diverse genera and species [1]. Traditional techniques for identification and systematics of cyanobacteria have relied primarily on observed morphological characteristics. However, morphology may change depending on environmental conditions (e.g. [2]) and the diversity of strains within a culture may be suppressed by selective culturing conditions [3]. Photosynthetic pigment content, lipid composition, differentiated cell structures and other characters may also be subject to change because of the variable expression of cyano-

* Corresponding author. Tel.: +44 (151) 231 2204; Fax: +44 (151) 298 2821; E-mail: V.A.SAUNDERS@livjm.ac.uk bacterial gene products in culture [4.5]. The limitations of such methods have prompted the development of molecular biological techniques for identification of cyanobacteria.

The randomly amplified polymorphic DNA (RAPD) technique in conjunction with the polymerase chain reaction (PCR) has been used to characterise cyanobacteria in pure culture and in symbiotic associations [6.7]. This approach has been employed to identify many organisms to the strain level of classification [8.9], but the requirement for axenic cultures restricts the applicability of the RAPD technique. Genetic characterisation of cyanobacterial strains has also been undertaken using restriction fragment length polymorphisms of the intergenic spacer of the genes encoding phycobilisome subunits (*cpc BA*) [10].

In this study we have developed a PCR-based

0378-1097/97 \$17 00 © 1997 Federation of European Microbiological Societies. Published by Elsevier Science B.V. PII S 0 3 7 8 - 1 0 9 7 (9 7) 0 0 2 4 9 - 8

technique that requires only small quantities of relatively impure DNA, and can distinguish between the various genera and species of cyanobacteria by using the variability of the ribosomal DNA spacer region. A number of cyanobacterial 16S rRNA genes and the internal transcribed spacer (ITS) region between 16S and 23S rRNA genes have been sequenced for taxonomic purposes [11 16]. The high interspecific variability reported for the spacer makes it an appropriate candidate for restriction fragment length polymorphisms (RFLPs). The approach we have therefore taken has been to amplify the 16S-23S rDNA spacer from cyanobacteria by PCR and test various restriction enzymes to find those producing RFLPs useful in identification. The primers used recognise conserved regions of cyanobacterial rDNA in the flanking 16S and 23S genes [12].

2. Materials and methods

2.1. Cyanobacterial cultures

The cyanobacterial strains used are described in

Table 1

Cyanobacteria used and the sizes of rDN/	fragments amplified by primer	pairs 1/18, 14/18 and 17/18
--	-------------------------------	-----------------------------

Strain	Source*	Amplified rDNA (bp) ^b			
		Primer 1/18	Primer 14/18	Primer 17/18	
Un cel u ar					
Chroucoccus pre cottu	BSM	1865	555	310	
Gloeocapsa sp	BSM	1910	560	322	
Microcy iis aeruginosa	BSM	1905	585	320	
Syne hococcus PCC 7942	LJMU	2100	735	425	
Syne hococcus CCAP 1479/7	LU	2400	1010	665	
S) nechococcus WH 7803	LU	2300	930	625	
Syne hocystus PCC 6803	UW	1940	665	390	
F lamentous non-heterocystous					
O illat ria animalis	BSM	2200	840	575	
Ph rmidium autumnale	BSM	2200	800	535	
Pseudanabaena sp LJM 001	LJMU	2075	735	450	
F lament us heterocystous					
Anabaena ATCC 27892	LU	1875,2100 (2300,3642)	515,730 (665,857,960,1100)	435	
Anabaena ATCC 27892 R3 2	LU	1875,2100 (2300,3642)	515,730 (665,857,960,1100)	435	
Anabaena vlindrica	UW	1890 (2150,2230)	575 (723,797,948,1036,1217)	485	
Anabaena flos-aquae D0848	UD	1836 (2065)	535 (797,887,995)	430	
Anabaena PCC 7120	UW	1875,2100 (2300,3642)	515,730 (665,857,960,1100)	435	
Anabaena sp D0697	UD	1836 (2065,2230)	530 (695,750,887,995)	430	
Aphanizomenon gracile	BSM	1780,2350	450,858 (329)	440	
Chlorogloeopsis Chlorogloea fritschii	LJMU	2018,2185 (2300)	593 (730.980)	480	
Fischerella musicola	BSM	2018 (2105,2340)	530 (730,930)	475	
Gloeoetrichia sp	UW	1940,2300 (2465)	494,845 (1029,1170)	586	
Nostoc commune	BSM	1970.2000,2190 (2460,3943)	560,790 (705,985,1230)	535	
Nostoc ellipsora	UW	1875,2100 (2300,3642)	515.730 (665.857,960,1100)	435	
Nostoc linckia D0734	UD	1930 (2190,2400)	530.686.790 (930,1050.1182)	445	
Nostoc MAC PCC 8009	UW	2090 (1979, 2320)	710 (630, 960)	495	
Nosioc MAC R1	LU	2090 (1979,2320,1876)	710 (630,960,540)	495	
Nostoc muscorum	BSM	2195 (1875,2005,3642)	780 (510.675,756,857,960,1100)	435	
Nostoc piscinale D0201	UD	1890,2190 (2410,3743,4133)	490,675.760 (880,985,1182)	490	
Nostoc planctonicum	BSM	1875 (2050)	515 (675,740,867)	520	
Nostoc sp D0740	UD	1890 (2190,2500)	490,745 (675,930,1100)	445	
Tolvpothrix distoria	BSM	2175 (1821)	682 (602)	440	

*LJMU, UW, BSM, LU, UD: Liverpool John Moores University, University of Warwick. Biobred Sciento Manchester, Lancaster University and University of Durham, respectively.

۰.

Data are the means of three determinations. Figures in parentheses are secondary fragments.

Table 1. Strains were grown in BG11 liquid medium or MN liquid medium for *Synechococcus* WH 7803 [17] in an illuminated orbital incubator at $36 \pm 1^{\circ}$ C bubbling with CO₂ enriched air. Strains were maintained on BG11 or MN agar as appropriate.

2.2. DNA preparation for PCR

Total genomic DNA was prepared according to a modified method of Porter [18]. A pelleted 10 ml aliquot of mid to late exponential phase culture was washed and resuspended in 0.5 ml of sucrose buffer (10^o (w/v) sucrose, 100 mM EDTA pH 8.0, 50 mM Tris-HCl pH 8.0). Lysozyme (to 5 mg ml⁻¹) was added and, following incubation at 37°C for 30 min, 6 μ l of proteinase K (10 mg ml⁻¹) and 60 μ l of 10% (w/v) sodium dodecyl sulfate were added. The mixture was incubated at 50°C for 2 h, chilled on ice and extracted with an equal volume of phenolchloroform-isoamylalcohol (25:24:1), followed by chloroform-isoamylalcohol (24:1). The supernatant was added to 1/3 volume of 10.5 M ammonium acetate. DNA was precipitated by the addition of isopropanol (-20°C) followed by centrifugation for 5 min at room temperature. The pellet was washed with 70° (v v) ethanol, dissolved in TE buffer (10 mM Tris-HCl, 1 mM EDTA, pH 8.0) and stored at -20° C. In a second method, a 1 ml aliquot of mid to late exponential phase culture was pelleted by centrifugation, the cells were washed, resuspended in 0.1 ml of TE buffer and used directly for PCR.

2.3. PCR and digestion of amplified DNAs

Each 100 μ l reaction mixture for PCR contained 40 pmol of each primer, 2.5 units of *Taq* polymerase (Bioline), 0.2 mM deoxyribonucleoside triphosphates (dATP, dCTP, dGTP and dTTP), 3 mM MgCl₂ and 0.1 μ g cyanobacterial DNA or 1 μ l TE-buffered culture. The reaction mixture was overlaid with mineral oil, and thermal cycling was achieved in a Perkin Elmer Cetus thermal cycler according to the following programme: initial denaturation at 94°C for 3 min, 35 cycles of 94°C for 1 min, 55°C for 1 min and 72°C for 3 min. Primers 1, 14, 17 and 18 as described by Wilmotte [12] were used. DNA negative controls consisted of the standard PCR reaction mixture, but with the DNA template replaced by an equal volume of water. Amplified DNA (10 µl. approximately 0.5-0.8 µg) was digested without further purification with various restriction enzymes. DdeI. Taql. Alul. Sau3A. HaeIII. Pvul. Xbal. Sall. Sacl. Aval. BamHI. HindII. HindIII. Pstl. Bg/II. EcoRV (from Boehringer Mannheim. Germany), and Hinfl. Hhal, Hpall, Rsal (from Sigma, UK) according to the manufacturers' instructions. The products were separated on gels containing 2% (w/v) agarose NA (nucleic acids) +1% (w/v) agarose MP (multi purpose). Appropriate molecular mass markers (123 bp) (Sigma, UK) were run on each gel. Each band visualised on a gel was measured using the Phoretix ID (version 2.51) package of Phoretix International Ltd, UK. Analysis of these data was performed by calculating pairwise genetic distances from a binary matrix. Distances were calculated using the 14 algorithms supplied by the RAPDistance package (version 1.03) of Armstrong et al. [19]. The resulting dendrogram was constructed using the neighbourjoining method.

3. Results and discussion

A number of primer pairs were tested for amplification of the 16S-23S rDNA spacer with a view to generating products useful for discriminating between cyanobacterial strains. The PCR was performed either directly on cyanobacterial cultures or on DNA extracted from them. In each case identical results were obtained. The choice of primers was critical to the use of spacer variation for species identification. In line with the work of Nelissen et al. [16] we have found that primer 18 is very cyanobacteriaspecific, preferentially amplifying cyanobacterial DNA, even in the presence of contaminating eubacterial DNA (data not shown). Primer pair 1/18 (which amplifies 16S rDNA. ITS region, tDNA^{ile} (or tDNA^{ile} and tDNA^{ala}) and 5' end of 23S rDNA) and 14/18 (which amplifies 3' end of 16S rDNA, ITS region, tDNA^{tle} (or tDNA^{tle} and tDNA^{ala}) and 5' end of 23S rDNA) amplified the rDNA to give multiple DNA bands of different intensities for a number of the species tested. The intense bands have been classified as primary products. the weaker variable bands as secondary products (Table 1 and Fig. 1A). Such classification of product





M 1 2 3 4 5 6 7 8 9 10 11 12 13 14 15

B

Fig 1 Agarose electrophoresis gels of amplified rDNA spacer products and RFLP patterns of selected cyanobacterial strains. A: PCR amplification using primer pairs 17/18, 14/18 and 1/18. Lanes: M, 1 kb DNA ladder: 1, Synechococcus PCC 7942; 2, S)nechocystis PCC 6803, 3, Oscillatoria animalis, 4, Nostoc muscorum and 5, Anabaena cylindrica. B: RFLP patterns of rDNA spacer amplified using primers 17 and 18 and digested by Taq1. Lanes. M, 123 bp DNA ladder: 1, Synechococcus PCC 7942; 2, Sinechococcus CCAP 1479/7; 3, Sinechococcus WH 7803 4, Sinechocistis PCC 6803; 5, Microcystis aeruginosa; 6, Gloeocapsa sp.; 7, Chroococcus prescottir: 8. Oscillatoria animalis 9, Phormidium autumnale; 10, Pseudanabaena sp LJM 001; 11, Aphanizomenon gracile; 12, Gloeotrichia sp.; 13, Tolypothrix distorta. 14, Chlorogloeopsis fritschu; 15, Fischerella musicola.

profiles has been reported in the amplification of the 16S-23S rDNA spacer of other bacteria (e.g. [20]). Weak bands were common in the heterocystous genera *Nostoc* and *Anabaena*, which may reflect rDNA heterogeneity for heterocyst differentiation. An explanation for the secondary amplification products could be that heteroduplexes form, containing conserved homologous sequences at the 3' and 5' ends separated by highly variable ITS sequences. These heteroduplexes would migrate more slowly in gels than the expected homoduplexes because of the single-stranded regions [21].

The pattern and length heterogeneity of the amplified bands was generally sufficient to resolve the different genera and species with the exception of Nostoc ellipsora, Anabaena PCC 7120, Anabaena ATCC 27892 and Anabaena ATCC 27892 R3.2 (which carries cyanophage AN23 DNA), which all shared the same sized fragments. However, a different RAPD DNA profile was generated in each case (data not shown). The same primary fragment of 2090 bp (with primer 1/18) or 710 bp (with primer 14/18) was generated for both strains of Nostoc MAC, with two weak secondary fragments in each case. However, strain R1 showed a further weak fragment, using either primer pair. For the unicellular and nonheterocyst-forming filamentous cyanobacteria a single amplified rDNA product was obtained using either of these primer pairs. Such variability in products generated may be due to differential specificity of primers, variation in copy number and microheterogeneities of the 16S-23S rDNA spacer regions



Fig 2. Diagrammatic representation of RFLP patterns of amplified rDNA spacer products (using primers 17 and 18) from selected strains of cyanobacteria with the restriction enzymes *HinfI*, *DdeI*, *AluI* and *TaqI*. The numbers below the tracks correspond to the RFLP patterns in Table 2.

146

Table 2

RFLP patterns and pattern types obtained for selected cyanobacterial strains

Strain	Pattern type ^a	HinfI	Ddel	Alul	Taql	_
Unicellular					-	
Chroococcus prescottii	Α	H7	D7	A7	T-	
Gloeocapsa sp	В	H6	D6	A6	T6	
Microcystis aeruginosa	с	H5	D5	A5	T5	
Sunechococcus PCC 7942	D	HI	DI	Al	T1	
Synechococcus CCAP 1479/7	E	H2	D2	A2	T2	
Synechococcus WH 7803	F	H3	D3	A3	T3	
Svnechocystus PCC 6803	G	H4	D4	A4	T4	
Filamentous non heterocystous						
Oscillatoria animalis	н	H8	D8	A8	T8	
Phormidium autumnale	I	H9	D9	A9	Т9	
Pseudanabaena sp LJM 001	I	H10	D10	A10	T10	
Filamentous heterocystous						
Anabaena ATCC 27892	К	H17	D17	A18	T16	
Anabaena ATCC 27892 R3 2	К	H17	D17	A18	TI6	
Anabaena PCC 7120	К	H17	D17	A18	T16	
Anabaena cylindrica	L	H18	D18	A19	T17	
Anabaena flos-aquae D0848	м	H19	D19	A21	T18	
Anabaena sp D0697	N	H19	D19	A20	T18	
Aphanizomenon gracile	0	H20	D20	A22	T19	
Chlorogloeopsis Chlorogloea fritschii	Р	H23	D23	A25	T22	
Fischerella musicola	Q	H24	D24	A26	T23	
Gloeoetrichia sp	R	H22	D22	A24	T21	
Nostoc commune	S	H12	D12	A12	T12	
Nostoc ellipsora	K	H17	D17	A18	T16	
Nostoc linckia D0734	Т	H14	D15	A15	T14	
Nostoc MAC PCC 8009	U	H11	D11	A11	T11	
Nostoc MAC RI	U	H11	DH	A11	T11	
Nostoc muscorum	v	H16	D17	A17	T16	
Nostoc piscinale D0201	w	H13	D13	A13	T13	
Nostoc planctonicum	x	H15	D16	A16	T 15	
Nostoc sp D0740	Y	H14	D14	A14	T14	
Tolypothrix distorta	Z	H21	D21	A23	T20	

Details of the patterns (H1 H24, D1 D24, A1 A26 and T1-T23) for amplified (primer pair 17/18) rDNA digests are given in Fig. 2. Pattern type derives from the pattern combination with the four test restriction enzymes.

among cyanobacteria. The pattern of amplified products with primer pair 14/18 enabled clearer discrimination between species than that with primer pair 1/ 18.

Analysis of length polymorphism and RFLPs of the amplified rDNA spacer using primer pair 17/18 has generally permitted discrimination between the cyanobacteria tested to genus and species level. This primer pair (which amplifies tDNA^{ile} (or tDNA^{ile} and tDNA^{ala}), ITS region and 5' end of 23S rDNA) amplified rDNA from all the strains tested, generating a single band (Table 1 and Fig. 1A). This amplified DNA was digested with 20 different restriction enzymes to identify RFLPs. Most of the enzymes did not cut the DNA or only very small fragments were cut from the ends. However, *Hin*fI, *Dde*I, *Alu*I and *Taq*I showed potentially useful discrimination among the different genera and species (Figs. 1B and 2). In most cases, a different RFLP pattern was generated (generally of three bands) for each species tested, with each restriction enzyme used. The organism could be assigned an RFLP pattern type on the basis of the four enzymes used (Table 2). For strains of the same species the same pattern type was obtained. Different species and genera had different pattern types, with the exception of *N. ellipsora, Anabaena* PCC 7120, *Anabaena* ATCC 27892 and *Anabaena* ATCC 27892 R3.2.



Fig. 3 Dendrogram indicating genetic distances among cyanobacterial strains. The genetic distance from the node, based on the RFLP profiles of 30 strains, is indicated on each branch.

An identical length (435 bp) and RFLP pattern type (K) were obtained for these cyanobacteria, indicating that they may be in the same species group. This is

consistent with results using primer pairs 1/18 and 14/18 where identical product profiles were obtained for these strains. In addition pattern type K showed

very high similarity to pattern type V for Nostoc muscorum. Strains of the genus Nostoc are closely related to, and often confused with, those of the genus Anabaena [22], particularly when morphological characteristics are used for identification. From this study, morphologies of N ellipsora and Anabaena PCC 7120 were very similar (data not shown), although previously PCC 7120 had been reported to be closer to Anabaena than to Nostoc, when using this criterion [17]. By contrast, on the basis of DNA-DNA hybridisation data and hybridisation patterns with highly repetitive (STRR) DNA sequences the strain has been renamed Nostoc sp. PCC 7120 [23,24]. This confusion highlights some of the difficulties in identification and taxonomy of cyanobacteria, which is exacerbated by continued use of traditional nomenclature and the botanical code. A dendrogram (Fig. 3) constructed on the basis of pairwise genetic distances using the combined RFLP data from the four restriction enzymes further indicates the similarity between Anabuena and Nostoc strains. In addition a low similarity is revealed between the freshwater Svnechococcus PCC 7942 and the marine Svnechococcus WH7803, in line with hybridisation data [25]. Since detailed biochemical and morphological studies were not carried out on some of the strains here, it is not possible to conclude on the correspondence between the genotypic characterisation and other properties of the organisms. However. it is noteworthy that the heterocystous strains form a cluster, excluding all the other strains. This is consistent with 16S rRNA sequence trees (e.g. [16]).

Thus PCR amplification and RFLP analysis of the 16S-23S rDNA spacer yield potentially useful markers for the identification of cyanobacteria. In addition more than one cyanobacterial strain can be detected in a mixed culture (data not shown). With more data it should be possible to generate a database of band patterns, in order to identify unknown isolates by the matching of pattern type and determining similarity to laboratory strains from genotypic analysis. Furthermore, the use of cells as the starting material for the PCR reaction, a procedure already reported for cyanobacteria (e.g. [15]) and confirmed here, should facilitate the development of a rapid and reproducible method for distinguishing between cyanobacteria.

Acknowledgments

We thank Professor N.G. Carr. University of Warwick. Professor B.A. Whitton. University of Durham and Dr. R.J. Smith. Lancaster University. for provision of certain cyanobacterial strains.

References

- [1] Rippka, R. (1988) Recognition and identification of cyanobacteria. Methods Enzymol. 16⁻⁻, 3–6⁻⁻.
- [2] Evans, E.H., Foulds, I. and Carr, N.G. (1976) Environmental conditions and morphological variation in the blue-green alga *Chlorogloea fruschii*. J. Gen. Microbiol. 92, 147–155.
- [3] Doers, M.P. and Parker, D.L. 1988) Properties of Microcysii ueruginosa and M. flos-aquae (Cyanophyta) in culture: taxonomic implications. J. Phycol. 24, 502-508.
- [4] Kenyon, C.N., Rippka, R. and Stanier, R.Y. (1972) Fatty acid composition and physiological properties of some filamentous blue-green algae. Arch. Microbiol. 83, 216-236.
- [5] Klein, S., Champman, D.J. and Garber, E.D. (1973) Chemotaxonomy of the Oscillatoria-P iormidium complex. Biochem. Syst. 1, 173–177.
- [6] Neilan, B.A. (1995) Identification and phylogenetic analysis of toxigenic cyanobacteria by multiplex randomly amplified polymorphic DNA PCR. Appl. Environ. Microbiol. 61, 2286-2291.
- [7] Eskew, D.L., Caetano-Anoiles, G., Bassam, B.J. and Gresshoff, P.M. (1993) DNA amplification fingerprinting of the symbiosis. Plant Mol. Biol. 21, 363–373.
- [8] Welsh, J. and McClelland, M. 1990) Fingerprinting genomes using PCR with arbitrary primers. Nucleic Acids Res. 18. 7213–7218.
- [9] Welsh, J. and McClelland, M. 1991) Genomic fingerprinting using arbitrarily primed PCR and a matrix of pairwise combinations of primers. Nucleic Acids Res. 19, 5275-5279.
- [10] Bolch, C.J.S., Blackburn, S.I., Neilan, B.A. and Grewe, P.M. (1996) Genetic characterization of strains of cyanobacteria using PCR-RFLP of the *cpcB.4* intergenic spacer and flanking regions. J. Phycol. 32, 445–451.
- [11] Wilmotte, A., Turner, S., Van de Peer, Y. and Pace, N.R. (1992) Taxonomic study of marine oscillatoriacean strains (Cyanobacteria) with narrow trichomes. II. Nucleotide sequence analysis of the 16S ribosomal RNA. J. Phycol. 28. 828-838.
- [12] Wilmotte, A., Van der Auwera, G. and De Wachter, R. (1993) Structure of the 16S ribosomal RNA of the thermophilic cyanobacterium *Chlorogloeopsus* HTF (*Mastigocladus laminosus* HTF') strain PCC ~518 and phylogenetic analysis. FEBS Lett. 317, 96-100.
- [13] Wilmotte, A., Neets, J.M. and De Wachter, R. (1994) Evolutionary affiliation of the marine nitrogen-fixing cyanobacterium *Trichodesmium* sp. strain NIBB 1067, derived by 16S

ribosomal RNA sequence analysis. Microbiology 140, 2159-2164

- [14] Neilan, B.A., Cox, P.T., Hawkins, P.R. and Goodman, A.E. (1994) 16S ribosomal RNA gene sequence and phylogeny of toxic Microcystin p (Cyanobacteria) DNA Seq. 4, 333 337
- [15] Nelissen, B. Wilmotte, A., Neefs, J.-M. and De Wachter, R. (1994) Phylogenetic relationships among filamentous helical evanobacteria investigated on the basis of 16S ribosomal RNA gene sequence analysis, Syst. Appl. Microbiol. 17, 206-210
- [16] Nelissen, B. De Baere, R., Wilmotte, A and De Wachter, R. (1996) Phylogenetic relationships of nonaxenic filamentous cyanobacterial strains based on 16S rRNA equence analysis. J Mol Evol 42, 194-200
- [17] Rippka, R. Deruelles, J., Waterbury, J.B., Herdman, M. and Stanier, R.Y. (1979) Generic assignments, strain histories and properties of pure cultures of cyanobacteria. J. Gen. Microbiol 111, 1-61
- [18] P rter. R D (1988) DNA transformation. Methods Enzymol. 167, 703 714
- 19] Armstrong, J., Gibbs, A., Peakall, R. and Weiller, G. (1994)

The RAPDistance package. ftp://life.anu.edu.au (150.203.38.74) pub/molecular_biology/rapdpack.exe.

- [20] Jensen, M.A., Webster, J.A. and Straus, N (1993) Rapid identification of bacteria on the basis of polymerase chain reaction-amplified ribosomal DNA spacer polymorphisms. Appl. Environ. Microbiol. 59, 945-952.
- [21] Jensen, M.A. and Straus, N (1993) Effect of PCR conditions on the formation of heteroduplex and single-stranded DNA products in the amplification of bacterial ribosomal DNA spacer regions. PCR Methods Appl. 3, 186-194.
- [22] Caudales, R. and Wells, J.M. (1992) Differentiation of freeliving Anabaena and Vostoc cyanobacteria on the basis of fatty acid composition. Int. J. Syst. Bacteriol. 42, 246–251.
- [23] Lachance, M.A. (1981) Genetic relatedness of heterocystous cyanobacteria by deoxyribonucleic acid-deoxyribonucleic acid reassociation. Int. J. Syst. Bacteriol. 31, 139-147
- [24] Mazel, D., Houmard, J., Castets, A.M. and Tandeau de Marsac, N. (1990) Highly repetitive DNA sequences in cyanobacterial genomes. J. Bacteriol. 172, 2755-2761.
- [25] Wood. A.M. and Townsend. D. (1990) DNA polymorphism within the WH 7803 serogroup of marine *Sunechococcus* spp. (Cyanobacteria). J. Phycol. 26, 576-585.